UNIVERSITY OF SOUTHERN QUEENSLAND



AN EVALUATION OF REDD+ IN COMMUNITY MANAGED FORESTS: A CASE STUDY FROM NEPAL

A Dissertation Submitted by

Shiva Shankar Pandey

MSc (Biodiversity Management), MA (Sociology and Anthropology), BSc (Forestry)

For the award of

DOCTOR OF PHILOSOPHY

2014

CERTIFICATION OF DISSERTATION

I certify that the ideas, research works, results, discussions and conclusions reported in this dissertation are entirely my own effort, except where otherwise acknowledged. I also certify that the work is original and has not been previously submitted to earn academic awards.

Signature of Candidate Shiva Shankar Pandey Date

Endorsement

Signature of Principal Supervisor Professor Geoffrey J Cockfield

Signature of Associate Supervisor Dr Tek Narayan Maraseni Date

Date

LIST OF PUBLICATIONS AND AWARDS

List of journal papers during the PhD study

- 1. **Pandey, S.S.**, Maraseni, T.N., Cockfield, G. and Gerhardt, K. 2014. Tree species diversity in community managed and national park forests in midhills of Central Nepal. *Journal of Sustainable Forestry* 33,796-813 (*Rank* "*B*" *Journal in ERA 2010*)
- 2. Maraseni, T.N. and **Pandey, S.S.** 2014. Can vegetation types work as an indicator of soil organic carbon? An insight from native vegetation in Nepal. *Ecological Indicators* 46, 315-322 (*Impact factor 3.2*)
- Pandey, S.S., Maraseni, T.N., and Cockfield, G. 2014. Carbon stock dynamics in different vegetation dominated community forests under REDD+: A case from Nepal. *Forest Ecology and Management* 327, 40–47 (*Impact factor 2.76; Rank "A" Journal in ERA 2010*)
- 4. **Pandey, S.S.**, Cockfield, G. and Maraseni, T.N. 2014. Dynamics of carbon and biodiversity under REDD+ regime: A case from Nepal. *Environmental Science & Policy* 38, 272–281 (*Impact factor 3.5*)
- Pandey, S.S., Cockfield, G. and Maraseni, T.N. 2013. Major drivers of deforestation and forest degradation in developing countries and REDD+. *International Journal of Forest Usufructs Management* 14, 99–107

Awards/Scholarships during the PhD study

- USQ Publication Excellence Awards for: Pandey, S.S., Cockfield, G. and Maraseni, T.N. 2014. Dynamics of carbon and biodiversity under REDD+ regime: A case from Nepal. Environmental Science & Policy 38, 272–281
- 2. *Student scholarship* to attend IUFRO Conference on Forests for People. Traverse City, Michigan on May 19–23, 2013
- 3. Full travel grants to attend "Major groups-Led Initiative in support of the United Nations Forum on Forests (UNFF) under the theme Forests & Economic Development: Positioning Forests to Contribute to Green

Economy". National Institute of Tropical Botany, Rio de Janeiro, Brazil on March 18–22, 2013

List of conference papers during the PhD study

- Cockfield, G., Dhakal, A., Maraseni, T.N. and Pandey, S.S. 2013. Factors influencing carbon sequestration potential in Nepalese forests and agro-forest. In: *International conference on forests, people and climate: changing paradigm.* Institute of Forestry, Tribhuvan University, Departmente of Forest Research & Survey, Government of Nepal and Faculty of Science, University of Copenhagen. Pokhara, Nepal (August 28–30, 2013)
- Pandey, S.S., Cockfield, G., Maraseni, T.N., and Subedi, B. 2013. Carbon enhancement in community based forestry: A case from early REDD+ project Nepal. In: Burns, R.C. and Highsmith, J. (Eds.) *Book of Abstracts*. IUFRO Conference on Forests for People. Traverse City, Michigan (May 19–23, 2013) p.55 http://www.recpro.org/assets/Conference_Proceedings/ffp_abstract_book_fin al.pdf
- Pandey, S.S., Cockfield, G., and Maraseni, T.N. 2013. Conservation of the forests with linking livelihood of local communities and REDD+ in Nepal. In: Burns, R.C. and Highsmith, J. (Eds.) *Book of Abstracts*. IUFRO Conference on Forests for People. Traverse City, Michigan (May 19–23, 2013) p.56 http://www.recpro.org/assets/Conference_Proceedings/ffp_abstract_book_fin

al.pdf

4. Pandey, S.S., Cockfield, G. and Maraseni, T.N. 2013. Economic development activities in community forests and REDD+ in Nepal. In: Major groups-Led Initiative in support of the United Nations Forum on Forests (UNFF) under the theme *Forests & Economic Development: Positioning Forests to Contribute to Green Economy*. National Institute of Tropical Botany, Rio de Janeiro, Brazil (18–22, March 2013) jointly organized by Friends of Siberian Forests, International Tropical Timber Organisation, Government of Brazil and Government of Germany.

ABSTRACT

Deforestation and forest degradation contribute between 10 and 25% of total annual greenhouse gas emissions. The REDD+ program for reducing emissions from deforestation and forest degradation and promoting forest conservation, sustainable management of the forests and enhancement of forest carbon stocks is one mechanism developed in an attempt to mitigate greenhouse gas emissions. Various REDD+ initiatives have been trialled in developing countries, including for community forests (CFs), which are an increasingly common form of resource management. Through the program, incentives are provided to community forest user groups (CFUGs) to encourage changes in management practices likely to increase sequestration stocks. There is, however, limited knowledge about the factors responsible for enhancing carbon stocks in CFs, the likely trade-offs within communities and the potential for increasing sequestration stocks.

The overarching goal of this research is to evaluate the impacts and potential of REDD+ projects in CF systems. Results from this study provide information for the design and development of programs to increase sequestration and conservation benefits in developing countries. This study estimated carbon stocks and change in carbon stock, technical potential (maximum stocks), key factors affecting carbon stock and trade-offs between gains in sequestration and other foregone community benefits. The study covered 105 CFUGs operating within five major dominant vegetation types. Annual data of carbon pools comprising above and below ground biomass were used to analyse carbon stocks and stock changes. Where sufficient data and models for key species were available, the potential carbon stock was estimated. Social, economic and management data, including a review of existing relevant documents, key informant survey and focus group discussion were used to identify major drivers of forest carbon stock changes in CFs and added community effort and foregone cost added for REDD+. Total costs of REDD+ participation were compared with the potential carbon benefits to enable trade-offs to be identified.

This study found variations in sequestration rates between CFUGs. Key variables were species type, canopy cover, elevation, age, forest scale, agriculture landholding size, disturbance levels, biomass extraction and the use of alternative energy sources.

In comparing present carbon stock with the technical potential of carbon stock in forests, the study identified significant potential for REDD+ projects to increase carbon stock in CFs.

On the negative side, changes in management practices added costs to communities, either through loss of forest products or through additional REDD+ activities, to the extent that the pilot REDD + projects were generally not economically beneficial for CFUGs. However, they could be made more beneficial with a reduction in the opportunity cost of community engagement (through scheduling) and the bundling of other non-carbon benefits together with carbon benefits. Outcomes could be improved through reducing 'leakages' resulting from a high dependency on forest resources through strategies such as the promotion of alternative energy sources (e.g. improved cooking stove and biogas).

ACKNOWLEDGEMENTS

First of all I would like to express my sincere gratitude to my supervisors Professor Geoff Cockfield and Dr Tek Narayan Maraseni for their guidance, supervision, coaching and encouragement throughout my PhD. Their continuous support, insightful comments and suggestions have been vital to the completion of this project and to the publication of five papers in peer review journals. It would not be possible for me to publish quality papers as lead author without their constructive suggestions and contribution on manuscripts. Similarly, I gained valuable insights through writing papers with them, particularly in analysing data and contributing to the paper on which I am second author.

I would like to acknowledge the University of Southern Queensland (USQ), Office of Research and Higher Degree for offering me the USQ post-graduate scholarship which was instrumental in enabling me to join the university and move further in my academic career. Similarly, I want to thank my research centre the International Centre for Applied Climate Sciences (ICACS) and Faculty of Business, Education, Law and the Arts (BELA) research staff and administrative staff for their cooperation and support.

I would also like to thank Dr Bhishma Prasad Subedi, Executive Director of the Asia Network for Sustainable Agriculture and Bioresources (ANSAB) for his encouragement and continuous support to enhance my career. Similarly I would like to thank all the team members of the Biodiversity, Ecosystem Services and Climate Change department of ANSAB, especially Dr Nabin Joshi, Sanjeeb Bhattarai, Dr Kalyan Gauli, Rijan Tamrakar, Shiva Subedi, Sagar GC and Kabiraj Praja, for their continuous support and being with me always. Similarly, I would like to thank Dr. Bhaskar Singh Karky, resource economist of the International Centre for Integrated Mountain Development (ICIMOD), for his cooperation and sharing of knowledge, and Mr. Tibendra Raj Banskota for his support regarding GIS data processing. I am indebted to the Federation of Community Forestry Users Nepal (FECOFUN) and three watershed (Kayerkhola, Ludikhola, Charnawati) level REDD+ network members and district forest office staff for their support during field work. Similarly,

I would like to acknowledge the cooperation of members from the REDD+ pilot project which was implemented under financial assistance of the Norwegian Agency for Development Cooperation (NORAD).

I would like to thank Dr Badri Basnet, Dr Dev Raj Paudyal, Dr Arun Dhakal, Dr Arjun Neupane, Dr Rohini Prasad Devkota, Sanjib Tiwari, Suman Aryal, Hemang Sharma, Arjun KC, Gobinda Baral and Rohanmuni Bajracharya for their cooperation, encouragement and support throughout my PhD journey at USQ. I would also like to thank colleagues of the Nepalese Association of Toowoomba (NAT) executive committee and all Nepalese communities of Toowoomba for their cooperation and moral support. I would also like to thank Dr Kathryn Mary Reardon-Smith for editing and proof-reading this thesis.

Finally, a word of gratitude and appreciation to my family members, especially to my father Mohan Bahadur Pandey and mother Komal Pandey for their continuous encouragement and brothers (Gouri Shankar Pandey and Shankar Pandey), sisters (Swosthani Pandey, Saraswati Pandey and Kiran Pandey), sister-in-laws (Goma and Sakuntala), brother-in-laws (Prashuram and Madan), nieces (Elina, Garima, Ayushma, Jebina, Prinshi) and nephews (Ayam and Prince) for their continuous support. Similarly, I would like to thank my father-in-law Ram Bahadur KC and mother-in-law Indira KC for inspiration and support. Last but not least, my beloved wife Sangita and son Suyogya deserve my wholehearted thanks for their enormous patience and motivation throughout this study.

TABLE OF CONTENTS

	CERTIFICATION OF DISSERTATION	ii
	LIST OF PUBLICATIONS AND AWARDS	<i>iii</i>
	ABSTRACT	v
	ACKNOWLEDGEMENTS	vii
	TABLE OF CONTENTS	ix
	LIST OF TABLES	<i>xiii</i>
	LIST OF FIGURES	xvi
	LIST OF APPENDICES	xviii
	ABBREVIATIONS	xix
1.	INTRODUCTION	1
1.1.	Background	2
1.2.	Statement of the problem	3
1.3.	Contentions to be tested in the study	6
1.4.	Objectives of study	6
1.5.	Justification of the study	7
1.6.	Scope and significance of the study	7
1.7.	Structure of the thesis	8
1.8.	Conclusions	9
2.	A REVIEW OF THE CARBON SEQUESTRATION DYNAMICS	
		AND
	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY	AND 11
2.1.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY	AND 11 12
2.1. 2.2.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures	AND 11 12 12
2.1. 2.2.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions	AND 11 12 12 12
2.1. 2.2.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry	AND 11 12 12 12 12 12
2.1. 2.2.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy	AND 11 12 12 12 12 12 12
2.1. 2.2.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities	AND 11 12 12 12 12 12 12 12 12 12 12 12
2.1.2.2.2.3.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests	AND 11 12 12 12 12 12 12 12 12 12 12 12
2.1.2.2.2.3.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12
2.1.2.2.2.3.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF 2.3.2. Carbon pools measurement methods	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12 12
2.1.2.2.2.3.2.4.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF 2.3.2. Carbon pools measurement methods Factors affecting carbon stock changes in CF	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12
2.1.2.2.2.3.2.4.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF 2.3.2. Carbon pools measurement methods Factors affecting carbon stock changes in CF 2.4.1. Possible biophysical factors	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12
2.1.2.2.2.3.2.4.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF 2.3.2. Carbon pools measurement methods Factors affecting carbon stock changes in CF 2.4.1. Possible biophysical factors 2.4.2. Possible socio-economic factors	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12
2.1.2.2.2.3.2.4.	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF 2.3.2. Carbon pools measurement methods Factors affecting carbon stock changes in CF 2.4.1. Possible biophysical factors 2.4.2. Possible socio-economic factors 2.4.3. Possible political factors	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12
 2.1. 2.2. 2.3. 2.4. 2.5. 	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF 2.3.2. Carbon pools measurement methods Factors affecting carbon stock changes in CF 2.4.1. Possible biophysical factors 2.4.2. Possible political factors 2.4.3. Possible political factors 2.4.3. Possible political factors	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12 12
 2.1. 2.2. 2.3. 2.4. 2.5. 	ECONOMICS OF REDD+ IN COMMUNITY FORESTRY Introduction Climate change and role of forests in mitigation measures 2.2.1. Climate change and share of forestry sector in global GHGs emissions 2.2.2. Global forest management practices and community forestry 2.2.3. Forestry sector in climate change mitigation policy 2.2.4. REDD+: introduction, challenges and opportunities Carbon pools and carbon pool measurement in community forests 2.3.1. Carbon pools in CF 2.3.2. Carbon pools in CF 2.3.2. Carbon pools in CF 2.3.4. Possible biophysical factors 2.4.1. Possible biophysical factors 2.4.2. Possible biophysical factors 2.4.3. Possible political factors 2.4.3. Possible political factors 2.4.3. Possible political factors 2.5.1. Costs due to sacrificed benefits of communities for REDD+ mechanism	AND 11 12 12 12 12 12 12 12 12 12 12 12 12 12

2.6.	Revi	ew of research methods related to CF studies	34
2.7.	Mon	itoring methods of the REDD+ activities	35
2.8.	Carb	on prices in the existing market place	
2.9.	Revi	ew of existing models for forest growth prediction	37
2.10). Theo	retical frameworks related to REDD+ in CF study	39
2.1	1. Conc	lusions	41
2	DESE	ADCH DESICN AND METHODS	12
J.	KESE	ARCH DESIGN AND METHODS	
3.1.	Intro	duction	44
3.2.	Desc	ription of the study areas	45
	3.2.1.	Demographic information of the study area	45
	3.2.2.	Precipitation and temperature of the study area	47
	3.2.3.	Vegetation types of the study areas	47
	3.2.4.	Forest management practice in the areas	47
	3.2.5.	<i>REDD</i> + project activities in the study areas	48
3.3.	Meth	od used to estimate stocking rate of vegetations, carbon stock an	ıd
	carbo	on stock changes in CFs	49
	3.3.1.	Inventory of vegetation and carbon pool	49
	3.3.2.	Analysis of vegetation and carbon pool	54
3.4.	Meth	od used to analyse potential growth of carbon stock in CFs	59
	3.4.1.	Estimation of biomass carbon in undisturbed forests in S. robusta forests	59
	3.4.2.	Estimation of carbon stock growth in S. robusta forests under CF	60
	3.4.3.	Gap in carbon stock in CFs with mixed broadleaf, Schima-Castanopsis, Pi Rhododendron-Quercus forests	ine and 61
35	Math	and used to analyse demographic data forest management pract	icas
5.5.	chan	and behaviour and costs and benefits consequences of CEUGs for	r the
	RED	D+	, inc 61
	3.5.1	Review of relevant documents	
	3.5.2.	Focus group discussion	
	3.5.3.	Key informant survey	63
3.6.	Meth	ods used to analyse factors affecting carbon stock changes	63
3.7.	Meth	ods used to analyse trade-off between carbon benefits and comm	nunity
	bene	fits	67
	3.7.1.	Costs of CFUGs for REDD+	67
	3.7.2.	Analysis of total costs and carbon benefits in REDD+ CFs	71
	3.7.3.	Carbon gain estimation in CFs	71
	3.7.4.	Price used in carbon revenue estimation	71
3.8.	Conc	lusions	72
1	DVN	MICS OF CADRON STOCK IN DEDD - COMMUNITY	
4. DYNAMICS OF CARBON STOCK IN REDD+ COMMUNITY EODESTS			72
	TUNE	4D T D	

4.1.	Introduction	74
4.2.	Distribution of different size trees in CFs	74
4.3.	Biomass and carbon stock dynamics in sample plots	
	4.3.1. Total biomass carbon in the reference year 2010 and year 2013	77
	4.3.2. Change in carbon stock in sample plots in CFs between 2010 and 2013	81
4.4.	Biomass and carbon stock in individual CF	
4.5.	Conclusion	85
5.	POTENTIAL BIOMASS CARBON GROWTH IN COMMUNITY	Y
	FORESTS	
5.1.	Introduction	
5.2.	Technical potential biomass carbon in Shorea robusta forests	
	5.2.1. Biomass carbon stock in undisturbed forests	
	5.2.2. Biomass carbon growth in CFs	
	5.2.3. Carbon growth in undisturbed forests, dense CFs and sparse CFs	95
5.3.	Biomass carbon growth in other vegetation types	
5.4.	Conclusion	
6.	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC	TORS
6.	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS GES 99
6. 6.1.	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES
6. 6.1. 6.2.	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction
6. 6.1. 6.2.	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction 100
6. 6.1. 6.2. 6.3.	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction 100
 6.1. 6.2. 6.3. 6.4 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES
 6.1. 6.2. 6.3. 6.4. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS [GES 99 100 eduction 100 105 109
 6.1. 6.2. 6.3. 6.4. 6.5. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction 100 105 109 115
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction 100 105 109 117
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS Introduction Evidence of changed behaviour of communities related to biomass re- in the forests Changed practices of CFUGs for the REDD+ project Elevation of the forests Proximity of forests from road head Proximity of forests from settlement Stand age in CFs	TORS IGES 99 100 eduction 100 105 109 115 117 119
 6. 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction 100 105 105 115 117 119 119
 6. 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction 100 105 105 115 117 117 119 120
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS [GES
 6. 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS [GES 99 100 eduction 100 105 105 109 117 117 117 119 119 120 122 122
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS [GES 99 100 eduction 100 105 109 115 117 119 119 120 122 122 123
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS IGES 99 100 eduction 100 eduction 105 105 105 115 117 117 119 119 120 121 122 122 123 124
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS [GES 99 100 eduction 100 eduction 100 105 109 115 117 117 119 119 120 122 122 123 124 125
 6.1. 6.2. 6.3. 6.4. 6.5. 6.6. 6.7. 6.8. 	CHANGED BEHAVIOUR OF THE COMMUNITIES AND FAC AFFECTING CARBON STOCK AND CARBON STOCK CHAN IN THE REDD+ CFS	TORS [GES 99 100 eduction 100 105 109 115 117 119 119 120 122 122 122 123 125 126

		6.8.10. 6.8.11.	Fodder collection	128 129
	6.9.	Concl	usion	130
_				
7	•	TRAD	E-OFF BETWEEN SACRIFICED BENEFIT TO	122
		COMIN	IUNITIES AND CARDON DENEFTIS IN CFS	133
	7.1.	Introd	uction	134
	7.2.	Contr	ibution of communities for REDD+	134
	7.3.	Chang	ged forest product use practices of communities for REDD+	136
	7.4.	Concl	usion	141
8	•	DISCU	SSION OF FINDINGS	143
	8.1.	Introd	uction	144
	8.2.	Comr 144	nunity-based forest management regime, rural livelihoods and	REDD+
	8.3.	Tree s REDI	ize distribution in community based forest management under D+ mechanism	145
	8.4.	Chang	ges in carbon stock in CFs under REDD+ mechanism	147
	8.5.	Poten	tial carbon growth in CFs	151
	8.6.	Key f	actors affecting carbon stock in CFs	156
	8.7.	Trade	-off between communities sacrificed benefits and carbon benef	fits in
		CFs		161
	8.8.	Sensit	ivity analysis of foregone benefits of communities for REDD+	-
		mecha	anism	164
	8.9.	Perfor	mance of REDD+ projects in CFs	165
	8.10	. Conte	mporary issues in REDD+ and REDD+ CFs	167
	8.11	. Ideal	CF for REDD+ benefits	171
9	•	SUMM	IARY, CONCLUSIONS AND RECOMMENDATIONS	177
	9.1.	Introd	uction	178
	9.2.	Sumn	nary of major findings	178
		9.2.1.	Distribution of different sized trees in forests and carbon stock dynamics	in CFs by
		9.2.2	vegetation types Potential growth of carbon stock in CFs	179
		9.2.3.	Key factors affecting carbon stock in CFs	
		9.2.4.	Model CFs for REDD+ projects	180
		9.2.5.	Trade-offs between foregone community benefits and carbon benefits in R CFs	&EDD+ 180

9.3.	Contentions made before the study	
9.4.	Research contributions	
9.5.	Research implications	
	9.5.1. Implications for policy making level	
	9.5.2. Implications for project implementation level	
9.6.	Limitations of the study	
9.7.	Suggestions for further research	
REFE	RENCES	
APPE	NDICES	

LIST OF TABLES

Table 2.1 Ownership of the global forests 14
Table 2.2 Price of carbon dioxide equivalent (CO2e) in three different carbonmarketing mechanisms where LULUCF sector is included
Table 3.1 Demographic information and size of forests in the study areas
Table 3.2 Average meteorological information of closely located stations from three watersheds in 1976-2005
Table 3.3 Wood specific gravity used in analysis 56
Table 3.4 Dry to green weight ration used to convert green biomass to dry biomass of sapling 57
Table 3.5 Methods used for analysing factors affecting carbon stock and change in carbon stock in CFs 65
Table 3.6 Items and data sources used to estimate real time benefits and costs of the communities in CFs for six year (2006 -2012 i.e. three years before and three years during the REDD+ projects)
Table 3.7 Unit price of forest products in 2012 in the study areas
Table 4.1 Comparison of stocking rates of woody vegetation by species and their strata in the REDD+ CFs (in year 2010 and 2013) in the study sites75
Table 4.2 Comparison of change in stocking rates (from 2010 and 2013) of woody vegetation by vegetation types and their strata in the REDD+ CFs in the study sites
Table 4.3 Comparison of biomass C (both above and belowground, in year 2010and 2013) by vegetation and their strata in CFs in the study areas

Table 4.4 Pair-wise comparison of p value (assuming unequal variance, two tail) for having similar mean carbon stock (difference in mean =0) in CFs by vegetation and canopy cover differences in 2010
Table 4.5 Comparison of changes in biomass C in three year (between 2010 and 2013, both above- and-belowground) by vegetation types and their strata in CFs in the study areas
Table 4.6 Pair-wise comparison of p- value (assuming unequal variance, two tail)for having similar mean carbon stock (difference in mean =0) for havingsame changes in biomass carbon in CFs (from 2010 to 2013) by vegetationand canopy cover
Table 4.7 Average carbon stock (MgC/ha) in different canopy forests within different altitudes estimated from sample plots
Table 4.8 Average biomass carbon (above and below ground, in year 2010 and2013) in CFs by dominant vegetation types in the study areas
Table 4.9 Change in biomass carbon (above and below ground, between 2010 and2013) in CFs by dominant vegetation types in the study areas
Table 5.1 Dynamic models of relative total biomass accumulation in dense S. robusta forests 91
Table 5.2 Dynamic models of relative total biomass accumulation in sparse S. robusta forests 92
Table 5.3 Gaps between 10 th percentile and 90 th percentile value of carbon stock in both dense and sparse canopy types CFs by vegetation
Table 6.1 Summary of the changed practices in CFs for REDD+ in the study areas 106
Table 6.2 Proximity (Road to plots) and average C-stock (MgC/ha in year 2010) and change in C-stock (between 2010-2013, MgC/ha) in forests analysing 21 plots located at farthest away and 21 at closest from the road head
 Table 6.3 Proximity (settlement to plots) and average c-stock (MgC/ha in year 2010) and change in C-stock (between 2010-2013, t C/ha) in forests analysing 21 plots located at farthest away and 21 at closest from the local settlement 117
Table 6.4 Average age of forest stand, and C-stock in year 2010 (MgC/ha) and C- stock change between 2010 and 2013 (MgC/ha) in CFs by vegetation types 119
Table 6.5 Per capita forest areas (ha/person) difference and carbon stock in year 2010 and change in carbon stock between 2010–2013 in CFs by vegetation 120

Table 6.6 Household level landholding size and C-stock in year 2010(MgC/ha) and C-stock change between 2010–2013 (MgC/ha) in CFs by vegetation types 121
Table 6.7 Household level livestock unit and C-stock in the year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types. 122
Table 6.8 Proportion of biogas using households and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types
Table 6.9 Proportion of Petroleum energy (LP gas, Kerosene) using householdsand C-stock in year 2010 (MgC/ha) and change in C-stock between 2010 –2013 (MgC/ha) in CFs by vegetation types
Table 6.10 Change in quantity of biomass extraction (both timber and firewood) and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types
Table 6.11 Grazing (Live Stock Unit days in a year) and C-stock in year 2010(MgC/ha) and C-stock change between 2010–2013 (MgC/ha) in CFs by vegetation types
Table 6.12 Caste heterogeneity and C-stock in year 2010 (MgC/ha) and C-stock change between 2010–2013 (MgC/ha) in CFs by vegetation types
Table 6.13 Quantity of grass collection and C-stock in year 2010 (MgC/ha) and C- stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types 128
Table 6.14 Quantity of fodder collection, and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types 129
Table 6.15 Quantity of litter collection, and C-stock in year 2010 (MgC/ha) and change in C-stock between 2010 – 2013 (MgC/ha) in CFs by vegetation types 130
Table 7.1 Participant number (person/ha/yr) in meetings and general assembly and forestry activities costs in the CFUGs (US\$/ ha/yr)
Table 7.2 Cost of the CFUGs (US\$/ ha/yr) by vegetation type due to changed forest management practices after REDD+ in the study sites
Table 7.3 Average annual forest product collection and income from NTFP in CFUGs by vegetation types (3 years before and during 3 years of REDD+) in the study sites

Table 7.4 Average annual cost (US\$/ha) of the CFUGs due to changing forest product use behaviour for the REDD+ by vegetation types of the study sites 138

Table 7.5 Sacrificed benefits of communities for the REDD+ in CFs by vegetation types in the study area 140
Table 8.1 Comparison between our estimates and other estimates for closelyrelated forests for carbon stock (MgC/ha)
Table 8.2 Theoretical maximum carbon stocks by years in CF and undisturbed S. robusta forests 153
Table 8.3 Age category of CFs by dominant vegetation types 153
Table 8.4 Scale of REDD+ project for CFs and possible issues

LIST OF FIGURES

Figure 2.1 Greenhouse gas emissions (CO ₂ -equivalent) measured in 1970, 1990 and 2010 by five sectors in four economic regions (Source: Victor <i>et al.</i> 2014) 13
Figure 2.2 Development of the REDD+ mechanism in UNFCCC
Figure 2.3 Theoretical aspects for REDD+ in community forestry
Figure 3.1 Map showing the study area with three watersheds: Kayerkhola watershed (271–1,618m asl) in Chitwan district; Ludikhola watershed (418–1,401m asl) in Gorkha district; and Charnawati watershed (652–3,238m asl) in Dolakha district
Figure 3.2 Distribution of sample plots in <i>S. robusta</i> , mixed broadleaf, <i>Schima</i> <i>Castanopsis</i> , <i>Pine</i> and <i>Rhododendron-Quercus</i> dominated forests in Charnawati watershed Dolakha (A), Kayerkhola watershed Chitwan (B) and Ludikhola Watershed Gorkha (C) of the study areas
Figure 3.3 Model of a composite sample plot used to measure carbon pools during forest inventory
Figure 5.1 Cumulative biomass carbon in undisturbed <i>S. robusta</i> forests (<i>adopted from Rautienen 1995</i>)
Figure 5.2 Mean Annual Increment (MAI) and Current Annual Increment (CAI) in undisturbed <i>S. robusta</i> forests (<i>adopted from Rautienen 1995</i>)
Figure 5.3 Cumulative biomass carbon in dense S. robusta dominated CFs93
Figure 5.4 Cumulative biomass carbon in sparse S. robusta dominated CFs93

Figure 5.5 Mean Annual Increment (MAI) and Current Annual Increment (CAI) in dense <i>S. robusta</i> dominated CF
Figure 5.6 Mean Annual Increment (MAI) and Current Annual Increment (CAI) in sparse <i>S. robusta</i> dominated CF94
Figure 5.7 Cumulative biomass carbon in undistubed natural forests with uniform age; dense and sparse canopy CFs with average age of dominanat trees in <i>S. robusta</i> forests
Figure 5.8 Area under curves showing cumulative biomass carbon growth in undistubed natural forests(34,708.07); (A) dense CFs (10,651.03); (B) sparse CFs (8,304.57) in <i>S. robusta</i> forests
Figure 6.1 Observed forest fire incidence by proportion of total plots in different vegetation and canopy cover types
Figure 6.2 Observed livestock grazing incidence by proportion of total plots in different vegetation and canopy cover types
Figure 6.3 Observed fodder collection incidence by proportion of total plots in different vegetation and canopy cover types
Figure 6.4 Observed firewood collection incidence by proportion of total plots in different vegetation and canopy cover types
Figure 6.5 Observed timber collection incidences by proportion of total plots in different vegetation and canopy cover types
Figure 6.6 Scatter plots showing relationship between carbon stock (average in year 2010 and change between 2010- 2013) in <i>S. robusta</i> dominant forests by canopy types
Figure 6.7 Scatter plots showing relationship between carbon stock (in year 2010 and change between 2010 - 2013) in Mixed broadleaf forests by canopy (A- F)
Figure 6.8 Scatter plots showing relationship between carbon stock (in year 2010 and change between 2010-2013) in <i>Schima-Castanopsis</i> forests by canopy types (A-F)
Figure 6.9 Scatter plots showing relationship between carbon stock (average in year 2010 and change between 2010- 2013) in <i>Pine</i> forests by canopy types (A-F)
Figure 6.10 Scatter plots showing relationship between carbon stock (average in
year 2010 and change in carbon stock between 2010 and 2013) in <i>Rhododendron-Quercus</i> forests by canopy types (A-F)

Figure 6.12 Proximity of sample plots from settlement	18
Figure 8.1 REDD+ project performance cycle can be used during REDD+ evaluation (<i>adopted from Liu and Walker</i> (1998))1	65
Figure 8.2 Theoretical growth pattern of carbon stock in forests [Adopted from growth pattern of trees mentioned in literatures (Birch 1999; Clark & Clark	č
1999) which was found similar in this study]1	75

LIST OF APPENDICES

Appendix A: Checklist for key informant survey
Appendix B: Checklist used to collect forest related information from documents available with CFUGs
Appendix C: Checklist used to collect costs and benefits data from documents available with CFUGs
Appendix D: Checklist of the questions asked in group discussion
Appendix E: A format used to collect forest resource inventory data in each sample plot
Appendix F: Comparison of biomass (both above ground and below ground) using four year measurement data by vegetation types and canopy strata of CFs in the study areas
Appendix G: Average carbon stock (MgC/ha) in individual CF in the study areas 243
Appendix H: Summary of local practices, changes in CFs and local perceptions about REDD+ in the study areas
Appendix I: Definition of some terms used in this thesis

ABBREVIATIONS

3PG	Physiological Principle Predicting Growth Model	
AGTB	Above Ground Tree Biomass	
AGSB	Above Ground Sapling Biomass	
AFOLU	Agriculture, Forestry and Other Land Use	
ANSAB	Asia Network for Sustainable Agriculture and	
	Bioresources	
BEF	Biomass Expansion Factor	
BGTB	Below Ground Tree Biomass	
BGSB	Below Ground Sapling Biomass	
DoF	Department of Forests	
С	Carbon	
CAI	Current Annual Increment	
CBA	Cost Benefit Analysis	
CBD	Convention on Biological Diversity	
CDM	Clean Development Mechanism	
CER	Certified Emissions Reduction	
CF	Community Forest	
CFUG	Community Forest User Group	
cm	Centimetre	
CO_2	Carbon Dioxide	
CO ₂ e	Carbon Dioxide Equivalent	
COP	Conferences of the Parties	
DANIDA	Danish International Development Agency	
DBH	Diameter at Breast Height	
DHM	Department of Hydrology and Meteorology	
EU-ETS	European Union Emissions Trading System	
FAO	Food and Agriculture Organisation	
FCPF	Forest Carbon Partnership Facility	
FECOFUN	Federation of Community Forestry Users Nepal	
FGD	Focus Group Discussion	
FSC	Forest Stewardship Council	
FullCAM	Full Carbon Accounting Model	

g	Gram		
GA	General Assembly		
GHG	Greenhouse Gas		
GIS	Geographic Information System		
GoN	Government of Nepal		
GPS	Global Positioning System		
Н	Height		
ha	Hectare		
HB	Herb Biomass		
НН	Household		
ICIMOD	International Centre for Integrated Mountain		
	Development		
ICS	Improved Cooking Stove		
IPCC	Intergovernmental Panel on Climate Change		
kg	Kilogram		
LB	Litter Biomass		
LPG	Liquefied Petroleum Gas		
LSU	Livestock Standard Unit		
LULUCF	Land Use, Land Use Change and Forestry		
MAI	Mean Annual Increment		
MFSC	Ministry of Forests and Soil Conservation		
Mg	Mega gram		
mm	millimeter		
MMF	Morgen-Mercer- Flodin		
MPFS	Master Plan for Forestry Sector Nepal		
MRV	Measurement, Reporting and Verification		
MSY	Maximum Sustained Yield		
NARC	National Agriculture Research Coulcil		
NARMSAP	Natural Resource Management Sector Assistance		
	Programme		
NORAD	Norwegian Agency for Development Cooperation		
NSCFP	Nepal Swiss Community Forestry Project		
NTFP	Non-Timber Forest Products		
ppm	Part Per Million		

REDD+	Reducing Emissions from Deforestation and Forest
	Degradation,Sustainable Management,
	Conservation and Enhancement of Forest Carbon
	Stock
PES	Payment for Ecosystem Services
\mathbb{R}^2	Coefficient of Determination
REL	Reference Emission Level
SBSTA	Subsidiary Body for Scientific and Technological
	Advice
SD	Standard Deviation
SOC	Soil Organic Carbon
UN	United Nations
UNFCCC	United Nations Framework Convention on Climate
	Change
UNFF	United Nations Forum on Forests
UN-REDD	United Nations REDD+
US\$	United States Dollar
VDC	Village Development Committee

CHAPTER ONE

INTRODUCTION

1.1. Background

Forest area loss is considered to be one of the major sources of greenhouse gas emissions (GHGs) (Fearnside 2000; UNFCCC 2014). Deforestation and forest degradation contribute 10–25 % of global anthropogenic emissions (Stern 2007; Van der Werf *et al.* 2009). To reduce deforestation and forest degradation, "REDD+" (reducing emissions from deforestation and forest degradation, conservation of the forests, sustainable management of the forests and enhancement of the carbon stock in the forests) has been trialled in developing countries (Olander *et al.* 2012; Streck 2012). REDD+ is expected to offer offsetting opportunities to Annex I countries (emitting more GHGs and with an obligation to reduce) for achieving their set targets of emission reduction while the non-annexed developing countries get income generating opportunities from their forest carbon enhancement performance (Angelsen 2009; Ghazoul *et al.* 2010).

The REDD+ program was instigated at the thirteenth session of the United Nations Framework Conventions in Climate Change (UNFCCC) Conference of Parties (COP 13) (Angelsen *et al.* 2012) and it is considered to be a primary agenda of the UNFCCC (UNFCCC 2007, 2010, 2011; Okereke & Dooley 2010; Szolgayová *et al.* 2014). Since 2007, many developing countries have initiated REDD+ readiness activities and demonstration projects (Cerbu *et al.* 2011). These initiatives are taking place under a number of bilateral and multilateral funding arrangements including the United Nations REDD+ (UN-REDD+) program and the World Bank's Forest Carbon Partnership Facility (FCPF). These projects are implementing pilot activities for the REDD+ at both project and national levels (Angelsen *et al.* 2009; Blom *et al.* 2010; Hajek *et al.* 2011; Leggett & Lovell 2012).

While implementing REDD+ mechanisms, safeguarding social or environmental benefits and developing credible Monitoring, Reporting and Verification (MRV) guidelines have been identified as key issues which need to be addressed by researchers (Angelsen *et al.* 2012). These issues were raised during negotiations and the Subsidiary Body for Scientific and Technological Advice (SBSTA) requested that developing countries be provided with guidance and support to address them (Johns *et al.* 2008; UNFCCC 2011). As each developing country is unique in terms of its economic status (Lee 2005), each has different capacity to adopt new technologies (Comin & Hobijn 2004), each has specific forest management policies and practices (FAO 2011), and each may need context specific REDD+ mechanisms to address these issues.

Increased carbon stock in forests is a major outcome of the REDD+ mechanism and this needs to be estimated and accounted for as a product for carbon markets and the focus of forest owners and their project activities (Fry 2011; Angelsen *et al.* 2012). The rates of carbon stock change in forests may differ with different natural factors related to forests and socio-economic factors related to forest owners. This is important information when designing REDD+ policies and projects. There is a range of tenure arrangements from forests which are government managed, to those managed by individual people or companies, to forests managed by local communities or indigenous people (Siry *et al.* 2005; FAO 2011). The proportions of both government and individual/company-owned forests are higher in developed countries whereas community-owned forests are more common in developing

countries (White & Martin 2002; Agrawal *et al.* 2008). Therefore, community forestry, a process whereby specific community forest users protect and manage state forests in some form of partnership with the government (Hobley 1996), is important for the REDD+ mechanism.

The area of community forests is increasing in Asia and Africa in order to conserve forest, fulfil subsistence needs for forest resources and improve livelihood opportunities of the forest dependent communities (Sunderlin *et al.* 2008). Nepal, a Himalayan country, provides a good example of community forests (CFs) with rights of forest management and sustainable use of more than 25% of national forests (1664 917 ha) afforded to 17,808 community forest user groups (CFUGs) representing more than 35% of the population (CFD/DoF 2013). The country has piloted Forest Stewardship Council (FSC) certification and has 21 CFUGs that have met the FSC forest management standards (Kanel 2006). A REDD+ community forestry (CF) pilot project was initiated by government and non-government organisations in 2009 (MFSC 2011). CFs can increase carbon stocks and contribute in sequestering atmospheric CO₂ (Maraseni et al. 2005) and the REDD+ interventions could be instrumental in improving both people's livelihood and carbon sequestration capacity of forests. However, REDD+ may require additional efforts and costs from communities, and poor forest dependent peoples would only be able to sacrifice the benefits they currently gain from forests if they get appropriate compensation from REDD+. Therefore there is nothing to be gained in implementing REDD+ in CFs if REDD+ carbon benefits are less than costs. Alternatively, CFUGs are able to improve the condition of forests after decade long management efforts and start to generate income by selling forest products (MFSC 2013) when benefits exceed costs. This is important for developing countries like Nepal where a total of 83% of people live in village areas, about 76% of households are involved in agriculture based economy (CBS 2011b) and farming practices are interrelated to forest resource extraction (Adhikari & Nagata 2004). Similarly, a total of 77% of total household energy demand is sourced from fuel wood in Nepal (WECS 2010) with forests the primary source.

In this context, there is a need to evaluate the REDD+ pilot projects to understand the carbon stock in CFs, factors affecting carbon stock and carbon changes, the maximum potential growth capacity of forests and the trade-off between carbon benefits and sacrificed benefits of the communities in REDD+ CFs. This knowledge may help to design REDD+ and similar programs. This PhD research strives to make a contribution on these issues by analysing an ongoing REDD+ pilot project in Nepal. In the next section, some of the issues related to this study are discussed.

1.2. Statement of the problem

The main policy problem driving this research is that forests are one of the key sectors in mitigating the rise in the level of GHGs (specifically, CO_2) in the atmosphere (Pachauri & Reisinger 2007; Stern 2007). Deforestation and forest degradation are main causes of CO_2 emissions with a global forest area loss of 5.2 million ha per year during 2000–2010 (FAO 2011). In order to reduce deforestation and forest degradation in developing countries, the REDD+ mechanism evolved after the Bali Action Plan (2007) and Copenhagen Accord (2009) as a potentially suitable market based mechanism (Grainger & Obersteiner 2011). There are possibilities to

change the existing forest management practices of local communities to increase future carbon stocks through the REDD+ incentive based mechanism. CF is an important target form of management because forest dependent people have a history of collectively working and satisfying their subsistence forest product needs while improving forest quality (Yadav *et al.* 2003; MFSC 2013). In pursuit of the potential economic benefits of a future REDD+ mechanism, CFUGs could change their existing practices; however, this may add costs or increase sacrificed benefits and may change governance practices (Phelps *et al.* 2010). Therefore there is not appropriate to implement REDD+ in CFs if REDD+ carbon benefits will be less than sacrificed benefits. From a welfare economics perspective (Hochman & Rodgers 1969), these costs could and should be fully compensated in the REDD+ mechanism (Ghazoul *et al.* 2010).

CFUGs comprise different community groups in terms of castes and forest resource dependency, and also encompass a range of vegetation types (Springate-Baginski *et al.* 2003). According to the decision of the General Assembly (GA) of a CFUG and with the approval of the district forest office, communities can also change their existing use practices and forest management activities to further improve forests and generate greater REDD+ carbon benefits. These changed activities and practices may involve additional costs to these communities which need to be estimated and identified. Knowledge about the trade-off between a community's sacrificed benefits and the carbon benefits gained, and so also whether or not the REDD+ approach is appropriate for CFs, is limited. The following research gaps in relation to this problem were identified in the literature.

• Knowledge gaps in forest carbon enhancement in CFs

Most previous studies use governance as well as forest cover and the income of local users as indicators to evaluate the outcomes of CF management (Yadav *et al.* 2003; Acharya 2004; Gilmour *et al.* 2004; Sunderlin 2004; Sapkota & Odén 2008). However, there are few and limited studies about carbon enhancement with the incentive motivation of REDD+ in CFs. One limitation is that there is little attention given to the disaggregation of the carbon storage performance of forest species. CF systems in Nepal vary in terms of vegetation composition and species dominance (CFD/DoF 2013) so are likely to have different capacities for carbon growth with the adoption of REDD+ activities. Some studies related to carbon accounting in forests have been carried out in government managed and protected area systems (Singh *et al.* 1994; Wang *et al.* 2008; Sharma *et al.* 2010; Solberg *et al.* 2010; Usuga *et al.* 2010; Köthke *et al.* 2014). However, for designing an appropriate REDD+ project and benefit distribution of the carbon benefits within CF systems, the carbon enhancement status of different CFs needs to be estimated.

• *Knowledge about gaps between actual carbon stock in CFs and potential growth in undisturbed forests*

Quantification of the baseline or reference level of carbon stock is important to assess changes in GHG emission reduction projects (IPCC 2006). However, the literature on studies of tree carbon stocks in CF and in natural undisturbed forests,

which would provide a benchmark for the full potential of these forests, is limited (Resosudarmo *et al.* 2012). If silvicultural activities are not appropriate, harvesting of forest products in higher quantities than the sustainable yield can reduce the biomass stock of forest stands (Foley *et al.* 2005; Hoover & Stout 2007; Chiang *et al.* 2008). Although, there is a specified provision of forest product harvesting and silviculture activities in operation plan of CFUGs; technical knowledge of communities and their implementation practices affect the outcomes (Pagdee *et al.* 2006). Therefore, the management practices of forest users in CF can lead to differences in carbon growth in managed forests compared to undisturbed forests. Estimation of carbon stock in both situations (i.e in CFs and without disturbances) is important to understanding the impact of CF. These difference will provide an idea about the maximum potential carbon benefits in CFs with REDD+ activities.

• *Knowledge gaps of major factors affecting carbon enhancement to design project activities to be targeted for REDD+ CF*

Forest based CO_2 emissions may be driven by various human induced factors including the conversion of forest land to agricultural land, illegal harvesting, infrastructure development, forest fires, encroachment and grazing (Geist & Lambin 2002; Pandey et al. 2013). Previous studies have found that the growth capacity of forests is strongly influenced by management practices (Hoen & Solberg 1994; Foley et al. 2005; Harmon et al. 2009). The sequestering of atmospheric CO₂ may depend on a forest's proximity to roads (Angelsen & Kaimowitz 1999) and settlements (Laurance et al. 2006), the rate of extraction of forest products (Harmon et al. 2009), livestock grazing (Blackmore & Vitousek 2000), agriculture landholding size (Adhikari et al. 2004), alternative energy sources (Katuwal & Bohara 2009), demographic factors (Jha & Bawa 2006), size of forests (Rudel et al. 2005) and age of forest stands (Pukkala et al. 2009). These may also differ in CFs where local communities are carrying out various forestry activities according to local knowledge and needs and the location of the forest area (MFSC 2013) but also in response to REDD+ economic possibilities. CFUGs may change existing practices in response to REDD+ incentives which may then change the status and carbon stock of the forests to an extent that is yet to be identified and analysed. Understanding of the major factors responsible for carbon enhancement in CFs is important in the design and implementation of REDD+ projects.

• *Knowledge gap about trade-off between community's sacrificed benefits and carbon benefits for REDD+ project in CFs*

An evaluation of the pilot project, particularly in terms of the trade-off between community foregone sacrificed benefits and carbon benefits, is important for the design and implementation of the REDD+ project. However, this has been found to be missing in the literature. Each community forest user group experiences real time benefits and costs in the management of a CF. Changes in existing costs (sacrificed benefits and added activities) and benefits (use of products, use of Non-Timber Forest Products (NTFPs) and ecosystem services including carbon) to communities due to the REDD+ (difference between before and after REDD+) is not well

understood in the available literature. It is important that the community sacrificed benefits (cost) involved in enhancing carbon stock in CFs is evaluated holistically and that the REDD+ program is modified if necessary to ensure that communities are not disadvantaged through their participation.

1.3. Contentions to be tested in the study

This research was based on several contentions during data collection, analysis and interpretation:

- Stocking rates of trees is affected by silviculture operations (Nyland 1996) therefore forest biomass (and therefore carbon stocks and sequestration rates) in CFs is affected by management practices. These include harvesting and other disturbance practices. Due to these disturbances, actual potential carbon stocks in CFs may be much lower than technical potential carbon stocks in undisturbed natural forests.
- Carbon stocks and sequestration rates are affected by various biophysical and socio-economic factors which include altitude, age, forest canopy cover, species type, size of forests, caste heterogeneity, agriculture land holding size, forest product extraction and use of alternative energy sources (Rudel 1989; Smith *et al.* 1997; Cochrane & Schulze 1999; Pahari & Murai 1999; Geist and lambin 2002; D'Amato *et al.* 2011).
- REDD+ incentives may be insufficient (Ghazoul *et al.* 2010), especially in the long term, to offset the economic losses from changing management practices of communities.

1.4. Objectives of study

The overall aim of the study is to evaluate the feasibility of the REDD+ project in CF systems in Nepal by considering the gain in carbon stock in relation to costs incurred by communities. The objectives (with research sub-questions) are to:

- a) Estimate carbon stocks and annual carbon stock changes in REDD+ CFs by vegetation type:
 - What is the total carbon stock in CFs by dominant vegetation types for a reference year of the REDD+ project?
 - What are the dynamics of carbon stocks in REDD+ CFs by different vegetation types?
- b) Estimate the technical potential carbon stock of undisturbed forests and actual carbon stock in CFs by dominant vegetation types:
 - What is the technical potential of carbon growth in CFs by dominant vegetation types?
 - What is the actual carbon growth in CFs by dominant vegetation types?
 - What is the gap between actual and potential carbon stocks in CFs by dominant vegetation types?

- c) Identify and analyse key factors affecting carbon stock changes in CFs:
 - What is the occurrence of biomass reduction activities after REDD+ in CFs?
 - What are the key biophysical and socio-economic factors affecting carbon stock changes in REDD+ CFs?
- d) Identify and estimate trade-offs between carbon benefits and net sacrificed community benefits for REDD+ in CF:
 - What community benefits are sacrificed to enhance carbon stocks in the REDD+ CFs?
 - How much community benefit is sacrificed to enhance per unit amount of carbon stock in the REDD+ CFs ?

1.5. Justification of the study

This study investigates the impact of REDD+ in CF because, while the UNFCCC has agreed to use REDD+ as a climate change mitigation option to be implemented in developing countries with an incentive mechanism (UNFCCC 2007), it is not clear that it will be an effective implementation mechanism; CF in some developing countries is recognised as a successful forest management system which respects traditional practices, fulfils local needs and generates incomes for the local people (Gilmour *et al.* 2004; Nurse & Malla 2006). Nepal is one of the leading countries which have been successfully implementing CF from last three decades (MFSC 2013).

This study selected to study CFs in Nepal because the Nepalese CF provides a successful model for adoption in other developing countries (Brown *et al.* 2002); the present CF system in Nepal has been developed from long experience (CFD/DoF 2013; MFSC 2013); government and nongovernment organisations in Nepal have been involved in REDD+ pilot initiatives in CF; and the Government of Nepal has established a separate institutional structure (REDD- Forestry and Climate Change Cell) under the Ministry of Forests and Soil Conservation (MFSC) to facilitate REDD+ activities. Moreover, a multi-partner REDD+ project in CF was implemented in Nepal from 2009 which provided ground for researching community efforts and carbon stock relationship (MFSC 2011). Finally the author has had a personal involvement in CF and earlier worked on the REDD+ pilot initiatives in Nepal. Although this study was conducted in Nepal, the overall framework of the research could be replicated and the knowledge generated from the study may be applicable in other developing countries.

1.6. Scope and significance of the study

On average, the global surface temperature has increased by 0.2°C per decade over the past 30 years (Pachauri & Reisinger 2007). Changes in the global temperature are resulting in changes to ecological systems and a shift in the habitats of many plants and wildlife (Walther *et al.* 2002) and an increase in extreme climatic events (Beniston & Stephenson 2004). Therefore, global negotiations are around the agenda of halting global temperature increases below 2°C above the pre-industrial period to reduce climate change risks, impacts and damages (Pachauri & Reisinger 2007; Meinshausen *et al.* 2009). Among different climate change mitigation options, the UNFCCC has agreed upon a REDD+ incentive based mechanism to be implemented in developing countries but scientific inputs are lacking (Caplow *et al.* 2011). This is important in CFs because the participation of local communities and the safeguarding of socio-economic benefits are issues for REDD+ (UNFCCC 2011) and the proper management of forest resources is suggested for poverty reduction and environmental sustainability by United Nations Forum on Forest (UNFF 2011). Without knowledge of the trade-offs between community's added foregone costs and carbon benefits, REDD+ may lead to perverse incentives and may not address, but exacerbate, existing social, environmental and climate change problems. Therefore, this study is crucial to policy decisions.

Reducing poverty of the forest users through the sustainable use of forest resources is one of the key aims of CF (Mahanty *et al.* 2009). Existing CFs provide subsistence products to the communities that conserve forests but may not be enough to bring these communities out of poverty especially in rural area where limited economic options are provided (Edwards 1996; Subedi 2006). Previous studies suggest implementing economically oriented CF to address poverty issues in developing countries (Edwards 1996; Gilmour *et al.* 2004; Subedi 2006). This study provides learning and new knowledge for consideration during the design and implementation of REDD+ and other incentive based conservation projects in CF.

As a least developed country, the government of Nepal is also moving towards REDD+ initiatives under multinational (UN REDD, World Bank forest carbon partnership facility -FCPF) and bilateral (US government, Norwegian government) funding schemes. The Government of Nepal has agreed to facilitate a nested approach (both national and subnational level) to REDD+ projects (MFSC 2011). However, the effective implementation of the REDD+ project needs a fair benefit distribution mechanism (Hoang *et al.* 2013) which accounts for foregone costs. The finding of this study can help to make appropriate policies that consider both community costs and carbon benefits in CF.

1.7. Structure of the thesis

This thesis is divided into nine chapters. In the first three chapters, the introduction, review of relevant literatures and research methodology used for this study are included. The introduction chapter provides the background, research problem, objectives and significance of the study. Chapter Two reviews the existing literature on carbon sequestrations dynamics and the economics of REDD+ in CFs. In this chapter, knowledge gaps, theoretical perspectives and issues related to REDD+ and CFs are identified. In Chapter Three, the detailed methodologies used to address gaps identified in the literature review and specified in Chapter One are presented. The fourth, fifth, sixth and seventh chapters give the results of the study. Chapter Four gives the dynamics of the stocking rates of different sized trees, carbon stocks in CFs for the reference year (2010) and changes after REDD+ project activities in CFs. Chapter Five gives actual carbon stock and potential maximum carbon stock growth in CFs. Chapter Six investigates changes in the behaviour of communities with REDD+ and key factors affecting carbon stock differences in CFs and Chapter Seven looks at the trade-offs between sacrificed benefits of communities for carbon benefits

under the REDD+ mechanism. Finally, Chapter Eight and Nine discuss and summarise the major findings of the study, its research contribution and implications and makes suggestions for future research.

1.8. Conclusions

The REDD+ mechanism for developing countries is a promising concept which provides offset options to developed countries and income generation opportunities to poor forest dependent communities. CF is a participatory forestry practice that may demand additional efforts and changes in the existing practices of communities. For its long term success, an effective design and implementation of REDD+ in CFs is required; this could be better if it is based on knowledge about existing carbon stocks, the carbon stock increment potential, factors affecting carbon stock differences in CFs and trade-offs between community foregone sacrificed benefits and carbon benefits.

This chapter identified problems and gaps in the literature. Since REDD+ is new mechanism, knowledge generated from field based studies is limited. The broad aims and specific objectives of the study supported with research questions and contentions are listed. Justification and scope, as well as the significance of the study are also highlighted. Outcomes of this study may be useful to assess possible carbon benefits in other CFUGs while designing and implementing the REDD+ projects in future. Now many institutions are working on the REDD+ mechanism in Nepal and can be benefited from this study to design and implement better REDD+ projects. Issues related to carbon sequestration dynamics in forests, factors affecting carbon stocks, economics of REDD+ in CF, methodologies related to carbon pool measurements are reviewed in existing literatures and presented in Chapter 2.

CHAPTER TWO

A REVIEW OF THE CARBON SEQUESTRATION DYNAMICS AND ECONOMICS OF REDD+ IN COMMUNITY FORESTRY

2.1. Introduction

Chapter one explains about major knowledge gaps on possible outcomes of the REDD+ projects in CFs, research questions for this study and its significance. This chapter includes a review of some relevant issues about climate change, the role of forests in mitigating climate change and mechanisms for reducing GHG emissions from forestry sectors. This chapter includes ten sections. The next section reviews the literature related to climate change and the role of forests in mitigation measures. The third section gives information about carbon pools and carbon pool measurement practices in forests, the fourth section includes reviews regarding possible factors affecting carbon stock changes in CFs, the fifth section covers the economics of REDD+ in CFs while the sixth section includes a review of research methods in CF. In the seventh section, the chapter covers existing knowledge about carbon prices in global markets and, in the eighth section, reviews existing yield models for forest growth prediction. The ninth section highlights information about theoretical aspects related to REDD+ in CFs and the final section provides a conclusion to the chapter.

2.2. Climate change and role of forests in mitigation measures

2.2.1. Climate change and share of forestry sector in global GHGs emissions

Average global temperatures increased in the late 20th century. The rate of global surface temperature increase has been about 0.2°C per decade over the past 30 years (Hansen et al. 2006; Pachauri & Reisinger 2007). Global temperature increase and climate change are used interchangeably in literatures (Stern 2007; Shi et al. 2010; Cook et al. 2013). Anthropogenic activities are considered the main causes of climate change (Tett et al. 2002; Cook et al. 2013). According to the fifth assessment report of the Intergovernmental Panel on Climate Change (IPCC), GHG emissions from 2000 to 2010 have grown at about twice the rate observed in the decade from 1970. Of all anthropogenic GHGs, CO_2 was responsible for about 55% of total emissions in the decade from 1970 and 65% of emissions in the decade to 2010 (Victor et al. 2014). Out of five major sectors examined, Agriculture, Forestry and Other Land Use (AFOLU) contributed 24%. Within total emissions of each economic region, the AFOLU related emission is higher in lower income regions than higher income (Figure 2.1). Therefore, addressing emissions related to AFOLU in developing countries is important.



Figure 2.1 Greenhouse gas emissions (CO₂-equivalent) measured in 1970, 1990 and 2010 by five sectors in four economic regions (Source: Victor *et al.* 2014)

Within AFOLU, deforestation and forest degradation contribute significant emissions. The forestry sector is important, storing huge quantities of carbon stock. At present, forests occupy 30% of the global land area and hold double the amount of carbon that is in the atmosphere (Canadell & Raupach 2008; FAO 2010b). This shows the important contribution of the forests in mitigating climate change through reducing atmospheric carbon; however, there is also significant risk of stored carbon being returned to the atmosphere through forest disturbance (Canadell & Raupach 2008). Deforestation and forest degradation contribute 10-25 % of total global emissions (Stern 2007; Van der Werf et al. 2009). Those emissions related to deforestation and forest degradation come mainly from human activities (Kasischke et al. 1995; Nepstad et al. 1999; Hurteau et al. 2008). This is mostly in developing countries where forests are used for economic development (Koop & Tole 2001; Ewers 2006). For example, in Nepal, the area of forests decreased by 1.7% per annum from 1978 to 1994. Shrub areas of Nepal is incressing from 4.7% in 1978/79 to 10.6 % in 1994 of country areas which shows that degradation is going on in some of the existing forests (FAO 1999).

2.2.2. Global forest management practices and community forestry

Clarity in ownership is an important factor in achieving the aim of sustainable forest management (Dudley *et al.* 2005; Jin & Sader 2006; Siry *et al.* 2005). At present, global forests are mainly under three ownership systems, namely public, private and community. The majority of forests are owned by government, followed by private (individual or company) and community (White & Martin 2002; FAO 2011;). Forest areas owned by government are generally larger in developed countries than in developing countries, while the opposite is the case for community owned forests which are larger in developing countries. Similarly, a larger proportion of private forests are in developed countries than in developing countries are working collectively to manage their surrounding forests (Gilmour & Fisher 1991; Bray *et al.* 2005; Pagdee *et al.* 2006) through mutually agreed plans and decisions (Nurse & Malla 2006).

Categories	Government (%)	Community (%)	Private form (%)
Global forest	77	11	12
Developing countries	71	22	7
Developed countries	81	3	16
Tropical countries	71	19	10

 Table 2.1 Ownership of the global forests

Source: White & Martin (2002)

All forests contribute to global climate change mitigation (Houghton 2005; Maraseni et al. 2005; Peichl & Arain 2007) although the quantity and trends of carbon sequestration capacity vary (Kasischke et al. 1995; Lambin 1999; Ayres 2000; Koop & Tole 2001; Pregitzer & Euskirchen 2004; Houghton 2005). A global study conducted in major forested countries found that a significant proportion of government managed forest areas have logging concessions, particularly in Canada, Democratic Republic of Congo, Central African Republic, Gabon, Equatorial Guinea, Malaysia, Cambodia and Indonesia (White & Martin 2002) indicating a possible risk of increasing deforestation and forest degradation. CF recognises the traditional forest resource use rights of local and indigenous communities who are living in areas surrounding forests and have a history of using forest resources for their basic needs (Arnold 1991; Nurse & Malla 2006). CF is expanding in developing countries (White & Martin 2002) and there are now more than 10% of global forests and over 22% forests in developing countries under this form of management (Nurse & Malla 2006). This proportion is likely to increase with the present global prioritisation of participatory forest management approaches which emphasise equity and livelihood needs, in line with the aims of decentralization, cost effective management, capacity of the local people to be a best manager and proximity to resources (Brown 1999; Gilmour 2003; Gilmour et al. 2004).

The concept of CF evolved in the 1970s with recognition of the interdependent relationship between local communities and forests with local communities using forest products for various needs including fodder, fuel, foods, timber and fibre (Davidar *et al.* 2008; Metz 1994). In Nepal, the Government allocates a certain area

of national forest to an organized group of local communities who live around a forest and use forest products for their subsistence (Acharya 2002; Gilmour *et al.* 2004). After three decades of CF practices, Nepal has devolved forest management authority of more than 25% of its forested areas to 17,808 CFUGs involving more than 35% of the total population of the country (CFD/DoF 2013). Now, CFUGs are able to improve forest status and some of them have initiated to get economic benefits from sustainable harvesting of forest products (MFSC 2013).

2.2.3. Forestry sector in climate change mitigation policy

The scientific community has identified that the maximum allowable increment in mean global temperature of 2°C above the preindustrial level to minimise future climatic impacts (Meinshausen 2006; Randalls 2010) and that this is possible by limiting GHGs emissions to 550 ppmv (parts per million by volume) CO_2 equivalent (Randalls 2010). The UNFCCC agreed to develop a collaborative plan between developed and developing countries, linked to the sustainable development agenda (UNFCCC 2010). For this, the UNFCCC through the Kyoto protocol (1997) set emission reduction targets for industrialised countries and provided options to achieve the target by adopting three mechanisms; these are the Clean Development Mechanism (CDM), joint implementation and emissions trading (Kyoto-Protocol 1997).

The IPCC fifth assessment report (2014) clearly identifies the forestry sector as one of the key sectors responsible for GHG emissions (Victor et al. 2014) while conserving and reducing forest biomass loss can provide a relatively cheap form of climate change mitigation (Stern 2007). Therefore, forestry sector emissions were included in the UNFCCC, a key intergovernmental global body that facilitates discussions on global concerns and solutions related to climate change among parties, starting from the fifth Conference of Parties (CoP) (Kyoto-Protocol 1997) (Figure 2.2). The Kyoto Protocol (1997) included afforestation (plant and develop new forests on land where forests had not existed for a long period) and reforestation (plant and develop forests where forests had previously existed) project activities under the CDM mechanism. However there were few afforestation and reforestation (A/R) projects under the CDM mechanism with only 55 A/R projects of a total of 7,531 CDM projects registered with the UNFCCC by 2014 (UNFCCC 2014). This could be a result of the cumbersome process involved in getting a project registered under the CDM (Hayashi & Michaelowa 2007) and the comparatively long time needed for forestry projects to sequester a significant quantity of atmospheric CO₂.

Although natural forests and their role in mitigating climate change were not included in the CDM, deforestation and forest degradation of natural forests was highlighted as the main source of GHG emissions in developing countries (Gibbs *et al.* 2007) and global forest loss was recognised as massive in tropical countries including Brazil and Indonesia (Gibbs *et al.* 2007; Van der Werf *et al.* 2009). Costa Rica, Papua New Guinea and Central African Forest Commissions raised strong concerns about avoided deforestation at the 11th session of the Conference of Parties (COP 11) and requested consideration of avoided deforestation by the UNFCCC. However, provisions for rewarding countries which reduce deforestation and forest degradation were only considered after COP 13 (UNFCCC 2007).
Reducing emissions from deforestation and forest degradation (REDD) was included in the Bali Action Plan at COP 13 (UNFCCC 2007). The concept of reducing deforestation and forest degradation, conservation, sustainable management of the forests and enhancement of forest carbon (REDD+) was introduced in COP 14 in Poznan (2008) and agreed in COP 15 in Copenhagen (2009). Inception of the REDD+ was supported by pressure from countries associated with the Rainforest Alliance, countries of the African Forest Commission and the advice of UNFCCC's SBSTA. To put these decisions in practice, COP 15 encouraged developed countries to support developing countries for readiness and demonstration activities for REDD+.

While developing strategies and planning for the REDD+ is underway, the literature highlights a possible threat to biodiversity and the rights of local and indigenous communities (Angelsen 2009; Kanowski *et al.* 2011; Thompson *et al.* 2011). At the 16th COP in Cancun, parties agreed to develop appropriate guidelines for safeguarding these social and environmental principles while developing REDD+ projects in developing countries (UNFCCC 2011). The 19th COP adopted the 'Warsaw framework' to move forward on REDD+ which includes provisions for developing: (1) national and sub national level forest monitoring systems; (2) guidelines for safeguarding social and environmental benefits; and (3) financial and technical support for developing countries (UNFCCC 2014).

It is now already seven years after the REDD+ concept evolved which is coming through various discussions in different COPs at the UNFCCC (Figure 2.2).





(Source: UNFCCC websites:

http://unfccc.int/resource/docs/convkp/kpeng.pdf; http://unfccc.int/land_use_and_climate_change/lulucf/items/3063.php; http://unfccc.int/meetings/montreal_nov_2005/meeting/6329/php/view/decisions.php;

http://unfccc.int/meetings/cancun_nov_2010/meeting/6266.php; http://unfccc.int/resource/docs/2011/cop17/eng/09a02.pdf; http://unfccc.int/resource/docs/2012/cop18/eng/08a01.pdf; and http://unfccc.int/key_steps/warsaw_outcomes/items/8006.php)

http://unfccc.int/resource/docs/2007/cop13/eng/06a01.pdf; http://unfccc.int/meetings/poznan_dec_2008/meeting/6314.php; http://unfccc.int/meetings/copenhagen_dec_2009/meeting/6295.php;

2.2.4. **REDD+:** introduction, challenges and opportunities

a) **REDD+:** an introduction

REDD+ is considered to be a relatively cost effective, easy and quick way to mitigate climate change (Clements 2010; Angelsen *et al.* 2012; Gardner *et al.* 2012). The market based mechanism is expected to incentivise developing countries to undertake forest management activities to enhance carbon stocks in forests. The REDD+ will be voluntary for any developing country which can participate and present a verifiable quantity of forest carbon enhancement (Gibbs *et al.* 2007). It must have additionality in emission reduction or sequestration with REDD+ project activities. The additionality means that quantity of reduced emissions or increased sequestration which would not have happened without the REDD+ project activity or at the business-as-usual scenario. It includes five forestry activities for reducing forestry related carbon emissions and increasing sequestration of atmospheric carbon in developing countries. These are: (1) reducing emissions from deforestation; (2) reducing emissions from forest degradation; (3) conservation of the forests; (4) sustainable management of forests; and (5) enhancement of forest carbon stocks (UNFCCC 2010; Angelsen *et al.* 2012). Brief descriptions of these activities follow.

<u>Reducing emissions from deforestation</u>: The Marrakesh accord defined deforestation as a direct human-induced conversion of forested land to non-forested land (Angelsen *et al.* 2008). Reducing deforestation is a primary aim of the REDD+ (de Jong *et al.* 2010). Government or forest owners should limit the conversion of forest areas to other land use systems from the reference forest area (UNFCCC 2007). The reference level of historical deforestation is estimated for projecting business-asusual scenarios against which the reduction of deforestation is monitored for REDD+ outcome.

<u>Reducing emissions from forest degradation</u>: Forest degradation activity is a reduction of biomass stock and productivity of a forest. Forest cover in degraded forests is reduced from the reference level (Van der Werf *et al.* 2009). In REDD+, establishing baseline scenarios, developing a reference level and monitoring forest biomass degradation part is difficult compared to the deforestation situation (Herold & Skutsch 2011). Most forests face some sort of degradation due to human activities including CFs (GoN/MFSC/REDD-Cell 2014); however, levels of forest degradation can differ.

<u>Conservation of forests</u>: In the context of REDD+, the conservation of forests means conserving forests in as close to natural condition as possible. In this activity, maintaining the existing biomass stock of forests is expected to ensure the carbon sequestration capacity of forests (Herold & Skutsch 2011). In CFs, strict conservation without allowing people to harvest forest products may not be possible because the fundamental objective of CF is to improve forests and provide subsistence forest products to CFUGs (GoN 1993). If a CFUG has decided on strict conservation without arranged alternatives, that may either encourage leakage or deteriorate local people's socio-economic prospects, which would be against REDD+ concepts and UNFCCC agreements (Brown *et al.* 2008; UNFCCC 2011).

<u>Sustainable management of forests</u>: Sustainable management of forests means the conservation and sustainable use of forest resources without reducing existing productivity and biomass stock. This activity includes sustainable use practices (i.e. use within the limit of yield) that create zero or positive carbon balance in the long-run (Herold & Skutsch 2011).

<u>Enhancement of forest carbon stocks</u>: The enhancement of forest carbon stocks is also a key part of the REDD+ mechanism. For this, activities which ensure reductions in the extraction of forest resources, plantation activities, the promotion of alternative sources of forest products and reduce pressure on forests can be designed and implemented (Geist & Lambin 2002; Nagendra 2007).

b) Challenges and opportunities on REDD+

Although, the REDD+ is a promising mechanism, the design and implementation of this mechanism is challenging (Minang & van Noordwijk 2013). It may require changes to existing historical practices and the building of political consensus between key actors in policy arenas to achieve multilevel coordination (from global to local communities) and manage complex flows of information and payments in the context of large future uncertainties about climate mitigation regimes while ensuring supply to meet strong immediate demands for food, fuel and fibre (Angelsen *et al.* 2012). The REDD+ has a supposition of win-win policy frameworks that help to attain dual goals of reduction in GHG emissions and reduction of poverty in developing countries (Angelsen *et al.* 2009). For effective implementation, appropriate REDD+ policy, projects and implementation strategies may be needed. However there are various challenges indicated and issues raised in global discussions and the literature which need to be addressed. Key issues and challenges associated with REDD+ are discussed below:

<u>Technical challenges in developing a credible MRV mechanism</u>: Measurement, Reporting and Verification (MRV) of the REDD+ project performance is important to show carbon sequestration performance and to claim credits. Making a credible MRV mechanism considering transparency, accountability and sustainability is an important issue for the REDD+ framework (Angelsen *et al.* 2009). The use of external experts in the MRV process is possible, but may increase the costs and therefore the price of carbon credits in REDD+ schemes. On the other hand, the full or partial participation of local people in the MRV process can be cost effective, transparent and sustainable (Rist *et al.* 2010; Danielsen *et al.* 2011); therefore the potential of using locally based monitoring systems in REDD+ is emphasised (Skutsch 2005; Danielsen *et al.* 2011). Participatory monitoring may be more suited to the CF system as local communities that are managing a forest may be involved in forest resource monitoring (Ojha *et al.* 2009). However, an appropriate verification mechanism would need to be developed.

During design phase of the REDD+ project, a baseline scenario and a historical Reference Emission Level (REL) of deforestation and forest degradation of the country is identified on the basis of historical data (Herold & Skutsch 2011). In REDD+ projects, REL and change in carbon credits is estimated and verified to make the carbon credit marketable.

<u>Developing equitable benefit sharing mechanism</u>: Multiple stakeholders are involving in forest management in developing countries including government agencies, communities and the private sector (FAO 2010b). The combined efforts of all stakeholders in REDD+ projects are expected to generate carbon credits from their forest management. Creating an environment by involving all stakeholders including local communities, accounting contributions of rural people and designing an equitable benefit distribution without promoting perverse incentives are important for an effective REDD+ mechanism (Lindhjem *et al.* 2011).

<u>Social and environmental safeguards:</u> The possible exclusion of local people and indigenous people from the decision making process of the REDD+ projects was raised in UNFCCC's COP 16 and an agreement was made for social and environmental safeguards (UNFCCC 2011). It can however be difficult to secure the rights of local communities and indigenous people in the REDD+ process (Singh 2008; Angelsen *et al.* 2009; Karousakis 2009; Grainger *et al.* 2009; Ghazoul *et al.* 2010; Springate-Baginski *et al.* 2010; Corbera & Schroeder 2011; Lyster 2011; Thompson *et al.* 2011). This is important because a huge number (827 million) of people living in developing countries are hungry with 14.3% of the global population undernourished (FAO 2013). Many local and indigenous people living in these countries have long relied on forest resources for their livelihood (White & Martin 2002). This dependency has been recognised in CF programme in which local communities are engaged in forest improvement to fulfil household level of forest products needed.

As forest ecosystems are rich in terrestrial biodiversity in comparison to other ecosystems (CBD 2001), CF has been contributing to improved forest ecosystem condition (Dev *et al.* 2003; Thoms 2008). However poor people are motivated to generate income from forest management (Subedi 2006) and therefore their management can promote certain tree types which have higher carbon growth potential. This may result in the loss of biodiversity in these forests.

Although CF is contributing both to improved livelihoods of poor communities and condition of forests, REDD+ carbon oriented forestry practices may impact upon livelihood benefits and environmental conservation. If REDD+ projects do not consider the livelihood needs of local communities during project design and implementation, they may create a scarcity of forest resources to those communities (Chhatre and Agrawal 2009; Caplow *et al.* 2011). Similarly, the REDD+ project with carbon-oriented activities may promote limited fast-growing species without conserving multiple species and ecosystems. Maintaining multiple species is important for securing biodiversity benefits in abiding by the Convention on Bio Diversity (CBD 1992). Species-focused management will create instability within local societies and biodiversity loss which may ultimately lead to adverse outcomes for the REDD+ activities. Therefore safeguarding both community livelihood (social safeguard) benefits and biodiversity (environmental safeguard) is important for REDD+ projects.

<u>Addressing possible leakage and non-permanence</u>: Carbon leakage in REDD+ projects (i.e. conserving or increasing carbon stock in project areas by shifting source of biomass extraction to outside project areas) may lead to an increase in emissions (Phelps *et al.* 2010; Olander *et al.* 2012). This is possible in CF if communities are encouraged to avoid extraction of forest products while no alternatives to forest products are provided. Leakage issues could be addressed by considering potential emissions displacement, implementing possible activities and adopting proper leakage monitoring. Similarly, securing the permanency or irreversibility of emissions reductions for a REDD+ project is also a challenge (Angelsen 2009). Warranting tenure rights, providing alternative sources of forest products and benefits with increasingly less dependency on forests will ensure long-term emissions reduction (Angensen *et al.* 2008; Phelps *et al.* 2010). If governments take back forests from CFUGs or communities cannot get adequate monetary benefits from REDD+, extraction of forest products may increase again to similar levels to those experienced before the REDD+ activities.

Besides the above mentioned challenges, REDD+ has several associated opportunities for participating countries particularly in securing poverty reduction, biodiversity and other environmental co-benefits in developing countries (Huettner 2012). Key opportunities of REDD+ are:

- The REDD+ mechanism has potential to provide significant new opportunities for developing countries to receive funding for forest restoration projects and programs (Alexander *et al.* 2011).
- Carbon credits may be generated by reducing forestry sector GHG emissions and increasing the capacity of forests to sequester additional CO₂ in developing countries. Since GHG emissions in developing countries are low, these countries should be able to sell carbon units gained from forest management and get monetary benefits from high emissions countries (Kyoto-Protocol 1997).
- REDD+ mechanisms will provide opportunities to developed countries to offset their emissions in developing countries. This mechanism will help to achieve their commitments and emission reduction targets by purchasing carbon credits from developing countries (Phelps *et al.* 2010).
- REDD+ can reverse degradation in forests and deforested areas, and increase vegetation cover in denuded areas (Laurance 2008).
- REDD+ can generate various co-benefits or non-carbon benefits particularly conservation of biodiversity if there are multiple species promotion strategies (Kanowski *et al.* 2005; Grainger *et al.* 2009; Karousakis 2009) and enhance other ecosystem services (Millennium Ecosystem Assessment 2003).

c) **Present REDD+** projects in practices

Establishment of a mechanism and arrangements which enable the mobilisation of financial resources from developed countries was a key emphasis in the Copenhagen Accord (UNFCCC 2009). After Copenhagen, different funding schemes such as the World Bank funded Forest Carbon Partnership Facility (FCPF), UN-REDD and other bilateral approaches were initiated (Miles 2010).

The FCPF and UN-REDD programs are major funding schemes to support pilot REDD+ projects. Under the FCPF programme, 44 developing countries encompassing 17 in Africa, 16 in Latin America, and 11 in the Asia-Pacific region are carrying out readiness activities for REDD+. The FCPF has two separate but complementary funding mechanisms, namely the Readiness Fund (preparation for

REDD+) and the Carbon Fund (to achieve its strategic objectives). The FCPF has raised about US\$648 million including US\$258 million for the Readiness Fund and US\$390 million for the Carbon Fund (FCPF/World-Bank 2013). Similarly, the UN-REDD programme supports 53 partner countries across Africa, Asia-Pacific, Latin America and the Caribbean directly, through funding, or indirectly, by helping them to participate in meetings and discussion forums. By June 2014, total funding for these two streams was US\$195.7 million (UN-REDD 2014). In addition, there are several projects operating under bilateral funding schemes including USAID and NORAD. These projects focus on raising awareness of REDD+ and climate change issues and the REDD+ pilot projects are designing and implementing awareness raising and capacity building of local communities as important priority activities.

By October 2009, there were more than 79 REDD+ readiness projects and over 100 REDD+ demonstration projects operating in 64 developing countries (Cerbu *et al.* 2011). These numbers should continue to increase given continuing focus in the global negotiation process and funding support from donors. Readiness projects include policy formation, establishment of REL, institution building and organisational capacity building in developing nations. Demonstration project activities include designing projects, identifying the drivers responsible for deforestation and forest degradation and appropriate activities to address those drivers. At the pilot stage, most of the REDD+ projects were initiated in those countries which had higher deforestation rates, no significant socio-political issues and better governance. Consequently, fewer REDD+ projects are to be found in Africa while more have been instigated in Brazil, Indonesia and Cambodia (Cerbu *et al.* 2011).

d) **REDD+** in community based forest management

With decades of experience, local communities have developed a system for planning, managing and harvesting forest resources and receiving socio-economic benefits in CF (McDermott & Schreckenberg 2009; MFSC 2013). Many studies have highlighted a possible linkage between CF and REDD+ mechanism and suggested that the learnings and existing institutional set up of CFs be used in the upcoming REDD+ mechanism (Agrawal & Angelsen 2009; Hayes & Persha 2010; Fry 2011; Larson 2011; Alcorn 2014).

Since CF can contribute to mitigating climate change (Maraseni *et al.* 2005), these forests will be important in REDD+ incentive mechanisms in the future. While they were mainly within country, there were Payment for Ecosystem Services (PES) mechanisms functioning in Latin America before the UNFCCC agreement on 2007 (Sills *et al.* 2009; Alcorn 2014). These local PES mechanisms involved local communities in the maintenance of multiple services through the reduction of deforestation and forest degradation. These mechanisms was found contextual (both beneficial as well as not beneficial) from a pro-poor perspective (Grieg-Gran *et al.* 2005). As previously noted, while various pilot projects were initiated after the REDD+ concepts were agreed in the UNFCCC (2007), there is limited literature available on the lessons derived from the pilot REDD+ mechanism in CFs. There are several unanswered questions, particularly: Are all CFs contributing similarly to CO_2 emissions reduction? What is the potential carbon stock increment in existing CFs? What key factors influence carbon stock changes in CFs? How much additional community effort (in term of monetary value) is used for REDD+ in CFs? Answers to these questions may help to design an effective REDD+ projects in CFs.

After inception of the REDD+ concept in global climate change discussions, the government of Nepal facilitated national agencies to initiate pilot activities and a collaborative project was initiated in three watersheds covering S. robusta mixed broadleaf forests at lower altitude areas to Rhododendron-Quercus forest at high altitudes. This pilot project covers 105 CFs and is jointly implemented by the International Centre for Integrated Mountain Development (ICIMOD), Asia Network for Sustainable Agriculture and Bioresources (ANSAB) and Federation of Community Forestry Users Nepal (FECOFUN) with financial support from 2009-2013 from the Norwegian Agency for Development Cooperation (NORAD). The project implemented various activities including alternative energy promotion (biogas, improved cooking stock), training for appropriate silviculture operation, fire control activities and plantation activities to foster biomass carbon stock (ANSAB/ICIMOD/FECOFUN 2013). These projects activities potentially altered community forest management practices and carbon stock in forests. While the carbon stock in forests has changed (ANSAB/ICIMOD/FECOFUN 2013), the drivers of the change, including the role of dominant species in the differences observed, are unclear. Understanding this will help in raising awareness about potential carbon stock changes and enable future REDD+ projects and policies to focus on the key factors influencing carbon stock change. Moreover, there is little clarity about the potential for carbon growth in CFs and trade-offs of communities' present benefits for REDD+ carbon benefits. For REDD+ projects to make a long term contribution to carbon sequestration and community benefit, local communities may need to receive proper compensation for their changed behavior and additional efforts made for the REDD+ mechanism. Local forest dependent people are poor, therefore the project should emphasis the need to return higher benefits to them (Grieg-Gran et al. 2005; Alcorn 2014).

2.3. Carbon pools and carbon pool measurement in community forests

2.3.1. Carbon pools in CF

Vegetation removes carbon dioxide from the atmosphere through the natural process of photosynthesis and stores the carbon (C) in leaves, branches, stems, bark and roots (Johnson & Coburn 2010). Living biomass in the forests sequesters atmospheric carbon dioxide (CO_2) while dead wood, litter and soil store carbon. Carbon pools in forests are grouped into three major types, namely: (1) biomass (above-ground and below-ground biomass); (2) dead organic matter (dead wood, fallen twigs and leaf litter); and (3) soil organic matter (MacDicken 1997; Houghton 2005; IPCC 2006). All three components of the carbon pool can be considered for carbon accounting in CFs and the extent to which community activities reduce these pools, thereby affecting carbon stock, can be measured. Most CFUGs collect dead branches and leaf litter (Metz 1994; Maskey *et al.* 2006) from CFs and it can be assumed that communities leave little of this biomass in the forests for decomposition and formation of soil carbon. Therefore, it is unlikely that CF would change soil organic carbon (SOC) in the short time period because the accumulation of dead and dry matter in forests and changes in land use practices are the main causes for SOC change (Laurance *et al.* 1999; Guo & Gifford 2002). Therefore it can be assumed that there is a constant or negligible change in SOC in CFs during the three year period of this study.

2.3.2. Carbon pools measurement methods

In REDD+ mechanism, baseline scenarios of carbon in forests without REDD+ initiatives and possible carbon stock added in forests after the REDD+ project activities is estimated (Eckert *et al.* 2011). Therefore estimation of REL is key step for carbon projects which is defined by analysing historical deforestation and forest degradation data of the project areas or a country (Herold&Skutsch 2011; Pelletier *et al.* 2011). The performance of the REDD+ project is estimated from a verifiable change in emissions over the REL (Pelletier *et al.* 2011).

Two types of approaches namely gain-loss and stock different are suggested by IPCC (2006) to estimate carbon stock change in projects. Gain-loss methods estimate annual change in biomass from gain or increment in biomass stock in the forests (i.e. growth) and loss or removal of biomass from the forests (i.e. extraction of biomass). Similarly, stock difference method estimates changes in biomass between two definite points of measurements. The biomass gain-loss method is applicable for all tiers (i.e. tire 1 method- is more general and use global or continental data; tire 2 method - use country specific data and tire 3 method- use forest specific data) while stock-difference method is more suited to tiers 2 and 3. This is because the stock-difference method provides more reliable estimates in case of large increases or decreases of biomass and accurate forest inventories are carried out (IPCC 2006).

Stock difference methods can be better in forest carbon monitoring which gives changes between stock at baseline and after the REDD+ project activities. To estimate carbon stock in forests at certain point, most of the studies have used either remote sensing (Steininger 2000; Dong *et al.* 2003; Basuki *et al.* 2009; Eva *et al.* 2010) or field inventory (Haripriya 2000; Dieter & Elsasser 2002; Mani & Parthasarathy 2009; Alves *et al.* 2010; Sharma *et al.* 2010). However, field measurement involving local communities is taken as a better approach to estimate biomass carbon in CFs at project level (Karky & Skutsch 2010). This is because, communities may trust the data generated by field inventory with their participation and local people involvements reduces costs of inventory (Karky & Skutsch 2010). In field inventory, measurement of various carbon pools i.e. litter and herbs, saplings and trees is done by establishing random plots in the forests (MacDicken 1997; Subedi *et al.* 2010).

2.4. Factors affecting carbon stock changes in CF

Two terminologies, deforestation and forest degradation, are commonly used in discussing forestry sector emissions. "Deforestation" is the conversion of forest to another land use or long-term reduction of tree canopy cover below a 10% threshold (FAO 2001). The Marrakesh Accord further defined deforestation thresholds between 10 to 30% of forest cover (Boyd & Schipper 2002). Similarly, "forest degradation" is defined as a process that leads to a temporary or permanent deterioration in stocking rates (Grainger 1993; Corbera *et al.* 2010). This can occur

in forests while overall forest canopy cover is still above deforestation threshold levels (10–30%).

Either proximate (visible) or underlying are responsible for deforestation and forest degradation in developing countries (Lambin *et al.* 2001; Geist & Lambin 2002). Most of these factors are site specific (Scrieciu 2007) and might be different according to countries and locations within country (Bhattarai & Hammig 2001). Identifying possible drivers for a particular country and specific forest management systems can facilitate the design, planning and effective implementation of REDD+ projects.

Although all CFs in a country follow the same guiding principles and policy provisions, there is also a possibility that a range of forest management outcomes will be achieved across the individual CFs (Thompson et al. 1990; Baral & Subedi 2000; Pokharel 2011). There might be various factors driving these differences which are important to know for designing and implementing REDD+ projects in CFs. As a key objective of REDD+ is to generate additional carbon benefits through providing incentives (Angelsen 2009), CFUGs may change the subsistence use aims of CFs, influencing CFs to change their use practices and forest management priorities to benefit from the REDD+ mechanism. Although previous studies identify different success factors for CFs, including: (1) biophysical and/or resources; (2) social-economic; and, (3) socio-political (Ostrom 1990; Gibson et al. 2000; Agrawal & Ostrom 2001; Varughese & Ostrom 2001; Foley et al. 2005; Agrawal & Angelsen 2009), there is limited knowledge about success factors for REDD+. The factors influencing the success of CFs may or may not work in the same way in REDD+ project in CFs. However, those factors could provide a basis on which to assess ongoing REDD+ projects. Similarly, drivers identified for deforestation and forest degradation in government managed forests with different socio-political contexts may be different than those operating in CFs. For example, the main drivers of deforestation and forest degradation in tropical countries are reported to be infrastructure development, agriculture expansion, wood extraction, forest fire, demographic, forest quantity harvest, income, proximity, policy and institutional differences (Cochrane & Schulze 1999; Pahari & Murai 1999; Bahuguna 2000; Blackmore & Vitousek 2000; Roy 2003; Grau et al. 2005; Balch et al. 2011; IPCC 2014); these may not be similarly important for CFs in Nepal. The previous studies cover global or regional levels, while project specific studies are limited, particularly in relation to CFs. However, these studies can provide a useful knowledge base for the design of appropriate activities for REDD+ projects which address these factors and increase carbon stock in forests. Possible factors are discussed below.

2.4.1. Possible biophysical factors

Brief descriptions of biophysical factors which might play a role in carbon enhancement in CF are given below.

a. <u>Canopy cover differences</u>: Forest canopy is one of key determinants that affects the growth and survival of plants (Smith *et al.* 1997; D'Amato *et al.* 2011) and so can make significant difference to the quantity of carbon stock in CFs. Canopy cover creates different micro-climates in forests by obstructing the penetration of light and precipitation to the forest floor thereby influencing tree growth (Jackson

1994; Khanna 2004). If a species is a light demander, it may not grow well under the closed canopy of a forest while the opposite is the case for shade bearer species (Khanna 2004). In general, very dense forests may not allow all trees to reach their potential stem diameter due to high competition for space, minerals and food (Smith *et al.* 1997; D'Amato *et al.* 2011). In the case of CFs, user communities carry out different forest management activities which may create different crown densities, thereby affecting the quantity of biomass carbon.

- b. <u>Altitudinal variation</u>: Altitude affects the distribution of different vegetation types in CFs (Jackson 1994; Vetaas & Grytnes 2002). However, there are contradictory findings regarding biomass carbon quantity variations along altitude gradients. A study in Brazil found increasing biomass with increasing altitude in tropical moist forests (Alves *et al.* 2010) while biomass was higher in lower altitude than higher altitude forests in the Himalaya region (Singh *et al.* 1994). However, forest products such as fuel wood use practices in CFs is different with altitude in Nepal (FAO 1999), which may affect biomass carbon stocks in forests. It would helpful in the design of REDD+ projects and benefit sharing mechanisms if possible relationship between carbon stock and altitudes in CFs were known.
- c. <u>Climatic patterns</u>: Intensity of rainfall, drought and temperature affect seed production, germination and establishment of seedlings in forests (Everham III *et al.* 1996). More rainfall, short duration of drought and appropriate temperatures help to germinate more seedlings and support their establishment and physiological activity (Yang *et al.* 2006). Weather affects photosynthesis processes and therefore biomass enhancement in forests in their natural state. However, different vegetation types are adapted to different climatic patterns (Jackson 1994). In the short term, climatic patterns may not vary greatly and climate change may have negligible effects on species growth.
- d. <u>Dominant tree species in forests</u>: Many studies carried out in government managed forests and protected areas have found that biomass carbon stocks in forests are affected by dominant vegetation types (Lodhiyal & Lodhiyal 2003; Rossi *et al.* 2009; Tan *et al.* 2011). However, given the different management regime, these findings need to be verified for CFs. There are various dominant vegetation type forests found in Nepal and these forests are also under community management (GON 2002). These forests differ in growth characteristics and wood density, both of which affect biomass carbon (Kauppi *et al.* 1992; Jandl *et al.* 2007). In global classification systems, these forests can be classified into tropical, sub-tropical, lower temperate, upper temperate, subalpine and alpine forests (Schmitt *et al.* 2009). Brief descriptions of the forest types in Nepal are given below (Jackson 1994):
 - Shorea robusta forests: This forest type is close to the tropical forests of global classification systems (Woo *et al.* 1991). *S. robusta* is a large tree up to 45 m height. This is a slow growing tree species; however, it can grow quickly at the younger stages under favourable conditions. *S. robusta* forest is dominant in Terai low land, including the Siwalik hills, and is distributed up to 1,500 m asl (metres above sea level). It is deciduous (leaves fall for a short time) and a light demander. Seedlings of *S. robusta* are susceptible to dieback if there is frequent frost, drought and/or fire. This is a popular timber tree and

highly preferred for building houses in the country (Troup 1921; Jackson 1994).

- Mixed broadleaf forests: Similar to *S. robusta* forests, mixed broadleaf forests are close to the tropical forests in global classification. This forest type includes species such as *Lagerstroemia parviflora* (Botdhairo), *Terminalia spp, Anogeissus latifolia* (Bajhi), *Dalbergia sissoo* and *Syzygium cumini* (Jamun). *L. parviflora* is a large, deciduous, light-demanding tree which is distributed from Terai and Bhabar zones in Terai to about 1,200 m altitude. It is not affected by browsing by cattle and able to recover quickly after fire. As with *S. robusta, L. parviflora* is a valuable timber tree. *S. cumini* (Jamun) is a fairly fast growing species which is found in wide ranges from Terai to 1,600m. *A. latifolia* (Bajhi) occurs from Terai to 1,700 m elevation, has a moderate growth rate and is unpalatable to livestock and not damaged by browsing. *Dalbergia sissoo* is a fast growing strong light demanding tree found up to 1,100 m asl (Troup 1921; Jackson 1994).
- Schima-Castanopsis forests: These forests are more similar to the subtropical category in global classifications occurring in areas with a comparatively colder climate than tropical forests but with minimum temperature greater than those experienced in temperate climates (Woo et al. 1991). Schima wallichi is a large tree with a moderately shade-tolerant character which is generally found between 900 and 2,000 m asl together with Castanopsis species. There are three Castanopsis species found in different distribution ranges; Castanopsis indica, between 1,200 and 2,900 m asl, Castanopsis hystrix between 1,000 and 2,500 m asl and Castanopsis tribuloides between 450 and 2,300 m asl. These species are palatable and used as fodder for livestock (Boojh & Ramakrishnan 1983; Jackson 1994).
- Pine forests: Similar to *Schima-Castanopsis* forest, pine forest is closer to the sub-tropical forest category. *Pinus roxburghii* (occurring from 900–1,950 m asl), *Pinus walichiana* (from 1,800–3,600 m asl), planted *Pinus patula* (from 1,500–2,500 m asl) and *Tsuga Dumosa* (from 2,100–3,600 m asl) are the major species found in this category. Pine trees are not palatable to cattle and are used for timber but they are soft and have lower market values than *S. robusta* (Jackson 1994).
- *Rhododendron-Quercus* forests: This forest type is closer to temperate forests and is found in areas experiencing a cold climate and covered by snow at certain times of the year. This forest is composed of three different tree types, *Rhododendron, Quercus* and *Lyonia* species. *Rhododendron* species grow from 1,500 to 3,300 m asl. They reach 15 m in height and the flower is used to make a pickle. *Quercus* species are generally found between 1,700 and 3,800 m asl and are used as fodder. *Lyonia ovalifolia* is a deciduous shrub and occurs from 1,300 to 3,300 m asl (Jackson 1994; Singh *et al.* 2003)
- e. <u>Age of the forest stands</u>: As vegetation exhibits a sigmoidal growth curve (Weiner & Thomas 2001), there is increasing carbon stock found in forests with age (Lodhiyal & Lodhiyal 2003). However, communities manage and use forest

resources in CF systems and CFs could be non-uniform by age category. Therefore both carbon stock and the age of forest stands in CFs could differ in terms of the natural sigmoidal growth pattern. In addition, this relationship has not been verified for all vegetation types. On the other hand, young forests have higher growth rates and capacity to sequester biomass carbon (Pregitzer & Euskirchen 2004; Bradford & Kastendick 2010) which may affect carbon benefits in a CF. Therefore, the age of forest stands could be one factor to be considered while designing and implementing the REDD+ project in CFs.

2.4.2. Possible socio-economic factors

Various socio-economic factors which have been shown to influence carbon enhancement in forests may or may not be applicable in CFs. Possible factors include:

- a. <u>Size of forests (Per capita forests available in individual CF)</u>: The size of CFs in Nepal range from less than 1 ha to over 2,500 ha (CFD/DoF 2013). Per capita forest allocation is listed as a success factor for CF (Agrawal & Angelsen 2009) and may influence carbon stock changes in CFs. Previous studies highlighted that the larger the per capita forest area available, the higher the possibility of success of CF (Agrawal & Angelsen 2009). However, this may not be true in all cases as communities may take greater care of a forest if it is smaller but management is linked to incentives. It has been reported that communities are conserving forests where there is a link with income (Subedi 2006). Clarity about the effects of per capita forest area on carbon stock and change in carbon stock in CFs is lacking in the existing literature.
- b. Population growth: The relationship between the status of forests and population growth accords with Malthus' theory which makes three assumptions: (1) that food is a limiting factor for human populations; (2) that populations grow geometrically; and (3) that the land required to feed a growing population increases in a linear way (Seidl & Tisdell 1999). A growing population demands food which cannot be sufficiently derived from existing agricultural land, and this need is met by expansion of forests converted to arable land. However, the Malthus theory was developed in the 18th and 19th century and the present context is different. Now population growth control techniques are adopted all over the world (Robinson & Ross 2007) and appropriate technologies to increase agricultural production have been developed and applied (Lal 2006) that may be changing the assumptions on which Malthus theory is based. Ives and Messerli (1989) identified population pressure is main cause of deforestation in the Hill areas of Nepal. This was found in other literatures as well (Armenteras et al. 2006, Pahari & Murai 1999) but the relationship between population growth and deforestation trends may be weaker now where social, cultural, political and economic aspects of countries play a major role (Mather & Needle 2000). Therefore, we can assume that the relationship between people, agriculture and forests in CFs are context specific and this needs to be considered while designing REDD+ projects.
- c. <u>Caste variations</u>: In Nepal, there are 125 different caste groups (CBS 2011b) which has created a social hierarchy within villages. In the social hierarchy, these

castes come under Brahmin, Chhetri, Ethnic Groups and Dalit (Gurung 2003; Lynn *et al.* 2008), each of which may have different cultural and forest product needs. These diverse people are living in a village and organizing as a CFUG for the management of a forest (CFD/DoF 2009). Earlier research findings indicate that highly heterogeneous groups cannot make consensus decisions and implement those decisions (Blair 1996) thereby affecting forest management outcomes. However, this finding was opposed by Varughese and Ostrom (2001) who found no relationship between heterogeneity and forest improvement in CFs. Since carbon stock is a new product with likely benefits and opportunities, CFUGs may have changed existing practices including decision making processes. However, there are limited studies regarding the possible effect of community heterogeneity on carbon stock changes in CFs.

- d. <u>Accessibility of forests</u>: There is no clarity about whether access to markets (distance from forest to road) and access to forests (distance from settlement to forests) affect biomass cabon in CFs. Close location (i.e. high accessibility to markets from road connections) increases deforestation (Angelsen & Kaimowitz 1999). This is because market demands for timber encourage local people to harvest more from forests. However, there is the opposite finding also that access to market helps to increase biomass carbon in forests by providing jobs to villagers and facilitating the adoption of alternative energy sources thereby reducing pressure on forests (Chhatre & Agrawal 2009). Similarly, settlement proximity to forests can have both positive and negative effects on biomass carbon. People can harvest more if the forest is located close to their house (Laurance *et al.* 2006), reducing biomass carbon. But this is less likely in the case of CFs where local communities aim to improve forests with shared goals and collective action (Ostrom 1990 and 1999) that can be effectively applied in nearby forests.
- e. Forest management practices: Management practices can help to differ carbon stocks in forests (Dixon 1997; Foley *et al.* 2005; Hoover & Stout 2007; Harmon *et al.* 2009). Silvicultural operations (i.e. thinning, pruning, weeding, and cleaning) and protecting forests from disturbances (i.e. illegal harvesting, grazing and forest fire) are major forest management practices which may affect biomass growth (Thompson *et al.*1990) and also CO_2 emission/sequestration rates. Thinning frequencies in forest may help to increase the carbon sequestration rate because forest stands with regular thinning have higher growing stock compared to stands which are clear-cut on short rotations (Khanna 2004) and undisturbed old growth forests (Thornley & Cannell 2000; Harmon *et al.*2009). However, there is evidence for reduced aboveground carbon in thinned stands compared to un-thinned stands (Chiang *et al.* 2008; Campbell *et al.* 2009). Despite these insights, empirical evaluations of CF practices and outcomes are lacking (Hoover & Stout 2007) therefore forest management practice change for the REDD+ and carbon benefits needs to be evaluated.
- f. <u>Recurrence of forest fire (anthropogenic fire)</u>: Frequency of forest fire is considered an important factor that can affect forest carbon stock in CFs. Forest fire releases not only stored carbon from forests but also damages germination capacity and increases the mortality of trees and seedling (Arthur *et al.* 1998; Barlow *et al.* 2003). Moreover, forest fire damages the overall productivity of

forests (Mehta 1996). A study conducted in east Amazon found that the unburned areas had stored 256 Mg/ha of living biomass whereas only 24 Mg/ha was stored in burned areas (Cochrane & Schulze 1999). However, if the fire is controlled and prescribed, it can be helpful in improving forests (D'Antonio 2000). Therefore, fire might play a role in carbon stock changes in CF; therefore it could be important to know about the incidence of fire and possible effects on carbon stocks in CFs.

- g. <u>Extraction of forest products</u>: Illegal harvesting of trees for timber is considered a main driver of deforestation (Geist & Lambin 2002). Existing studies related to forests products extraction and deforestation cover mostly government managed tropical forests (Barbier & Rauscher 1994; Barbier *et al.* 1995). In CFs where local communities are guarding against illegal harvesting this may not be a problem. This is because communities are managing forests and operating management activities according to their own decisions (Gilmour & Fisher 1991; Schlager & Ostrom 1992; Acharya & Gentle 2006). In CFs, there is provision for sustainable harvesting of forest products by keeping the harvested quantity below the level of incremental increase in the forests (CFD/DoF 2009). As minimizing the harvesting of forest products is critical to maximizing the aboveground carbon stock (Harmon *et al.* 2009; Nunery & Keeton 2010), knowledge about the harvesting behaviours of communities and the quantity of extraction can be important in the design of REDD+ projects in CFs.
- h. <u>Grazing</u>: Grazing is also an important factor which plays a role in forest carbon sequestration (Vinton *et al.* 1993) particularly by reducing biomass in the forest (Blackmore & Vitousek 2000). A study conducted in natural pasture land in Africa concluded that grazing land subject to heavy grazing pressure has significantly reduced vegetative cover and biomass yield, particularly on steep slopes (Mwendera *et al.* 1997). However, the opposite finding has also been reported: that controlled grazing aids forest improvement (Smith *et al.* 1997; D'Amato *et al.* 2011). In CFs, there may be different grazing provisions and practices which may affect carbon stock. Possible relationships between grazing practices and carbon stock need to be analysed when designing REDD+ projects.
- i. <u>Average household level private agriculture land holding in CFUGs</u>: Forests are the main source of fuel wood and fodder in rural areas (Adhikari *et al.* 2004). The collection by local communities of dry biomass from the forests for fuel wood is common practice in Nepal where 77% of household energy demand is fulfilled from forests (WECS 2010). However, while people collect fuel wood from nearby forests (Metz 1990), it is possible to grow trees on private land to reduce the pressure on CFs. Therefore, the average agriculture land holding size and alternative sources of fodder, grasses and timber may play a role in increasing carbon stocks.
- j. <u>Average household level livestock size in CFUGs</u>: The relationship between livestock numbers in households and forest conservation status was found to be both positive and negative in different literatures. A study found that deforestation has increased the scarcity of fodder and grass supply from forests and therefore communities have reduced numbers of livestock (Barraclough & Ghimire 1995). However, another study found that the conservation of forests has

reduced fodder and grass supply and reduced community livestock numbers (Dhakal *et al.* 2005). Given these contradictory findings, analysis of livestock numbers and effects on carbon stock in CFs is needed.

k. <u>Alternative energy use in CFUGs</u>: Wood is a major source of global energy supply. Global production and use of round wood was about 3300 (10⁶) m³ in 1999 and 55% of wood is used for fuel (Parikka 2004). Fuel wood use is comparatively high in developing countries i.e total 90% of global fuel wood consumption (Parikka 2004). In Nepal, over 80% of the country's energy is derived from fuel wood (WHO 2006). Biogas (cattle dung is used) and liquefied petroleum gas (LPG) are the main alternative household energy sources used at the household level in Nepal (WECS 2010). To reduce the use of fuel wood, improved cooking stoves (ICS) have also been designed and promoted in communities because traditional cooking stoves consume 30–40% more firewood (Bhattacharya *et al.* 2000) than ICSs (Dhakal & Raut 2010). Theoretically, the use of alternative energy sources should increase carbon stock in CFs but this need to be analysed.

2.4.3. Possible political factors

- a. <u>International and national policy</u>: International and national policy mechanisms are the key to changing carbon stocks in forests. Decisions of UNFCCC Conference of Parties (CoP) sessions influence the number of forest carbon projects in developing countries. Based on conference decisions regarding the incentive provisions of REDD+, developing countries have proposed different strategies and policies for biomass carbon increments in forests (FCPF/World-Bank 2013). As the national REDD+ strategy is common guiding document, it could have similar provision for a particular forest management system in the country such that the policy itself would not differentiate between carbon stocks of CFs. Therefore the policy factor was not included in analysis of carbon stock changes in CFs.
- b. <u>Tenure issue of forests:</u> Clarity on the tenure of forests is important in the REDD+ mechanism; therefore clarifying tenure must be an initial step in the REDD+ project. If forests are under the ownership of multiple stakeholders, they may have less effective management than those with clear community ownership (Duchelle *et al.* 2014; Dunlop 2009). Besides tenure clarity, economic benefit to local communities is also important in making the REDD+ effective (Resosudarmo *et al.* 2014). However, CF in Nepal has clear policy provision that ownership of forest products lies with the community while ownership of land lies with government (GoN 1993). Carbon rights are associated with land tenure; hence it is important to clarify this issue to ensure a fair mechanism when a CF enters the performance–based market (Karsenty *et al.* 2014).
- c. <u>Governance mechanism of CFUG</u>: Good policy, appropriate institutional structure and deliberative decision making processes are important aspects in forestry governance (Cadman 2009). There are clear policy provisions about the structure and function of CFUGs (CFD/DoF 2009) and common to all CFUGs of the country. However, the decision making process in a CFUG may differ in practice. Therefore the decision making process of CF may affect carbon quantities in forests under the REDD+ project.

2.5. Economics of REDD+ in CFs

GHGs are considered as a commodity with CO_2 equivalent the unit used in carbon trading mechanisms (Hepburn 2007; Laurance 2008). The REDD+ project is focused on changing the existing behaviour of relevant actors to reduce deforestation and forest degradation and preventing future threats of deforestation and forest degradation (Lin *et al.* 2012). Through these activities, greater carbon quantities are added in forests compared to the historical trend for carbon stocks (Fry 2011); these quantities are monitored and verified in order to estimate carbon credits and additionality.

Poverty eradication is a millennium development goal (Sachs & McArthur 2005) and poverty reduction has been envisioned as one of the major co-benefits that REDD+ might bring (Vatn & Angelsen 2009). However, concerns about the relationship between communities' sacrificed benefits and REDD+ benefits need to be assessed. Since forest resources are important for the livelihood of local communities, the REDD+ mechanism should consider their needs and ensure that the outcomes are profitable for them. Poor communities are conserving forests with an expectation of livelihood support in Nepal (Subedi 2006); therefore, the contribution of the REDD+ mechanism to poverty reduction co-benefits is important, together with any biodiversity, water and recreation benefits in CFs. Moreover, there could be an expectation that REDD+ will bring additional benefits to local communities which will be higher than communities' associated foregone costs. However, there is no clarity in this regard. To assess the REDD+ project in CFs, Cost Benefit Analysis (CBA) could be done by establishing a business-as-usual case (the without REDD+ project situation) and assessing changes with a REDD+ project by disaggregating the individual activity value (Pearce et al. 2006).

The Master Plan for Forestry Sector (1988) and tenth five year plan 2002–2007 of Nepal (NPC/GON 2002) are focusing on an economic development agenda and the forestry sector is mentioned as a possible sector to consider. In addition to products based income from forests, the PES is also a possible option to incentivise the local poor for their forest management contributions (Subedi & Singh 2008; Pandey *et al.* 2011). Each CFUG can put in place different conservation efforts by following their economic development agenda (Adhikari & Nagata 2004; Maskey *et al.* 2006). Since the REDD+ mechanism needs to safeguard the needs and good practices of communities, analysing changes in costs and benefits for REDD+ projects is important. Therefore, all direct costs sacrificed due to changed forest management and use practices are equally important. However, the optional value of land in CFs does not need to be considered in the Nepalease context because CF lands are owned by the government and communities are not permitted to change the land use system (GON 1993). In the following sections, a review of possible forest product benefits and contributions of communities to REDD+ is presented.

2.5.1. Costs due to sacrificed benefits of communties for REDD+ mechanism

Though protecting forests by providing legally ensured ownership and supplying forests products for the subsistence household needs of local communities was a main aim of CF in its inception phases (Gilmour *et al.* 2004; Sunam 2011), it has expanded towards the economic development of local people from sustainable use of forest resources through to enterprise approaches (Subedi 2006; Adhikari *et al.* 2007). Local communities are extracting various products and generating income in CFs (Subedi 2006; Adhikari *et al.* 2007). However, communities may need to change such benefit practices to deliver REDD+ carbon benefits that can add costs of communities. The following are possible key benefits obtained in CFs which may be changed for REDD+ projects:

- Forest products benefits (Yadav *et al.* 2003; Adhikari & Nagata 2004; Cooke *et al.* 2008; Davidar *et al.* 2008;):
 - Extraction of timber: Timber is one of main building materials collected from forests. In a CF, the executive committee decides to give a certain amount of timber to its members for their own use. CFUGs can also sell the timber if they have excess quantity of the annual allowable timber harvest in a year. As an institution, a CFUG should maintain records of annual timber harvest from a forest (CFD/DoF 2002).
 - Extraction of fire wood: As noted, fire wood is a very important forest resource in rural areas. Rural communities extract wood and trees to fulfil household energy demand. Each CFUG has different firewood demand and they are extracting different quantities from forests (Agarwal 2001; Adhikari *et al.* 2007).
 - Extraction of fodder and grasses: CFUGs have different provisions for collecting fodder and grasses in CF to feed their livestock. Since each individual household has only a small agriculture landholding (the national average is <0.5 ha per household) (CBS 2011a), they may not able to supply enough fodder and grasses from their own land and therefore they depend on CF.
 - Extraction of litter: Litter is fallen leaf and twigs which is collected from the CFs and use as bedding material for livestock. Decomposed litter, after it is used as bedding material, is used as agricultural manure (Adhikari & Nagata 2004). As litter is dried and fallen leaves, most CFUGs allow this to be collected in forests.
 - Extraction of non-timber forest products (NTFPs): NTFPs are also a source of income for rural communities. People collect various parts such as leaves, fruits, roots and stems of NTFP plants from forests (Edwards 1996). Usually, NTFPs collected from CFs are sold and the annual sale of each CF is recorded because financial reporting is made compulsory by CF guidelines (CFD/DoF 2009).

- Grazing livestock (Buffalo, Cow, **Buffalo** and Goat) 0 in forests: Livestock is one of the important income generating options of rural communities particularly in agriculture based economies (Dhakal et al. 2005; Das & Shivakoti 2006). Seasonal grazing of livestock in the forest occurs in CFs (Dev et al. 2003) but if grazing pressure is heavy (more than the carrying capacity of the forests) it may degrade forests (Blackmore & Vitousek 2000). Most CFUGs are gradually reducing grazing intensity compared to earlier levels (Acharya 2002). This might be changed further for REDD+ which could increase the sacrificed benefits of communities.
- Social benefits: The REDD+ project may help to build greater social capital than CF particularly in terms of institution building, networking and linkage, and awareness raising (Kanel & Kandel 2004; Pokharel 2011; ANSAB/ICIMOD/FECOFUN 2013). As the REDD+ mechanism is new, it may be necessary to develop new institutional arrangements at various levels from local to international to facilitate the incentive mechanism, improve connections with various relevant institutions to explore opportunities and bring new knowledge which will be helpful to individuals and their society. However, social benefits may not provide direct economic incentives to poor people therefore it may not motivate poor communities to sacrifice existing forest product related benefits.
- Carbon benefits: Carbon trade is evolving under the global climate change policy process after the Kyoto protocol (Kyoto-Protocol 1997; Woerdman 2000). There is the possibility of generating additional income from carbon trade for CFUGs because local communities can generate additional carbon stock in their forests (Dev *et al.* 2003; Maraseni *et al.* 2005). Carbon credits generated in CFs may get premium prices in markets in future if global communities recognise that these forests have maintained co-benefits including biodiversity, the water cycle and the traditional practices of local communities in addition to carbon sequestration.
- Climate change adaptation: Forests provide food and fibre to surrounding poor communities (Wunder 2001; Sunderlin *et al.* 2005). If people cannot grow sufficient agricultural crops due to climatic conditions (e.g. prolonged drought, flooding) or other causes (e.g. war), they go and collect food supplements from forests (Falconer 1990). CFs can provide food supplements during times of food scarcity but to do so the diversity of species in forests must be maintained (Shrestha & Dhillion 2006). If there has been a reduction in the diversity of plants to stands dominated by only a few species with high biomass stock, a forest may not provide adequate food supplements to poor people and enable them to survive during adverse conditions.

2.5.2. Costs due to added efforts of communties in forest management

Communities are making efforts to conserve and improve forests in CFs through two means: 1) decision making processes (i.e. General Assembly (GA) and meetings), and 2) forest management (i.e. silviculture operation, plantation, fire control and guarding against illegal activities) (Acharya 2004; CFD/DoF 2009). The annual total value of these contributions can differ between CFs. Average annual changes in costs from before to after REDD+ project activities indicate total foregone or sacrificed benefits to communities. As CFUGs are operating following activities in CF practices (CFD/DoF 2009), they may have changed existing practices that can add to the sacrificed benefits due to REDD+:

- Executive committee meeting (number)
- Plantation
- Forest fire control
- Measuring & monitoring forest-C stocks
- Silviculture operation
- Harvesting of forest products (timber and NTFP)
- Guarding

2.6. Review of research methods related to CF studies

In order to understand the socio-economic aspects of CFUGs, forest management and forest product use practices, and factors associated with forest management, people-related data are required. In previous studies, social data are collected by random sampling methods in which issue-specific data is collected by using household surveys, focused group discussions and key informant survey tools (Ostrom 1990; Agrawal & Ostrom 2001; Varughese & Ostrom 2001; Adhikari et al. 2004; Maskey et al. 2006; Agarwal 2009). Household surveys through face to face interviews are considered the better option because the interviewer can explain questions if interviewees do not understand and bring the facts. This is effective in the case of Nepal where adult men usually go away from their village in search of paid work (Seddon et al. 1998) and the individuals who stay at home in rural areas are comparatively illiterate. Focus group discussion is another data collection tool used in social research where the perceptions of communities are collected from representative members of the different community groups (Pandit & Bevilacqua 2011). The key informant survey method is also used in social research (Marshall 1996) to gain deeper understanding of issues and to cross-check with other research approaches. Key individuals who work on issues or are associated with the issues are selected as informants and asked their opinions.

All these three methods are valid approaches when conducting social research related to the REDD+ in CFs. Estimation of cost and benefits for the REDD+, identification of key possible factors responsible for carbon increments in REDD+ CFs and exploring the perceptions of local people are possible by using these tools and methods. In many cases, mixed methods (i.e. questionnaire survey and/or FGD and/or key informant survey with checklist) are used to get complete data with cross-verification (Johnson & Turner 2003). Careful development of the data collection

tools (questionnaire and check list) to ensure all possible variables are covered and can be analysed in the social research is important (Bernard 2000). Detailed data collection methods and analysis used in this study are discussed in the methodology chapter.

2.7. Monitoring methods of the REDD+ activities

Differences in carbon stock in forests with REDD+ project activities are estimated in order to understand the project performance (IPCC 2006). Remote sensing data such as SPOT satellite data and Landsat imagery can be useful to estimate historical rates of deforestation and estimate the carbon stock changes due to reduction in deforestation rates (Eckert *et al.* 2011; Hett *et al.* 2012). In order to estimate forest degradation, high resolution satellite imagery, ground-based inventory and well developed statistical models with relationship between the reflectance pattern of images and above ground biomass are required (Eckert *et al.* 2011; Sloan & Pelletier 2012). Although biomass analysis from remote sensing data including Landsat, ASTER and SPOT is often used, LiDAR has been increasingly employed in recent years (Gibbs *et al.* 2007; Avitabile *et al.* 2012). This is because LiDAR data gives the vertical structure of the forest on the basis of distance from the sensor to the ground (Næsset *et al.* 2011; van Leeuwen *et al.* 2011) and there are a strong correlation between LiDAR data and above ground biomass (Drake *et al.* 2003; Næsset *et al.* 2011).

Although monitoring of the sustainable management of forests is difficult, it can be done with pre-defined principles, criteria and indicators available at the global level such as those developed by the International Tropical Timber Organisation (ITTO), the Forest Stewardship Council (FSC) and the Montreal Process (McDonald & Lane 2004). These indicators include some aspects of the carbon cycle and ecosystem services, but are not directly relevant to the REDD+ concept. Existing indicators for REDD+ (Næsset *et al.* 2011; van Leeuwen *et al.* 2011; Sloan & Pelletier 2012).

Similar to the other REDD+ activities, monitoring of conservation and carbon enhancement can be done against a baseline identified from historical data and used to compare the carbon increment in forests under REDD+ activities. The baseline is adjusted to take in account the national circumstances of a country mainly by considering development adjustment factors. If a country need massive infrastructure, their deforestation rates could be higher. However, Nepal is already listed as a country with late transition state of deforestation and improving forests status (Hosonuma *et al.* 2012) there is likely to be increased carbon stock rather than a reduction and such possibilities are taken into account .

In CFs, deforestation activity is unlikely because local communities are not allowed to change forests into other land use practices. CFUGs are allocated a patch of forest to improve and are allowed to use a certain proportion of the increased amount of forest products (GoN 1993; CFD/DoF 2009). Detailed estimation methods used in this study are discussed in the methodology Chapter.

2.8. Carbon prices in the existing market place

Carbon offsetting is becoming a popular means of taking action to reduce GHG emissions and addressing global climate change. It neutralises a unit of CO_2e (carbon dioxide equivalent) emitted in one place by avoiding the release of a unit CO_2e elsewhere or sequestering a unit of CO_2e that would have otherwise remained in the atmosphere (Taiyab 2006). There are two market mechanisms, namely compliance market and voluntary market, for carbon offsetting and mitigating global climate change (Hamilton *et al.* 2008; Kollmuss *et al.* 2008; Peters-Stanley & Yin 2013).

Compliance markets are created and regulated by mandatory international, regional, and sub-national carbon reduction schemes such as the Clean Development Mechanism (CDM) regulated by the Kyoto Protocol and the European Union's Emissions Trading Scheme (EU-ETS). These systems provide options to countries (which have accepted limits for GHG emissions under UNFCCC) to meet their emission reduction targets. At present, EU-ETS does not accept credits from land use, land use change or forestry (LULUCF) projects, which are considered temporary in nature, but there are suggestions that such temporary credits may be included in the next phase of the scheme (Graichen 2005; Schlamadinger *et al.* 2005). By comparison, the CDM is a common mechanism which includes afforestation and reforestation projects; this mechanism follows a procedure to ensure the Certified Emissions Reduction (CER) by third party validation and verification (Hamilton *et al.* 2008; Kollmuss *et al.* 2008; Peters-Stanley & Yin 2013). Currently, market prices of CO₂e in CDM are reduced compared to 2008 (Table 2.2).

Voluntary markets are not regulated by mandatory schemes; however, they facilitate companies, government, organisations, organisers of international events and individuals to purchase carbon offsets for their carbon emissions on a voluntary basis. This mechanism provides opportunities for companies and private institutions to neutralise their emissions if this is of interest. The price for emission reductions in the voluntary market is largely driven by buyers' willingness to pay and sellers' willingness to accept payment (Taiyab 2006; Peters-Stanley & Yin 2013). The price of CO_2e on the voluntary market ranges according to projects and buyer interests (Table 2.2).

Table 2.2 Price of carbon dioxide equivalent (CO_2e) in three different carbon marketing mechanisms where LULUCF sector is included

Compliance market	Price at 2008 (US\$/MgCO ₂ e)	Price at 2012 (US\$/MgCO ₂ e)
California climate action	US\$5 to 14 /MgCO ₂ e depending	Auction price is set at
	upon location and guarantee	US\$10/MgCO ₂ e. It offer auction four
	(Capoor & Ambrosi 2009)	times a year at three prices levels: 40,
		45 and 50. With 5% plus inflation
		each year (World-Bank 2013)
CDM	US\$10-23 /MgCO ₂ e (Capoor &	US\$ 0.26 /MgCO ₂ e December
	Ambrosi 2009)	(World-Bank 2013)
Voluntary market	US\$1.20-46.90/MgCO ₂ e; forest	US\$5.9 /MgCO ₂ e (Peters-Stanley &
	management US\$7.7 /MgCO ₂ e	Yin 2013)
	(Hamilton et al. 2009)	

Prices are comparatively higher in voluntary markets than through the CDM. Carbon credits generated from forests do not reach the higher rates of energy projects

because forestry projects have higher uncertainties compared to energy projects regarding emissions reductions (Chomitz 2000). The REDD+ is a new likely mechanism but yet to be developed a market mechanism. However, some pilot carbon projects have been priced: e.g. US\$4–10/MgCO₂e in Uganda (Peskett *et al.* 2011); US\$5/MgCO₂e in Cambodia (Sasaki *et al.* 2013); and in early 2013 REDD+ project received an offer in a range of US\$7 and 8/MgCO₂e for forward REDD+ credits (World-Bank 2013). This shows the possibilities of the REDD+ mechanism on the market. In Nepal, the REDD+ project has distributed payment differently, with weighted payments of 60% based on social aspects and 40% on environmental aspects including 16% for carbon increment and 24% for biomass stock (Maraseni *et al.* 2014). This approach considers socio-economic safeguards while less emphasis is given to the added carbon.

2.9. Review of existing models for forest growth prediction

The quantity of carbon stock in a forest differs according to climatic, ecological and site specific factors (Houghton *et al.* 2001; Zhao & Zhou 2006). Forests sequester additional carbon with biomass growth (MacDicken 1997; IPCC 2006; Ordóñez *et al.* 2008). Therefore growth projections under undisturbed conditions and under community management are important when planning carbon stock enhancement projects in CFs. Projecting the growth of biomass carbon in different situations (undisturbed and community management) may need time series data. However, the use of the most relevant existing model is a better option for projecting the growth of forests if no time series data are available (Landsberg & Waring 1997). Broadly, the empirical forest yield model, ecological gap model, process model, and hybrid model are used to estimate yield in forestry (Monserud 2003).

Among these models, the empirical yield model predicts expected yield under a certain management regime over time from a direct statistical relationship. There are several empirical models developed in Nepal. These include models to predict: the total and merchantable volume and biomass of trees (Sharma & Pukkala 1990); early growth of S. robusta (Acharya & Acharya 2004); tree stem biomass and volume of S. robusta in central Nepal (Laamanen et al. 1995); diameter growth of S. robusta trees (Korhonen et al. 1992); fresh biomass and volume of trees in community managed forests (Tamrakar 2000); stand volume by age of S. robusta dominated forests stand (Rautiainen 1995); biomass estimation for 9 years Pinus roxburghii trees (Applegate et al. 1988a; 1988b); and biomass of 10 years old mix young stand of Pinus roxburghii and broadleaf species (Mohns et al. 1989). Yield models are simple, more efficient and provide better estimates than other forest growth models (Weiskittel et al. 2011). However, they require substantial site specific and species specific timeseries data generated either by repeated inventories or chronosequences to estimate parameters (Monserud 2003). The ecological gap model is an individual tree based model designed to study tree population dynamics in forest patches. This model predicts dbh growth by age based on population dynamics (Risch et al. 2005) and requires information regarding gaps, recruitment, growth and mortality of individual trees in forest patches (Pabst et al. 2008). The gap model is used to analyse disturbances and effects on vegetation types, sizes and stocking rates (Mailly et al. 2000). The process model develops key processes and underlying causes of

productivity including photosynthesis, respiration, nutrient cycles, climate effects and water stress to predict yield (Monserud 2003). For example, the 3-PG (Physiological Principle in Predicting Growth) model is one of the process based models adopted in growth prediction (Landsberg & Waring 1997). On the other hand, a hybrid model is a mixture of both empirical and process based yield models which merge the best features of both (Monserud 2003). The Triplex model is an example of a hybrid model (Peng *et al.* 2002).

Although both the 3-PG model and Triplex models are tested to predict forest yield, mostly the 3-PG model was used for single species plantation forests (Sands 2000; Almeida et al. 2004) and Triplex model was used for mixed vegetation (Zhang et al. 2008). Both these models need various parameters which may not be available for all forest types, particularly in developing countries like Nepal. Required parameters for this model include climatic data (average precipitation, temperature and relative humidity), photosynthetically active solar radiation (atmospheric absorption factor, solar radiation fraction), gross primary production of a forest (maximum canopy conductance, stomata conductance, canopy boundary layer conductance, coefficient for conductance to vapour pressure deficit, minimum temperature for growth of the forests, maximum temperature for growth of the forests, optimum temperature for growth of the forests, nitrogen factor for tree growth), soil carbon and nitrogen (lignin-nitrogen ratio, lignin for leaf, fine root, coarse root, branch, and wood) and soil and water data (depth of soil layer, relative root density for layers, fraction of water flow to stream, fraction of water flow to deep storage, fraction of deep storage water to stream, maximum soil water) (Peng et al. 2002). These parameters are different for each vegetation type and may need long term research to identify.

Similarly, the Full Carbon Accounting Model (FullCAM) developed for estimating Australia's national greenhouse gas emissions and removals due to land-use, land-use change and forestry provides a comprehensive approach (Paul & Polglase 2004). Although this is national model, it can provide a framework to adopt in other countries. This also uses parameters of the 3-PG model.

As mentioned earlier, both field inventory and remote sensing data is being used in above ground biomass prediction (Avitabile *et al.* 2012). Use of remote sensing data (satellite imagery and LiDAR data) in REDD+ carbon stock monitoring is a better option (Hett *et al.* 2012; Romijn *et al.* 2012). LiDAR data give the vertical structure of biomass in forests and are therefore better for estimating forest degradation (Gibbs *et al.* 2007). However, there are possibilities of missing small patch of disturbances in land cover data (Avitabile *et al.* 2012). Therefore, high resolution images and proper interpretation of images with field verification is important for accuracy. However, developing countries need to build their capacity to use remote sensing based data in forest monitoring (Romijn *et al.* 2012). In this context, empirical models could be better for predicting biomass in REDD+ CFs for Nepal.

A number of methods, including destructive, non-destructive or the use of existing allometric equations, are used to estimate biomass carbon. In destructive methods, all trees are harvested and weighted. Although, this method is relatively accurate, it is time and resource consuming (Nordh & Verwijst 2004; Tyagp *et al.* 2009). In the case of non-destructive methods, parts of trees are measured and a regression between vegetation cover and biomass performed which needs detailed measurement and analysis (Flombaum & Sala 2007). The use of existing species-specific

allometric equations is quicker. However, in order to develop biomass tables or allometric equations, initial data must be collected using the destructive method (Rautiainen 1995; Chave *et al.* 2005; Chapagain *et al.* 2013). Once these equations have been developed, they can be applied more broadly across large areas with similar climatic, topographic and management practices (Alamgir & Al-Amin 2008; Henry *et al.* 2010).

Repeated measurements and data collection are required to develop growth models (Vanclay & Skovsgaard 1997). Therefore, the establishment of permanent experimental plots is needed in order to establish carbon increments in forests (MacDicken 1997; Subedi *et al.* 2010). In short term studies, an existing model or yield table are the best option for estimating biomass carbon with one variable. A model developed by Rautiainen (1995) for *S. robusta* forests was found to be applicable in predicting biomass carbon growth for undisturbed forests. The data tested for this model was collected from undisturbed *S. robusta* forests with 20% of other species of Terai and Bhabar area in Nepal. This model has been used in other related researches (Gautam & Devoe 2006; Sapkota & Meilby 2009).

2.10. Theoretical frameworks related to REDD+ in CF study

Study related to CF and the REDD+ mechanism are linked with environmental, social and economic theories. More relevant theories of this study are forest governance particularly common pool resource management (Hardin 1968; Ostrom & Gardner 1993), forest yield (Smith *et al.* 1997; Khanna 2004), self- determination in human motivation (Deci & Ryan 2000), political economic theory or economic tranformation theory (Breisinger & Diao 2008) and Pareto improvement in welfare economic (Hochman & Rodgers 1969; Chou & Talmain 1996). Linkage of these theories in REDD+ is shown is figure 2.3.



Figure 2.3 Theoretical aspects for REDD+ in community forestry

In common pool resource management theories, the theory of "tragedy of the commons" says that the collective actions of individuals with common ownership inevitably lead to degradation of the common resources (Hardin 1968). This concept evolved during the early 1970s from studies investigating whether the degradation of common pool resources happens due to overuse practices of the communities; these studies found that external interventions were needed to control it (Ostrom 1990; Ostrom et al. 1994; Gibson et al. 2000). In opposition to this concept, another theory emerged that the common resources can be conserved and individuals work better if there is a mechanism of sharing rights, responsibilities and benefits of the common pool resources (Schlager & Ostrom 1992; Edwards & Steins 1999). For this to occur, proper communication and good governance mechanisms are important to facilitate the sustainable management of the common resources (Ostrom 1990; Agrawal & Ostrom 2001). However, there are still gaps in the literature about the reactions of communities and outcomes if some external motivating factors (such as monetary incentives from REDD+) come into the existing common pool management practices. If local communities start to invest additional efforts as a result of incentive oriented motivations, this cost and the possible benefits from these efforts may need to be assessed. If there are higher added costs than possible benefits of REDD+, understanding of the response of common pool resource managers toward their performance is important.

In the case of CFUGs, communities are operating in a complex social structure with a heterogeneous economy and forest product needs, where the collective efforts of all community members are expected. REDD+ interventions can motivate user communities to change existing activities (Caplow et al. 2011). They change those activities either through intrinsic motivation of their own or extrinsic motivation from external incentives (Deci et al. 1999; Deci & Ryan 2000; Gagné & Deci 2005). Extrinsic motivation impacts the autonomy of individuals. Therefore the selfdetermination theory (Robert 1991; Deci & Ryan 2000; Gagné & Deci 2005) assumes that people act from a combination of intrinsic and extrinsic motivations. Most of the present studies relate to motivation in regard to the human resources aspect of the organisation, but this may also be applicable in the CFUGs and REDD+ outputs. Incentives of REDD+ may be an extrinsic motivation factor for CFUGs and they can be motivated with the expectation of appropriate incentives. If communities are getting fewer incentives than foregone sacrificed benefits, their efforts may not be continued in the long term. In this situation, if their efforts are not driven by intrinsic motivation, the outcome will not be sustainable.

Within the REDD+ context, political economic theory, mainly transformation theory, is also relevant. As REDD+ is emerging as a new paradigm for forest management, it that may require new institutions, the involvement of new stakeholders, new knowledge about forest management and new incentive mechanisms, the complete transformation of existing mechanisms or the generation of new systems will be required for the effective implementation of REDD+ (Angelsen *et al.* 2012). This transformation may change existing power relationships between government authorities and Community Forest User Groups (CFUGs). Carbon benefits with financial flows from overseas may come through central government and, as such may run the risk of centralising power from CFUGs. This will be problematic due to current land tenure system of community forest management – a system where land tenure is with government (GoN 1993) and carbon payments may be linked with tenure rights (Corbera *et al.* 2011) therefore communities may not able to access

them directly. Similarly, the REDD+ mechanism may add to the institutional complexity associated with distributing financial benefits at a community level; and this can add further problems in heterogeneous communities where inequality in benefit sharing already exists in a CFUG (Malla *et al.* 2003; McDermott & Schreckenberg 2009). This may affect existing community forestry systems and drive further social instability.

Current annual increment (CAI) measures yield in terms of of tree volume. CAI is slower at juvenile age, higher at younger stages of the forest stands and gradually decreases to zero as a stand ages (Birch 1999; Husch *et al.* 2003). In forests, growth rates vary with species compositions, climatic factors, site factors, average stand age, disturbances and forest product extraction. In CF, community activities may have obstructed the natural growth of the forests. However, if a forest is managed conservatively, forest stands will reach an equilibrium stage after a certain time with CAI zero. The yield also depends on the silviculture operation and harvesting intensity. An appropriate silvicultural operation reduces competitors of the trees and increases the growth of remaining trees in the forest stands (Thornley & Cannell 2000; Poorter 2001; Ishii *et al.* 2008; Harmon *et al.* 2009).

2.11. Conclusions

The review of the existing literature indicates that the REDD+ incentive based mechanism in developing countries is likely to come into future global GHG reduction agreements. However, there are gaps in scientific knowledge, particularly about REDD+ in respect to CFs. In order to design a sustainable REDD+ mechanism in CFs, the following gaps identified in the document review need to be addressed in future research:

- <u>Potential carbon stock enhancement in CF:</u> There are limited studies on existing carbon stocks in CFs. Knowledge of the current carbon stock of CFs is important when designing an appropriate benefit distribution mechanism for REDD+. Similarly, the technical potential carbon enhancement in CFs is important to enable prediction of future scenarios of REDD+; this is not clear at present. If a forest is left without disturbances, the natural growth of forest stands can achieve an equilibrium stage with high levels of carbon stock; however, many human activities in CFs can prevent a forest from attaining this. Gaps between the technical potential carbon stock in forests without human disturbances and carbon stock with community interventions in CFs are not well understood.
- <u>Factors affecting carbon enhancement in CF</u>: The existing literature focuses mainly on the global perspectives of the REDD+ mechanism. Those studies cover factors affecting deforestation and forest degradation in national contexts. However, there is limited knowledge about the local potential factors affecting carbon stocks in CFs as most of the present studies have limited coverage of this regime.
- <u>There is a gap in project level research on REDD+ demonstration activities</u>: There are limited studies available on REDD+ demonstration projects. The

lessons from these REDD+ demonstration project activities and outcomes are important in developing the REDD+ policy and institutional set up.

• <u>Trade-off between sacrificed community benefits and REDD+ benefits is</u> <u>unclear</u>: As overall aim of the REDD+ mechanism is to provide incentives for forest improvement performance in developing countries. At present, there is no clarity in the literature about trade-offs between community foregone sacrificed benefits and carbon benefits, particularly in CFs of differing dominant vegetation types. Knowledge about the benefits to local communities sacrificed to enhance unit carbon credit in CFs is important when designing an appropriate REDD+ incentive mechanism, but is currently lacking.

In a nutshell, the identification of factors associated with a higher rate of carbon sequestration in CFs and lower trade-offs in terms of community benefits is important for the design of REDD+ in CF. Analysis of the trade-off between forgone sacrificed benefits of communities and carbon benefits would help in identifying whether REDD+ is beneficial for CFs or not and which types of species dominated CFs are more beneficial. This field based study expected to contribute on these gaps in knowledge to help improve the ongoing approaches and practices of REDD+ projects to achieve both community and carbon sequestration benefits. To address this research issue and specified objectives, the next chapter will detail the methodology used for this study.

CHAPTER THREE

RESEARCH DESIGN AND METHODS

3.1. Introduction

The previous chapter gives a review of existing knowledge about climate change, the role of forests in mitigating climate change and REDD+ in CFs, relevant theories and knowledge gaps. This chapter presents the methodology used to address the research problems identified in Chapter One. The Overarching goal of the study was to answer four major questions: 1) What are the carbon stock dynamics in CFs with REDD+ program; 2) What is the technical potential and actual growth of carbon stock in CFs; 3) What are the possible factors affecting carbon stock changes in CFs; and 4) What are the trade-offs between carbon stock and community foregone sacrificed benefits in REDD+ CFs.

Both biophysical and socio-economic data were used in this study. Biophysical data includes carbon pools, canopy structure, vegetation measurements (tree, sapling and regenerations), average age of the dominant trees in plots, altitudinal location of the forests above sea level, proximity of the forests from settlement and road head, and incidences of biomass reducing factors in forests. Similarly, socio-economic data includes demographic information of CFUGs (population, caste group), average household level agriculture land holding, average household level livestock and change after the REDD+ project, average household level forest products use and changes in quantity after the REDD+, average number of households using alternative energy, grazing practices (livestock unit in forests), cost and benefits of communities due to changed forest management practices and changed forest product use practices for the REDD+ in CFs. The carbon pool data were collected from February to May of each year of 2010, 2011, 2012 and 2013, and the data related to socio-economic aspects of communities was collected from November 2011 to March 2012. During carbon pool measurement, observation and recording of evidence of biomass reducing factors namely forest fire, livestock grazing, fodder collection, firewood collection and timber harvesting was done for all plots to analyse changes in such incidences after REDD+ project.

This chapter has been organized into eight sections. The next section gives a description of the study areas covering demographic information, location map, precipitation and temperature, vegetation types, forest management practices and REDD+ project activities in CFs. The third section gives methods used to estimate stocking rates of vegetation, carbon stock and change in carbon stock in CFs. The fourth section explains the methods used to analyse the potential growth of carbon stock in the REDD+ CFs. The fifth section includes methods used for analysing demographic information, forest management practices, changed behaviour and cost and benefits of CFUGs for REDD+. The sixth section describes methods used to analyse factors affecting carbon stock and carbon stock changes in CFs. The seventh section gives methods about analysing trade-offs between carbon benefits and community benefits. The final section concludes the chapter.

3.2. Description of the study areas

The study area covers 105 CFUGs in three watersheds, one watershed in each of three districts namely, Kayerkhola watershed in Chitwan, Ludikhola watershed in Gorkha and Charnawati watershed in Dolakha (Figure 3.1). These areas represent diverse social (people with ethnic differences) and ecological (altitudes and vegetation differences) aspects. The area is the only REDD+ project that demonstrates REDD+ pilot project activities in CF in Nepal. The demonstration REDD+ pilot project was implemented for four years (2009–2013) jointly by the International Centre for Integrated Mountain Development (ICIMOD), Asia Network for Sustainable Agriculture and Bioresources (ANSAB) and Federation of Community Forest User Group (FECOFUN) under the financial assistance of the Norwegian Agency for Development Cooperation (NORAD). Unique characteristics and other details about the study areas are discussed in this section.

3.2.1. Demographic information of the study area

Among three watersheds, the Kayerkhola watershed represents the tropical lower altitudes terai, where 16 CFUGs are managing 2,381.8 hectares of forest area. In these 16 groups, diverse caste groups (Chepang, Tamang, Dalit, Newar, Gurung, Brahmin and Chhetri) are involved in forest management. Ludikhola watershed, which represents both the tropical and sub-tropical physiographic region, covers 31 CFUGs and their 1,887.7 hectares of forest area. Magar, Gurung, Tamang, Dalit, Brahmin and Chhetri caste groups are major residents in the Ludikhola watershed. Charnawati watershed represents subtropical and lower temperate physiography and includes 58 CFUGs representing Chhetri, Brahmin, Newar, Tamang, Thami and Dalit communities. These CFUGs are managing 5,996.1 hectares of forest area. Among caste groups of the study areas, Dalits, Chepang, Thami, and Tamang are more forest-dependent and they are also listed as socially underprivileged groups in the country (CBS 2011b). Comparatively, Charnawati watershed is larger in term of populations and forests areas and also in per capita forest distribution, while Kayerkhola has the smallest population and the Ludikhola watershed the smallest area of forest (both total and per capita area) (Table 3.1).

S N	C:tog	CFUGs	Households	Population	Forests	Per capita
3. N.	Sites	(1NO.)	(190.)	(190.)	area (na)	iorest (na)
1	Kayerkhola	16	3650	22090	2381.8	0.108
2	Ludikhola	31	4000	23197	1887.7	0.081
3	Charnawati	58	7730	48504	5996.1	0.124
	Total	105	15380	93791	10265.6	0.109

Table 3.1 Demographic information and size of forests in the study and	reas
--	------

Note: CFUG= Community forest user group; No. = Number (Source: Operation plan and Constitution of CFs)



3.2.2. Precipitation and temperature of the study area

Precipitation and temperature are key climatic factors affecting the growth of vegetation. This study area represents various forests with different temperature ranges and rainfall. Temperatures in the study area range from a maximum of $24.3-36.6^{\circ}$ C to a minimum of $-1.2-7.6^{\circ}$ C. The average temperature of the Kayerkhola watershed in Chitwan is comparative warmer followed by Ludikhola, Gorkha and colder in Charnawati, Dolakha. Average annual rainfall is more than 1700 mm in all study areas (Table 3.2). The study areas receive precipitation in all but 1-2 months (December-January) of the year are dry (DHM 2013).

 Table 3.2 Average meteorological information of closely located stations from three watersheds in 1976-2005

S.N.	District	Ave annual Temperature range (°C)	Meteorological station	Ave annual Rainfall (mm)	Meteorolog ical station
1	Chitwan	Min 7.6; Max 36.6	Rampur	2009.6	Rampur
2	Gorkha	Min 5.4; Max 28.5	Syangja	1736.7	Gorkha
3	Dolakha	Min-1.2; Max 24.3	Jiri	2104.4	Charikot

Note: Min denotes average of the minimum temperature and Max means average of the Maximum temperature; *Source: Marahatta et al.* (2009)

3.2.3. Vegetation types of the study areas

The study areas cover a diverse ecology due to different physio-climatic characteristics. Dominant forest types ranges from *Shorea robusta* forests at lower altitudes through *Schima-Castanopsis* to *Rhododendron–Quercus* in higher altitude areas. Even though *S. robusta* mixed tropical and sub- tropical deciduous are major forest types in Chitwan and Gorkha, associated species are different: *Lagerestroemia parviflora, Mallatus phillipinensi* and *Terminelia tomentosa* are dominant associates in Chitwan whereas *Schima wallichii* and *Castanopsis indica* are the most common associates in Gorkha. While few *S. robusta* and *Schima-Castanopsis* forests are found in lower altitudes in Dolakha, the major vegetation types are *Schima-Castanopsis* and *Rhododendron* forests, extending high up to *Quercus* forest, representing the dominant lower temperate forests. Dolakha site has also some plantation forests with *Pine* species.

3.2.4. Forest management practice in the areas

Though the Forest Act 1993 and 1995 legislation mention government managed forests, private forests, religious forests, protected forests, leasehold forests and CF as major forest management systems of Nepal, the Master Plan for Forestry Sector gave priority to CF (MFSC 1988). CF is now considered a successful model for

conserving forests, raising the awareness of local people, decentralising forest governance practices and proving socio-economic benefits at the local level (Acharya 2002; Adhikari *et al.* 2007; Pagdee *et al.* 2006). CF is expanding in Asia, Africa and other developing countries (Brown *et al.* 2002; Nurse & Malla 2006; FAO 2011) and the learnings from CFs are important for future policy processes intended to enable the REDD+ mechanism (Bluffstone *et al.* 2013).

CFs in the study areas have been started at different times with different enabling efforts. Dolakha and Gorkha districts are located to the hilly region of Nepal where CFs was started at earlier stages than in the terai low altitude areas of Chitwan (CFD/DoF 2013). Joint efforts of government institutions and other supporting institutions in the areas were important for this. Nepal Swiss Community Forestry Project (NSCFP) (a bilateral project of Swiss government and Nepal government) and Asia Network for Sustainable Agriculture and Bioresources (ANSAB) (a nongovernmental organisation) were working to develop and promote CFs in Dolakha. They were supporting district forest offices and local communities to form CFUGs, organize communities, develop forest management plans and constitutions, develop and operate sustainable forest based enterprises and conduct plantation and silviculture activities. In Chitwan, the Danish International Development Agency (DANIDA) funded bilateral project the Natural Resource Management Sector Assistance Programme (NARMSAP) was working for CFs. Similarly, Care Nepal (a non-governmental organisation) and NARMSAP were working to advance CFs in the Gorkha district (DoF/MFSC 2005). Additionally, ANSAB together with Federation of Community Forestry Users Nepal (FECOFUN) initiated Forest Stewardship Council (FSC) certification in 2003 and brought a total of 11 CFUGs into the forest management certification pools in Dolakha (Biggs & Messerschmidt 2005).

3.2.5. REDD+ project activities in the study areas

During the project period (2009–2013), various activities including carbon monitoring, capacity building and institution building, interventions for reducing biomass extraction from the forests, promotion of alternative energy, plantation and improvement activities of forests were carried out. More specifically, the project implemented the following activities in CFs in the study areas:

- a. <u>Carbon stock monitoring in CFs</u>: Measurement of carbon pools was carried out from February–May in 2010, 2011, 2012 and 2013.
- b. <u>Capacity building and institution development:</u> The project organised training and interaction programmes at local and national level involving local communities, government and non-government organisation representatives who were working in forestry sectors. For raising awareness of the participants on various contemporary issues of REDD+, training and interaction events were designed covering subject matters related to the climate change mitigation role of forests, carbon monitoring mechanism and ongoing discussions and the possible mechanism of the REDD+. The project facilitated the formation of a loose watershed level REDD+ network by organising all CFUGs of that particular watershed and enhancing their participatory decision making processes in

designing and implementing all REDD+ activities. The project also helped to revise the constitution and forest management plans of CFUGs incorporating those activities (ANSAB/ICIMOD/FECOFUN 2013).

- c. <u>Interventions for reducing biomass extraction</u>: The project encouraged local communities to reduce the extraction of biomass, mainly the quantity of timber, firewood, grasses and fodder harvested from the forests, to better conserve and enhance carbon stocks in the CFs. The project distributed improved cooking stove (ICS) to the communities and provided subsidies to install and use biogas energy. The project also facilitated the building of biogas plants and ICS by establishing linkages to the government subsidy programme (ANSAB/ICIMOD/FECOFUN 2013).
- d. <u>Improvement of the forests status:</u> The project assisted communities in the design and implementation of forest fire control, and the reduction of grazing and illegal harvesting activities. The project organized practical training for local communities about silviculture operations to enhance biomass stock in forests. It provided seedlings and technical guidance to carry out plantation activities (ANSAB/ICIMOD/FECOFUN 2013).

3.3. Method used to estimate stocking rate of vegetations, carbon stock and carbon stock changes in CFs

3.3.1. Inventory of vegetation and carbon pool

In order to estimate carbon stock in CFs for the reference year (2010) and changes in carbon stock after the REDD+ project activities in the following years (2011, 2012 and 2013), forest inventory data were obtained by collaborating with the project implementing institutions. Before measurement commenced, hands-on training and coaching was arranged for crew members to ensure their capability in generating accurate and consistent data in the field every year. The following steps were taken for carbon pool measurement:

a. Stratification of the forests

Stratification of the forests increases the precision of the inventory data (McRoberts *et al.* 2002); therefore, a stratified random sampling design was used in the study. Percentage of canopy cover and dominant vegetation types were taken as criteria for stratification. All forests of the study area were divided into two strata by canopy cover i.e. dense (i.e \geq 70% canopy cover) and sparse (i.e <70% canopy cover). Satellite images (Geoeye) taken in November 2009 were analysed using GIS software IRDAS IMAGINE to identify forest areas with dense and sparse canopy cover types. Out of the total 10,265.8 ha of forests, areas of dense and sparse forests were estimated (7,436.8 ha and 2,829 ha, respectively).

b. Distribution and lay out of the sample plots

After stratification, a total of 490 composite plots including 95 plots in sparse canopy and 395 in dense canopy forests were proportionally established for annual carbon pool inventory. These sample plots covered >0.08% of forest areas of each stratum which is within the range of sampling intensity (0.05–7.8%) used in other forest inventory related studies (Brown & Lugo 1992; Magnussen & Boudewyn 1998; Bongers *et al.* 1999; Specht & West 2003).

These sample plots were distributed randomly in CFs covering all five dominant vegetation types; namely *S. robusta*, Mixed broadleaf, *Schima-Castanopsis*, *Pine*, and *Rhododendron-Quercus*. Hawth's analysis tool, developed by Beyer (2004) for ArcGIS to create random points in ecological and landscape level researches (Lanier *et al.* 2007; Haines *et al.* 2012), was used to distribute sample plots on the map (Figure 3.2).


Coordinates of the centre points of plots generated by Hawth's tool were loaded into a GPS and located in the field. After locating the points, circular composite plots were established on the ground for carbon pool measurement. A circular plot was used because it is easier to establish, especially in repeated measurements and commonly use in forest inventory (Jalonen *et al.* 1998). Moreover, the circular plot reduces errors with smaller perimeter than a rectangular plot of the same area (Goldsmith & Harrison 1976). With a smaller perimeter, a circular plot could have fewer trees and other measure of biomass within the plot boundaries (i.e. partially inside and partially outside the plots) that make easier for the crew to decide which materials to include in the plot and thereby increase the precision of estimates of biomass carbon.

Composite plots with different radius subplots were established for measuring different carbon pools in the forests at the GPS located points (Figure 3.3). From the central point, sub-plots with radii of 0.56 m, 1m, 5.64 m and 8.92 m were established to measure different carbon pools. Details about carbon pools and measurement techniques are discussed in later sections.



Figure 3. 3 Model of a composite sample plot used to measure carbon pools during forest inventory

Since, the study areas were not perfectly flat, slope distance was converted into horizontal distance which is the usual practice in forest inventory (Waddell 2002). This is because if the length was not converted it would represent a smaller area than in reality in two dimensions. Conversion was made by using the trigonometric relationship of horizontal distance, slope angle and slope distance (Horizontal distance=cosine slope angle*slope distance).

c. Information of plot

After locating and establishing the plots, dominant vegetation, altitude (above sea level) of the plots, incidence of biomass reducing events, average age of the dominant trees, and permanent reference marks of the plots were recorded at the beginning of the measurement using a pre-prepared form (Appendix E). Key biomass reducing events, namely forest fire, grazing and forest product (timber, fuel wood and fodder) harvesting in the plots for that particular year, were recorded. Before recording, it was first confirmed, with the local people who were present during the forest inventory in the plot, whether the incidence had happened in that particular

year or not. Similarly, the average age of the dominant trees of the plot was asked of local people and forest technicians because they know starting time of protection activities and growth patterns, and can estimate present age (Griffith & Ram 1943; Rautiainen 1995). Quantitative techniques such as root system analysis or ring count were not applied because due to practical difficulties; such techniques require a lot of effort (Rautiainen 1995) and were not possible for this study.

d. Forest carbon pool and its measurement

According to IPCC Good Practice Gudelines (2006), above ground biomass, below ground biomass, litter and debris, and SOC are the key carbon pools found in forests. However, the guidelines provide flexibility to enable countries to decide which carbon pools to include or exclude in their national GHG accounting process based on shares in total emissions (IPCC 2006). Therefore, tree (above ground and below ground), sapling (above ground and below ground), herb (all living plants except trees, sapling and regenerations), litter (dry fallen leaves and twigs) were considered as carbon pools and measured for four consecutive years (2010, 2011, 2012 and 2013) in this study.

Inventory in the plots was scheduled for the same week (almost the same days of the month) of each year to avoid potential effects on the quantity of biomass in the plots due to different growth seasons of trees and herbs (Wright & Cornejo 1990; Moser & Hoveland 1996; Vaganov *et al.* 1999; Brienen & Zuidema 2005) and litter fall (Wright & Cornejo 1990). Smaller plots were used for both herbs and litter than for the other pools. This is because herbs and litter are smaller and need a rigorous process to be followed for destructive sampling. There are practices to establish smaller plots for inventory of small size plants (CFD/DoF 2002). A brief description of measurement techniques used in this study follows:

<u>Tree measurement</u>: A plant with \geq 5 cm DBH is considered a tree. Trees sequester atmospheric CO₂ and store higher amounts of carbon compared to other biomass pools (Turner *et al.* 1995; Finér *et al.* 2003). During the field inventory, the names of tree species were recorded, and the height and DBH of trees were measured in the sample plots with radius 8.92 m. Diameter tape was used to measure the DBH of each tree at 1.3 m from ground because this is common in forest inventory practices (CFD/DoF 2002; Husch *et al.* 2003). Clinometer and compass were used to measure the angle and linear tape was used to measure distance from the bottom of a tree to the standing point of the crew member from which angle was measured to estimate tree height.

<u>Sapling measurement</u>: A woody plant with DBH 1 to 5 cm was considered a sapling. The names and DBH of saplings within plots of radius 5.64 m were recorded.

<u>Herb measurement:</u> Herbs were uprooted from 0.56 m diameter plots and fresh weight taken in the field. Of the total fresh weight of these thoroughly mixed herbs on each plot, a sample portion of 100 g was taken and sent to the National Agriculture Research Coulcil (NARC) laboratory of Nepal for dry weight estimation in the first year. Since destructive sampling was carried out in plots, the locations of the 0.56 m radius plots were shifted 5.64 m from north in second year, and the same distance in clockwise direction in following years i.e. perfect east in third year and

perfect south in fourth year. Herbs were uprooted and only fresh weights were taken in the second, third and fouth years.

<u>Litters measurement</u>: Litter biomass (fallen leaves, small twigs) was measured in the same plots (with radious 0.56 m) in which the herbs were measured. Similar to herbs, all the litter was collected from the plot and the fresh weight taken. From the collected litter, a well mixed sample of 100 g weight was sent to the NARC laboratory for biomass estimation by the oven dry method. Only the fresh weight of litter in the plots was taken in the second, third and fouth years.

e. Excluded pools in measurement

<u>Soil organic carbon (SOC)</u>: SOC is an important part of the forest carbon pool (IPCC 2006); however, this study excluded this pool. This is because SOC does not change over short periods of time under the same land use practices (Fearnside & Barbosa 1998; Houghton *et al.* 2000; Guo & Gifford 2002). Moreover, there is limited chance of humus accumulation in forests managed by communities where all fallen leaves and litter are used for cattle bedding materials and coarse woody debris are used for fuel-wood (Metz 1994; CFD/DoF 2009; Maskey *et al.* 2006) with no biomass left on the ground to decay. Therefore, the study assumed that there would not be much variation in SOC within the short timeframe, and that it remained almost constant during the study period compared to forests that have no collection.

<u>Harvested wood products</u>: Most of the harvested wood goes for fuel wood (even if it is timber) after a certain time; therefore, this study has excluded harvested wood from the measurements. Communities are allowed to harvest timber trees if they need to build or renovate houses (Adhikari *et al.* 2007) and old decaying wood has to be replaced. Most of the communities replace old timber with fresh during renovation and the old timber goes for firewood. Therefore, the study assumed that the quantity of harvested wood used remained almost constant in community households during the study period.

<u>Regeneration</u>: Number of recruits (plants with less than 1 cm diameter) was counted within 1 m radius plots in the field. The study excluded this in the biomass estimations because plants of this size contribute only a very small amount of biomass compared to other pools and destructive sampling to estimate regeneration biomass was not conducted.

3.3.2. Analysis of vegetation and carbon pool

a. Stocking rates of different size trees in forests

Physical and temporal distribution of trees in forests were analysed by canopy differences in each forest type. Particularly, stocking rates (i.e. numbers of regenerations, saplings and trees per hectare) available in CFs were analysed. This analysis was expected to generate knowledge about the distribution of different sized trees in forests to design a plan for maintaining normal forests i.e. larger stocking rates of smaller sized trees and gradual reduction towards larger sizes in the forest stands (Smith *et al.* 1997; Montagnini & Ashton 2010). Average stocking rates of

each vegetation category in plots for the reference year 2010 and changes between 2010 and 2013 (after the REDD+ project activities) were estimated and compared.

b. Biomass carbon in forests

Biomass carbon in forests for a particular year was estimated in four steps: first of all the biomass of each pool in a plot was estimated; secondly, total biomass in each plot was derived by adding all individual biomass pools; similarly, total biomass was converted to carbon using the carbon fraction value (0.47); and the biomass carbon estimated for each forest type and individual CFs. Detailed analysis methods used in the study were as follows:

Estimation of above ground tree biomass (AGTB)

Though destructive sampling is reliable if samples adequately represent the stand, it is impractical, time consuming and expensive due to its destructive nature and the rigorous process involved (Verwijst & Telenius 1999; Tyagp *et al.* 2009). Therefore, this study used existing allometric equations (previously developed from destructive sapling from similar forests) combined with forest inventories, which is the method most commonly used to assess forest biomass (Chave *et al.* 2005; Alamgir & Al-Amin 2008; Henry *et al.* 2010). In tree allometric equations, the relationship between several dendrometric variables have been established for estimating wood biomass (Chave *et al.* 2005; Henry *et al.* 2010; Henry *et al.* 2013). Most equations use either DBH alone or DBH and height or DBH, height and the specific gravity of wood as variables to estimate tree biomass.

The equation 1 suggested by Chave *et al.* (2005) for the moist forest stands is the most relevant equation for the study sites and used to estimate AGTB because : a) the equation was developed using a large dataset (with >5 cm DBH trees) including Asian forests; b) the equation uses three variables i.e. wood specific gravity, tree height and DBH (all of three variables are important determinants of the carbon quantity); 3) the equation has been tested in a variety of climatic situations and different equations have been recommended for specific climates; 4) the study sites fall under the moist category of forests with average annual rainfall exceeding 1,700 mm during 1976–2005 (Marahatta et al. 2009) and more than 1990 mm in 2011, and the sites receiving rainfall in most months with few exceptions (DHM 2013). This allometric equation was used to estimate AGTB in other studies related to carbon accounting (Chave et al. 2008; Mani & Parthasarathy 2009; Sharma et al. 2010). Most importantly, the Ministry of Forests and Soil Conservation Nepal has recommended this equation be used to estimate tree biomass and it is included in the forest carbon measurement guidelines, which were developed together with forestry experts working in the REDD+ activities in Nepal (MFSC 2011).

The equation suggested by Chave *et al.* (2005) for the moist forests which was used in this study is:

 $AGTB = 0.0509 * \rho D^2 H$ (1)

Where, *AGTB* = aboveground tree biomass [kg];

- ρ = wood specific gravity [gcm⁻³];
- D = tree diameter at breast height (DBH) [cm]; and H = tree height [m].

Of the three variables required for equation (1), two variables (i.e. diameter and height of the trees) were obtained from the inventory data while the wood specific gravity of all the tree species was obtained from Nepal specific publication (MPFS 1988) (Table 3.3).

Species	Specific gravity (gcm ⁻³)
Sal (Shorea robusta)	0.88
Botdhangero (Lagerstroemia parviflora)	0.85
Saj (Terminelia tomentosa)	0.95
Guras (Rhododendron arboreum)	0.64
Chilaune (Schima wallichii)	0.69
Khote sallo (Pinus roxburghii)	0.65
Katus (Castanopsis spp)	0.74
Other species found in tropical forests	0.72
Other species found in sub-tropical and lower	0.59
temperate forests	

Table 3.3 Wood specific gravity used in analysis

The AGTB values of all individual trees within a plot were estimated separately and added to calculate the total AGTB in that particular plot. The sum of all the individual tree biomass values (in kg) of a sampling plot was divided by the area of a sampling plot (250 m^2). This gives the biomass of a plot in kg/m², which value was then converted into megagrams per hectare (Mg /ha). Finally, total estimated AGTB of the plot was converted into carbon stock by multiplying by the carbon fraction 0.47 which is recommended for whole trees (IPCC 2006) and used in other biomass carbon studies (Gibbs *et al.* 2007; Zanchi *et al.* 2013)

Above ground sapling biomass (AGSB)

The Nepalese specific biomass table was used to estimate AGSB. Although biomass table for mid-hill forests were discussed in Thompson (1990), biomass tables developed by Tamrakar (2000) are more applicable in this study. This is because the table developed by Tamrakar (2000) was particularly for community forests of Nepal and gives green biomass for various species with different tree diameters. There were missing biomass data for some species; therefore, values recommended for mixed vegetation types were used. After estimating the green biomass of saplings, this value was then converted into dry biomass by multiplying by the species-wise conversion fraction (dry weight to green weight ratio). Species-wise dry (weight after deducting moisture content) to green weight ratio was taken from the relevant literature (Table 3.4). For saplings for which a fraction was not available in the literature, mean values of closely related species were used as suggested in the literature (Baker et al. 2004; Ngugi et al. 2011). Average values for trees of the same Family were used; for example: the average value of Pinus wallichiana and Pinus roxburghii was used for Tsuga spp.; the value of Alnus nepalensis for Betula spp.; the value of Lyonia ovalifolia for Rhododendron spp.; and the value of Albizia for Dalbergia sisoo.

Tree Dry to green Source weight ratio*		Source	Information source (Forest types)	Study area (country)
Pinus roxburghii	0.58	Shrestha et al. (2006)	Sub-tropical forests	Central Nepal
Alnus nepalesnsis	0.57	Shrestha et al. (2006)	Sub-tropical forests	Central Nepal
Shorea robusta	0.517	Jain & Singh (1999),	Tropical	Central and North east India
		Kataki & Konwer (2002)	_	
Terminalia tomentosa	0.50	Jain & Singh (1999)	Sub-tropical forests	Central India
Lyonia ovaliforlia	0.613	Jain & Singh (1999)	Sub-tropical forests	Central India
Albizzia lebbeck	0.537	Kataki & Konwer (2002)	Tropical and sub-tropical	North east India
Pinus wallichiana	0.45	Kataki & Konwer (2002)	Sub-tropical	North east India
Schima wallichi	0.545	Bhatt & Tomar (2002)	Sub-tropical and lower	North east Indian mountain
Quercus spp	0.627	Kataki & Konwer (2002)	sub-tropical and lower temperate	North east Indian mountain

Table 3.4 Dry to green weight ration used to convert green biomass to dry biomass of sapling

Note: *Dry to green ratio was obtained from the relationship between fresh weight immediately after cutting and constant weight obtained with laboratory drying at 103 ± 2 °C.

Below ground tree biomass (BGTB) and below ground sapling biomass (BGSB)

At present, the above ground biomass estimation method is well developed compared to below ground root biomass estimation (MacDicken 1997). However, root biomass has been considered as an important carbon pool to be included in the carbon accounting process (Hertel et al. 2009; Kenzo et al. 2010). As observing and measuring root biomass in forests is difficult (Vogt et al. 1998; Titlyanova et al. 1999), a relationship between shoot biomass and root biomass has been identified and used to estimate belowground root biomass (Cairns et al. 1997; Mokany et al. 2006). Root to shoot ratio is different according to climatic and physiographic factors of forests (Cairns et al. 1997), therefore this study has used two different ratios according to altitude. This is because altitude affects the vegetation type, temperature, soil type and humidity of a particular site (Jackson 1994; Leuschner 2000; GoN 2002). Mokany et al. (2006) has identified different ratios for different forests. Above ground biomass was multiplied by 0.27 for the forests located below 2000 m altitude, as suggested for tropical and subtropical forests (Singh and Misra 1979 cited in IPCC 2006), and 0.3 for forests above 2000 m altitude, which was the average for temperate broad leaf and temperate conifer forests identified by Mokany et al. (2006). It also represents various sites, vegetation types, live and dead roots, both healthy and unhealthy trees (Mokany et al. 2006); therefore, average ratios were used to convert average shoot biomass to average root biomass in plots. These ratios have been suggested by IPCC (2006) for use in national GHG inventories.

Litter biomass (LB) and herb biomass (HB)

The biomass of both litter and herbs was estimated separately from: (1) the fresh weight of samples in a plot; (2) the dry-weight and fresh-weight ratio of samples (in which the dry weight of samples was obtained using oven dry methods in the laboratory); and (3) the sample plot area to hectare expansion factor (the area of a hectare/area of a sample plot) as shown in equation (2)

	$B = w_{fresh} * r * ha Expansion \qquad \dots $
where;	
В	= biomass of leaf litter or herb [Mg/ ha];
W fresh	= weight of the fresh field sample measured in plot [Mg];
r	= ratio of oven dry-weight of sample to fresh weight of the sample (dimensionless); and
ha Expansion	= hectare expansion factor [area one hectare/area of plot] [ha];

Total biomass/carbon

Total biomass was estimated by adding all individual biomass pools (AGTB, AGSB, BGTB, BGSB, LB and HB). Total biomass was converted into carbon by multiplying by the carbon fraction 0.47 (IPCC 2006) and later converted into CO_2 equivalent by multiplying by 3.667 (the ratio of the molecular weight of carbon dioxide (44) to carbon (12)).

Carbon stock in dominant vegetation types

Biomass carbon data in all sample plots were grouped by vegetation type. Average biomass carbon in the reference year 2010 and changes in biomass carbon (2010–2013) in each forest type were estimated.

Carbon stock for individual CFs

Since sample plots were representative of forests, the total carbon stocks estimated within sample plots were extrapolated to derive the total carbon stock of CFs. Species types differ in CFs with altitude and canopy differences (Garkoti & Singh 1995; Alves *et al.* 2010;). Therefore, the average carbon stock for each canopy type (i.e. <70% and \geq 70%) within each altitude category (i.e. <1000 m, \geq 1000–2000 m and \geq 2000 m) was estimated from plot data. Total carbon stock in a particular CF was estimated from carbon stock for each forest category and area of that forest type category within a CF.

Species dominated CFs were identified on the basis of the proportion of areas of dominant vegetation types. For example, if the majority area of a CF was *Schima-Castanopsis* vegetation, the forest was categorised as *Schima-Castanopsis* type.

Carbon stock changes in CFs by REDD+ project activities was estimated adopting the stock difference method of the IPCCC (2006). This was estimated from two times within the carbon stock data (i.e. at reference year in 2010 and after project interventions in 2013) and the time difference.

3.4. Method used to analyse potential growth of carbon stock in CFs

Potential growth of biomass carbon in natural forests without anthropogenic disturbances and average biomass carbon growth under community management have been estimated to understand the gap between these two forest management regimes. Forests exhibit natural growth patterns in an undisturbed condition (Oliver & Larson 1990) while these trends can be distorted by human activities in CFs. Growth patterns by age of stands were estimated. In this analysis, both the above and below ground biomass of trees and saplings were considered because these two contribute most to the total biomass carbon in forests (Finér *et al.* 2003; Turner *et al.* 1995).

3.4.1. Estimation of biomass carbon in undisturbed forests in *S. robusta* forests

There are several models developed for forest growth estimation but the local empirical model was used due to the availability of data for parameters and the potential future application of the study. In order to estimate the "theoretical maximum" or "technical potential" growth of biomass carbon, the yield table (stem volume) developed for natural uniform *S. robusta* forests in Terai in Nepal

(Rautiainen 1995) was used. The model used data generated from a fully stocked *S. robusta* stand (<20% other species) under undisturbed conditions (Rautiainen 1999; Rautiainen *et al.* 2000). The model data has also been adopted in other studies (Gautam & Devoe 2006; Sapkota & Meilby 2009) and the climate of the experimental plots used in the model is similar to the *S. robusta* dominated forest sites in this study (Kayerkhola, Chitwan and Ludikhola, Gorkha). The age of the *S. robusta* trees in this model refers only to the age after die-back phase and 10–25 years of dieback age was not considered (Rautiainen 1995).

Above ground stem volume was converted to biomass carbon by multiplying by the wood density of *S. robusta* (0.72) as suggested in the literature (FAO. 1997; Reyes *et al.* 1992) then the carbon fraction. This stem biomass carbon was then converted into total AGB carbon by multiplying by 1.59. This Biomass Expansion Factor (BEF) was used to estimate the total biomass for *S. robusta* forests in India where trees with diameters from 5 to 80+ cm were available (Haripriya 2000). A similar BEF (1.6) was found in tropical broad leaf forests in Costa Rica (Segura & Kanninen 2005). Total biomass carbon was then estimated by adding the AGB and BGB values. As mentioned, BGB was estimated from the root to shoot ratio. The total biomass carbon (Mg/ha) by age was then plotted to show the patterns of carbon growth of natural undisturbed *S. robusta* forests.

3.4.2. Estimation of carbon stock growth in *S. robusta* forests under CF

The growth pattern of trees follows a sigmoid curve (Weiner & Thomas 2001); therefore the increase in biomass carbon with age follows a similar pattern. Time series field data are needed to derive such a growth curve (Foody *et al.* 1996); however, the collection of time series data is generally not feasible due to financial and resource constraints and many researchers have instead used chronosequences (van Kanten *et al.* 2005; Gehring *et al.* 2005; Dhakal 2013;). Therefore, this study used chronosequence plots of *S. robusta* dominated stands and analysed these to see the growth patterns. The study assumed that the chronosequence plots were derived from similar ecological and climatic conditions and there is minimum natural variation between plot characteristics.

Age and biomass carbon stock (tree and sapling) in plots were presented separately according to canopy types (dense and sparse). In order to identify the 'best fit' model and its parameters, data (total tree biomass carbon and age of dominant trees in respective plots) collected from *S. robusta* dominated forests were fitted into the CurveExpert 2.0 software. CurveExpert is a comprehensive curve fitting system which models X and Y data using a toolbox of linear regression and nonlinear regression models (Kuehne *et al.* 2014; O'Brien *et al.* 2014). This software has inbuilt models that can be used to compare the models and choose the best curve. The software helps to analyse data and identify parameters for the models (Jia *et al.* 2014). A best fit model was selected to estimate the growth patterns in the particular forests. In order to select the 'best fit' model, all biomass carbon estimates in the plots by average age of the dominant trees in both dense and sparse forests were grouped and non-linear regressions (between the carbon stock and age) run using the software. In total, six models (Logistic model, Richards's model, Gompertz model

Weibull and Morgen-Mercer- Flodin (MMF)) were found to explain sigmoidal growth curves in both dense and sparse forests. Among them, the best performance of the models was estimated by comparing differences in the coefficient of determination (\mathbb{R}^2). If two or more models had similar \mathbb{R}^2 value, the less complicated model (i.e. having fewer parameters) was chosen (Perrin *et al.* 2001).

After selecting the best fitted model, potential biomass carbon growth in dense and sparse CFs by age was estimated. The biomass carbon growth patterns (mean annual increment-MAI, current annual increment-CAI and cumulative increment by age) in CFs were compared with carbon accumulation in undisturbed forests. As mentioned in the literature, CAI and MAI are used in forest management (Davis *et al.* 2001; Powers 2001), where MAI is the average annual increase in biomass carbon of forest stands up to the specified age and the CAI is increase in biomass carbon in a single year (Powers 2001). Similarly, area under the curve was estimated by using the CurveExpert 2.0 for all undisturbed, dense and sparse forest stands to assess differences (the gap) in biomass carbon growth between forest categories.

3.4.3. Gap in carbon stock in CFs with mix broadleaf, *Schima-Castanopsis*, *Pine* and *Rhododendron-Quercus* forests

There are limited publications available regarding carbon stock in undisturbed forests by age for mix broadleaf, *Schima- Castanopsis, Pine* and *Rhododendron-Quercus* forests; therefore, the potential carbon stock increment in dense and sparse canopy forests in these vegetation types were estimated by comparing highest and lowest carbon stock values among the sample plots. As suggested by Tol (2008) and Bornmann *et al.* (2013), estimated maximum and minimum carbon stock data were converted to percentiles to facilitate ranking.

The 90th percentile carbon value in sample plots was estimated and considered as being the maximum growth potential of CFs whereas the 10th percentile carbon value in the sample plots was treated as the minimum level of carbon stock in the forests. Although this is not the technical potential stock, this could indicate potential space available for growth in CFs.

3.5. Methods used to analyse demographic data, forest management practices, changed behaviour and costs and benefits consequences of CFUGs for the REDD+

This study used document review and social research methods namely focus group discussion (FGD) and key informant survey to collect demographic data, forest management practices, change practices and cost and benefits of CFUGs for the REDD+. The following sections give details about each method:

3.5.1. Review of relevant documents

CFUGs, project implementing organisations (ICIMOD, ANSAB, FECOFUN), the department of forests and district forest offices are agencies that potentially are able to maintain the various documents related to forest management decisions, activities and outcomes. In order to collect data related to forest management and socioeconomic activities at all 105 CFUGs, a list of relevant documents was prepared. Among them, baseline studies and project progress reports were collected from organisations; CFUG constitutions, operational plans, project implementing monitoring reports and district forest resource related publications from the department of forests and district forest offices as well as meeting decisions and activity records from community forest users committees. These documents provided information related to demographic overview, changes in the condition of forests and management and use practices. A separate checklist was developed and used to collect data related to general information of CFs and costs and benefits to communities for REDD+ (Appendix B and Appendix C). Annual costs and benefits of CFUGs related to forest management were collected for six consecutive years; i.e. three years before and 3 years during the REDD+ (2006–2012). Before requesting these documents and data, a clear explanation about the research purpose (i.e. purely academic) of the study and assurances of confidentiality of the data and its anonymous use were given.

3.5.2. Focus group discussion

FGD is considered a more active and dynamic social discussion which provides opportunities to achieve a cumulative understanding of the identified problem (Kitzinger 1994; Tashakkori & Teddlie 2003). This is popular among social scientists when conducting applied research (Stewart *et al.* 2007). This study used the FGD discussion method with two categories of participants; i.e. representative members of CFUGs by dominant vegetation type and with CFUG executive committees. These were carried out from December 2011 to March 2012.

As mentioned by Kitzinger (1994) and Stewart *et al.* (2007), the study conducted 12 FGDs (i.e. 3 in each vegetation dominated CFs with random selection) to explore reactions and experiences of people for REDD+. Following the method of other studies, 9–15 individuals representing all types of members (including executive committee member, ethnic people, female) were invited to a group discussion (Adhikari *et al.* 2004; CFD/DoF 2009; Maskey *et al.* 2006). From these FGDs, the perceptions of communities regarding changed behaviour due to REDD+ activities, perceptions on forest management, cost and benefits for REDD+, factors affecting carbon stock change in CFs and suggestions for making the REDD+ effective were collected using a checklist (Appendix D). The checklist was developed by including all possible factors, tested in the field and revised to make sure that tool would adequately capture all data. The final revised tools were used to facilitate discussions in a logical flow.

As suggested in the literature (Tashakkori & Teddlie 2003), FGDs with executive committees in all 105 CFUGs were organised to collect complementary information

to the data obtained from the documents review method and also for triangulation. Some of the required data, such as quantitative data about grass, litter and firewood collection as well as grazing at CFUGs (Appendix B and Appendix C), were not available in the documents, in which cases FGD data were used in analysis. It was assumed that communities have knowledge about activities and practices even if this has not been recorded.

3.5.3. Key informant survey

A key informant survey is taken as a reliable method to get accurate information (Miller & Cardinal 1994). This was done by involving 20 individuals who were working in CFs and the REDD+ project. Among them, five individuals were from government, five from non-governmental organisations, five from local leaders (including teachers and CFUG leader) and five forest based enterprise workers. These people provide more information and deeper insights about issues from their personal skills, positions and experiences (Marshall 1996) and are generally consulted in CF related studies (Dongol *et al.* 2002; Richards *et al.* 2003; Springate-Baginski *et al.* 2003; Maskey *et al.* 2006; Adhikari *et al.* 2007; Sapkota & Odén 2008). A list of major factors or drivers responsible for reducing biomass in CFs (directly or indirectly) was prepared by reviewing the relevant literature (Appendix A) and provided to key informants for comment and to finalise the list of relevant factors in the study areas as in other studies (Elmendorf & Luloff 2001). Then they were administered in focus group discussion.

3.6. Methods used to analyse factors affecting carbon stock changes

A change in the level of individual factors was compared with changes in carbon stock in CFs (by vegetation types) with the REDD+ project intervention. The final list of relevant factors of the study areas and analysis method was as follows:

a. Evidences of change in biomass reducing factors in CFs with REDD+

The frequency of biomass reducing factors (namely forest fire, livestock grazing, fodder collection, firewood collection and timber harvesting) which were observed and recorded in the inventory plots were analysed for each year. The proportion of plots where each incidence was recorded for each year were analysed and plotted as line charts by vegetation type to identify changes after the REDD+ project.

b. Carbon stock in different elevation forests

Elevation of sample plots (m asl) was recorded using GPS during the inventory. The elevation of plots was compared with carbon stock in the reference year (2010) and carbon stock changes (2010–2013) to see possible effects. Scatter plots were developed by canopy cover types (dense and sparse) for all vegetation types to demonstrate the relationship. The relationship was assessed using the coefficient of determination (\mathbb{R}^2) in regression models.

c. Carbon stock in different proximity of the forest from motor-able road and from settlement

The distance of sample plots from closely located driveable road was estimated by analysing GIS data. The data related to forest canopy cover and location of forests was obtained from satellite imagery and GPS points from field measurement. Similarly, data related to roads and settlements were obtained from the Department of Survey, Government of Nepal. The locations of sample points, settlements and roads were identified and distances of the plots from closely located roads and settlements were estimated using ArcGIS.

All vegetation-wise plots were grouped into two categories (i.e. 21 plots located at extremely far and 21 at extremely close distances from the road-head) and estimates of average carbon stock in the reference year (2010) and changes in carbon stock between 2010 and 2013 for each category used to analyse the possible effect of proximity from the road head on carbon stock in the forests.

Following the same process, the effect of distance from settlement on carbon stock and change in carbon stock was estimated and compared.

d. Carbon stock in different aged forests

The average age of dominant trees in a forest stand and its relationship with carbon stock in the year 2010 and change in carbon stock from 2010–2013 was compared. CFs were grouped into older and younger age categories by vegetation. Average carbon stock and changes (2010–2013) between older and younger age categories by vegetation type were compared to investigate possible effects.

e. Carbon stock with different socio-economic factors

The effects of possible factors (namely, per capita forest areas, caste heterogeneity, household level landholding, average change in livestock, biogas use, petroleum energy use, biomass extraction (timber and fuel wood), grazing, quantity of grass and fodder collection, and litter collection) on carbon stock and changes in carbon stock in CFs were analysed individually (Table 3.5). All CFs were grouped into two categories (i.e higher and lower scale (value)) of the factor by vegetation type and carbon stocks analysed. Numbers of CFs by dominant vegetation types differ. However, two groups of equal numbers of CFs with higher and lower scales of factors were created for analysis. Therefore, *S. robusta* mix broad leaf dominated CFs were grouped into 10 higher and 10 lower, *Schima-Castanopsis* dominated CFs into 10 higher and 10 lower, *Pine* dominated CFs were grouped into 8 higher and 8 lower and *Rhododendron-Quercus* dominated CFs into 4 higher and 4 lower (Table 3.5).

S.N.	Factors	Data source	CF category	Descriptions of analysis
1.	Per capita forest areas	Population and areas of CF from documents	Smaller Larger	Estimated per capita forest in each CF and estimated carbon stock and change in carbon stock of smaller and larger size
2.	Caste heterogeneity	Household number of specific caste group (i.e. Brahmin-Chhetri, Ethnic, Dalit) in a CFUG was collected from community records.	Lower Higher	Estimated caste heterogeneity situation of each CF [the below equation was used which was also used in other similar studies (Sapkota & Odén 2008; Varughese & Ostrom 2001). A= $1 - \sum_{i=1}^{n} (P_i)^2$ Where, Pi = the proportion of total population in the i th caste group, and A= Caste heterogeneity index, (varies from 0–1, where 0 is perfectly homogeneous and 1 is highly heterogeneous)]
3.	Average household level landholding size	CFUG records and verified from the FGDs	Smaller Larger	As for point 1, estimated average land holding in each CF and estimated carbon stock and change in carbon stock of smaller and larger size
4.	Average change in livestock	Average livestock size in CFUG. Average number of Buffalo, Cow/Ox, Goat in CFs (in per HH) from records and FGD	Change small Change large	 Estimated average livestock size in CFUG in particular year by using livestock standard units (LSU) in which: 1 buffalo = 1 unit, 1 cow=0.7 unit and 1 goat =0.1 unit (Thapa & Paudel 2000; Thorne & Tanner 2002) Estimated change in livestock size in CF from before (average of 3 years) to after (average of 3 years) the start of the REDD+ project
5.	Biogas use	Biogas using households in each CF from records and FGDs	Lower proportion Higher proportion	Estimated proportion of biogas using households (biogas using HHs/Total HH in CF) in each CF.
6.	Petroleum energy use	Petroleum energy (Liquefied Petroleum Gas and Kerosene) using household in CFs from records and FGDs	Lower proportion Higher proportion	Estimated proportion of Petroleum energy using households (biogas using HHs/Total HH in CF) in each CF.

Table 3.5 Methods used for analysing factors affecting carbon stock and change in carbon stock in CFs

7.	Biomass extraction*	Total average annual extraction of fuel wood and timber from records and FGD	Less change in extraction More change in extraction	1 2 3 4	 Estimated average annual extraction of timber biomass [volume (m³/yr) was converted into biomass by multiplying by wood density.] Estimated average annual extraction of fuel wood (Bhari) in each CF multiplied by conversion factor (1bhari=30 kg) suggested in Nepal (Neupane 2003; Pokharel 2008) Estimated total annual extraction (sum of both timber and fuel wood) of biomass in each CF Estimated average change in total annual extraction from before (average of 3 years) to after (average of 3 years) the start of the REDD+ project
8.	Grazing	Total number of buffalo, cow/ox and goat grazing in forests for a whole year was collected from records and FGDs	Smaller change Larger change	A 1 1 a E a	Annual livestock grazing in CF [Number of each grazing ivestock types converted into LSU. Average annual grazing ivestock size in each CF was estimated using a conversion factor as stated in point 4 in this table.] Estimated changes in grazing from before (average of 3 years) to after (average of 3 years) the start of the REDD+ project
9.	Quantity of grass and fodder collection	Total annual grass collection in each CF from record (decision to collect) and FGDs	Smaller change Larger change	C v k E a	Converted Bhari to ton [Unit of grass was Bhari in study sites which was converted into tons from the relationship: 1 Bhari=25 cg (Adhikari <i>et al.</i> 2004; Sunam 2011)] Estimated changes [Grazing from before (average of 3 years) to after (average of 3 years) the start of the REDD+ project]
10.	Litter collection	Same as grass (point 9 of this table)	Smaller change Larger change	((E y	Converted Bhari to ton [Unit of litter was Bhari/Doko which was converted into tons from the relationship: 1 Bhari=20 kg (Adhikari <i>et al.</i> 2004; Sunam 2011)] Estimated changes in litter collection from before (average of 3 years) to after (average of 3 years) the start of the REDD+ project

Note: * Among all carbon pools included in this study, trees constitute a major part of the aboveground biomass carbon in forests (Turner *et al.* 1995) and extraction of fuel wood and timber from the forest reduces these stock

3.7. Methods used to analyse trade-off between carbon benefits and community benefits

3.7.1. Costs of CFUGs for REDD+

The trade-off between additional costs to communities (including foregone benefits, effort/contribution made for forest management) for the REDD+ and carbon revenue was analysed to investigate the cost and benefit aspects in REDD+ CFs. Additional costs to communities were estimated from real time cost and benefit differences between 3 years before the REDD+ (July 2006- July 2009) and 3 years during the REDD+ (July 2009- July 12). The estimated cost was compared against possible carbon revenue. The carbon revenue was estimated from carbon stock change (2010–2013) and the expected market price of the carbon credits.

In CFs, forest products extraction and grazing are the main direct benefits to communities (Bahuguna 2000; Adhikari *et al.* 2007; Cooke *et al.* 2008; Sapkota & Odén 2008), while the labour contribution of communities to forest management is the main cost (Bhattarai 2011). The data related to both benefits and contributions were obtained by reviewing records and other documents maintained by communities and projects and FGDs (Table 3.6); these were analysed in the following ways:

Table 3.6 Items and data sources used to estimate real time benefits and costs of the communities in CFs for six year (2006 -2012	i.e.
three years before and three years during the REDD+ projects)	

	Item heading	Data collection (annual)
Benefits	Timber	Reviewed records of CFs and asked species-wise annual total extraction of timber (m ³)
	Grasses	Asked annual collection (Bhari)
	Fodder	Asked total annual collection of fodder in CF (Bhari)
	Litter	Asked annual quantity collection in CF (Bhari)
	Fuel wood	Reviewed records of CFs and asked annual collection of fuel wood (Bhari)
	NTFPs	Total annual income from records
	Livestock grazing (buffalo, cow, goat)	Asked annual total number of buffalo, cow and goat grazing days in CF and Additional grass
		required for one day during stall feeding livestock
Contribution	General assembly	Reviewed records of annual total man-days and other related cost in CF
	Executive committee meeting	Reviewed records of annual total man-days and other related cost in CF
	Plantation	Reviewed records and asked annual labour and other materials cost in CF
	Forest fire control	Reviewed records of annual labour and other materials cost in CF
	Silviculture operation	Reviewed records of annual labour and other materials cost in CF
	Harvesting of forest products	Reviewed records and asked annual costs in CF
	Guarding	Reviewed records and asked annual labour cost in CF

a) Costs of communities due to changed forest product use behaviour or sacrificed benefit for REDD+

Although the fundamental aim of CF is to provide forest products for subsistence needs and improve forests, forest product use practices may have changed with the REDD+ project. These changed practices have added costs to communities which were estimated to determine sacrificed benefit. The annual quantity of all forest product use in each CF was collected for all 6 years and changes in the quantities of forest product harvested were estimated from the difference between the average of annual quantity of harvest in the 3 years before the REDD+ and the average of annual harvest during the 3 years of REDD+ project period. Total changed benefits were estimated by multiplying the change in quantity by a per unit price. The study used the 2012 price to estimate the changes in relation to the latest value. Similarly, change in contribution was estimated from the average of annual contribution cost (i.e labour and material cost).

The unit prices of timber, firewood, grasses, fodder and litter in year 2012 were different in sites (Table 3.6). Timber prices were highest for *Shorea robusta* and less for other timber trees (Pine, *Schima* and *Castanopsis*). Prices were collected from 15 random individuals from each study site and the average used. Local sawmill owners and carpenters were the key respondents for timber prices while local restaurants and firewood sellers provided firewood prices. Sawmill owners and carpenters were less than 15 in number in all three sites therefore local people who were familiar with the timber business were also consulted. Similarly, prices per unit of grasses (*Bhari*), fodder (*Bhari*) and litter (*Bhari/ Doko*) were asked from 15 local people at each site and the average used. Although grasses, fodder and litter were not commonly sold in the study areas, the price was estimated from willingness of communities to buy a unit of product and/or considering the time value required for collection from the forests.

	Price (US\$) in 2012						
Forest product	Gorkha	Chitwan	Dolakha				
Timber (per m ³) -Shorea robusta	1221.47	1495.67	0.00				
Dalbergia sissoo	0.00	373.86	0.00				
Terminalia alata and Adina cordifolia	0.00	274.16	0.00				
-Other spp**(per m ³)	186.96	207.73	207.73				
Fire wood per Bhari	1.80	2.35	1.76				
Grass/fodder per Bhari	1.30	1.42	1.30				
Litter per Bhari	0.71	0.82	0.71				

Table 3.7 Unit price of forest products in 2012 in the study areas

Note: 1 US\$= 85 NRs (exchange rate of 2012); ** other spp includes Schima wallichii, Pinus spp, Alnus nepalensis,

In the case of NTFPs, CFUGs sell them and gain income. Total annual incomes of each CF from the NTFPs were collected from community records.

Grass and fodder requirements are higher for buffalo, followed by cow/ox and goat, respectively (Das & Shivakoti 2006). The price of grazing benefit in each year in CFs was estimated for each livestock type (Buffalo, cow/ox, goat) by identifying additional grass required to feed each livestock during a non-grazing day and the price of the grass. During group discussions, total grass required for individual buffalo, individual cow/ox and individual goat in a day during grazing day and non-grazing day was asked. Those differences provided the total additional quantity of grasses required for each individual livestock type during a non-grazing day. Separate estimation of costs was made for each livestock type (buffalo or cow/ox or goat) and added to get the total cost in CF due to stall feeding. For example, total annual non-grazing days of additional buffalo used in a CF for REDD+, the additional units of grasses required for those buffalo during a non-grazing day and the unit price of grasses. A similar process was followed for ox/cow and goat.

b) Cost of CFUGs due to reduced forest product harvesting practices for REDD+

With reductions in the quantity of forest products harvested to increase REDD+ carbon benefits, CFUGs had reduced the cost of harvesting. The reduced harvesting cost was a benefit to CFUGs. In hill areas, individual members of CF who need timber themselves do harvesting and logging and pay the required costs. Therefore, hill CFUGs have no institutional harvesting costs applied before or after the REDD+ project. In contrast, CFUGs in terai harvest timber and distribute it to those individuals who need it. Therefore, harvesting costs were considered only for terai CFs. Annual labour costs and other material costs incurred for timber harvesting in each terai CF were collected. The difference between the average cost (3 year before and 3 year during REDD+ project) of timber harvesting in terai CFUGs (i.e. Kayerkhola) were estimated to investigate any changes.

c) Cost to communities due to changed forest management activities or changed contribution for REDD+

Man-days (wages) and other cost (foods and materials) spent for GAs and executive committee meetings, plantation, forest fire control, silviculture operation, and guarding activities were analysed for individual CFs for each year. As for the forest products prices, the wage rates of year 2012 were used to estimate the total changed value of contributions for the REDD+ project. Although communities have different skills, they could have different opportunity costs. However, CFUGs have set practices that they schedule meetings and GAs during free time for the majority of people (that includes days off for the job holders). Furthermore, the majority of participants are farmers and there are limited opportunities for income and paid jobs in rural areas. Therefore, the study assumed the same opportunity cost equivalent to agricultural labour for both skilled and non-skilled participants. Average wage rates (in 2012) for the study areas were estimated from 15 random respondents from each district.

3.7.2. Analysis of total costs and carbon benefits in REDD+ CFs

The total real time costs or sacrificed benefits of each CF for REDD+ was estimated by adding the costs for changed forest products use and costs for changed forest management practices, and subtracting added benefits. In many studies, a per hectare basis is used for carbon stock estimation for forests (Tan *et al.* 2010; Maraseni *et al.* 2010; Maraseni & Cockfield 2011; Maraseni & Xinquan 2011); therefore all net sacrificed benefit or cost of communities are converted into a cost per hectare basis. Similarly, estimated carbon stock changes in CFs by species dominated forests are also converted into carbon dioxide (CO₂) equivalent per hectare to understand community added costs for each unit of carbon stock increment.

3.7.3. Carbon gain estimation in CFs

The study used the 100 year time horizon rule to estimate carbon sequestration potential in CF. Actual CO₂ sequestration from the atmosphere in the forests was estimated by dividing the total estimated carbon by a constant 100 (Fearnside 1997). The 100 year time horizon is a timeframe over which temporary carbon must be stored in a forest to be considered equivalent to a permanently avoided emission (Costa & Wilson 2000; Fearnside 2000; Fearnside 2002). This is important in forestry projects because there is uncertainty about carbon saving in forestry. Trees can easily be cut or burned so that their accumulated carbon is re-released. However, this is not so in the case of energy projects which save fossil fuel usage and avoid related emissions permanently in a sense that if once saved, they can never be lost again. Therefore carbon estimation in forestry related carbon projects need some adjustments within a time horizon. In forestry projects, 100 years could be a reasonable timeframe to ensure permanence (i.e. longevity of a carbon pool and the stability of its stocks under the given management and disturbance environment in which it occur) which was discussed for GHGs emissions estimation in different literatures (Fearnside 2002; IPCC 2006; Maraseni 2007; UNFCCC 2007; Kollmuss et al. 2008).

3.7.4. Price used in carbon revenue estimation

Compliance markets (mandatory and regulatory that include CDM and EU-ETS) and voluntary carbon markets (non-regulatory but operate with agreement between seller and buyer) are now functioning to offset GHGs emissions. The market price of carbon stocks on these markets fluctuates. During the first commitment period of the Kyoto protocol (2008–2012), prices were higher at the beginning and lower in later years. The CDM market price was US\$23/MgCO₂e in 2008 (Capoor & Ambrosi 2009) and fell to US\$0.26/MgCO₂e in 2013 (World-Bank 2013). On the voluntary market, carbon price ranged from US\$1.20–46.9/MgCO₂e in 2008 (Hamilton *et al.* 2009) and US\$5.90/MgCO₂e in 2013 (Peters-Stanley & Yin 2013). Though the European Union Emission Trading Scheme (EU-ETS) does not accept forestry sector carbon at present, there is the possibility this will be included in future schemes;

EU-ETS had higher prices (i.e above \$45/MgCO₂e ton) in 2008 which also fell (i.e \$6/MgCO₂e) in 2013.

In addition to the CDM and the voluntary mechanism for afforestation and reforestation projects, the REDD+ mechanism is evolving as a market mechanism. The REDD+ mechanism was initiated as a voluntary market mechanism and one of the pilot projects was offered US\$7-8/MgCO₂e for forward credit in 2013 (World Bank 2013). However, this price may not compensate all costs, global negotiations are striving to bring better incentives to those REDD+ projects which are generating both environmental and socio-economic benefits. This is more likely to get priority for the carbon credits generated particularly in community managed forests of least developed countries and may generate a premium price. Three assumptions have been made in this study and three possible prices used for carbon credit revenue; 1) an optimistic situation for a better future carbon market mechanism giving priority to CF (securing biodiversity and community socio-economic aspects); an optimistic price of US\$20/MgCO₂e was used; 2) continuing with the present situation with REDD+ carbon credits priced as mentioned in World Bank (2013); a price of US\$7/MgCO₂e was used; and 3) a worst situation of the carbon market which assumed that REDD+ CFs got the lowest value mentioned in the range of voluntary market prices; a price of US\$1.2/MgCO₂e was used for analysis.

3.8. Conclusions

This chapter described the complete methodology adopted for this study. In order to achieve the goals of the study, both primary and secondary data were analysed. The primary data used included carbon pools, biophysical and socio-economic aspects of CFs, and changed practices of CFUGs for REDD + projects. Similarly, the specific gravity of tree species, dry to fresh weight ratio or moisture content in the wood of various tree species, default values and ratios for carbon fraction and above to below ground ratio and data analysis techniques were taken from the relevant literature. Methods used to analyse the distribution of different sized trees, estimate carbon stock and change in carbon stock, estimate gaps in current CFs compared with the potential growth of the forest without disturbance, analyse factors and their possible effects on carbon stock and change in carbon stock in CFs and analyse trade-off between community foregone sacrificed benefits and carbon gains were presented. All of these analyses were made by dominant vegetation types. Results of the analysis are presented in the following chapters.

CHAPTER FOUR

DYNAMICS OF CARBON STOCK IN REDD+ COMMUNITY FORESTS

4.1. Introduction

Chapter 3 discussed the methodology used in this study. This chapter presents the dynamics of distribution of different size trees, carbon stock and changes in carbon stock in the study area REDD+ CFs. The chapter focuses on results associated with two questions: What is the present distribution pattern of trees by diameter size class in REDD+ CFs? What is the carbon stock and change in carbon stock in REDD+ CFs by vegetation type? It attempts to answer these questions by analysing forest inventory data.

This chapter is divided into different four sections. The next section gives an overview of stocking rates of different sized trees in CFs, the third section gives knowledge of biomass and carbon stock dynamics in the sample plots by dominant vegetation type and the fourth section gives knowledge of biomass and carbon stock in individual CFs by dominant vegetation and the last section concludes the chapter.

4.2. Distribution of different size trees in CFs

Stocking rates of the forests are analysed and presented in Table 4.1 and Table 4.2. Forest vegetation were divided into regeneration (rooted to <1 cm dbh), sapling ($\geq 1-5$ cm dbh) and tree (≥ 5 cm dbh) and analysed by vegetation type.

The inventory for the reference year 2010 showed that regeneration, sapling and tree sized plants were distributed in a pattern of a higher stocking rate of smaller sized plants and gradual reduction towards larger ones. Among five vegetation types, as expected, the highest numbers of regeneration were found in dense forests in all vegetation types except S. robusta dominated forests. In S. robusta forests, regeneration numbers were higher in sparser strata in year 2010 which could be a result of the species' light demanding characteristics and the open space available to foster seedlings on the forest floor. Similarly, a gradual decrease in the stocking rate of regeneration was found in forests from dominant S. robusta vegetation to mixed broadleaf, Schima-Castanopsis, Pine and Rhododendron-Quercus forests. In the case of saplings, these were highest under dense canopy cover in the mixed broadleaf forests, Schima-Castanopsis forests and Rhododendron-Quercus forests, while in S. robusta dominated and *Pine* dominated forests sparse canopy cover tended to have greater stocking rates of saplings. As expected, tree stocking rates in the year 2010 were highest under dense canopy in all forests. Among the different vegetation types, the stocking rate of trees was highest in S. robusta forests followed by Schima-Castanopsis, Rhododendron-Quercus, mixed broadleaf and Pine forests respectively.

Table 4.1 Comparison of stocking rates of woody vegetation by species and their strata in the REDD+ CFs (in year 2010 and 2013) in the study sites

Dominant	Type of	No of	Stoking Reg ¹ (g rate of No/ba)	Stoking Sanling	g rate of 2 (No/ba)	Stoking Tree ³ (rate of
vegetation type	strata	Plot	Year 2010	Year 2013	Year 2010	Year 2013	Year 2010	Year 2013
	Dense	154	54161	41112	1738	2324	1336	1716
Shorea robusta	Sparse	42	58614	47164	2093	2272	1035	1327
	Total	196	55115	42409	1814	2312	1271	1632
Maria	Dense	53	40260	34071	1281	1647	639	973
Mixed	Sparse	10	36943	32803	880	1280	514	824
broadleaf*	Total	63	39733	33871	1217	1588	619	949
C - 1 , 1 ,,	Dense	98	31815	24991	1815	2565	1082	1597
Schima-	Sparse	18	27070	27070	1572	1511	706	1178
Castanopsis	Total	116	31078	25313	1778	2402	1023	1532
	Dense	30	19533	18365	977	1530	642	976
Pine	Sparse	12	11412	16720	1167	2009	503	845
	Total	42	17213	17895	1031	1667	602	939
וווומ	Dense	60	18153	15605	2370	2485	1066	1467
Rhoaoaenaron-	Sparse	13	3185	17149	2300	1931	653	920
Quercus	Total	73	15487	15880	2357	2386	993	1370
	Dense	395	38652	30565	1734	2257	1086	1493
Total	Sparse	95	36809	33894	1778	1943	798	1129
	Total	490	38295	31211	1742	2196	1030	1422

Note:*species include *Lagerstroemia parviflora* (Botdhairo), *Terminalia* spp, *Anogeissus* latifolia (Bajhi), *Dalbergia* sissoo, *Syzygium cumini* (Jamun); ¹regeneration means woody plants with <1cm dbh; ²sapling means woody plants with 1-5cm dbh and ³trees means woody plants with \geq 5cm dbbh; dense means forests with \geq 70% canopy cover; sparse means forests with <70% canopy cover

After three years of the REDD+ project intervention, there was a negative change in the stocking rate of regeneration found in all forests except sparse *Schima-Castanopsis* forests, *Pine* forests and *Rhododendron-Quercus* forests where numbers increased (Table 4.2). One possible reason behind that increase could be the favourable condition of the forest floor for germination. Open space available in sparse forests admits sunlight and rain water at ground level which may encourage germination and growth of new seedlings. However, sparse canopy types of *S. robusta* forests and mixed broadleaved forests may have had negative changes in regeneration because they already have been abundant in the reference year. Good seed years for particular species could be another reason for such differences in the stocking rates of regeneration and saplings.

A positive change in the stocking rate of saplings was found in all forests with both canopy categories except sparse canopy of *Schima-Castanopsis* and *Rhododendron-Quercus* forests. There could be several reasons for that exception, including: less recruitment of the new saplings with positive changes in regeneration, thinning and cleaning activities of communities; harvesting of smaller sized trees from sparse canopy forests for wooden handled agricultural equipment and fencing in middle and higher altitudes (*S. robusta* is preferred for wooden handles due to its strength; however people harvest *Schima* and *Castanopsis* trees in areas where *Shorea* is not available); and sparse canopy sites were located near walking trails where people could easily harvest and remove smaller trees. The stocking rate of saplings was

higher in sparse canopy *Pine* forests followed by dense canopy types of *Schima-Castanopsis* forests which are comparatively fast growing by nature.

The stocking rates of trees increased in all CFs but the increase rates differed (Table 4.2). *Rhododendron-Quercus* dominated forests with sparse canopy had less positive change in tree numbers followed by *S. robusta* dominant sparse canopy forest. This could be due to site quality of forests, slow growth character of the species and the forest products use and management practices of communities. The highest stocking rate of trees was observed in *Schima-Castanopsis* forests which is a fast growing species. Communities may also have changed harvesting practices. Communities were using forest resources to fulfil their daily needs but did harvest trees for timber only if user members needed to build or renovate a house. In general people do not build houses every year so they do not harvest timber trees on a regular basis. *S. robusta* is considered to be a high value timber tree and there was some illegal harvesting by outsiders observed in the forests.

Table 4.2	Comparison	of change	in st	ocking	rates	(from	2010	and	2013)	of	woody
vegetation	by vegetation	n types and	thei	r strata	in the	REDI	D+CF	s in	the stu	dy :	sites

	Type		Regener	Regeneration ¹		ng ²	Tree ³	
Dominant vegetation type	of strata	No of Plot	Change (No/ha)	SD	Change (No/ha)	SD	Change (No/ha)	SD
Shorea robusta	Dense	154	-13049	35560	586	1703	380	374
shorea robusia	Sparse	42	-11450	27906	179	1653	292	261
	Total	196	-12706	34005	498	1696	361	354
Mixed	Dense	53	-6189	32570	366	983	334	145
broadleaf	Sparse	10	-4140	20642	400	1028	310	166
including Sal*	Total	63	-5862	30857	371	982	330	148
Sohima	Dense	98	-6824	26555	750	1626	515	411
Castanonsis	Sparse	18	0	33615	-61	825	472	349
$\begin{array}{c cccccc} & Total & 196 & -12700 \\ \hline Mixed & Dense & 53 & -6189 \\ broadleaf & Sparse & 10 & -4140 \\ including Sal* & Total & 63 & -5862 \\ Schima- & Dense & 98 & -6824 \\ Castanopsis & Total & 116 & -5765 \\ Dense & 30 & -1168 \\ Pine & Sparse & 12 & 5308 \\ Total & 42 & 682 \\ \end{array}$	-5765	27713	624	1555	509	401		
	Dense	30	-1168	21265	553	923	334	310
Pine	Sparse	12	5308	20203	842	925	342	173
Vegetation typestrataPlotChange (No/ha)SDChange (No/ha)SDChange (No/ha)Shorea robustaDense154-13049355605861703380Sparse42-11450279061791653292MixedDense53-618932570366983334broadleafSparse10-4140206424001028310including Sal*Total63-586230857371982330Schima- CastanopsisDense98-6824265557501626515Dense98-68242655553923334PineSparse18033615-61825472Total116-5765277136241555509Dense30-116821265553923334PineSparse12530820203842925342Total4268220931636922337Dense60-2548172431151258401Sparse131396410920-3691271267Total7339317435291265377Dense395-8087300215231499407Total5parse95-2915270221651355331Total490-7084 <t< td=""><td>276</td></t<>	276							
Rhadadandran	Dense	60	-2548	17243	115	1258	401	195
Quercus	Sparse	13	13964	10920	-369	1271	267	273
Quercus	Total	73	393	17435	29	1265	377	215
	Dense	395	-8087	30021	523	1499	407	341
Total	Sparse	95	-2915	27022	165	1355	331	269
	Total	490	-7084	29508	454	1478	392	329

Note: *species include *Lagerstroemia parviflora* (Botdhairo), *Terminalia* spp, *Anogeissus latifolia* (Bajhi), *Dalbergia sissoo*, *Syzygium cumini* (Jamun); ¹regeneration means woody plants with <1cm dbh; ²sapling means woody plants with 1-5cm dbh and ³trees means woody plants with \geq 5cm dbbh; dense means forests with \geq 70% canopy cover; sparse means forests with <70% canopy cover; SD means standard deviation; - sign means negative changes; no sign means positive changes

4.3. Biomass and carbon stock dynamics in sample plots

Carbon stock and its changes are analysed and presented by dominant vegetation types. Different results were observed in plots by vegetation type in year 2010, 2011,

2012 and 2013 (Appendix F). These data have been further analysed and results for the reference year 2010 and changes with REDD+ intervention are presented in the sub-sections below.

4.3.1. Total biomass carbon in the reference year 2010 and year 2013

Average carbon stock (both above and below ground) was higher in dense canopy forests than in sparse. The quantity of biomass carbon was also found to differ by dominant vegetation type (Table 4.3). CFs with *S. robusta* dominant plots had, on average, higher quantities of carbon stock followed by *Rhododendron-Quercus* forests, *Pine* forests and *Schima-Castanopsis* forests, respectively. Carbon stock in sample plots ranged from 0.6 MgC/ha in sparse canopy *Pine* forests to 345.1 MgC/ha in dense canopy mixed broadleaf forests in year 2010; these values increased in year 2013 (Table 4.3).

The quantity of carbon stock was higher in dense forests than sparse. For example, in *Pine* forests, dense canopy forests had 11.3 to 230 MgC/ha which was 10.7 to 45.5 MgC/ha higher than sparse canopy in year 2010. This increased to 18.4 to 127.0 MgC/ha in year 2013 which is 15.1 to 56.5 MgC/ha higher than sparse forests. These results indicate that carbon stock in CFs can increase with intervention.

Average carbon stock in dense canopy forests of each dominant vegetation type differed significantly (p < 0.05) from sparse canopy forests in both years 2010 and 2013 (Table 4.3). Since sparse forest areas were smaller than dense, the numbers of sample plots were also less. Therefore, the Shapiro-Wilks normality test was carried out for sparse category forests before t-tests were conducted for year 2010 data. The test confirmed the normal distribution of samples with p value > 0.05 (i.e. 0.19 for *S. robusta*, 0.59 for mixed broadleaf, 0.11 for *Schima-Castanopsis*, 0.08 for *Pine* and 0.06 for *Rhododendron-Quercus* forests.

Dominant	True of	No of	Bio	mass carbon in	year 2010	Bi				
vegetation type	strata	Plot	Avg biomass C (MgC/ha)	Minimum (MgC/ha)	Maximum (MgC/ha)	P-value	Avg biomass C	Minimum (MgC/ha)	Maximum (MgC/ha)	P-value
	Dense	154	129.9	2.9	326.6		146.8	1.3	364.3	
Shorea robusta (sal)	Sparse	42	89.2	2.3	198.9	0.00	97.8	6.3	203.8	0.00
	Total	196	121.2	2.3	326.6		136.3	1.3	364.3	
	Dense	53	118.0	6.0	345.1		136.9	12.8	356.9	
Mixed broadleaved*	Sparse	10	69.5	8.0	165.6	0.03	79.1	14.7	174.1	0.01
	Total	63	110.3	6.0	345.1		127.7	12.8	356.9	
	Dense	98	95.2	10.8	277.7		116.0	30.4	300.3	
Schima- Castanopsis	Sparse	18	48.3	8.8	126.3	0.00	54.1	9.9	110.9	0.00
	Total	116	87.9	8.8	277.7		106.4	9.9	300.3	
	Dense	30	103.0	11.3	230.6		127.0	18.4	225.3	
Pine	Sparse	12	62.4	0.6	185.6	0.02	70.5	3.3	185.9	0.00
	Total	42	91.4	0.6	230.6	0.05	110.8	3.3	225.3	0.00
Dhododau duau	Dense	60	114.7	21.1	254.8	0.00	121.7	18.8	254.5	0.00
Khoaoaenaron-	Sparse	13	48.2	3.7	151.7	0.00	54.2	6.4	174.4	0.00
Quercus	Total	73	102.9	3.7	254.8		109.7	6.4	254.5	
	Dense	395	115.3	2.9	345.1		132.5	1.3	364.3	
Total	Sparse	95	70.4	0.6	198.9	0.00	78.1	3.3	203.8	0.000
	Total	490	106.6	0.6	345.1		122.0	1.3	364.3	

Table 4.3 Comparison of biomass C (both above and belowground, in year 2010 and 2013) by vegetation and their strata in CFs in the study areas

Note:*species include Lagerstroemia parviflora (Botdhairo), Terminalia spp, Anogeissus latifolia (Bajhi), Dalbergia sissoo, Syzygium cumini (Jamun)

A pair wise comparison of the probability of having the same mean between different forests by vegetation types and canopy cover types was carried out (Table 4.4). Results indicate that mean carbon stocks in year 2010 differed significantly between dense and sparse canopy samples (within each species dominated forest). Comparison between different dominant vegetation types indicated that the mean carbon stock of each forest category also differed significantly from the mean of at least three other forest categories. These differences were higher in the dense canopy of *S. robusta* forests (i.e. significantly different from six other vegetation types) followed by sparse canopy of *Schima- Castanopsis* and sparse canopy of *Rhododendron-Quercus* forests (i.e. significantly different from five other forests) and lowest in dense canopy of *Pine* forests (i.e. significantly different from three other forests) (Table 4.4).

Dominant vegetation types	S. robusta sparse	Mix broadleaf dense	Mix broadleaf sparse	Schima- Castanopsis dense	Schima- Castanopsis sparse	<i>Pine</i> dense	<i>Pine</i> sparse	Rhododendron- Quercus dense	Rhododendron- Quercus sparse
S. robusta dense	0.00	0.38	0.01	0.00	0.00	0.03	0.00	0.17	0.00
S. robusta sparse		0.05	0.32	0.59	0.00	0.32	0.12	0.05	0.02
Mix broadleaf dense			0.03	0.09	0.00	0.34	0.00	0.82	0.00
Mix broadleaf sparse				0.18	0.28	0.11	0.75	0.03	0.34
Schima-Castanopsis dense					0.00	0.52	0.00	0.07	0.01
Schima-Castanopsis sparse						0.00	0.40	0.00	1.00
Pine dense							0.03	0.39	0.00
Pine sparse								0.00	0.47
Rhododendron-Quercus dense									0.00

Table 4.4 Pair-wise comparison of p value (assuming unequal variance, two tail) for having similar mean carbon stock (difference in mean =0) in CFs by vegetation and canopy cover differences in 2010

Note: Bold p-value shows significant difference (<0.5)

4.3.2. Change in carbon stock in sample plots in CFs between 2010 and 2013

On average, carbon stock was found to increase between 2010 and 2013 in all forest types in the REDD+ CFs. In plot-wise analysis, the mean of carbon stock changes were higher in dense canopy forest then sparse. Specifically, these changes were higher in dense canopy *Pine* forests followed by dense canopy *Schima-Castanopsis* forests, dense canopy mixed broadleaf forests, and dense canopy *S. robusta* forests, respectively. Among all dense canopy vegetation types, dense *Rhododendron-Quercus* forests showed less increase, while in sparse canopy forests, mixed broadleaf forests showed the highest rate of increase followed by *S. robusta*, *Pine, Rhododendron-Quercus*, and *Schima-Castanopsis* forests.

Changes ranged from extreme values of -29.0 MgC/ha (negative) to 49.6 MgC/ha (positive) in the experimental plots. Higher positive change was estimated in dense canopy *Pine* forests followed by dense canopy *Schima-Castanopsis* forests and *S. robusta* forests, while lower positive changes were estimated in sparse canopy *Rhododendron-Quercus* forests. On the other hand, higher negative changes were found in dense canopy *S. robusta* forests followed by dense canopy *Pine* forests. These negative changes were mainly due to forest use provisions and regular silvicultural activities (removal of malformed trees) of communities' excess to yield in many cases except in *S. robusta* forests which is commercial timber tree and illegal activities may also have been going on due to heterogeneous communities and road access.

Similar to the analysis of carbon stock in the reference year 2010, the normality test for changes in carbon stock data was carried out for sparse vegetation. The Shapiro-Wilks test showed those samples were normally distributed in all vegetation types with p-value > 0.05 (i.e 0.47 for mixed broadleaf, 0.07 for *Pine*, 0.09 for *Rhododendron-Quercus*, 0.26 for *S. robusta*, 0.97 for *Schima-Castanopsis*). Comparison of mean changes in carbon stock between canopy covers (dense and sparse) within each vegetation type found significant differences in the majority of cases except between mixed broadleaf forests and *Rhododendron-Quercus* forests (Table 4.5).

Table 4.5 Comparison of changes in biomass C in three year (between 2010 and 2013, both above- and-belowground) by vegetation types and their strata in CFs in the study areas

Dominant vegetation type	Type of strata	No of Plot	Change in biomass C	Minimum (MgC/ha)	Maximum (MgC/ha)	SD	P-Value
	Dansa	154	(NIGC/IIA)	20.0	18 5	21.0	0.02
Shorea robusta	Sparse	134	87	-29.0	40.5	10.6	0.02
Shorea robusia	Total	196	15.1	-29.0	44.5	21.6	
	Dense	53	18.9	-20.7	45.2	16.5	0.13
Mixed broadleaf *	Sparse	10	9.6	-16.1	41.4	16.7	0110
	Total	63	17.4	-20.7	45.2	16.8	
	Dense	98	20.9	-17.4	49.1	16.5	0.00
Schima- Castanopsis	Sparse	18	5.7	-15.4	27.6	11.5	
1	Total	116	18.5	-17.4	49.1	16.7	
	Dense	30	24.0	-21.9	49.6	17.4	
Pine	Sparse	12	8.1	-3.4	27.4	9.7	0.00
	Total	42	19.4	-21.9	49.6	17.1	
Dhododau duau	Dense	60	7.0	-28.0	39.5	16.5	0.77
Rhoaoaenaron-	Sparse	13	5.9	-16.7	22.7	9.8	
Quercus	Total	73	6.8	-28.2	39.5	15.5	
	Dense	395	17.2	-29.0	49.6	19.4	
Total	Sparse	95	7.8	-26.4	44.9	15.6	0.00
	Total	490	15.4	-29.0	49.6	19.1	

Note: *species include *Lagerstroemia parviflora* (Botdhairo), *Terminalia* spp, *Anogeissus latifolia* (Bajhi), *Dalbergia sissoo*, *Syzygium cumini* (Jamun); Bold p-value shows significant difference (< 0.5)

In pair-wise comparison among dominant vegetation types, mean carbon stock in sample plots of a particular vegetation type were found to be significantly different from others (either with different canopy cover or with different vegetation types) (Table 4.6). Changes in carbon stock in dense canopy *Pine* forests were highly significantly different to those from all other forest types except dense canopy *S. robusta, Schima-Castanopsis* dense and mixed broadleaf forests.

Table 4.6 Pair-wise comparison of p- value (assuming unequal variance, two tail) for having similar mean carbon stock (difference in mean =0) for having same changes in biomass carbon in CFs (from 2010 to 2013) by vegetation and canopy cover

Dominant vegetation types	Shorea sparse	Mixed broadleaved	Mixed broadleaved	Schima- Castanopsis	Schima- Castanopsis	Pine dense	Pine sparse	Rhododendron- Quercus dense	Rhododendron- Quercus sparse
		dense	sparse	dense	sparse				
Shorea dense	0.02	0.49	0.22	0.10	0.00	0.06	0.01	0.00	0.00
Shorea sparse		0.01	0.87	0.00	0.48	0.00	0.90	0.65	0.51
Mixed broadleaf dense			0.13	0.49	0.00	0.20	0.31	0.00	0.00
Mixed broadleaf sparse				0.07	0.52	0.03	0.80	0.65	0.54
Schima-Castanopsis dense					0.00	0.40	0.00	0.00	0.00
Schima-Castanopsis sparse						0.00	0.55	0.73	0.96
Pine dense							0.00	0.00	0.00
Pine sparse								0.74	0.58
Rhododendron-Quercus dense									0.77

Note: Bold p-value shows significant difference (< 0.5)

4.4. Biomass and carbon stock in individual CF

Carbon stock in CFs was estimated from the relationship between the area occupied by a forest type (based on altitude category and canopy category) and carbon stock estimated for that type (Table 4.7).

Altitude of different forests by each canopy type in CFs	Carbon stock in 2010 (MgC/ha)	Carbon stock in 2011 (MgC/ha)	Carbon stock in 2012 (MgC/ha)	Carbon stock in 2013 (MgC/ha)
<1000m (≥70% canopy) (N=223)	125.06	133.54	136.32	140.64
<1000m (<70% canopy)(N=66)	74.07	78.79	79.22	82.74
1000-2000m (≥70% canopy)(N=78)	87.40	102.83	110.29	117.09
1000-2000m(<70% canopy) (N=16)	62.93	64.34	70.68	72.57
>2000m (≥70% canopy)(N=94)	115.44	122.68	123.04	126.08
>2000m (<70% canopy)(N=13)	60.76	58.48	58.95	62.65

 Table 4.7 Average carbon stock (MgC/ha) in different canopy forests within different altitudes estimated from sample plots

Comparing the plot-wise analyses for particular forest type, higher carbon stocks were estimated in dense canopy forests than sparse. Among dense canopy forests, forest at altitudes <1000 m had higher carbon stocks in all four years followed by those at altitudes of >2000 m and 1000–2000 m. In sparse canopy types, higher carbon stock was estimated in forests at altitudes of <1000 m followed by 1000–2000 m and >2000 m, respectively (Table 4.7).

For individual CFs, higher carbon stocks were estimated in large *S.robusta* forests in the reference year 2010 and 2013, followed by *Pine* forest, *Schima-Castanopsis* forests and *Rhododendron* forests, respectively. This study estimated different carbon stock within a particular forest type; for example, carbon stock in *S.robusta* dominated CFs ranged from 76.9 to 124.8 MgC/ha in year 2010. Such differences were also evident in the year 2013 (Table 4.8).

Table 4.8 Average biomass carbon (above and below ground, in year 2010 and2013) in CFs by dominant vegetation types in the study areas

CF by	Number	Carbon stock in 2010			Carbon stock in 2013			
dominant vegetation	of CF	Average (MgC/ha)	Minimum (MgC/ha)	Maximum (MgC/ha)	Average (MgC/ha)	Minimum (MgC/ha)	Maximum (MgC/ha)	
Shorea forest	37	114.1	76.9	124.8	128.5	86.8	140.4	
Schima- Castanopsis	43	86.0	62.9	123.1	101.6	72.6	138.4	
Pine	16	89.0	63.9	122.9	106.6	74.3	138.3	
Rhododendron- Quercus	9	77.4	62.9	98.5	88.0	72.6	106.9	

Since each CF encompasses multiple categories of forest patches, vegetation categories were grouped on the basis of dominant vegetation type in terms of area coverage. All CFs had positive changes in carbon stock but the estimated levels of changes differed. Annual changes in biomass carbon in CFs were highest in *Pine* forests followed by *Schima-Castanopsis* forests and lowest in *Rhododendron-Quercus* forests. The study showed the changes varied from 1.9 to 9.7 MgC/ha in a year (Table 4.9 and Appendix F). This indicates that the rates of carbon stock change in the CFs can increase with appropriate interventions. Communities were implementing a range of forest management activities and these, and the use of forest resources, might need review to plan for better intervention.

CF by dominant	Number	Annual	Minimum	Maximum	SD
vegetation	of CF	change	(MgC/ha)	(MgC/ha)	
Shorea forest	37	4.7	3.0	5.2	0.5
Schima-Castanopsis	43	5.1	2.0	9.7	2.1
Pine	16	5.8	2.1	9.4	2.5
Rhododendron-Quercus	9	3.3	1.9	5.6	1.3

Table 4.9 Change in biomass carbon (above and below ground, between 2010 and2013) in CFs by dominant vegetation types in the study areas

4.5. Conclusion

Forest type and canopy cover differences maintained different quantities of biomass carbon stock and exhibited different increment rates with REDD+ activities. Dense canopy forests had the highest density of trees and therefore maintained a higher carbon quantity in the reference year 2010 compared to sparse canopy forests. In term of increments, all CFs across the pilot REDD+ project sites contributed to increases in carbon stocks; however, both total carbon stock and the carbon increments varied by vegetation types. The CFs with *S. robusta* and mixed broadleaf forests were shown to have higher carbon stocks than either *Schima-Castanopsis* or *Pine* forests in the reference year of the REDD+ project. However, average increments in carbon stocks after the REDD+ activities were highest in *Pine* and *Schima-Castanopsis* dominated forests than others. This shows that *Schima-Castanopsis* and *Pine* forests are able to get greater benefit from the REDD+ project in terms of additional carbon benefits.

In plot wise analysis, some plots showed a reduction in biomass carbon in the years following implementation of the REDD+ project. This shows that there are still forest biomass extraction activities going on in the forests as per the provisions and practices stipulated in the operational plans of CFUGs. However, in case of *S. robusta* forests, being commercial species illegal activities might have contributed to some extent in reducing biomass stock. If CFUGs could stop such activities, they would be able to produce more biomass carbon in their forests. But consideration of community forest resource needs and possible alternative means of accessing these may be needed before such action. In the next chapter, the potential growths of carbon stock in CFs are investigated to identify gaps in present carbon stock in CFs.

CHAPTER FIVE

POTENTIAL BIOMASS CARBON GROWTH IN COMMUNITY FORESTS
5.1. Introduction

The previous chapter presented results for stocking rates, carbon stock and change in carbon stock of the different forest types in the study. The overarching goal of this chapter is to analyse the gap between carbon stock growth under undisturbed natural conditions (i.e. "technical potential growth") and growth under CFs. This information is important in designing REDD+ projects in CFs. This chapter describes the current annual increment (CAI), mean annual increment (MAI), cumulative biomass carbon in both undisturbed forests and CFs (both dense and sparse canopy types).

The chapter has been divided into four sections. The next section gives the technical potential biomass carbon in *S. robusta* forests, the third section covers biomass carbon growth in CFs for all other vegetation types and the final section concludes the chapter.

5.2. Technical potential biomass carbon in *Shorea robusta* forests

5.2.1. Biomass carbon stock in undisturbed forests

The growth rate of biomass carbon is highest at the early stage of tree growth and least at older ages in undisturbed *S. robusta* forests (Figure 5.1). However, total carbon stock in this study was 807.69 Mg/ha at age 80 but only 76.53 Mg/ha at age 9. The Mean Annual Increment (MAI: the cumulative biomass carbon up to a certain age divided by that age) and the Current Annual Increment (CAI: the growth observed in a stand in a specific one year period) were found to be equal at the age of 37 after the die-back period (Figure 5.2). Therefore the maximum sustained yield was obtained at age 36 after the die-back period in undisturbed natural forests. This result indicates that, if the aim of forest management is to maximise economic returns from wood production, rotation age should be at age 36 because growth rates will gradually reduce and be less beneficial after 36 years of age. However, in the case of carbon sequestration, old age forests can make a positive contribution in terms of net CO₂ flux. Under the likely REDD+ incentive mechanism, old age forests will also be conserved.



Figure 5.1 Cumulative biomass carbon in undisturbed *S. robusta* forests (*adopted from Rautienen 1995*)



Figure 5.2 Mean Annual Increment (MAI) and Current Annual Increment (CAI) in undisturbed *S. robusta* forests (*adopted from Rautienen 1995*)

5.2.2. Biomass carbon growth in CFs

<u>Best fit growth curve for CFs:</u> Coefficients of determination (\mathbb{R}^2) calculated for dense *S. robusta* forests were 0.84 using the Richards, Gompertz, Weibull and Morgen-Mercer-Flodin (MMF) models and 0.83 using the logistic and Ratkowsky models. Although four models gave similar \mathbb{R}^2 values, the deviation of parameter value was comparatively less (i.e. less range gaps in 95% confidence) in the Gompertz model as was the number of parameters (i.e. three) (Table 5.1 and 5.2). Therefore the best fitted curve explaining biomass growth in dense *S. robusta* forests was given by the Gompertz model with parameters: relative maximum change (a) = 215.36; initial value of the parameter (b) =1.21; and growth rate parameter (c) = 0.06 (Table 5.1). Similarly, for sparse forests, \mathbb{R}^2 values for all models were similar but the Gompertz

model again used fewer parameters used and was selected as the best to estimate the biomass carbon growth in sparse forests. The parameters estimated and used for sparse forests were a=172.82, b=1.45 and c=0.06 (Table 5.2).

		Parameters								\mathbf{R}^2
Models	Models Fitted equations		a		b		c		d	
	-	Value	Range (95% confidence)	Value	Range (95% confidence)	Value	Range (95% confidence)	Value	Range (95% confidence)	
Richards	Y=a/(1+exp(b- c*x))^1/d	215.34	193.21 to 237.48	-5.29	-874.63 to 864.04	0.06	0.02 to 0.10	0.001	-1.29 to 1.30	0.84
Logistic	Y=a/(1+b*exp(- c*x))	207.05	196.41 to 217.70	9.69	6.18 to 13.20	0.09	0.07 to 0.10			0.83
Gompertz	Y=a*exp(-exp(b- c*x))	215.36	201.33 to 229.39	1.211	0.97 to 1.45	0.06	0.05 to 0.07			0.84
Weibull	Y=a-b*exp(-c*x^d)	223.32	188.37 to 258.28	241.04	153.39 to 328.69	0.01	-0.01 to 0.04	1.29	0.65 to 1.94	0.84
MMF	$y=(a*b+c*x^d)/(b+x^d)$	-0.11	-46.96 to 46.74	518.72	-1138.70 to 2176.13	249.47	193.64 to 305.29	1.85	0.88 to 2.82	0.84
Ratkowsky	y=a/(1+exp(b-c*x))	207.06	196.41 to 217.70	2.27	1.91 to 2.63	0.09	0.07 to 0.11			0.83

Table 5.1 Dynamic models of relative total biomass accumulation in dense S. robusta forests

Note: Gompertz model is found better as it has only three parameter and high R^2

					Parame	ters				\mathbf{R}^2
Models	- Fitted equations	Fitted equations a			b		c		d	
Withdels		Value	Range (95% confidence)	Value	Range (95% confidence)	Value	Range (95% confidence)	Value	Range (95% confidence)	-
Richards	Y=a/(1+exp(b- c*x))^1/d	171.27	117.64 to 224.91	-1.13	-34.06 to 31.80	0.07	-0.02 to 0.16	0.07	-2.06 to 2.20	0.81
Logistic	Y=a/(1+b*exp(- c*x))	158.85	134.83 to 182.87	17.23	1.09 to 33.37	0.11	0.06 to 0.15			0.81
Gompertz	Y=a*exp(-exp(b- c*x))	172.82	136.74 to 208.90	1.43	0.88 to 1.97	0.06	0.03 to 0.09			0.81
Weibull	Y=a-b*exp(-c*x^d)	167.85	121.20 to 214.49	165.12	93.69 to 236.56	0.002	-0.01 to 0.01	1.83	0.61 to 3.06	0.81
MMF	$y=(a*b+c*x^d)/(b+x^d)$	8.99	21.74 to 39.72	5206.58	26818.59 to 37231.75	186.22	111.90 to 260.54	2.50	0.55 to 4.46	0.81
Ratkowsky	y=a/(1+exp(b-c*x))	158.85	134.82 to 182.87	2.85	1.91 to 3.78	0.11	0.06 to 0.15			0.81

 Table 5.2 Dynamic models of relative total biomass accumulation in sparse S. robusta forests

Note: Gompertz model is found better as it has only three parameter, high R^2 and less ranges in parameter

<u>Biomass carbon in CFs:</u> Using the Gompterz model and estimated parameter values, growth trends of biomass carbon in dense and sparse canopy in dominant *S. robusta* CFs was projected and curves plotted. Both dense and sparse canopy forests show similar trends i.e. high increment at early ages and less in older age stands (Figure 5.3 and 5.4). In CFs, the estimated carbon quantity at age 9 in dense canopy forests was 31.05 Mg/ha while it was 16.93 Mg/ha in sparse forests (i.e. equates to a 14.12 Mg/ha gap). The carbon stock increased to 209.94 Mg/ha in dense and 169.13 Mg/ha in sparse canopy types by age 80 (i.e. a 40.81 Mg/ha gap). Comparing dense and sparse canopy forests, higher gaps were indicated in old aged forest stands than in young stands.



Figure 5.3 Cumulative biomass carbon in dense S. robusta dominated CFs



Figure 5.4 Cumulative biomass carbon in sparse S. robusta dominated CFs

MAI and CAI curves in dense and sparse canopy forests with *S. robusta* forests were found to differ in CF (Figure 5.5 and 5.6). Biological rotation in CFs currently occurs earlier in dense forests than sparse forests. The CAI and MAI curves intersect at age

30 in dense forest compared to 34 years in sparse forests, after the die-back period, which indicates that increasing the rotation of silvicultural operations is a possibility in dense forests.

While comparing meeting points of MAI and CAI curves for undisturbed forests (40 years, carbon yield 12.09MgC/ha) (Figure 5.2) and dense forests (29 years, carbon yield 4.20 MgC/ha) (Figure 5.5), there were huge differences in meeting point (year) and carbon yields. They are expected as dense forests have over \geq 70 % crown cover and have higher number of smaller trees whereas undisturbed forests may have higher number of larger trees.



Figure 5.5 Mean Annual Increment (MAI) and Current Annual Increment (CAI) in dense *S. robusta* dominated CF



Figure 5.6 Mean Annual Increment (MAI) and Current Annual Increment (CAI) in sparse *S. robusta* dominated CF

5.2.3. Carbon growth in undisturbed forests, dense CFs and sparse CFs

There is a remarkable gap in cumulative biomass carbon between dense and sparse canopy CFs within undisturbed uniform (i.e. same age) forests. In Figure 5.7, the area of curve AOF is the theoretical maximum potential carbon stock and the area of BoF is the maximum potential gain in dense CFs. Similarly, the area of EOF is the potential gap between sparse and dense CFs and the combined areas of EOF and BOF is the maximum growth potential of CFs to attain the theoretical maximum (Figure 5.7). The gap between undisturbed forest and CFs was less at early stages and higher in older aged forests. As stated, a possible reason behind this increased gap in later aged stands could be the harvesting practice of CFs where communities might be harvesting larger trees rather than those that are smaller. Most of the CFs were highly degraded at the time of handover and therefore site quality might not be favourable for higher growth rates of early age forests in sparse forests. Both smaller and larger aged trees were fewer in number in CFs than its technical potential.



Figure 5.7 Cumulative biomass carbon in undistubed natural forests with uniform age; dense and sparse canopy CFs with average age of dominanat trees in *S. robusta* forests

Results from CurveExpert analysis indicated that the area under the curves was larger in undisturbed forests (i.e. O1O2CB or o1o2cb = 34708.07 followed by dense (i.e. O2O3AB = 10651.03) and sparse forests (i.e. o2o3ab = 8304.57) (Figure 5.8 A &B). These differences show that area of lines showing by undisturbed forests have 2.26 times higher than dense canopy forests and 3.18 times higher than sparse canopy forests, further explaining the gaps.





Figure 5.8 Area under curves showing cumulative biomass carbon growth in undistubed natural forests(34,708.07); (A) dense CFs (10,651.03); (B) sparse CFs (8,304.57) in *S. robusta* forests

5.3. Biomass carbon growth in other vegetation types

The highest possible biomass carbon in CFs has been estimated based on the 90th percentile value in plots while the lowest has been estimated in 10th percentile value in plots. The differences between those two extreme values indicate that there is potential for CFs to increase biomass carbon. Among four different forest vegetation types, the highest gap between these two limits were found in dense mixed broadleaf

forests followed by dense *Rhododendron-Quercus* forests and the lowest difference was apparent in sparse *Schima-Castanopsis* forests (Table 5.3).

Forest types	Total plots	10 th percentile value	90 th percentile value	Difference
Sparse mixed broadleaf	10	7.59	133.83	126.24
Dense mixed broadleaf	53	26.88	224.04	197.16
Sparse Schima- Castanopsis	18	8.65	89.48	80.83
Dense Schima- Castanopsis	98	28.91	177.49	148.58
Sparse Pine	12	13.68	102.35	88.67
Dense pine	30	36.35	175.87	139.52
Sparse Rhododendron- Quercus	13	3.56	139.67	136.11
Dense Rhododendron- Quercus	60	39.46	208.43	168.97

Table 5.3 Gaps between 10th percentile and 90th percentile value of carbon stock in both dense and sparse canopy types CFs by vegetation

5.4. Conclusion

Based on the results presented in this chapter, it is apparent that there are possibilities to increase biomass carbon in both dense and sparse canopy type CFs. Increment potential is higher in sparse forests than dense. Comparison with theoretical maximum growth patterns indicates that differences between sparse and dense CFs are higher in old age forests than early stage forests. This shows that CFs may be harvesting or losing old age trees from their stands. However if REDD+ activities are designed to ensure benefits to CFUGs, they may reduce the harvesting of trees and help to reduce the gap between current and maximum potential growth of forests.

To reduce these gaps and increase the biomass carbon of CFs, CFUGs need to develop plans that consider the factors affecting carbon stock increments in forests. The possible factors affecting biomass carbon in CFs are discussed in the next chapter.

CHAPTER SIX

CHANGED BEHAVIOUR OF THE COMMUNITIES AND FACTORS AFFECTING CARBON STOCK AND CARBON STOCK CHANGES IN THE REDD+ CFs

6.1. Introduction

The previous chapter presented the potential carbon growth in undisturbed forests and CFs by canopy cover and vegetation type. This chapter describes changed practices of local communities for the REDD+ project and factors affecting carbon stock in CFs. This chapter also presents evidence of changed community practices in forests, different possible factors and possible relationships with carbon stock and change in carbon stock in the respective experimental plots in CFs.

This chapter is divided into eight sections. The next section gives evidence of the relationship between changes in community behaviour and biomass reduction in the forests. In the third, fourth, fifth, sixth and seventh sections biomass quantity in year 2010 and change in carbon stock between 2010 and 2013 is presented in relation to different elevation forests, different proximity from road head, different proximity from settlement, different average age of the dominant trees in CFs and various socio-economic aspects of CFs (including per capita forests available, household level landholding, size of household level livestock (Livestock Unit), biogas using households in CFs, petroleum energy using households in CFs, biomass extraction, grazing, caste heterogeneity (index) in CFs, quantity of grass collection, fodder collection, litter collection) respectively. The final section concludes the chapter.

6.2. Evidence of changed behaviour of communities related to biomass reduction in the forests

Evidences of biomass reducing incidences (forest fire, livestock grazing, fodder collection, fuel wood collection, and timber collection) recorded during annual carbon pool measurement (between 2010 and 2013) in CFs showed decreasing trends in later years of the REDD+ project (Figure 6.1, 6.2, 6.3, 6.4 & 6.5). The decreasing rates in the frequencies of these incidences differed by vegetation types and by year in which measurement was taken. These evidences of incidences are presented individually in the sub-sections below.

Forest fire

Comparatively forest fire incidences were observed in higher proportion of plots in dense canopy of forests at the reference year 2010 except Rhododendron-Quercus forests. The incidence of fire fell over the following years and was least in the year 2013. Communities contributed to controlling the incidence of forest fire in dense canopy mixed broadleaf forests and were able to reduce fire frequency from nearly 40% of plots in the year 2010 to 10% in the year 2013 (Figure 6.1). There were still high forest fire incidences in sparse canopy Schima-Castanopsis and Pine forests which may require additional fire control efforts by these communities.



Figure 6.1 Observed forest fire incidence by proportion of total plots in different vegetation and canopy cover types

Grazing

Grazing incidence was highest in both dense and sparse canopy *Rhododendron-Quercus* forests followed by sparse canopy *S. robusta* forests and lowest in dense canopy *Shorea* forests and dense canopy *Schima-Castanopsis* forests. Grazing in all forests gradually reduced in the later years of the study, indicating that local communities were controlling livestock grazing after the REDD+ project activities commenced. A higher level of livestock grazing (in more than 30% of plots) was found in sparse canopy of *Rhododendron-Quercus* forests in 2013 which means that communities were still sending their livestock into CFs; this practice may need to be controlled to achieve maximise REDD+ carbon benefits (Figure 6.2).



Figure 6.2 Observed livestock grazing incidence by proportion of total plots in different vegetation and canopy cover types

Fodder collection

A higher proportion of plots had fodder collection in both sparse and dense canopy mixed broadleaf forests in 2010 followed by dense canopy *Schima-Castanopsis* forests; this practice was lowest in sparse canopy *Schima-Castanopsis* forests. Fodder collection decreased slightly in the year 2013 in all forests, but the rates of decrease were highest in *Schima-Castanopsis* forests, followed by sparse canopy *S. robusta* forests and lowest in sparse canopy *Pine* forests (Figure 6.3).



Figure 6.3 Observed fodder collection incidence by proportion of total plots in different vegetation and canopy cover types

Fire wood

A higher proportion of plots in dense canopy *S. robusta* forests had firewood collection incidences followed by dense canopy *Schima-Castanopsis* forests while this was lowest in dense canopy mixed broadleaf, sparse canopy *Pine* and sparse canopy *Schima-Castanopsis* forests. There seems to be a very slight decrease in firewood collection during the REDD+ project, although this trend is not evident in the dense *Rhododendron-Quercus* forests (Figure 6.4).



Figure 6.4 Observed firewood collection incidence by proportion of total plots in different vegetation and canopy cover types

Timber

A lower proportion of plots had timber harvesting incidences in all vegetation types (i.e. less than 10% of plots) (Figure 6.5). The study results showed a reduction in timber harvesting in plots (with proportion of timber harvesting plots least in the year 2013 in all vegetation types) while there were no regular reduction trends. In the reference year 2010, a highest proportion of timber harvesting plots was found in dense canopy *S. robusta* forests followed by dense canopy *Schima- Castanopsis* forests and lowest in dense canopy mixed broadleaf forests.

According to local communities, there were no regular tree harvesting practices in their CFs because they only decide and allow tree harvesting for timber if someone from one of their member groups demands timber to build or renovate their house. But that type of timber demand does not come every year and timber harvesting could be seen only when and where communities had demand and decided to cut trees (Annex H).



Figure 6.5 Observed timber collection incidences by proportion of total plots in different vegetation and canopy cover types

6.3. Changed practices of CFUGs for the REDD+ project

With facilitation of the REDD+ project, CFUGs have changed some of their existing behaviours, mainly: decision making processes; forest management practices; forest product collection; and forest biomass reduction activities, in order to increase biomass carbon in forests (Table 6.1).

According to Focus Group Discussions (FGDs), most CFUGs make decisions through participatory processes by organising meetings of the executive committee and General Assembly (GA) of CFUG members. The numbers of both the meeting and GA are increased and the agenda of both meeting and GA have been expanded by adding awareness raising and processes for the REDD+ mechanism and activities. The level of practice change was found to be greater in S. robusta, mixed broadleaf and *Rhododendron-Quercus* forests than before REDD+ project. All CFUGs have increased their carbon increment goals following the REDD+ project intervention. In order to achieve these goals, the project provided technical guidance about forest management activities and material support about planting to CFUGs. Now, most CFUGs operate regular silvicultural activities (mainly removing dead, dying, diseased and malformed trees from forests) except in Rhododendron-Quercus forests and do plantation activities in all Community Forests (CFs) except S. robusta forests. Similarly, communities have reduced the quantity of timber harvesting from CFs with the highest reduction in S. robusta, followed by the mixed broadleaf and Schima-Castanopsis forests. Most CFs have made provisions to collect dry and fallen branches and trees for fuel wood on certain days of each month except in *Rhododendron-Quercus* forests. Fodder collections are open during certain months in

some lower altitude CFs and litter collections are mainly open for *Schima-Castanopsis* and *Rhododendron-Quercus* forests. There was no reduction in the quantity of tradable NTFPs harvested in CFs. Majority CFs conduct activities to control possible damage from forest fires, illegal harvesting and grazing. More specifically, care was increased to control forest fire in *S. robusta*, mixed broadleaf forests and *Pine* forests and illegal harvesting in *S. robusta* and mixed broadleaf forests. In order to reduce the dependency on forest products in CFs, communities have reduced numbers of unproductive livestock, increased plantations on private unproductive agricultural lands and the use of alternative energy (Improved Cooking Stove-ICS, and biogas plants).

Activities	Changes (common for all CFs)	Specific by species dominated CFs
1. Decision making process Executive committee meeting	 Executive committees are more inclusive, representing all castes and economic groups Numbers of meeting are increased Agenda of meeting are expanded to 	No more differences between vegetation types however small forests and <i>Schima-</i> <i>Castanopsis</i> and <i>Pine</i> forests have no much change because
	 include REDD + activities and plan Number of participants in meeting is increased Now members of the committee are more motivated to contribute to forestry activities 	they are less heterogeneous (community) and were doing well even before the REDD+ project
General assembly (GA)	 Increased numbers Increased number of participants Agenda of GA has been expanded to awareness, planning and capacity building for REDD+ activities and benefits sharing 	No more differences between vegetation types
2. Forest		
management		
Aim	- Expanded from improvement of forests and sustainable use of forest products to maintaining ecosystem services including forest carbon stocks	After the REDD+ concept evolved in the study areas, all CFs aimed to go for increasing forest carbon stock
Silviculture operation	 Silviculture operations are being operated annually on a more regular basis in most of CFs Clearing and thinning is done to create a more conducive environment to grow trees 	More regular in CFs with S. robusta mix broadleaf, Schima-Castanopsis vegetation
Plantation	 Plantation activities in some of the CFs (both in forests and uncultivated private land) are increased Project supported to give technical knowledge, seedling for plantation All labour for plantation activities was done by users 	Comparatively less in <i>S. robusta</i> forests

Table 6.1 Summary of the changed practices in CFs for REDD+ in the study areas

collection	from	
CFS Timber	 Quantities of timber harvesting is less than before Timber is distributed only to needy user members Quantities of timber sold is decreasing in most of CFs Casta comp Rhod (about) 	tity of harvests has reduced particularly in (<i>S. robusta</i>) and mid- de (<i>Pine</i> and <i>Schima-</i> <i>unopsis</i>) (about 70%) pared to higher altitude <i>lodendron-Quercus</i> at 40%) CFs
Fuel wood	 Mostly allow collection of dry and fallen wood Dry and fallen branches/ trees are collected on certain days in a month Green wood obtained from silvicultural open operations are distributed for fuel wood Collection of bent and forked green trees for fuel wood is permitted during and f silviculture operation 	a only for 1-2 days in month in <i>S.robusta</i> , and <i>Schima-</i> <i>unopsis</i> forests whereas for all days in <i>lodendron-Quercus</i> ts to collect dead/dying fallen branches for fuel f n trees in <i>S. robusta</i> ts are not allowed to be ested
Fodder/grass	 Fodder and grass collection is made in a systematic way for REDD+ Green seedlings and poles are not in allowed to be cut in forests for fodder <i>Rhod</i> and grass supply Non woody grasses are allowed to be for co collected without damaging regeneration 	a to cut green ground es for the whole year higher altitude lodendron-Quercus ts whereas it is open ertain months in some at lower altitudes
Litter	- Litter collection is not controlled but the quantity of collection has been reduced - Colle Schin Rhod forest aim o make decon Pine leave decon prefe	action is higher in ma-Castanopsis and codendron-Quercus ts because the main of litter collection is to manure by mposing the leaves but needles and Shorea s take time to mpose and are not rred by communities.
NTFPs	 Income from NTFPs has been increased with access to market information Lokta (<i>Daphne bholua</i>), Wintergreen (<i>Gaultheria fragrantissima</i>) and Chiuri in the (<i>Bassia butyracea</i>) are major marketable NTFPs plants of the study areas Lokta and Wintergreen are not large wood trees and collection of them in a sustainable way may not affect to the overall biomass carbon in the forests. Chiuri is a tree but people collect fruits for butter that does not affect the carbon quantity Collection of NTFPs which are used for 	lle and higher latitude ts (mainly <i>Schima- anpsis</i> and <i>Pine</i>) get income from NTFPs e study areas

	level and income gained by selling these has not been changed but careful collection has been encouraged	
4. Forest biomass reduction activities Forest fire	 Preventive measures (awareness about possible causes and effects of forest fire and preparing firelines to reduce possible damage of forest fire) are carried out between February to March Users are gathered and worked together to control forest fire if these occur 	 Fire risk is higher in lower altitude forests (<i>S. robusta</i> and mixed broadleaf) and less in middle altitudes (<i>Schima-Castanopsis</i> forests) Pine forest is more sensitive due to flammable resin and communities are engaged in preventive measures. Participants get food or small amount of money to attend fire control activities in Terai CFs but these are fully voluntary in other forests.
Illegal harvesting	- Most of the CFs have developed guarding provision to control illegal activities, control forest fire and reduce damages to plantations	 In <i>S. robusta</i> forests, lower altitude CFs appoint 1 or 2 persons as guard or watchman for the whole year in most cases In hill areas, communities guard forests against illegal activities in rotation. They guard forests only for a certain months when there is high potential of illegal activity
Grazing	- Grazing is almost controlled in CFs. Few CFs have some grazing practices at the forest edge (i.e edge between forest and agricultural land and/or nearby settlement)	- In <i>Rhododendron-Quercus</i> forests, there are some grazing practices permitted. These are almost controlled in <i>Schima-Castanopsis</i> , <i>Pine</i> and <i>Schorea</i> mixed broadleaf forests.
Encroachment	- Encroachment has been fully controlled in CFs	
5. Other related		
Livestock	 Communities have reduced unproductive livestock (mainly cows) Number of goats is increasing in many CFs for income generation 	- Not much difference in pattern of CFs by vegetation types
Private plantation	 CFUG members have initiated plantation of fodder and grass on private marginal lands Project helped deliver seedling and 	- More in <i>Schima-</i> <i>Castanopsis</i> and <i>Pine</i> forests located at middle altitude forests

		technical knowledge about plantation		
Use of alternative energy	-	Improved Cooking Stove (ICS), Biogas and LG gas are the main changed alternative energy practices in CFs to reduce household level fuel wood consumption Project provided partial financial and technical support for the use of ICS and Biogas for households energy	-	Higher proportion of households using biogas in lower altitude (<i>S.</i> <i>robusta</i> and mixed broadleaf forests); less in middle altitude (<i>Schima-</i> <i>Castanopsis</i> and <i>Pine</i> forests) than higher altitude (<i>Rhododendron-</i> <i>Quercus</i> forests) Higher proportion of households adopted ICS in higher altitude forests whereas a higher proportion of households adopted Biogas at lower altitudes. This is also due to feasibility from both environmental and economic point of view.

(Source: FGD: 2012)

6.4. Elevation of the forests

A relationship between altitude and average carbon stock (MgC/ha) in the reference year 2010 and the change in carbon stock (MgC/ha) between 2010 and 2013 in plots, by vegetation and canopy types, were analysed and are presented in this section.

Although, the scatter plots of carbon stock and change in carbon stock by altitude variation in all plots of specific vegetation type showed no significant correlation, there were different indicative trends. In *S. robusta* forests, the biomass carbon stock in year 2010 indicated a slightly negative relationship with altitude whereas changes in carbon stock indicated a positive relationship (Figure 6.6). Mixed broadleaf forests seems to have a positive but slight relationship with altitude for both carbon stock in 2010 and change in carbon stock (Figure 6.7). The relationship between altitude and carbon stock in 2010 was slightly negative but positive with carbon stock change in *Schima-Castanopsis* forests (Figure 6.8). The opposite was true in the case of both *Pine* forests and *Rhododendron-Quercus* forests where carbon stocks seemed to be positive with altitude in the year 2010 but carbon stock changes were negative (Figure 6.9; Figure 6.10)

The relationship between altitude and carbon stock in the year 2010 was slightly positive in sparse canopy forests of all vegetation types but the relationship seemed to be negative with changes in carbon stock between 2010 and 2013. This was similar in dense canopy of Pine forests and *Rhododendron-Quercus* forests whereas it was almost opposite in *S. robusta* forests, mixed broadleaf forests and *Schima-Castanopsis* forests (Figure 6.6; 6.7; 6.8; 6.9 and 6.10).



Figure 6.6 Scatter plots showing relationship between carbon stock (average in year 2010 and change between 2010- 2013) in *S. robusta* dominant forests by canopy types



Figure 6.7 Scatter plots showing relationship between carbon stock (in year 2010 and change between 2010 - 2013) in Mixed broadleaf forests by canopy (A-F)



Figure 6.8 Scatter plots showing relationship between carbon stock (in year 2010 and change between 2010-2013) in *Schima-Castanopsis* forests by canopy types (A-F)



Figure 6.9 Scatter plots showing relationsnip between carbon stock (average in year 2010 and change between 2010- 2013) in *Pine* forests by canopy types (A-F)



Figure 6.10 Scatter plots showing relationship between carbon stock (average in year 2010 and change in carbon stock between 2010 and 2013) in *Rhododendron-Quercus* forests by canopy types (A-F)

6.5. Proximity of forests from road head

Location of plots (by distance) from nearby road head was found to differ in analyses of satellite imagery and road data (Figure 6.11). The results showed that average carbon stock in plots located further from the road head was higher in the year 2010 (Table 6.2). According to communities, road access had increased the harvesting of timber and firewood especially before community forestry started. This might be a reason for lower carbon stocks near roadheads in all vegetation forests except Pine. Pine forests were mostly plantation forests and plantation activities were greater at locations close to roads.

Average changes in carbon stock (between 2010 and 2013) were also found to be higher in area located close to the roadhead in all vegetation types except pine forests (Table 6.2). This shows that where there were higher carbon stocks in plots during the reference year these tended to have lower carbon increments over the study period. This indicates that these communities have conserved forest biomass with the REDD+ activities. However, in some plots, there was higher removal of biomass (higher negative value in the range of carbon stock estimates in plots in Table 6.2) which shows there were some over harvesting or illegal harvesting going on closer to road heads.

Table 6.2 Proximity (Road to plots) and average C-stock (MgC/ha in year 2010) and change in C-stock (between 2010-2013, MgC/ha) in forests analysing 21 plots located at farthest away and 21 at closest from the road head

Vegetation types	Ave. distance (m)	Range (m)	C-stock in year 2010 (MgC/ha)	Range (MgC/ha)	C- change 2010-2013 (MgC/ha)	Range (MgC/ha)
Shorea robsuta	40.5	1.5-92.7	108.5	13.3-200.8	15.8 [#]	-23.4-47.7
forests	1310.8	990.6-3892.1	118.4^{*}	15.0-265.4	14.8	-26.7-43.3
Mixed	290.4	29.5-493.6	105.3	8.0-345.1	17.2 [#]	-4.5-45.2
forests	2828.9	927.9-4360.0	127.0^{*}	27.1-310.3	15.8	-20.7-44.5
Schima-	174.3	28.9-292.6	89.5	10.9-269.7	$20.8^{\#}$	-15.4-43.3
Catstanopsis	2231.6	1170.0-4541.6	93.5 [*]	10.8-277.7	17.5	-17.1-39.0
Pine forests	205.2	20.7-384.5	102.9	11.9-230.6	15.3	-21.9-42.9
T the forests	971.4	452.6-3094.9	79.9	0.6-186.5	23.6	0.1-49.6
Rhododendron	275.2	63.6-554.9	73.4	13.3-174.0	$10.0^{\#}$	-18.8-29.8
-Quercus	2548.5	1747.3-3636.0	149.9 *	23.5-243.2	7.5	-28-33.3

Note: * higher average biomass carbon (above ground and below ground, in 2010) plots located at far from road head, # higher change in biomass carbon (above ground and below ground, between 2010 and 2013) in plots located at close from road head



CFboundary

6.6. Proximity of forests from settlement

Similar to the proximity to road heads, the location of plots from nearby settlements was found to differ in analysis of satellite images and settlement data (Figure 6.12). Average carbon stock was found to be higher in forest located close to settlements in all vegetation types except *Rhododendron-Quercus* forests and *Schima-Castanopsis* forests (Table 6.3). Carbon stocks in *Schima-Castanopsis* forests were similar in both close and far locations whereas these were more than twice as high in far locations in *Rhododendron-Quercus* forests. According to communities, *Rhododendron-Quercus* species were not useful for timber and they wanted to promote timber trees in their forests; therefore, they may have been more likely to harvest these from nearby locations.

Carbon stock changes were found to be higher in forests located far from the settlements especially in *S. robusta*, mixed broadleaf and *Rhododendron-Quercus* vegetation, whereas the was opposite occurred in the case of *Schima-Castanopsis* and *Pine* forests. Communities explained that, following the inception of CF, they have been working, and even more so since the REDD+ project began, in *Schima-Castanopsis* and *Pine* forests, which could be a reason for increased carbon stocks near some settlements (Table 6.3).

Vegetation types	Ave. distance (m)	Range (m)	C-stock in 2010(MgC/ha)	Range (MgC/ha)	C-change 2010- 2013(MgC/ha)	Range (MgC/ha)
S. robsuta forests	466.9	74.6-252.1	118.4 [#]	2.3-307.1	15.5	-26.6-47.2
	1361.0	1073.3-1987.4	113.7	7.8-322.0	18.0^{*}	-23.4-45.9
Mixed broadleaf forests	438.2	145.2-605.0	120.7#	8.0-345.1	14.6	-20.7-41.4
	1299.6	848.9-2265.8	114.1	6.0-310.3	18.6*	-19.5-45.2
Schima- Catstanopsis	268.8	65.1-385.5	92.6	19.4-228.3	22.8	-17.4-42.0
	1381.9	910-3118.2	93.7	25.4-269.7	10.4	-13.4-37.6
Pine forests	420.5 995.7	190.0-559.0 618.3-2842.9	97.2 [#] 85.6	0.6-230.6 28.1-184.4	19.5 19.3	-21.9-42.9 -16.1-49.6
Rhododendr on-Quercus	494.5	104.0-774.4	63.3	6.9-174.0	7.7	-18.6-39.5
~	2418.9	1697.0-3490.8	142.7	3.7-243.2	8.1 [*]	-28.0-33.3

Table 6.3 Proximity (settlement to plots) and average c-stock (MgC/ha in year 2010) and change in C-stock (between 2010-2013, t C/ha) in forests analysing 21 plots located at farthest away and 21 at closest from the local settlement

Note: #higher average biomass carbon (above ground and below ground, in 2010) plots located at close from settlement, * higher change in biomass carbon (above ground and below ground, between 2010 and 2013) in plots located at far from settlement



CFboundary

6.7. Stand age in CFs

Comparatively older trees were found in CFs with *S. robusta* and mixed broad leaf vegetation followed by *Schima-Castanopsis* while younger trees occurred in *Pine* and *Rhododendron-Quercus* vegetation. Carbon stocks in the year 2010 were higher in those CFs which had higher age class trees than lower age class but the positive changes in carbon stock was higher in the CFs which had lower age class of trees in all vegetation (Table 6.4).

Table 6.4 Average age of forest stand, and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 and 2013 (MgC/ha) in CFs by vegetation types

Vegetation	Age of the forest stand		C- stock 2010 (MgC/ha)		C- Change 2010- 2013 (MgC/ha)	
0	Average	SD	Average	SD	Average	SD
Shorea-mixed broadleaf (N=10)	25	4.08	112.26	12.59	14.40	0.97
	51	9.37	116.90	7.33	13.04	1.47
Schima-Castanopsis (N=10)	22	2.58	76.62	8.56	20.14 [#]	6.64
	52	8.14	96.92 [*]	15.20	13.28	5.81
Pine (N=8)	31	4.41	84.12	5.36	19.77 [#]	8.48
	42	8.45	93.95 *	21.11	14.82	6.07
Rhododendron. (N=4)	26	4.79	67.39	5.16	13.11 [#]	4.00
	42	11.9 0	86.60 *	16.11	8.01	1.73

Note: * Higher the age of forest stand of CFs had higher carbon stock in year 2010 # smaller the age of forest stand of CFs had higher positive changes in carbon stock between 2010 and 2013

6.8. Socio-economic aspects of CFUGs

There are several socio-economic aspects of communities which may affect carbon stock in CFs. According to the literature and consultation with experts working in the REDD+ projects and forestry sector in Nepal, per capita forest areas, household level land holding size, live-stock holding, bio-gas use, petroleum energy use, change in quantity of biomass (timber and firewood) extraction, change in grazing livestock, caste heterogeneity, change in quantity of grass collection, change in quantity of fodder collection, charge in quantity of litter collection are all possible socio-economic factors responsible for changing carbon stocks in CFs. Results of analysis about these factors in CFs by different vegetation types (i.e. the dominant vegetation type of CF) are presented in this section.

6.8.1. Per capita forest areas

Analysis of forest area per capita forests and its relationship with carbon stock in CFs was made on the basis of higher and lower size categories for each vegetation type. In the higher size category, *Schima-Castanopsis* forests had higher forest area per capita followed by *S. robusta*-mixed broad leaf forests; this was lowest in *Pine* forests. In the lower category, *Pine* forest had higher per capita area followed by *Rhododendron-Quercus* and the lowest was in *Shorea*-mixed broadleaf forest.

Carbon stock in the year 2010 was highest in CFs which had higher per capita forests available except the CFs with *Rhododendron-Quercus* vegetation (Table 6.5). There were few differences between the mean of larger and smaller per capita forest categories of the *Rhododendron-Quercus* forests which indicates that smaller forest s of this type were not under heavy pressure from communities. In term of changes, *S.robusta* and *Rhododendron-Quercus* forests had higher levels of change in larger per capita forests while the situation was different for the other two vegetation types. Smaller forests located at lower altitudes (*S.robusta*) and higher altitudes (*Rhododendron-Quercus*) experienced comparatively high pressure. But, there might be more care taken in the smaller forests dominated by *Pine* and *Schima-Castanopsis* vegetation located in the mid-hills.

Vegetation	Per capita fo in CF (rest area ha)	C stock in 2010 (MgC/ha)		stock in 2010 C-change 2010 (MgC/ha) 2013 (MgC/ha)	
	Average	SD	Average	SD	Average	SD
Shorea mixed	0.02	0.01	108.18	15.58	13.22	2.17
broadleaf (N=10)	0.20	0.04	118.21^{*}	10.23	14.63 [#]	1.43
Schima-Castanopsis	0.03	0.01	78.16	11.28	18.71	7.37
(N=10 in both case)	0.43	0.26	95.8 2 [*]	13.17	12.02	5.61
Pine (N=8)	0.05	0.02	84.96	15.98	18.52	7.77
	0.15	0.06	93.11 *	15.37	16.07	7.68
Rhododendron- Quercus	0.03	0.02	76.52	16.83	8.41	2.01
(N=4)	0.18	0.10	75.98	14.81	12.63#	4.62

Table 6.5 Per capita forest areas (ha/person) difference and carbon stock in year

 2010 and change in carbon stock between 2010–2013 in CFs by vegetation

Note: *Larger per capita forest had higher carbon stock in year 2010, # larger per capita forest had higher changes in carbon stock between 2010 and 2013

6.8.2. Household level landholding

Land holding size was found to differ between CFs. Land holdings were larger in communities associated with *Schima-Castanopsis* forests and *Pine* forests followed by *S. robusta* and mixed broadleaf forests and *Rhododendron-Quercus* forests. The carbon stock in year 2010 was higher in CFs with smaller sizes of household level land holding except for *Shorea*-mixed broadleaf forests. The sizes of household level land holdings in *S. robusta* and mixed broadleaf forests were smaller; this may have been because some of the settlements at these locations were comparatively new.

After the REDD+ activities, a higher positive change in carbon stocks was estimated in all CFs (across different vegetation types) which had larger household level agricultural land holdings. Specifically, these changes were higher in *Pine* forests followed by *Schima-Castanopsis* forests and lower in *Rhododendron-Quercus* forests (Table 6.6).

Table 6.6 Household level landholding size and C-stock in year 2010(MgC/ha) and C-stock change between 2010–2013 (MgC/ha) in CFs by vegetation types

Vegetation	Agriculture land in CF (ha/HH)		C- stock i (MgC	in 2010 /ha)	C-stock change 2010 -2013 (MgC/ha)	
	Average	SD	Average	SD	Average	SD
Shorea-mixed	0.50	0.12	114.34	8.04	14.09	1.12
broadleaf (N=10)	0.82	0.04	120.32	3.01	14.92[#]	0.41
Schima-Castanopsis	0.60	0.09	90.03 *	16.07	12.35	5.74
(N=10)	0.89	0.05	84.93	14.85	20.65#	6.47
Pine (N=8)	0.71	0.07	90.59 [*]	16.92	12.12	6.05
	0.89	0.05	87.48	15.42	22.48[#]	5.00
Rhododendron.	0.66	0.06	82.33 [*]	14.83	7.36	1.80
(N=4)	0.81	0.06	76 18	14 58	12.79#	4 4 9

Note: *Smaller per household landholding had higher carbon stock in year 2010, # larger per household agriculture land had higher changes in carbon stock between 2010 and 2013

6.8.3. Average change in livestock

After the REDD+ projects, the average number of livestock changed in CFs. The change in carbon stocks was higher in CFs where positive changes in livestock were estimated in all vegetation types except *Shorea*- mixed forests. According to the communities, people use forest resources for livestock in lower altitude areas (e.g. CFs with *Shorea* forests) where they have less per capita agricultural land and if they reduce livestock that increases biomass carbon in the forests (Table 6.7).

C-stock in 2010 **Change in Livestock** Livestock (per C-stock change unit (HH) in CF HH) in CF (MgC/ha) 2010-2013 before REDD (MgC/ha) Vegetation Average SD Average SD Average SD Average SD -1.08 1.07 1.84 1.65 114.23* 11.02 14.08# 1.54 Shorea-mixed 1.11 broadleaf (N=10) 0.54 1.26 1.31 112.54 11.05 13.83 1.54 -0.96 0.69 2.06 1.12 **85.16*** 18.73 12.56 3.74 Schima-Castanopsis (N=10) 0.91 0.47 1.94 80.81 16.09 16.90# 1.11 6.28 -0.240.84 1.65 1.43 **89.87*** 15.12 16.14 8.38 Pine (N=8) 1.39 18.45# 1.16 0.48 0.63 88.19 17.30 7.04 -0.53 0.88 1.35 0.06 82.22* 14.05 7.46 2.06 Rhododendron. (N=4) 0.38 0.35 2.03 0.73 74.14 10.93# 16.81 3.79

Table 6.7 Household level livestock unit and C-stock in the year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types

Note: *Reduced LSU at household had higher carbon stock in year 2010 [#] Increase in LSU had higher changes in carbon stock between 2010 and 2013

6.8.4. Biogas use

Higher proportions of households in *S.robusta* CFs were using biogas followed by *Schima-Castanopsis* and lowest in *Rhododendron-Quercus* CFs. Of various reasons, the applicability of present technology (which is better for lower altitude areas) and the economic aspect of people (more people could afford to invest in and operate biogas plants in these CFs) may be important.

In 2010, carbon stocks were higher in those CFs where more households were using biogas in *Schima- Castanopsis* and *Pine* forests. Change in carbon stock was higher in the CFs where more households were using biogas except in *S.robusta* forests (Table 6.8). There were similar changes in both higher and less biogas using CFs in *S.robusta* vegetation because communities only collect fallen dry branches for fuel wood and that may not affect carbon change in the forests. Biogas using households were also using firewood for major cooking as they think their biogas stove is not as effective.

Vegetation	Proportion of biogas using HH		C-stock in 2010 (MgC/ha)		C- change 2010- 2013 (MgC/ha)	
<u> </u>						
	Average	SD	Average	SD	Average	SD
Shorea-mix broadleaf (N=10)	0.03	0.02	116.01	8.15	14.32	1.14
	0.19	0.11	114.24	10.89	14.07	1.52
Schima-Castanopsis (N=10)	0.00	0.00	84.95	18.43	11.99	3.71
	0.11	0.13	102.62^{*}	14.63	15 . 57 [#]	6.49
Pine (N=8)	0.00	0.00	83.07	12.54	15.62	8.62
	0.01	0.02	95.00 [*]	17.06	18.98[#]	6.48
Rhododendron-						
Quercus (N=4)	0.00	0.00	78.37	14.44	10.11	4.74
	0.01	0.02	71.25	11.33	10.46 [#]	4.36

Table 6.8 Proportion of biogas using households and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types

Note: ^{*} Higher biogas using HHs proportion in CFs had higher carbon stock in year 2010 [#] Higher biogas using HHs proportion in CFs had higher changes in carbon stock between 2010 and 2013

6.8.5. Petroleum energy use

Similar to biogas energy use, a comparatively higher proportion of household in CFs with *S. robusta* vegetation were using petroleum energy (LP gas and Kerosene) followed by *Schima-Castanopsis*, while the lowest proportion was in *Pine* and *Rhododendron-Quercus* forests. According to local people, petroleum energy was costly in the mountain areas where *Pine* and *Rhododendron-Quercus* forests occur due to higher transportation cost. Therefore, only those people who can afford petroleum energy cost in mountain areas use this energy.

In 2010, there was little difference in carbon stocks between high and low petroleum energy using CFs. But changes in biomass carbon between 2010 and 2013 were higher in those CFs which had higher petroleum energy using households except for *Shorea* and mixed broadleaf forests (Table 6.9). In group discussions, people explained that they kept Liquid Petroleum gas but they used firewood for major cooking such as for feeding livestock, cooking and heating. Mostly, people used LP gas for tea making and other small and quick cooking purposes.
Table 6.9 Proportion of Petroleum energy (LP gas, Kerosene) using households and C-stock in year 2010 (MgC/ha) and change in C-stock between 2010 – 2013 (MgC/ha) in CFs by vegetation types

Vegetation	Prop. of Pe energy usi	troleum ng HH	C-stock in (MgC/	n 2010 Tha)	C- change 20 (MgC/)10-2013 ha)
	Average	SD	Average	SD	Average	SD
<i>Shorea</i> -mixed broadleaf (N=10)	0.08	0.05	116.53	9.59	14.40	1.34
× /	0.45	0.08	114.23	8.69	14.07	1.21
Schima-Castanopsis (N=10)	0.00	0.00	80.84	15.77	12.93	4.72
` ,	0.25	0.15	89.98 *	20.78	17.61 [#]	5.13
Pine (N=8)	0.00	0.00	81.88	9.76	17.21	8.63
	0.04	0.04	96.19 *	17.81	17.39 [#]	6.96
Rhododendron. (N=4)	0.00	0.00	76.52	16.83	8.41	2.01
· · /	0.05	0.04	75.98	14.81	12.63#	4.62

Note: ^{*} Higher Petroleum using HHs proportion in CFs had higher carbon stock in year 2010 [#] Higher Petroleum using HHs proportion in CFs had higher changes in carbon stock between 2010 and 2013

6.8.6. Biomass extraction

Communities are extracting firewood and timber as major forest products from their CFs (Table 6.10). Before the REDD+ activities, extraction was highest in *Shorea*-mixed broadleaf CFs followed by *Schima-Castanopsis* and lowest in *Rhododendron-Quercus* forests. Local communities harvest higher biomass in CFs where carbon stock was higher in year 2010.

After the REDD+ project, communities either reduced or maintained their levels of biomass extraction from forests. The reduction in the quantity of biomass (firewood and timber) extracted and the incremental increase in biomass were both greatest in *S. robusta* –mixed broadleaf CFs. The lowest reduction in biomass extraction was in *Rhododendron-Quercus* forests and lowest increase in biomass quantity was in *Pine* forests. Carbon stock changes (between 2010 and 2013) were highest in those CFs which had reduced biomass extraction by the greatest extent (Table 6.10).

Table 6.10 Change in quantity of biomass extraction (both timber and firewood) and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 - 2013 (MgC/ha) in CFs by vegetation types

Vegetation	Cha extra (before a REDD+	nge ction nd after -) (Mg)	Extrac befo REDD+	ction re -(Mg)	C- sto 2010 (M	ck in gC/ha)	C-stock change 2010-2013 (MgC/ha)		
	Ave.	SD	Ave.	SD	Ave.	SD	Ave.	SD	
Shorea-mixed broadleaf (N=10)	-11.92	11.31	18.53	21.09	121.58 [*]	2.52	15.10[#]	0.35	
	4.24	10.47	11.27	19.76	103.44	13.10	12.56	1.82	
Schima- Castanopsis (N=10)	-2.96	2.94	12.21	10.34	99.65*	14.15	19.30#	5.68	
	2.24	1.70	12.18	11.86	82.63	16.79	11.26	2.94	
Pine (N=8)	-5.39	6.28	6.47	7.01	91.26 [*]	19.12	19.12 [#]	7.59	
	0.08	0.36	10.03	12.28	86.81	12.36	15.47	7.59	
Rhododendron. (N=4)	-0.16	0.28	1.10	1.02	76.28	15.39	12.69 [#]	4.54	
	1.06	1.32	2.55	2.04	77.71	17.14	8.43	1.91	

Note: * Lower biomass extraction in CFs had higher carbon stock in year 2010; # Lower biomass extraction in CFs had higher changes in carbon stock between 2010 and 2013; Ave. means average; SD means standard deviation

The REDD+ pilot project supported the provision of biogas and ICS to reduce the quantity of firewood used. The project distributed ICS to 6% households and biogas plants to 3% of households in *S. robusta* mixed broadleaf CFs. Similarly, there were distributions of ICS to 11% and biogas to 1% of households of *Schima-Castanopsis* CFs; ICS to 10% and Biogas to 1% of households of Pine forests; and ICS to 15% and Biogas to 1% of households of *Rhododendron-Quercus* CFs (ICIMOD 2014). It is evident that ICS was distributed to more household in higher altitude CFs whereas biogas plants were supported to more households of lower altitude CFs. These interventions may have played some role in the increase in biomass carbon in these CFs.

6.8.7. Grazing

Livestock grazing was analysed by using LSU day in the forests. Before REDD+ project activities, highest LSU grazing occurred in *Schima-Castanopsis* forests and lowest in *Rhododendron-Quercus* forests. Higher levels of livestock grazing in CFs were associated with higher carbon stock in 2010. This means grazing in the forest had almost no impact on the carbon stocks maintained in CFs. Reductions in livestock grazing occurred in CFs after the REDD+ project; these were highest in *Shorea*-mixed broadleaf forests followed by *Schima-Castanopsis* forest and lowest in *Pine* forests. The study found higher reductions in livestock grazing in CFs which had higher carbon stock in year 2010. This shows the consciousness of the people in protecting forests before the REDD+ project in CFs which had more reduction in grazing due to project intervention. A higher positive change in carbon stock was found in the CFs where reductions in livestock grazing were higher (Table 6.11).

Table 6.11 Grazing (Live Stock Unit days in a year) and C-stock in year 2010 (MgC/ha) and C-stock change between 2010–2013 (MgC/ha) in CFs by vegetation types

Vegetation type	Change g (days/yr) (b after RE	grazing efore and CDD+)	Livestoc (days per y REI	k grazing ear) before DD+	C-stock i (MgC	in 2010 2/ha)	C-change in 2010 and 2013 (MgC/ha)		
	Ave	SD	Ave	SD	Ave	SD	Ave	SD	
Shorea-mixed broadleaf (N=10)	-12397.30	12495.70	23036.33	13742.10	116.83 *	10.13	14.40[#]	1.42	
	0.00	0.00	0.00	0.00	114.37	9.43	14.09	1.31	
Schima- Castanopsis (N=10)	-7903.10	5225.16	28496.57	16617.19	90.92 [*]	16.72	16.21 [#]	6.26	
	0.00	0.00	0.00	0.00	92.17	22.65	12.47	3.41	
Pine (N=8)	-567.75	1605.84	3090.00	8739.84	85.68	11.15	19.06 [#]	7.56	
	0.00	0.00	0.00	0.00	92.39	19.48	15.53	7.65	
Rhododendron- Quercus (N=4)	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	

Note: ^{*}Reduced grazing days in CFs had higher carbon stock in year 2010 [#]Reduced grazing days in CFs had higher increment in carbon stock between 2010 and 2013; Ave. means average; SD means standard deviation

6.8.8. Caste heterogeneity

The study found the highest index of caste heterogeneity in *Shorea*-mixed broadleaf forests followed by *Rhododendron-Quercus* forests and lowest in *Schima-Castanopsis* and *Pine* forests. Forest types may therefore also reflect caste heterogeneity. While in analysis, all CFs with higher carbon stock in the reference year 2010 were managed by heterogeneous communities except *S. robusta* and mixed broadleaf vegetation types. There change differences were fewer in carbon stock (between 2010 and 2013) between CFs with higher and lower caste heterogeneity indices. The change was highest in *Shorea* mixed broadleaf forests where there was the lowest caste heterogeneity index and almost same in *Rhododendron-Quercus* vegetation. In *Schima-Castanopsis* and *Pine* CFs, a higher positive change in carbon stock with higher caste heterogeneity was found (Table 6.12).

Vegetation type	Caste hetero index in	ogeneity CFs	Biomass C 2010 (MgC/	stock in) ha)	Change in biomass C (MgC/ha between 2010 and 2013)		
	Average	SD	Average	SD	Average	SD	
Shorea-mixed broadleaf (N=10)	0.12	0.10	115.87	9.99	14.30	1.39	
	0.63	0.03	110.86	15.70	13.60	2.19	
Schima-Castanopsis (N=10)	0.04	0.04	88.87	15.22	17.18	7.31	
	0.53	0.04	89.12 *	14.34	17.59 [#]	6.56	
Pine (N=8)	0.24	0.13	87.73	19.17	15.30	6.69	
	0.49	0.07	90.34 *	12.57	19.29 [#]	8.30	
Rhododendron-Quercus (N=4)	0.23	0.17	78.84	14.86	10.12	4.63	

Table 6.12 Caste heterogeneity and C-stock in year 2010 (MgC/ha) and C-stock change between 2010–2013 (MgC/ha) in CFs by vegetation types

Note: ^{*}Higher the caste diversity in CFs had higher carbon stock in year 2010 [#] Higher the caste diversity in CFs had higher changes in carbon stock between 2010 and 2013

0.07

79.66*

15.38

10.03

4.66

6.8.9. Quantity of grass collection

0.56

The quantities of changes in grass collection (positive and negative) in the REDD+ CFs are presented by vegetation type in Table 6.13. Average negative changes in grass collection were found to be highest in *Shorea*-mixed broadleaf forests followed by *Schima-Castanopsis* and *Pine* and lowest in *Rhododendron-Quercus* forests. Positive changes in grass collected were found to be highest in *Schima-Castanopsis* vegetation followed by *Rhododendron-Quercus* forests.

Before the REDD+, the highest level of grass collection was from *Shorea*-mixed broadleaf forests followed by *Schima-Castanopsis* forests; this was much reduced after the REDD+. CFs which had the highest reductions in grass collected during the REDD+ project had higher carbon stocks in the year 2010. One of the reasons behind that was due to less grass available in the higher biomass forests with dense canopy than sparse one. CFs which had no reduction or small increment in the quantity of grass collection had the highest increment of carbon stock of all the vegetation types (Table 6.13). This indicates that the forest management communities were not affecting biomass during grass cutting which could help to increase carbon stock in their forests by creating a better environment for tree growth.

T 7 4 4	Change : collectio	in grass on (Mg)	Grass co before (M	ollection REDD [g)	C-stock (Mg(in 2010 C/ha)	C-change (MgC/ha between 2010 and 2013)		
Vegetation	Ave	SD	Ave	SD	Ave	SD	Ave	SD	
Shorea-mixed broadleaf (N=10 in both case	-67.30	84.78	207.01	250.46	117.23	9.59	14.49	1.34	
	0.63	3.61	106.16	274.74	118.47	5.49	14.66 [#]	0.77	
Schima- Castanopsis (N=10)	-59.44	76.04	172.50	163.02	94.01 [*]	21.98	14.43	4.23	
	6.08	16.40	31.21	35.94	83.82	13.34	16.55 [#]	7.78	
Pine (N=)	-17.08	6.81	90.83	51.22	95.36 [*]	19.76	15.80	6.94	
	0.28	4.49	67.70	114.92	82.71	6.91	$18.80^{\#}$	8.34	
Rhododendron- Quercus (N=4)	-12.76	6.86	51.00	20.65	76.52	16.83	8.41	2.01	
	4.14	5.81	59.18	40.05	79.66	15.38	10.03[#]	4.66	

Table 6.13 Quantity of grass collection and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types

Note: * Less grass collection in CFs had higher carbon stock in year 2010 # Higher grass collection had higher changes in carbon stock between 2010 and 2013; Ave. means average; SD means standard deviation

6.8.10. Fodder collection

Fodder collection was reduced in CFs with both *Shorea*-mixed broadleaf and *Schima-Castanopsis* vegetation while there was little change in fodder collection in *Pine* and *Rhododendron-Quercus* forests for the REDD+ project. Before the REDD+ project activities, fodder collection was highest in *Shorea*-mixed broadleaf forests followed by *Schima-Castanopsis* forests.

People were collecting fodder in the sparse forests where less carbon stock was available but the highest collection was from the forests with high carbon stock. It seems the quantity of fodder collection has little effect on carbon stock changes in CFs (Table 6.14). This could be due to the consciousness of communities around protection of seedlings and saplings during fodder collection in all CFs.

Vegetation	Change i collectio	n fodder on (Mg)	Grass befor (collection e REDD Mg)	C-stock (MgC	in 2010 C/ha)	C-stock 2010 (Mg	x change -2013 C/ha)
	Ave	SD	Ave	SD	Ave	SD	Ave	SD
Shorea-mixed broadleaf (N=10)	-40.06	89.87	291.7	719.7	110.76	10.30	13.58	1.44
	0.30	0.95	3.5	11.2	117.03 *	10.22	14.46	1.42
Schima- Castanopsis (N=10)	-56.67	149.85	79.9	188.6	85.21	16.13	16.65	5.75
	0.13	0.42	22.8	72.2	85.94 [*]	19.19	14.00	5.29
Pine (N=8)	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Rhododendron- Quercus (N=4)	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00

Table 6.14 Quantity of fodder collection, and C-stock in year 2010 (MgC/ha) and C-stock change between 2010 – 2013 (MgC/ha) in CFs by vegetation types

Note: ^{*} Less fodder collection in CFs had higher carbon stock in year 2010 [#] Less fodder collection had higher changes in carbon stock between 2010 and 2013; Ave. means average; SD means standard deviation

6.8.11. Litter collection

Reduction of litter collection was highest in CFs with *Schima-Castanopsis* forests followed by *Shorea*-mixed broadleaf forests and lowest in *Pine* and *Rhododendron-Quercus* forests (Table 6.15). This was reduced according to the quantity collection before the REDD+ (i.e. the higher the quantity collected before, the higher the reduction after the REDD+). Litter collection was highest in *Schima-Castanopsis* forests followed by *Shorea*-mixed forests. According to the communities, *Pine* needles were not collected for livestock bedding because they do not easily decompose for manure (used for agriculture crops).

Carbon stocks were high in the year 2010 where litter collection was high in all the forests except *Schima-Castanopsis* where slightly different carbon stock was estimated. This is potentially because higher biomass was associated with higher litter production in the forests. Carbon stock changes between 2010 and 2013 were different in different vegetation types with changed quantities of litter collection. Although there are no clear trends in the results of any of the CFs, higher carbon stock changes were found in *Schima-Castanopsis* and *Rhododendron-Quercus* CFs, where litter collection was highly reduced (Table 6.15). This was because communities preferred not to collect *Pine* needles and *Shorea* leaves as litter as these take time to decompose.

Vegetation	Change in litter collection (Mg)		Litter col before RI (Mg)	lection EDD+	C-stock i 2010 (MgC/ha	in 1)	C-stock change 2010 -2013 (MgC/ha)	
	Ave	SD	Ave	SD	Ave	SD	Ave	SD
Shorea-mixed broadleaf (N=10)	-9.16	18.03	30.98	55.12	109.34	12.52	13.39	1.75
	1.54	3.91	19.67	42.94	117.36 *	10.26	14.51	1.43
Schima-Castanopsis (N=10)	-11.74	19.58	41.04	66.91	93.90	15.82	16.54 [#]	8.14
	0.97	1.16	18.00	22.25	90.65	18.98	14.62	5.80
Pine (N=8)	-1.83	2.79	12.81	11.02	80.95	10.79	16.58	7.76
	0.83	2.36	3.50	8.18	97.11 [*]	16.24	18.01	7.84
Rhododendron-Quercus (N=4)	-1.22	1.17	12.02	7.33	73.96	16.84	10.97 [#]	3.87
· · ·	0.79	1.14	3.62	4.18	82.22 *	14.05	7.46	2.06

Table 6.15 Quantity of litter collection, and C-stock in year 2010 (MgC/ha) and change in C-stock between 2010 - 2013 (MgC/ha) in CFs by vegetation types

Note: ^{*} Higher litter collection in CFs had higher carbon stock in year 2010 [#] Less litter collection had higher changes in carbon stock between 2010 and 2013; Ave. means average; SD means standard deviation

6.9. Conclusion

With REDD+ interventions, local communities reduced the incidence of biomass reducing factors which was observed during the annual inventory. There were various factors involved in changing biomass carbon from the reference year to after REDD+ interventions. Of these, the elevation of forests, dominant species, proximity of the forests to road head and settlement are major biophysical factors playing a role in the change in biomass carbon in CFs. Similarly, the size of the forests, size of agriculture land holdings, livestock herd size, the use of alternative energy sources (Bio-gas, Petroleaum energy) and biomass carbon in CFs.

Although CF is contributing to the improvement of forests under business-as-usual scenarios, they are able to further increase carbon stocks by changing existing behaviours associated with forest management and use. There is likely to be a change in existing practices of communities for REDD+ benefits, but those changes may increase costs or lead to the sacrifice of benefits for communities. These sacrificed benefits in the name of carbon benefits can be higher if communities' members have only one option if the source of forest products is CFs and they are highly dependent on forest products for their livelihood. These sacrificed benefits may need to be compensated by the REDD+ carbon benefits to make the REDD+ mechanism effective for long run. However, carbon growth in a forest has biological limitations that will reach at equilibrium stage and have no further scope. Those forests at the equilibrium stage may not get carbon payments based on REDD+; and additionally the communities may be contributing and expecting benefits. This situation may need to be addressed in the future. Estimation of such added costs is important in the

design of an effective REDD+ for CF. The next chapter presents evidence of tradeoffs between community foregone sacrificed benefits and carbon benefits for REDD+.

CHAPTER SEVEN

TRADE-OFF BETWEEN SACRIFICED BENEFIT TO COMMUNITIES AND CARBON BENEFITS IN CFs

7.1. Introduction

The previous chapter gave results for the major factors affecting carbon stock changes in CFs. This chapter describes the benefits sacrificed by communities to generate per unit carbon benefits in CFs. The overarching goal of this chapter is to identify and estimate trade-offs between carbon stocks and net sacrificed community benefits in the REDD+ CFs.

This chapter is divided into four sections. The next section gives an overview of the additional contributions made by communities for the REDD+. This section quantifies annual contributions made for meetings, plantation, guarding, fire control, silviculture operations and the harvesting of forest products for the REDD+. The third section presents findings about changes in forest resource use practices in CFs and total foregone benefits sacrificed by reducing forest resource extraction for the REDD+. The final section concludes the chapter.

7.2. Contribution of communities for REDD+

According to the group discussions, communities are putting additional effort into meetings, plantation, guarding, fire control, silviculture operations and the harvesting of forest products for the REDD+ project. These added efforts have costs. After identifying community efforts for each year, the differences between average annual costs in the 3 years before REDD+ and 3 years during REDD+ were estimated for CFs by dominant vegetation type (Table 7.1). Similarly, the change in costs due to changed forest management activities for REDD+ was estimated and is presented in Table 7.2. In management practices, CFs with dominant *Pine* species had highest costs followed by *S. robusta* forests, *Rhododendron-Quercus* forests and *Schima-Castanopsis* forests, respectively. Among management costs, the cost associated with executive committee meetings and GAs was higher than other activities in all vegetation types. The meeting costs were highest in *S. robusta* forest and lowest in *S. robusta* forests, while harvesting costs which applied predominantly in *S. robusta* forests were reduced after the REDD+ project.

Vegetation types	¹ N (pers	/leeting son/ha/yr)	² G (person	A /ha/yr)	³ Planta (US\$,	³ Plantation cost (US\$/ha/yr)		⁴ Guarding (US\$/ha/yr)		°fire control (US\$/ha/yr)		⁶ Forestry operation (US\$/ha/yr)		⁷ Harvesting cost(US\$/ha/yr)	
	Before	After	Before	After	Before	After	Before	After	Before	After	Before	After	Before	After	
Shorea ^a	1.45	1.90	1.45	1.67	0.54	1.38	3.89	6.61	1.09	2.31	1.47	2.62	15.16	8.41	
Schima ^b	0.88	1.00	0.82	0.92	0.57	1.45	0.63	0.84	0.21	0.33	0.36	0.74	0.00	0.01	
Pine ^c	1.45	1.72	1.58	1.74	0.20	0.66	1.08	1.62	0.35	0.64	0.88	1.62	0.05	0.06	
Rhododendron ^d	1.21	1.46	1.04	1.16	0.00	0.38	0.41	0.47	0.00	0.00	0.22	0.86	0.00	0.00	

Table 7.1 Participant number (person/ha/yr) in meetings and general assembly and forestry activities costs in the CFUGs (US\$/ ha/yr) by vegetation type (3 years before and during 3 years of REDD+) in the study sites

Note: ^a*S. robusta* mixed broadleaf such as *Lagerstroemia parviflora* (Botdhairo), *Terminalia* spp, *Anogeissus latifolia* (Bajhi), *Dalbergia sissoo*, *Syzygium cumini* (Jamun); ^b *Schima- Castanopsis*; ^cPine dominated; ^d*Rhododendron- Quercus*; ¹Participants in executive committee meeting; ²participants in general assembly in CFs; ³Labour cost and material cost used for plantation in CFs, ⁴Labour cost for guarding forests from illegal activities; ⁵cost for fire line preparation and forest fire control in CFs, ⁶labour and material cost for silviculture operation in CFs, ⁷labour and material costs for timber and fuel wood harvesting in CFs

Vegetation types	Meeting (US\$)	Plantation (US\$)	Guardin g (US\$)	Fire control (US\$)	Silviculture operation (US\$)	Harvesti ng (US\$)	Total cost (US\$)
Shorea ^a	5.15	0.84	2.71	1.22	1.15	-6.75	4.32
Schima ^b	1.96	0.88	0.20	0.12	0.38	0.00	3.56
Pine ^c	3.73	0.46	0.54	0.29	0.74	0.00	5.76
Rhododendron ^d	2.55	0.38	0.06	0.00	0.65	0.00	3.64
Total	3.19	0.79	1.00	0.47	0.68	-2.07	4.05

Table 7.2 Cost of the CFUGs (US\$/ ha/yr) by vegetation type due to changed forest management practices after REDD+ in the study sites

Note: : ^aS. *robusta* mixed broadleaf such as *Lagerstroemia parviflora* (Botdhairo), *Terminalia* spp, *Anogeissus latifolia* (Bajhi), *Dalbergia sissoo*, *Syzygium cumini* (Jamun), ^b *Schima- Castanopsis*, ^cPine dominated, ^d*Rhododendron- Quercus*; -sign is benefits or reduced costs of the communities; 85NRs=1US\$

7.3. Changed forest product use practices of communities for REDD+

Annual forest product use practices in each CF were estimated and average annual quantity used before and during REDD+ project was estimated and presented (Table 7.3). Most of the CFUGs in the study sites were using eight forest products, namely timber, fuel wood, grass, fodder, litter, grazing cow, grazing goat and NTFP income. Foregone sacrificed benefits due to reducing timber harvest were highest in *Pine* forests followed by *S. robusta* mixed broadleaf and lowest in *Rhododendron-Quesrcus* forests. Among all foregone sacrificed benefits items, timber was highest in the case of *S.robusta* forests and *Pine* forests, whereas stall feeding (reduced grazing) livestock was highest in *Schima-Castanopsis* forests and grass cutting was highest in *Rhododendron-Quercus* forests (Table 7.4). By contrast, CFs with *Pine* and *Schima-Castanopsis* vegetation types had positive changes in NTFP benefits. According to communities, NTFP business is increasing, especially the use of herbs and shrubs, which involve comparatively less biomass loss than trees.

Table 7.3 Average annual forest product collection and income from NTFP in CFUGs by vegetation types (3 years before and during 3 years of REDD+) in the study sites

Vegetation	¹ Timber	(m ³ /ha)	ha) Fire wood (Bhari/ha)		Grass (Bhari/ha)		Fodder (Bhari/ha)		Litter (Bhari/ha)		Cow grazing (numbers/ha)		Goat grazing (numbers/ha)		Income NTFP (US\$/ha)	
types	Before	After	Before	After	Before	After	Before	After	Before	After	Before	After	Before	After	Before	After
Shorea ^a	0.19	0.15	41.03	32.42	56.12	45.47	37.43	32.39	8.12	6.91	79.73	36.41	272.77	159.93	0.15	0.10
Schima ^b	0.10	0.09	11.62	8.47	24.45	19.70	7.84	3.53	13.87	12.83	60.67	43.65	118.93	87.34	1.10	11.70
Pine ^c	0.19	0.11	9.60	9.13	42.46	37.96	0.00	0.00	5.46	5.13	15.57	12.50	97.93	81.36	19.40	30.21
Rhododendron ^d	0.03	0.03	13.18	12.41	26.50	24.30	0.00	0.00	4.63	4.51	0.00	0.00	0.00	0.00	0.00	0.00

Note: ^a*S. robusta* mixed broadleaf such as *Lagerstroemia parviflora* (Botdhairo), *Terminalia* spp, *Anogeissus latifolia* (Bajhi), *Dalbergia sissoo*, *Syzygium cumini* (Jamun);, ^b *Schima- Castanopsis*; ^cPine dominated; ^d*Rhododendron- Quercus*; ¹Timber quantity includes total quantity of timber harvesting in CF in this table but the value (given in Table 7.4) was estimated separately because price rate is different for timber of different tree species.

Vege.	Timber	Fuel-	Grass	Fodder	Litter	¹ Cow	² Goat	NTFP	Total
types		wood				feeding	feeding	income	lost
Shorea ^a	39.64	15.31	14.58	6.97	0.98	28.90	39.02	0.00	145.41
Schima ^b	5.77	2.99	6.37	5.58	0.73	11.06	10.28	+0.12	42.65
Pine ^c	42.38	16.01	5.83	0.00	0.23	1.98	5.36	+0.13	71.66
Rhodo ^d	+1.59	1.02	2.85	0.00	0.09	0.00	0.00	0.00	2.37
Total	19.96	8.16	8.60	4.99	0.71	14.76	17.86	+0.08	74.96

Table 7.4 Average annual cost (US\$/ha) of the CFUGs due to changing forestproduct use behaviour for the REDD+ by vegetation types of the study sites

Note: ^a*S. robusta* mixed broadleaf such as *Lagerstroemia parviflora* (Botdhairo), *Terminalia* spp, *Anogeissus latifolia* (Bajhi), *Dalbergia sissoo*, *Syzygium cumini* (Jamun), ^b*Schima- Castanopsis*, ^cPine dominated, ^d*Rhododendron- Quercus;* ^{1,2}Cow and Goat feeding costs include added costs due to reduced grazing; + sign is increased benefits of the communities, no sign is lost benefits, 85NRs=1US\$

Total annual sacrificed forgone benefits, annual contribution, total added costs (sum of sacrificed benefits and contributions) and annual carbon gain differed in CFs by vegetation type (Table 7.5). Highest sacrificed community benefits were estimated in *S. robusta* and mixed broadleaf forests followed by *Pine* forest and lowest in *Rhododendron-Quercus* forests. In terms of contribution, communities who were managing *Pine* forests had the highest additional costs followed by *Shorea* and mixed broadleaf forests; the lowest were in *Shima-Castanopsis* forests. In total, the highest additional cost to communities (equal to US\$149.73) was for managing one hectare of *S. robusta* and mixed broadleaf forests after the REDD+ project whereas the lowest (equal to US\$6.01) was in *Rhododendron-Quercus* forests. However, these foregone benefits may change with change in forest product prices and labour wage rates. Possible changes in these prices are discussed in next chapter.

Associated with these additional costs of communities, different carbon quantities were gained in CFs. There was a similar trend observed for annual carbon gains and community contributions in CFs; this was highest in *Pine* forests followed by *Schima-Castanopsis* forests and lowest in *Rhododendron-Quercus* forests. In terms of total costs (sacrificed foregone benefits and contributions), CFs with *S. robusta* and mixed broad leaf forests had the highest costs (US\$149.73 per ha), followed by *Pine* forests (US\$ 77.42) while the lowest was in *Rhododendron-Quercus* forests (US\$ 6.01). On average, communities have invested efforts equivalent to US\$79.01 per year to gain 5.1 Mg of carbon in CFs. This equals an annual total cost to communities of US\$15.5 per unit of carbon (Table 7.5).

Considering the 100 year time horizon of the present carbon increment, this study estimated the highest carbon benefits (US\$4.75) in *Pine* forests followed by *Schima-Castanopsis* (US\$ 4.61) and lowest in *Rhododendron-Quercus* forests. On average, all CF communities have increased carbon quantities through REDD+ activities at an annual value of US\$1.32 per ha assuming the present price of US\$7 per MgCO₂e is static into the future. However, if the market mechanism for carbon price improves (assuming an optimistic rate of US\$20 price per MgCO₂e), this will rise to US\$3.76 per ha. On the other hand, if the carbon market worsens, the price could go down to the lowest range of the voluntary market (US\$1.2 per MgCO₂e) and CFUGs would

only get US\$0.23 per ha (Table 7.5). Among the different vegetation types, the highest income will likely be realised by *Pine* dominated forests followed by *Schima-Castanopsis* and the lowest by *Rhododendron-Quercus* forests.

Species domination in CF	Foregone benefits ¹ (US\$/ ha/yr)	Added contribution ² (US\$/ ha/yr)	Total Cost ³ (US\$/ ha/yr)	Average change ⁴ MgC/ha (from 2010- 2013)	Average annual change ⁵ MgC/ha	Ave change ⁶ MgCO ₂ e /ha	Annual carbon benefit ⁷ MgCO ₂ e /ha (in 100 yr time horizon)	C benefits ⁸ per ha at US\$ 20/MgCO ₂ e	C benefits ⁹ per ha at US\$ 7/MgCO ₂ e	C benefits ¹⁰ per ha at US\$ 1.2/MgCO ₂ e
S. robusta mixed	145.41	4.32	149.73	15.70	5.23	19.19	0.19	\$3.84	\$1.34	\$0.23
Schima-Castanopsis	42.65	3.56	46.21	18.84	6.28	23.03	0.23	\$4.61	\$1.61	\$0.28
Pines dominated	71.66	5.76	77.42	19.43	6.48	23.75	0.24	\$4.75	\$1.66	\$0.29
Rhododendron- Quercus	2.37	3.64	6.01	6.78	2.26	8.29	0.08	\$1.66	\$0.58	\$0.10
Total	74.96	4.05	79.01	15.4	5.13	18.82	0.19	\$3.76	\$1.32	\$0.23

Table 7.5 Sacrificed benefits of communities for the REDD+ in CFs by vegetation types in the study area

Note: ¹Foregone sacrificed benefits include costs due to change in forest products uses before and after the REDD+, currency conversion rates 1 US\$=85NRs is used (during data collection date February 2012),

²Added contribution includes costs added due to forest protection and forest development activities (meeting, fire control, guarding, plantation, silviculture operation),

³Total cost is sum of sacrificed benefits and cost for added contributions,

⁴ Average changes in total biomass carbon (both above ground and below ground) between 2010 and 2013

⁵ Average annual changes in carbon stock (MgC/ha)

⁶Carbon benefit (CO₂e) MgC/ha is estimated from biomass carbon multiplying by 3.67

⁷ Carbon benefit (CO_2e) MgC/ha per year considering permanence issue (for 100 year time horizon as suggested) (Fearnside 2002)

⁸Carbon benefit per ha per year assuming optimistic rate US\$20 per ton

⁹Carbon benefit per ha per year assuming continuation of present rate US\$7 per ton for REDD+

¹⁰Carbon benefit per ha per year assuming worse market condition and pessimistic rate US\$ 1.2 per ton

7.4. Conclusion

These results show that there is a higher trade-off of community benefits for potential carbon benefits for the REDD+ project. Communities' direct costs for REDD+ activities exceeds carbon benefits. The price of timber price and the labour cost for meetings and other forestry activities are main causes for the higher cost to communities. Highest cost were found in Schima-Castanopsis forests and lowest in Rhododerndron-Quercus forests while higher carbon benefits were obtained in Pine forests and lowest in Rhododerndron-Quercus forests. However, most of the CFUGs are organizing forestry activities and meetings and assemblies during the leisure time of the majority of people. These therefore have lower opportunity costs as people are doing these activities during their free time. Since the conservation and enhancement of carbon stocks in forests would be the main aim of CFUGs together with the country's commitment under REDD+, communities can decide on the level of efforts they wish to contribute. However, they are unlikely to increase their effort further if their additional costs exceed additional income from carbon in the long term. At that stage, communities may not be interested in REDD+. The next chapter presents a discussion about the results of the study by relating these to the various issues, theories and findings published in the existing literature.

CHAPTER EIGHT

DISCUSSION OF FINDINGS

8.1. Introduction

This chapter builds upon the findings in previous chapters to explore the state of knowledge and ongoing issues for the REDD+ mechanism and CFs.

The chapter are divided into various sections. Each section deals with a particular issue with regards to the REDD+ mechanism and CFs based on the results of this study and the relevant literature. The section 8.2 gives community-based forest management regime, rural livelihoods and REDD+; The section 8.3 gives tree size distribution in community based forest management under the REDD+ mechanism; the section 8.4 covers changes in carbon stock in CFs under the REDD+ mechanism ; the sections 8.5 discuss about potential carbon growth in CFs; the section 8.6 gives key factors affecting carbon stock in CFs ; The section 8.7 gives trade-off between communities sacrificed benefits and carbon benefits in CFs; the section 8.8 explains performance of REDD+ projects in CFs. The section 8.9 gives contemporary issues in REDD+, the section 8.9 gives REDD+ CFs and the Section 8.10 gives ideal CFs for REDD+ benefits respectively.

8.2. Community-based forest management regime, rural livelihoods and REDD+

While over 7% of global forests are under community management (FAO 2011), this form of forest management is increasingly popular in developing countries with more than 22% of all forests in developing countries under this form of management (Nurse & Malla 2006). This is especially the case in Asia and Africa (White & Martin 2002; Nurse & Malla 2006); over 25% of total forests in Nepal are community managed. The initial aim of CF was to improve forest condition by involving local communities in the management of local forests and to provide opportunities to meet their subsistance need for forests resources (Arnold 1991; Luckert 1999). Most of these forest management communities are poor and use forest resources for livelihood support (Sunderlin et al. 2005). Such communities use forest products in conservative ways which have helped to improve forests, meet their needs for forest products and generate income (Dev et al. 2003). However, the existing benefits derived from such practices can be potentially further enhanced by incorporating incentive mechanisms such as REDD+. As discussed earlier, the REDD+ mechanism offers opportunities to both increase carbon stock in forests and reduce local poverty. This is supported by the finding of a previous study that CFs can contribute in reducing atmospheric GHGs (Maraseni et al. 2005). Such incentives can encourage local communties and govement to initiate REDD+ projects in Nepal. In providing support for both communites and forests (Kellert et al. 2000), the proposed REDD+ mechanism would incentivise the conservation efforts of communitties. While the REDD+ mechanism is evolving and policy frameworks are being formulated at international and national levels (Cerbu 2011), special consideration is necessary for CF systems that respects the subsistance forest product needs of communities, generates income and engages communities at the local level (MFSC 2013). This study shows all CFs have potential to increase carbon stocks however the quantity of carbon stock changes varies; hence, there is an opportunity to provide incentive to forest dependent communities to increase carbon sequestration capacity of forests and generate socio-economic benefits.

8.3. Tree size distribution in community based forest management under REDD+ mechanism

The decline in stocking rates of large-diameter tree size classes often but not always follows a reverse-J-shaped distribution (Hitimana et al. 2004). The reverse-J shaped distribution indicates the equilibrium structure of forests with similar mortality rates across the entire range of diameters particularly in less disturbed uneven aged forests (Chen & Bradshaw 1999; Nyland 1996). However, some size trees could be missing in disturbed forests (Hitimana et al. 2004) resulting in a changes in the size distribution pattern. Similar to Hitimana et al. (2004), results presented in Chapter 4 indicate that most of the CFs in this study tended towards a reverse j-shaped distribution pattern of trees with different diameter size. However, the distribution pattern in the study CFs varied with vegetation type in the reference year of 2010. Tree size distribution in these forests also changed with REDD+ project activities over the 2010-2013 period. These differences could be the outcomes of forest management activities of CFUGs but may also be due to site factors (Hitimana et al. 2004). Site factors such as soil are unlikely to change over short timeframes (Paul et al. 2002) though they may reflect the impact of past management activities. This study found dense canopy CFs had higher stocking rates of larger trees than did sparse canopy CFs. As reported elsewhere (Connell et al. 1984; Smith et al. 1997), this finding might also be due to the lower productivity of sparse canopy sites. As mentioned by Acharya (2002), forests canopy structures have been improved after community management. As improvement of forest vegetation helps to increase the productivity of forest sites (Pretzsch 2010), gradual improvements in forest productivity can be expected even in highly degraded forests. Degradation due to community management has been less expected because CFs are bounded by requirements for forest protection and sustainable use under the Forest Act 1993 and Forest Regulation 1995. Moreover, CF management is highly connected to the livelihood options of local communities (Sunderlin et al. 2005) and discussions with local communities in the field indicated their need to manage forests well both for themselves and also out of respect for their ancestors.

The *Forest Act 1993* of Nepal allows local communities to use forest resources for their basic forest product needs (GoN 1993). However, CFUGs use forest products differently according to their socioeconomic situations and available supply sources (Adhikari *et al.* 2004). If extraction exceeds recruitment, that results in lower stocking rates. Therefore it can be said that if CFUGs change the extraction rate of trees in CFs, they can change the stocking rates of certain size trees. Under the REDD+ project, communities have changed a range of silvicultural activities including harvesting practices, thinning and plantation activities. These changed activities could be one reason for the observed change in the stocking rate of different sized trees in the forests. Communities carried out these operations mainly to improve forest conditions while fulfilling their subsistence needs for forest products during the inception phase of CFs in Nepal (Gilmour & Nurse 1991). However, this aim was gradually expanded to address the economic development of communities through active forest management practices aimed at increasing the

quantity of high value timber trees in forests (Acharya 2002; Acharya 2003) and carbon stock increment benefits from the REDD+ projects.

Different vegetation types in CFs show different distribution patterns of trees by size. This is because the germination and growth characteristics of particular vegetation types affect the stocking rates of each size of plants in forests (Troup 1921). In the 2010, regeneration was higher in S. robusta dominated forests followed by mixed Schima-Castanopsis, Pine and Rhododendron-Quercus broadleaf. forests. respectively. These species have different seed production and regeneration capacity. Some need specific seed treatment to increase germination; for example, Rhododendron-Quercus and Pine need specific seed treatment to increase the germination percentage. As noted in silviculture books, climatic factors such as rainfall and sunlight affect regeneration and the canopy cover of forest can inhibit germination by limiting the amount of sunlight and rainfall at ground level (Troup 1921; Jackson 1994). A study conducted in S. robusta forest in Nepal found that an optimum spacing and canopy cover is needed for germination, with both highly dense and sparse forests having lower recruitment rates (Sapkota et al. 2009). Additionally, community interests may also affect the recruitment of certain species; regeneration of broadleaf species Alnus and Schima, Lyonia spp were promoted in in Pine dominated forests where community preference favoured a broadleaf forest type (Timsina 2005). However, this study found that community aims have been altered by including REDD+.

Stocking rates of saplings and trees in CFs can be affected by harvesting and thinning practices of communities (Nyland 1996). According to the communities, most of the CFUGs are doing selective harvesting for timber and thinning to create better growth environments in the forests; these activities may affect the stocking rates of saplings and trees. If numbers harvested during those activities exceed recruitment, stocking rates would be decline whereas the opposite will occur if they harvest less. This may affect the reverse-J shaped distribution patterns of vegetation which may then skew the potential yield of forest products. Communities might be protecting timber species such as *S. robusta*, *Pine* and *Schima-Castanopsis* species in the study areas but conducting no regular management activities in *Rhododendron-Quercus* forests which are not useful for timber. Moreover, *S. robusta* is a restricted species to harvest (GoN 2000) which may be one reason to increase the stocking rate of larger tree sizes.

For the REDD+ project, communities may operate thinning activities and remove unhealthy or bent trees and thin dense regeneration to create a better environment for the remaining good condition trees as suggested in the literature (Smith *et al.* 1997). This is identified as a common forestry operation of communities which may be responsible for the decrease in the stocking rate of new recruits in 2013 compared to 2010. Stocking rates of new recruits had declined to a greater extent in *S. robusta* forests compared to those in *Rhododendron-Quercus* forests which had either marginally decreased or slightly increased. Possible reason for these differences could be that a good seed year for *S. robusta* occurs at three to five year intervals (Champion and Seth, 1968 cited in Rautiainen *et al.* 1997) and there is no new regeneration in forests during this interval. Similarly, dieback could be affecting *S. robusta* regeneration. Repeated dieback over 10–25 years, with inhibition of shoot growth and mortality of newly germinated seedlings, has been recorded for *S.*

robusta (Rautiainen 1995). Evidence of dieback differs with site (soil, climate, topography) and disturbance factors (forest fire, grazing). However, under community forest management, forest situations have improved with positive results for regeneration stocks. On the other hand, the occurrence of forest fire is higher in low land terai that also affects regeneration (Gentle 1997). However, with better management of communities, fire intensity and damages impacts were reduced, helping to improve regeneration patterns. With the possible incentive mechanism of REDD+, CFUGs may have started to increase the stocking rates of larger trees to increase carbon stock in their forests and therefore saplings and trees were found to be increasing in CFs. Communities are carrying out thinning activities in a cycle which differs from that in other vegetation and may have changed stocking rates. Stocking rates of saplings was slightly decreased in sparse forests in Schima-Castanopsis and Rhododendron-Quercus dominated stands. This could be due to growth characteristics of *Schima-Castanopsis* which is fast growing (Paudel 2005); hence, sapling size may convert into tree sizes in few years. On the other hand, communities may harvest more smaller trees for a range of purposes, such as making handles for agriculture equipment, rafters for cattle shed or building houses, causing a reduction in the stocking rates of sapling. Changes in the stocking rates of trees, the main carbon pool stock in forests, were positive in all dominant vegetation types. This shows that CFUGs are able to increase the number of larger trees in forests for carbon benefit.

The overall distribution of different trees sizes in CFs is good indicating their capacity to continue to produce additional biomass carbon for the long term under the REDD+ project. As found in Pandey (2007), the results indicated that the present forest management practice of communities may be promoting timber trees and fast growing species for REDD+ carbon benefits which may undermine the global aim of biodiversity conservation (CBD 1992). Most of the CFUGs have income generating interests in their forest conservation (Subedi 2006), therefore there are possibilities to reduce the diversity and habitat condition of endangered plants and wildlife, and provide less support for maintaining ecosystem services for carbon benefit. As suggested by Angelsen (2012), safeguarding these services is important in REDD+ projects and there is need to consider protection and to incentivise communities to ensure these co-benefits are realised while designing and implementing REDD+ projects in CFs.

8.4. Changes in carbon stock in CFs under REDD+ mechanism

CFUGs were found to be contributing to the conservation of carbon stocks and showing capacity to further increase carbon stocks in forests under REDD+ projects. However, each CF had different capacity to contribute carbon in delivering sequestration outcomes. As also reported by Turner *et al.* (1995) and Litton *et al.* (2003), this study found a higher carbon sequestration capacity in trees than sapling and other pools (Appendix F). Therefore, a forest having a higher stocking rate of trees can store higher carbon stocks than less stocking. However, saplings have the potential to grow vigorously whereas trees may soon mature and stop growing. The sigmoidal growth pattern of trees indicates the higher increment potential of saplings than old trees (Birch 1999); therefore, a forest with a large number of saplings could

have more carbon benefits in the long run. Among trees, size and vegetation characteristics affect biomass carbon because biomass carbon is a function of the height and diameter of trees and the specific gravity of wood (Chave *et al.* 2005). A higher stocking rate of larger trees was found in *S. robusta* forests, indicating their higher carbon stock than other forests in the year 2010. Wood specific gravity of *Shorea robusta* species is higher than that of *Pine* tree species (MPFS 1988).

Tree species in forests are different by altitudinal location i.e. S. robusta and mixed broadleaf forests are located at lower altitudes, Schima-Castanopsis and Pine forest are at middle altitudes and Rhododendron-Quercus in higher altitude areas (Manandhar 2002; Vetaas & Grytnes 2002). The study found that the forests at lower and higher altitude areas have higher carbon stocks than those at middle altitudes. As in a study conducted in the forests of Hawaii with an altitudinal range of 914 to 2,438 m asl (Aplet & Vitousek 1994), this study found an increase in biomass as altitude decreased in all forests types with the exception of *Rhododendron-Quercus* forests. Rhododendron-Quercus forests were located at higher altitude areas but had higher quantities of carbon stock than Pine and Schima-Castanopsis middle altitude forests, indicating that many factors before and after the inception of CFs play roles (which have been discussed in next sub-section). The lower carbon stock in middle altitude forests was probably due to higher human pressure on forest resources; these forests were probably highly degraded before CF management as has been contended elsewhere (Gilmour et al. 1989; Gilmour et al. 2004). Although people living at higher colder altitudes need more fuel wood energy (KC et al. 2011), the populations are comparatively small compared to those in middle and lower altitude areas (CBS 2011b). Therefore, total demand is less in high altitude areas. According to the guidelines for community forestry in Nepal (CFD/DoF 2009), a CFUG can extract a certain proportion of the total increment of forest products. However, they reduced extraction for REDD+ project. In sparse plots, illegal extraction of forest products occurred from outsiders before CFs was established; this has been reduced now through mechanisms which guard forests. As degraded forests have lower site productivity (Fox 2000), sparse forests are comparatively degraded and had less carbon increases.

While comparing results from closely-related studies (similar species, climate and altitudinal ranges), this study generated slightly different results (Table 8.1). Both S. robusta forests and mixed broadleaf forests are closely related to the tropical moist forests (Asia continent) category (IPCC 2006) and forest types of Bihar and Uttar Pradesh (Haripriya 2000). Estimated carbon stocks in both these forests (in year 2010) was within the range given in IPCC (2006), higher than the estimation of FAO (2010b) for South and Southeast Asian forests and the estimation of Chhabra et al. (2002) and Kaul et al. (2009) for Bihar forests. Similarly, Schima-Castanopsis forests of the study areas are closely related to the sub-tropical humid forests (Asia continent) category of IPCC (2006) and forests of Himanchal Pradesh (Chhabra et al. 2002; Kaul et al. 2009). By comparison, the carbon stock estimate for Schima-Castanopsis forests was within the range given in IPCC (2006), higher than the estimation of FAO (2010b), and less than the estimation of Chhabra et al. (2002) and Kaul et al. (2009) for Himanchal forests. In the case of Pine forests and Rhododendron-Quercus forests of the study areas, these were similar to the temperate forests (Asia continent) in the IPCC (2006) category and forests of Himanchal Pradesh in Chhabra et al. (2002) and Kaul et al. (2009) categories. Carbon stock estimated in Pine forests and *Rhododendron-Quercus* forests was within the range given in IPCC (2006), higher than the estimation of FAO (2010b) and less than the estimation of both Chhabra *et al.* (2002) and Kaul *et al.* (2009).

Estimated carbon stocks in all categories of forests in the study were within the given range of IPCC (2006) categories and less than the estimation of both Chhabra *et al.* (2002) and Kaul *et al.* (2009) for Himanchal and Uttar Pradesh forests except in *Shorea robusta* forests. A possible reason behind this exception could be better condition of the forests before the inception of community forestry (Nagendra 2002) which has then kept improving under community management (Acharya 2002). The situation of *Schima-Castanopsis, Pine* and *Rhododendron-Quercus* forests were the opposite where forests were highly degraded when CFs was initiated in the early 1970s (Gilmour & Fisher 1991; Gautam *et al.* 2003).

On average, the stock of carbon in forests with REDD+ projects is increasing in CFs which agrees with the common pool theory that collective action and shared goal of communities can improved common pool resources (Ostrom 1990). As the success of CFs is dependent on various factors (Acharya 2002), REDD+ project tends to increase different carbon stock in CFs with various characteristics.. Higher increases in carbon stocks are found in CFs where the dominant vegetation is *Pine*, *Schima*-Castanopsis and mixed broadleaf which have fast to moderate growth characteristics. The increment is comparatively less in the CFs with slow growing species Rhododendron-Quercus (Smith et al. 1997). While compiling the sampling inventory, we also documented the growth stage of the forests. Most of these forests were young and had not reached maturity. Therefore, the CAI of the forests are higher. We estimated 5.0 MgC/ha/yr for S. robusta forest, 5.8 MgC/ha/yr for mixed broadleaf forests, 6.2 for Schima-Castanopsis forests which is higher than the value of 2.0 MgC/ha/yr given in IPCC (2006) for closely related forests in tropical and subtropical forest categories. Similarly, our estimate for Pine forests was 6.4 MgC/ha/yr which was higher than the value given in IPCC (2006) for temperate forests (4 MgC/ha/yr) but the estimate was lower in the case of *Rhododendron-Quercus* forests (2.2 MgC/ha/yr).

Possible key reasons for such differences include the conservation oriented forest management activities of local communities (Pagdee et al. 2006); the availability of alternative sources of forest products; the growth characteristics of the vegetation (i.e. fast, moderate or slow growing); as well as the productivity of the study sites (Smith et al. 1997). Moreover, the REDD+ pilot project has facilitated training and awareness rasing activities, promoting alternative energy (e.g. improved low fire wood consuming cooking stock and biogas), planting and silviculture activities and changing forest product use practices and has improved forest management activities toward carbon benefits. According to the communities, extraction of forest products is a key driver reducing carbon stock but reduction in forest products harvest is only possible if there are alternative sources. Therefore, the development and promotion of appropriate alternative energy sources could be an important intervention in Rhododendron-Ouescus forests where carbon stock increments are lower. Based on increased carbon stock, Schima-Castanopsis and Pine forests can deliver more benefits and these forests may also receive greater emphasis for REDD+ carbon benefits.

This study (Litter included)		IPCC (2006) (Litter excluded)		Biomass in states forests* were estimated by Chhabra <i>et al.</i> (2002) (Litter excluded)		Biomass in states forests* were estimated by Kaul <i>et al.</i> (2009) (Litter excluded)		FAO (2010b) (Litter excluded)	
Forest types	Biomass Carbon (MgC/ha)	Closely related classification	Biomass Carbon (MgC/ha)	Closely related classification	Biomass Carbon (MgC/ha)	Closely related classification	Biomass Carbon (MgC/ha)	Closely related classification	Biomass Carbon (MgC/ha)
Shorea robusta	121	Tropical moist deciduous forests (Asia	107 (6-334)	Bihar (Shorea robusta and Acacia species)	111 245	Bihar (Shorea robusta and Acacia species)	53.75	Average of 18 countries including	85.6
Mixed broadleaf	110	continental)		Uttar Pradesh (Shorea robusta, Mixed broad leaved)			174.90	Nepal (South and Southeast Asia category)	
Schima- Castanopsis	88	Sub-tropical humid forests (Asia continental)	107 (6-334)	Himanchal (hard wood mixed with conifers, Pines)	136	Himanchal (hard wood mixed with conifers, Pines)	194.95		
Pine forest Rhododendro n-Quercus	91 102	Temperate continental forest (Asia)	73 (12-196)	Himanchal (hard wood mixed with conifers, Pines)	136	Himanchal (hard wood mixed with conifers, Pines)	194.95		

 Table 8.1 Comparison between our estimates and other estimates for closely related forests for carbon stock (MgC/ha)

Note: * Species available in the state is mentioned by Haripriya (2000)

8.5. Potential carbon growth in CFs

CFUGs are found contributing to increased carbon stock in forests under REDD+ pilot initiatives; however, each forest could have a limit to its capacity to produce carbon stock. The maximum capacity of a forest to gain volume in an undisturbed natural condition is called "theoretical maximum yield in forests" (Rautiainen 1995) which is the maximum technical potential to grow. An undisturbed forest is an ideal state where only natural factors are active (Rautiainen 1995). It can be assumed that if community extraction of forest resources and management activities were not allowed in the CFs of the study areas, those forests would also follow natural growth patterns by attaining theoretical maximum carbon stock with time. In other way it can be said that if communities stop their activities and fully protect a forest, it could attain theoretical maximum growth. For designing a REDD+ project activities, knowledge about potential carbon growth (i.e. gaps between theoretical maximum carbon stock in forests and actual carbon stock in CFs) would be useful.

This study showed high potential of CFs to increase carbon stock under the REDD+ mechanism. While comparing carbon stock in S. robusta CFs (both dense canopy and sparse canopy) and undisturbed forests (i.e. theoretical maximum) by the age of forest stands, a clear gap has been identified. This gap is higher in sparse canopy forests than dense category forests (Table 8.2). This is supported by a study conducted in spruce forests which also found that dense canopy forests have higher biomass than sparse forests types (Michalek et al. 2000) so it is obvious that sparse canopy forests have high potential to grow with REDD+ intervention. Vegetation forms a sigmoidal curve if yield data is placed in y-axis and age data in x-axis ranging from zero to maximum age (Birch 1999; Weiner & Thomas 2001). This is because tree growth generally follows a pattern that is slower at young age, gradually higher after certain age and slower at older age (Weiner & Thomas 2001). A similar pattern of sigmoidal curves has been estimated in both dense and sparse CFs in this study (Chapter 5) however gaps differ with the age classes of forests. In both sparse and dense canopy CFs, estimated gaps were less in young stands than in old age stands. This shows young forests could have less potential to gain net additional carbon stock than older stands in which the gaps were higher. The majority of forests were found with the dominant age of trees in the 30 to 49 year category (Table 8.3) where the potential to increase was 230.26 mg/ha (in dense stand) to 441.33 mg/ha (sparse stand) in S. robusta forests. In the case of other vegetation types, there is limited knowledge about the theoretical maximum carbon stock by age of mix broadleaf forests, Schima-Castanopsis forests, Pine forests and Rhododendron-Quercus forests; therefore, this study did not perform detailed comparisons between the theoretical maximum and actual carbon stock in CFs for these forest types. It is inappropriate to use the theoretical maximum value of S. robusta for those vegetation types because the growth characteristics are completely different (Jackson 1994). However considering the same policy provisions of CFs and community practices with similar intervention strategies in all CFs of the country (Acharya 2002), it can be assumed that CFs with those vegetation types might also have similar gaps between the theoretical maximum carbon stock and therefore have similar potential capacity to grow carbon stock. CFs dominated by young age stands have comparatively smaller gaps between the theoretical maximum carbon stock of that age and older aged stands (Table 8.2) which indicates there are missing old age trees

in CFs. There could be two reasons: either communities are extracting larger and old age trees legally or illegally or the number of old age trees were less in the forests when the communities started CFs (Gilmour & Fisher 1991). However, CFs may have reduced stocks of old and large trees because they are allowed to harvest some forest products while ensuring forest condition is maintained (CFD/DoF 2009).

 Table 8.2 Theoretical maximum carbon stocks by years in CF and undisturbed S. robusta forests

S. robusta	Present carbon stock (by Age)				Theoretical carbon stock (by Age)				Potential to increase (by Age)			
forests	10yr	30 yr	50 yr	70 yr	10 yr	30 yr	50 yr	70 yr	10 yr	30 yr	50 yr	70 yr
Dense	34.82	125.76	183.74	205.50	88.83	356.02	588.35	760.55	54.01	230.26	404.61	555.05
Sparse	19.59	95.48	147.02	167.37					69.24	260.54	441.33	595.18

Table 8.3 Age category of CFs by dominant vegetation types

	CFs numbers by age of the dominant vegetation							
Dominant vegetation type	10-19 yr	20-29 yr	30-39	40-49 yr	50-59 yr	60-69 yr		
			yr					
S.robusta mixed broadleaf	0	7	11	15	4	0		
Schima-Castanopsis	0	14	13	11	5	0		
Pine	0	2	9	5	0	0		
Rhododendron-Quercus	0	2	5	1	1	0		

As noted, communities are protecting forests, carrying out plantings and sustainably harvesting forest products, all of which could change the natural patterns of growth in CFs (Springate-Baginski et al. 2003). However, even if communities do the same activities in forests, site quality differences may affect biomass growth (Daniel et al. 1979; Smith et al. 1997). CF provision was not a priority during the inception phase of CF in Terai areas of Nepal (MPFS 1988); however, the government brought in a separate policy to engage local people in the management of degraded forests and started the handover in 2000 (GoN 2000). Therefore, it is obvious that CFs were initiated to protect degraded forests and so biomass carbon by age was less in CFs than in undisturbed S. robusta forests. After protection and management activities of CFUGs, a gradual improvement in forests has occurred (Pagdee et al. 2006; MFSC 2013) and reversed the condition of degraded forests (Singh & Chapagain 2006). Theoretical maximum carbon stock values used in this study were taken from the yield table developed for S. robusta forests of Nepal (Rautiainen 1995), developed using data from almost uniform forests with 20% other vegetation mix in Terai. The data used for the yield table was collected from sample plots located in Makwanpur, Rupandehi, Kaski, Bara, Nawalparasi, Chitwan sites. These sites represent diverse locations of S. robusta forests matching the Kaverkhola watershed and Ludikhola watershed sites of this study in topographical and climatic aspects (Rautiainen 1995). While this is the best suited data for this study, the yield table of undisturbed forests was developed for site quality I and II, the highest ranking sites based on production capacity (Rautiainen 1995). The sites covered in this study were not uniform and likely to be more variable in terms of their productive capacity. However, if the forests were left without human disturbances, both dense and sparse forests of study areas could potentially achieve similar amounts of carbon stock as is found in undisturbed forests.

Appropriate open space available in forests provides opportunity for regeneration for some tree species. This is because sparser forests could have higher levels of erosion of productive soils, while denser forests might experience higher levels of competition among trees for sunlight and nutrients that affect growth rates (Smith *et al.* 1997). As sparse forest allows rainfall to reach the forest floor it has higher soil erosion potential (Rogers & Schumm 1991) that may reduce soil productity. With more soil erosion and lower forest productivity, sparse forest could have less biomass carbon growth potential by age than dense one. However this is improving with community efforts.

Increased rates of carbon yield may affect the optimum rotation age of a particular forest. In forestry, biological rotation (Huang & Kronrad 2001) and economical rotation (Gardiner & Quine 2000) are in practice. The biological rotation age is the time when each stand reaches a maximum mean annual increment i.e. highest biological productivity where MAI and CAI coincide (Tanvir *et al.* 2002). In economic rotation, the maximum net present value of a forest is considered to decide the rotation age of the forest. However, in the case of carbon sequestration benefits, rotation age could be different than either of these. Higher carbon sequestration could be possible with an extension of the rotation age beyond the optimum economical and biological rotations of forests because a longer rotation age is better for carbon sequestration (Liski *et al.* 2001; Foley 2009). The biological rotation age for biomass carbon yield in dense canopy CF was found to be younger than that of sparse canopy

CFs. As growth is dependent on optimum space available on the forests floor, disturbance, site quality, topographic factors, edaphic factors and climatic factors (Khanna 2004), forests dominated by the same vegetation type have almost similar topography, climate and soil types. Silviculture operations (i.e. thinning - removing the less healthy trees from dense stands and Singling - keeping single healthy sprout from multiple coppices from a stump) may help to reduce completion and create a favourable growth environment (Khanna 2004) to promote growth into old age, thereby helping to expand the biological rotation age in dense stands. This is supported by a research carried out for Douglas-fir which found that dense forests have a cumulative MAI curve which is greater at an early age than that of sparse forests (Curtis 1995), meaning that optimal spacing is an important factor in obtaining maximum sustained yield (MSY). Therefore optimum stocking rates of trees in forest stands is expected to increase carbon stock and silviculture operation might be helpful for it. Similarly sparse CFs reach MSY at an age similar to that of undisturbed forests, however the quantity of carbon is less than the theoretical maximum level and needs to be carefully managed to ensure its protection from biomass reduction factors.

In the case of dominant mixed broadleaf forests, Schima-Castanopsis, Pine and Rhododendron-Quercus forests, there is only limited evidence in the literature for the estimation of theoretical maximum biomass carbon stock levels under undisturbed natural conditions. There is evidence from the available literature for specific ages: for example above ground biomass of Pinus roxburghii plantation aged 9 years (Applegate et al. 1988a; Applegate et al. 1988b) and 10 years (Mohns et al. 1989) but this does not give growth patterns across the range of ages. Therefore, this study compared the 90th and 10th percentiles of carbon stock values estimated in forests and found huge potential to increase carbon stock if the 90th percentile carbon stock value is taken as the potential carbon stock in CFs. This study could not locate age-specific carbon stock measurements in CFs to enable the analysis if age-wise gaps. However, it indicates carbon stock gaps in forests and differences within a vegetation type. These differences are mainly due to the size and age of trees in stands because dense canopy cover with young or small trees store less carbon than dense canopy cover with old or large trees (Nowak & Crane 2002; Nowak et al. 2002). This shows potential to increase carbon stock with REDD+ interventions in both dense and sparse canopy CFs.

Although, there is potential to increase carbon stock in CFs, it may not be possible to attain the theoretical maximum level. This is because the fundamental principle of CF is to enable local communities to participate in the management of forests and sustainable supply of forest resources to fulfil subsistence forest product needs (Adhikari *et al.* 2007) therefore it may not be possible to fully control the extraction of forest resources and to enforce strict protection rules in CFs. Rural communities are highly depended on CFs for their forest product needs (Adhikari *et al.* 2004). In designing and implementing REDD+ projects, proper mechanisms must be in place to ensure that the socio-economic benefits to communities are safeguarded as agreed internationally (UNFCCC 2011); such mechanisms should ensure the supply of forest use practices or by providing alternative means for fuel wood energy, promoting plantations on private uncultivated lands and the supply of grass and fodder. If the REDD+ project does not consider these aspects, communities may

suffer and pressure on other forests outside project areas may increase with increased extraction of forest resources leading to increased leakage in terms of carbon sequestration. Leakage due to displacement of forest resource extraction from REDD+ to non-REDD+ forests (Hetsch & Chang 2010) does not contribute to the overall mitigation of climate changes. On the other hand, only a limited number of tree species are fast growing and capable of higher contributions to carbon sequestration compared to slower growing species. Communities may prefer faster growing species under carbon oriented management programs, which may not be beneficial from a biodiversity point of view. Because limited species forests are less likely to provide habitat for various flora and fauna, such programs may contradict the aims of agreed biodiversity conventions (CBD 1992).

8.6. Key factors affecting carbon stock in CFs

The existing literature indicates that biophysical and socio-economic factors influence forest areas (Geist & Lambin 2002; Lambin *et al.* 2001; Nagendra 2007; Yang *et al.* 2006). Among these factors, species domination in a CF, elevation of forests, proximity of forests to roads and settlements, the average age of the dominant trees of forests stand, per capita forests area, the size of household level agricultural land holdings, average numbers of livestock, the proportion of households using biogas and the proportion using petroleum energy, livestock grazing, caste heterogeneity, the quantity of biomass (timber and firewood) extraction, the quantity of grass collected, the quantity of fodder collected and the quantity of litter collected in CFs are indicated as major factors for carbon stock differences. As CF is completely different in terms of forest management objectives, management practices and outcomes to other government and private forestry models (FAO 2011), these factors may not play similar roles in CFs.

The dominant tree species in a forest is one of the key factors that affect carbon stock in CFs. As found in the literature, differences in carbon stock are mainly due to the growth characteristics of vegetation in forests (Silver et al. 2004). CFs with dominant Pine vegetation had higher carbon stock increment rates in this study due to fast growing nature of the species (Jackson 1994; Khanna 2004). Similarly, the wood density of vegetation types in the study CFs differ (MPFS 1988) and wood density affects the quantity of carbon stock in the forests (Baker et al. 2004). Although, this is mostly due to natural characteristics, CFUGs can influence the composition of forests through silvicultural practices (Pandev 2007) also changing carbon stocks. Therefore policy makers may need to consider species composition while designing REDD+ project activities. Different stocking rates of trees in forests also affect carbon stock differences in CFs. Across all vegetation types, CFs had higher carbon stocks in dense canopy forests (i.e. high stocking rates of trees) storing higher amounts of carbon stock than sparse canopy forests. This is supported by a study conducted in urban forests in USA which found that large diameter and higher numbers of trees per unit area of forests contribute large quantities of carbon sequestration in forests (Nowak & Crane 2002). Degraded condition in forests at the starting time of CF also affects carbon stocks in CFs. The level of degradation was worse in the mid-hills of Nepal when CFUGs were formulated and initiated to protect degraded forests (Gilmour et al. 1989; Acharya 2002 therefore these forests had fewer larger tree than other CFs in the study. Forest elevation is another possible factor affecting carbon stock in CFs. According to the literature (Gégout et al. 2005), climatic variables and soil properties in forests differ with elevation gradients. These climatic variables affect the distribution and growth rate of each species (Khanna, 2004) and therefore affect carbon stocks. A study found that biomass amount was higher in a warm wet climate compared to cool dry types (Aplet & Vitousek 1994) which may indicate lower altitude warm weather could support higher biomass carbon in the study sites. Lower altitude tropical forests can also store higher carbon stock than temperate forests (Malhi et al. 1999; Gibbs et al. 2007). This study found similar results in the case of S. robusta mixed broadleaf forests which store higher carbon stock than other forests while Schima-Castanopis and Pine forests (subtropical forests) had lower carbon stock than Rhododendron-Ouercus forests (lower temperate). There is, however, a seemingly unusual outcome in the higher biomass stock in sub-tropical forests (middle altitude area) possibly due to the high pressure on forest resources in middle altitudes resulting in higher levels of degradation prior to the establishment of CFs (Gilmour et al. 2004; Gilmour et al. 1989). In these areas, people were forest dependent and forests were treated as open access resources before the inception of the CF model (Gilmour 1990). Similarly, there was an unequal distribution of the population during the 1980s (i.e. before CF) when 60% people were living in hill areas and the major population concentration was in the middle altitudes (Shrestha & Conway 1985). This reflects the existing CF situation with forests located in middle altitude areas tending to have less carbon compared to those at higher and lower altitudes and so having higher potential to increase with REDD+ projects.

The proximity of forests to road-heads and settlements could be a cause of decreasing biomass carbon because people could have easy access to extract forest resources from the forests. A study conducted in Uttarakhand Himalaya found higher forest degradation in forests accessible from roads (Singh & Singh 1997); however, these were government forests where outsiders might have increased the pressure on forest resource use. CF is different as local communities take responsibility and guard against illegal harvesting. On the other hand, road access can provide opportunities to generate income, particularly for rural poor communities, by connecting them to market centres (Jacoby 2000; Banerjee et al. 2012;) which may indirectly help to protect forests. If people have higher incomes, they can afford alternative energy source such as biogas (Gewali & Bhandari 2005), which helps to increase carbon stock in CFs. Therefore, proximity from road head may not have effect on carbon stock in CFs as was indicated in this study with no clear pattern of carbon stock difference found. Similarly, there were no effects on CF carbon stocks due to the proximity of forests to settlements found in this study. This is because communities have full ownership and the shared aim to protect forests in CF systems and they contribute to improving forests (Ostrom 1990; Ostrom et al. 1994). Similarly, CFUGs do only selective logging (i.e. logging only those trees which are an exploitable size from improved forest management perspectives) and aim to improve forest status in all areas (Yadav et al. 2003). This was the opposite in a study carried out in Myanmar that found that a nearby village extracted more forest products from closely located forest areas, reducing biomass (Mon et al. 2012), where ownership was not given to the local people who were dependent on the forest resources. Therefore proximity to settlements may not affect carbon stocks in CFs. According to forest policies (GoN 1993), CF is a forest management system where traditional local users get forest management responsibilities and manage forests in a systematic way. Since, they have ownership of the forest resources, communities can further improve forests for REDD+ carbon benefits through plantation and reducing extraction from CFs in forest areas.

This study indicates that the age of trees in a stand is another factor affecting carbon stock in CFs. A higher quantity of carbon stock was maintained in older aged forests in all vegetation types in study CFs; other studies (Luyssaert et al. 2008; Keith et al. 2009) also report that old age forests have higher carbon stock than younger one. As carbon stock changes were greater in younger forests and most CFs were not old, there was high potential to increase carbon stock. Increased rates could be different in CFs due to the age of trees in forests because trees form sigmoidal growth patterns by age (Weiner & Thomas 2001) and a CF could fall at certain stage on the curve (e.g. juvenile, higher and slow growth). If REDD+ incentive mechanisms only consider the carbon increment, a CFUG that is doing better in terms of its conservation activities and had well stocked mature forests could receive less payment than a CFUG which had extracted all the aged trees and had only young trees. This situations would disappoint the people who were doing a good conservation job before the implementation of REDD+ projects. It might not be ethical to award such a situation while neglecting the work of others. Therefore, the REDD+ in CF should create a payment mechanism that considers these aspects.

The size of forest allocations (per capita forests) in CFs may affect the carbon stock. It was found in most of the CFs that the larger the per capita forest had higher carbon stock. This was also supported by findings reported in the literature (Agrawal & Goyal 2001; Nagendra *et al.* 2005). Communities can extract a certain proportion of increased wood biomass that does not reduce existing carbon quantity in forests. However, if people extracted similar amounts of forest products from large and small CFs, larger forests can continue to increase carbon stock more than smaller forests. This was especially the case when alternative sources of forest resources were limited and communities' only option was to use CFs. However, there is no indication of effect of forest size in high *Rhododendron-Quercus* forests in this study where people have comparatively larger areas of non-arable land growing trees and supplying forest resources compared to lower altitudes (Maltsoglou & Taniguchi 2004). Similarly, CFs with larger per capita forests can build higher carbon stock without compromising forest product benefits in REDD+.

As agricultural land provides food and other income sources to rural people (Adhikari *et al.* 2004) and also produces forest products, CFs with larger average areas per household of agricultural land can have higher carbon stock. However, in the Terai, (*S.robusta* and broad leaf forests areas), agricultural land is highly productive (Maltsoglou & Taniguchi 2004) and productive land is not used to grow trees as it is in higher altitude areas. With REDD+ project activities, the growth of biomass carbon in forests was higher in all CFs associated with larger areas of agricultural land as was shown in Table 6.6; this is because larger agricultural land holding households in a CF are able to access additional supplies of forest resources (Chhetri 2005). For example, where CFUG members have larger areas of non-irrigated land, they plant grasses and trees to meet the household demand for forest products; this was mostly evident in the middle and high altitude areas.

The proportion of biogas using households may affect the carbon stock in CFs. However, biogas is more effective in lower altitude warmer areas and only applicable if the household has 2 to 3 cattle to operate the biogas plant (Singh & Sooch 2004; WECS 2010); only the rich members of S.robusta-mixed broadleaf CFUGs were adopting biogas. Biogas energy can fulfil household energy needs and reduce the use of firewood as mentioned in other studies (Pokharel & Chandrashekar 1994; Shrestha 2005); therefore, if a higher proportion of CFUGs are using biogas, this may result in higher carbon stock in the forests. In terms of feasibility, the REDD+ project could facilitate the installation of biogas plants by providing subsidies, technical support and linking with government subsidy programmes for alternative energy adoption (GoN 2009). Some households in CFUGs had installed biogas plants but were using firewood for cooking large quantity food items including livestock feeds and heating house during winter. This is because communities found biogas stoves less effective for both cooking and heating. This indicates that biogas may possibly contribute to a reduction in firewood consumption but there is a need for technology development particularly in the design of a more effective biogas stove for cooking. As the average per capita income of Nepalese people is less than \$2 a day (CBS 2011a) and most forest dependent people are poor, they may not be able to afford the cost of plant construction or to buy cattle to operate it. Upfront financial support would facilitate biogas adoption. Without this, leakage due to fuel-wood extraction from REDD+ CFs to fulfil household energy demand may require REDD+ projects to include restrictions to control this.

Proportions of petroleum energy (e.g. kerosene) or LP gas using households in CFUGs is another factor affecting carbon stock. These energy sources are used for cooking in the study areas. The study found that if the proportion of petroleum energy using households in a CFUG was high, it had a higher increment of carbon stock than lower ones. This shows the contribution of alternative energy for increasing biomass carbon in CFs. However, fossil fuel is not a major source of energy, contributing less than 12% energy (World Bank 2011c) to total energy demand of Nepal. In general, LP gas was used by rich families mostly in city areas rather than rural areas. However, burning LP gas and kerosene stoves also contribute to greenhouse gas emissions (Bhattacharya & Abdul Salam 2002) and it would be advantageous to promote biogas, electricity and solar energy as alternatives to fuel-wood energy rather than these energy sources in the REDD+ project.

The quantity of tree and fuel wood extracted is another key factor affecting carbon stock in CFs. Similar to the finding that a higher biomass extraction in forests reduces biomass carbon stock in forests (Manhas *et al.* 2006), this study found that communities which extract higher biomass quantity (i.e. timber and firewood) from a CF had less biomass carbon stock. In CFs, communities can harvest an allowable quantity of timber and fuel wood as prescribed in the CF guidelines 2009 (CFD/DoF 2009) and forest inventory guideline 2002 (CFD/DoF 2002). However, after the REDD+ project activities, communities had reduced biomass extraction and were able to increase carbon stock in forests most probably higher than in non-project CFUGs. Communities are fulfilling their biomass product needs by extracting in a conservative way where the extraction quantity is less than the growth increment of a forest. As noted, for safeguarding the crucial needs of local communities and reducing possible leakage, communities should follow sustainable use practices of forest products and adopt alternative energy source of fuel wood in REDD+ project
areas. The use of improved cooking stoves (ICS) is a cheaper option to reduce fuel wood use in CFGs. ICS can reduce wood consumption by 30–40% more than traditional stoves (Dhakal & Raut 2010). Therefore REDD+ project need to consider these aspects while planning for generating carbon benefits in CFs.

This study found no effect of caste heterogeneity in carbon stock changes in CFs. As stated in the literature, heterogeneous groups may have different interests and therefore have problems reaching a consensus decision on better forest management practices leading to the "tragedy of unmanaged common resources" (Hardin 1968). However, this study found the opposite, with heterogeneous groups showing better understanding, mutual trust and collective action, and doing better in terms of forest management defiance of "common pool theory" (Gautam et al. 2003; Poteete & Ostrom 2004). Carbon stocks have been maintained in both heterogeneous and nonheterogeneous CFUGs. Since, CFUGs have different forest product dependency and economic situations (Adhikari et al. 2004), the REDD+ decisions may create problems for highly forest dependent members of the community. As per economic transformation theory (Breisinger & Diao 2008), communities may need to change all economic means such as energy use, construction material, livestock and agriculture practices for new mechanism of REDD+ which may have negative consequences to them. Therefore it is expected in the REDD+ project will provide alternatives or compensation for them to ensure the project's success.

Climate change itself can have some effects on carbon growth but there is limited research and information on the impact of climate change on the carbon sequestration potential of forests in Nepal. However, most of the research in Nepal predicts both the rainfall and temperature will increase (Malla 2009; Shrestha & Aryal 2011). From these likely rainfall and temperature scenarios, it can be said that climate change would have a positive impact on the carbon sequestration capacity of forests (Araújo *et al.* 2005). Moreover, carbon fertilisation due to increased carbon dioxide in the atmosphere would further enhance this capacity (Schimel *et al.* 2001; Schlesinger & Lichter 2001). However, climate change is a long-term phenomenon and this research reports the results from a four year survey; therefore it can be said that there is very little effect of climate change on the results of this study. Based on the results, it can be safely assumed that any increase in carbon is solely because of REDD+ pilot project activities and the changed behaviour of communities.

In a nutshell, the capacities of CFs to increase carbon stock differ. This study has indicated age of stands, dominant species, stocking rates of trees, elevation of forests, size of forests, extraction of trees and fire wood in forests may all influence such differences. These factors are contextual; for example, factors affecting lower altitude *S. robusta* forest may not affect higher altitude *Rhododendron-Quercus* forests. Discussion with communities and observation in the field confirm that communities have successfully changed existing forest resource extraction behaviours and reduced damages. Project interventions, particularly in distributing ICS to reduce fuel wood consumption, plantation support, awareness raising activities, distributed incentives for carbon enhancement focusing on the poor and ethnic communities (ANSAB/ICIMOD/FECOFUN 2013), could play a role in this.

8.7. Trade-off between communities sacrificed benefits and carbon benefits in CFs

Since REDD+ is an incentive-based mechanism to increase the carbon sequestration capacity of forests in developing countries, CFUGs were adding additional costs for the REDD+ projects with expectation of possible monetary benefits. They increased their efforts in forest management activities and reduced forest product benefits for REDD+. CFUGs were carrying out different forest management activities mainly plantation activities, guarding against illegal harvesting, carrying out more fire control activities, operating silviculture activities (Yadav et al. 2003) and holding meetings and assemblies for decision making (Agarwal 2009) before REDD+ project activities. According to the communties, they have increased activities which have contributed to increased REDD+ carbon benefit. These activities have an added cost burden for CFUGs; however, the added costs differed for various vegetation dominated CFs. For example, meeting costs were highest in S. robusta-mixed broad leaf forests and lowest in Schima-Castanopsis forests. Possible reasons for higher costs in a CF may be the number of households involved in a CF and higher frequencies of meetings and other participatory activities. If large numbers of members are organized in a CFUG to manage forests, meeting costs will also be proportionally higher. Participation of all members is expected in CF activities and every member needs to attend assembly and forestry activities voluntarily (CFD/DoF 2009). For good governance in REDD+ CFUGs, the participation of all members is crucial to ensuring the effectiveness of the REDD + mechanism (Cadman & Maraseni 2011) therefore CFUGs have increased the numbers of meetings and discussions for REDD+. CFUGs have also changed existing practices according to the characteristics of particular forests, which is seen in sparse S. robusta mixed broadleaf forests and Schima-Castanopsis forests where more space is available for planting and communities were doing plantation and guarding forests to stop illegal activities. The cost of security was higher in S. robusta mixed broadleaf forest which is a highly valuable timber tree and was facing higher illegal harvesting. Among the CFs of the study, CFs located in terai areas were highly sensitive to forest fires. Communities in these areas were engaged in fire prevention measures by constructing fire lines and fire control activities, as found by Acharya (2002). Similarly, *Pine* forest is also sensitive to forest fire and communities have taken care to prevent forest fire by organizing additional activities after the REDD+ projects, particularly raising awareness about fire damage and preventive measures (ANSAB/ICIMOD/FECOFUN 2013).

After a long period of conservation and management efforts by communities, trees in CFs are maturing and are now ready to sell on local markets (MFSC 2013). In addition to subsistence use, such matured CFUGs can generate a cash income from the sale of timber without detriment to forests. However, CFUGs may go for REDD+ carbon benefits by reducing harvesting quantities. This will increase the level of sacrificed benefits in mature CFs if the objective of forest management has shifted from subsistence use to a mix of subsistence and income generation from selling timber on the local market. It is evident in the study area that communities have reduced these extraction practices to increase carbon stocks in the forests with the expectation that the REDD+ income would be higher than the forgone benefits. However, as noted, sustainability of these carbon-centric practices is yet to be

researched. A higher sacrificed timber benefits were found in *Pine* forests followed by *S. robusta* mixed broadleaf forests whereas increased timber benefits were recorded in *Rhododendron-Quercus* forests. According to the communities, *Pine* trees are used in the veneer industries and also considered better for furniture in hill areas. CFUGs were moving towards conversion of vegetation composition from *Pine* (conifer forests) to *Schima-Castanopsis* associated broad leaf forests before the REDD+ project (Timsina 2005). But they may have changed those practices and reduced the quantity of extraction for REDD+.

As mentioned earlier, S. robusta timber is comparatively durable and a preferred timber species in Nepal and consequently more valuable. Similarly, firewood of S. robusta is also expensive due to its high calorific value (Jackson 1994; Kataki & Konwer 2002). Therefore, total costs become higher in S. robusta forests than other forests even if they reduce the quantity harvested under the REDD+. Rhododendron and *Quercus* trees were mostly used for firewood, being a less expensive option compared to S. robusta trees. On the other hand, a higher cost for reducing benefits from grasses and litter was found in S. robusta-mixed broadleaf forests; this was lowest in Rhododendron-Ouercus forests. Moreover, communities have reduced livestock numbers and are increasing the use of grasses and agricultural residue to feed their livestock from private land as reported in the literature (Cooke 1998; Devendra & Sevilla 2002) helping to reduce grass and fodder extraction from CFs. According to the communities, they were buying agriculture residues (such as paddy hay, wheat hay and millet hay) and also reducing the number of livestock for the REDD+; there is a trend towards keeping few productive cattle, replacing high numbers of non- productive livestock, in the Hindu Kush region including Nepal (Tulachan & Neupane 1999; Bhattarai & Kindlmann 2012). Although grass and litter make no significant contribution to total carbon stock in the forests, communities have changed their existing practices.

All these changes have costs to communities and their motivation behind these changes could be linked with potential carbon income in the future. Due to these extrinsic motivations of potential REDD+ benefits, the intrinsic motivation of local communities to manage and protect forests could change the fundamental concept of CFs (Deci *et al.* 1999). However, if REDD+ fails to provide benefits as per the expectation of communities, they may not continue their effort to increase carbon stock in CFs and may reverse emissions. There would be a high risk of uncertainty, as mentioned in the literature, regarding the sustainability of forestry sector emissions reductions (Fearnside 2000; Millar *et al.* 2007). While implementing REDD+ activities, creating situations based on the self-determination by communities to choose one out of various options could be better for long term behaviour changes rather than external suggestions and guidance (Deci & Ryan 2000). Therefore, providing better options including freedom for communities to make their own decisions and appropriate benefits might be important in REDD+ CFs.

As Non-Timber Forest Products (NTFPs) constitute lower carbon stocks when compared to trees, the sustainable extraction of NTFPs was continued during the REDD+ project period. Promotion of NTFP businesses based on a sustainable supply of raw materials from forests is an important factor for consideration in REDD+ project design. This could provide income to communities that help to support the livelihood of rural people and reduce poverty at local levels (Edwards 1996). Currently, communities living near *Schima-Castanopsis* and *Pine* forests derive income from NTFPs businesses; this needs to be further explored and adopted in other possible CFs. If NTFP provisions are incorporated in future REDD+ strategies, this may ensure the conservation and sustainable use of these resources. This will also support to conserve biodiversity by linking with the income of local communities as it is found crucial approach to improve degraded forest and reduce poverty (Subedi 2006).

While analysing trade-offs between communities' sacrificed benefits and carbon benefits from REDD+, community foregone costs (monetary value of sacrificed benefits and added contributions of communities) was found to be higher than carbon benefits (monetary value of added carbon dioxide equivalent) in CFs with the REDD+ scheme. Communities received US\$ 3.76 in carbon benefits while sacrificing US\$ 79.01. This cost does not include the administrative costs including MRV required under REDD+. These administrative costs could be significant and more than small scale forests are able to afford them (Huettner 2012). This will make carbon credit expensive and developed countries that go for compliance market may have less interest in this credit. However voluntary market could have interests on it. In this pilot payment mechanism communities are receiving payment with 16% weightage on carbon increment (Maraseni et al. 2014) but if this payment is based on performance, CFUG would receive about 1/6th of the current payment. That would not be of interest of communities therefore carbon additionally based payment mechanism of the REDD+ may not be beneficial in CFs. In a vegetation type wide analysis, a similar trade-off pattern was found. This indicates that REDD+ may not be beneficial for communities from a carbon only point of view. However, communities attach cultural and social values to CF (Maskey et al. 2006), which have invaluable benefits for them. Similarly, they want to keep their forests protected for their descendants and to maintain the condition of forests over time. Conserved forests provide environmental benefits including biodiversity (Pandey 2007), watershed management and rainfall patterns (Lu et al. 2001) which is important for local people and downstream residents. It would be good to consider these benefits while designing REDD+ project and making decisions about the carbon price.

Although this study found that costs have increased due to wage rates paid to communities for increased numbers of meetings and forestry activities for REDD+, most of these activities are being organized during leisure time for the majority of participants when their opportunity costs are almost zero. For example, most CFUGs organise these activities on Saturdays when job holders are free and opportunity costs are low because there are limited opportunities for cash income in rural areas in Nepal (Kelkar & Nathan 2005); meetings and forest management activities also tend to be organised to avoid the farming season. This is important because more than 70% people living in mountain regions, more than 62 % in hill and more than 58% in terai are engaged in self-employed agriculture (CBS 2011a).

More specifically, while focusing only on added carbon value in CFs, *Rhododendron-Quercus* forests could be a lower priority for carbon focused REDD+ projects developers. But, these forests have important biodiversity co-benefits, providing habitat for pheasants, orchids and leopards (GON 2002) and economic co-benefits from NTFPs (Smith Olsen & Overgaard Larsen 2003). These values need to

be respected and proper conservation measures need to be adopted through an appropriate PES mechanism.

The carbon market has been highly volatile (Bushnell *et al.* 2013; Hepburn 2007) and continues to change. Therefore this study assumes three different prices: one is a very optimistic price, one the present price and one a pessimistic price. The cost gap was estimated at US\$ 75.25 (i.e. between actual cost US\$ 79.01 and benefits US\$ 3.76) in the optimistic situation. This gap could be further increased if we used either the present price or pessimistic price. The optimistic premium price can be expected for carbon credits of Nepalese CFs because they can generate various non-carbon benefits such as biodiversity conservation, watershed protection and socio-economic safeguards for poor and indigenous people as agreed in UNFCCC (UNFCCC 2011) but it is still inadequate. Bundling of all the multiple ecosystem service related benefits generated in CFs and appropriate payment for these would compensate for the added costs to local communities under REDD+. PES such as water services is in practice in Costa Rica where it has helped to reward people who are generating these services (Pagiola 2008). Such models of PES need to be explored and adopted for CFs.

Appropriate market mechanisms could help to bring fair benefits at local levels. Most REDD+ talks currently taking place are about the market based mechanism (Kanowski et al. 2011; Lederer 2011; Pirard 2008); however, this mechanism may not be beneficial for CFUGs due to the higher costs to communities. This study found that the costs to communities were higher than the present market price (World-Bank 2013) which is a similar result to that found by Maraseni et al. (2014). Therefore, opportunities for PES and promotion of sustainable NTFP enterprise activities would provide additional income in communities. These can contribute to poverty reduction at local levels as mentioned in United Nations millennium development goal (United-Nations 2007). However, as stated in the literature, it is important that an appropriate benefit sharing mechanism, proper institutional frameworks and governance practices are developed for the REDD+ mechanism (Phelps et al. 2010). The CDM mechanism is considered to be complex in terms of its administrative procedures and to have higher transaction costs resulting in limited expansion of projects (Thomas et al. 2004; Chadwick 2006; Thomas et al. 2010); the REDD+ mechanism needs to be simple and to involve lower transaction costs.

8.8. Sensitivity analysis of foregone benefits of communities for REDD+ mechanism

As mentioned earlier, two of the highly sensitive attributes which affect the foregone benefits of communities are changes in forest products prices and labour wage rates. In this study, 2012 prices or wage rates were used for these attributes. However, due to the expansion of agricultural land and the scarcity of forest resources, the price of forest products, particularly for timber, may increase faster than the inflation rate (Clark 2001; Apsey & Reed 1994). Similarly, due to exponential growth in migration; labour scarcity is growing year by year. For example, from 2001 to 2011 the number of people who emigrated increased by 25 times from 76,000 to 1.9 million (World Bank, 2011a, 2011b) and the rate is even higher in recent years.

Therefore, the labour wage is highly likely to increase in the future. As noted, the forgone benefit of REDD+ project is already much higher than the additional income from carbon. Increasing forest product price and labour wage rates will make this situation worse. Therefore, there is no sign of any improvement in forgone benefits, while the income from carbon is unlikely to increase to a level which will compensate for this.

8.9. Performance of REDD+ projects in CFs

While evaluating the REDD+ project, evaluation of increased carbon sequestration capacity of forests and benefits to communities is taken as outcomes in this study. As mentioned by Liu and Walker (1998), an evaluation cycle covering three components namely behaviour, performance and outcome, can be applied to the REDD+ project (Fig 8.1). Assessment of these components could provide an idea about the appropriateness of the REDD+ project for CFs.



Figure 8.1 REDD+ project performance cycle can be used during REDD+ evaluation (*adopted from Liu and Walker (1998)*)

Since REDD+ is a newer mechanism than the existing CF practices, CFUGs has changed existing forest management and use behaviour. This study found changed behaviour particularly in forest improvement activities (silviculture activities, plantation, controlling of ongoing biomass reducing activities), meeting processes to increase participation of users and committee members and deliberation about forest management and benefit sharing decisions, extraction of forest products and use of alternative energy sources (biogas) and low fuel wood using ICS. As stated in the literature, this behaviour had affected biomass carbon in forests (Agrawal *et al.* 2011;

Pandey *et al.* 2013). Although, general rules and guiding principles of CFs are the same in the study area, socio-economic factors can result in contextual differences and induce CFUGs to change those behaviours (Maskey *et al.* 2006; Horn *et al.* 2012). In this study, CFUGs with dominant *Schima-Castanopsis* and *Rhododendron-Quercus* forests in Dolakha aimed to generate income from NTFPs. In those forests, mainly wintergreen (*Gaultheria fragrantissima*) and Lokta (*Daphne* spp) related enterprise activities, in addition to forest management, were being carried out and forest management activities were designed and implemented accordingly (Acharya 2005). Since, the forest management practices of small scale forest owners are affected by their management goal (Hugosson & Ingemarson 2004), setting apropriate goals in REDD+ CF is important. Communities aiming only to promote carbon stock in the forests would promote limited numbers of fast growing trees species and change practices to reduce the extraction of forest products. This may not be beneficial in long term.

The results of changed behaviour of CFUGs under the REDD+ mechanism would be carbon stock increment and co-benefit. If communities had changed their behaviour in a positive ways for REDD+ activities, that may help to increase higher carbon stock. In addition to carbon stock, socio-economic and environmental non-carbon benefits are considered as equally important outcome of REDD+ (UNFCCC 2011). Similarly, support for the generation of possible non-carbon benefits such as biodiversity conservation in CFs (as mentioned in Pandey 2007), protection of forest ecosystems services (Millennium Ecosystem Assessment 2003) regulation of the water cycle (Daily 1997; Nelson et al. 2009), continued supply of wild food and nontimber forest products benefits (Edwards 1996; Aryal et al. 2009) is important. This study found that CFs were performing well in carbon stock benefit which was higher in middle altitude sub-tropical forests with dominant Schima-Castanopsis and Pine species but lowest in Rhododendron-Quercus forests. However, the sacrifices made by communities were higher than the potential income from the REDD+. For S. robusta- mixed broadleaf forests benefits sacrificed were highest while they were was lowest in Rhododendron-Quercus forests. Higher potential increment from carbon benefits perspectives for Pine and Schima-Castanopsis forests may attract REDD+ project developers more.

Since safeguarding non-carbon benefits is a key issue in REDD+ (Visseren-Hamakers et al. 2012), REDD+ CFs must be able to maintain non-carbon benefits. However, if the project does not provide incentives to maintain non-carbon benefits, poor people may only focus on carbon benefits because communities want income from their forest conservation efforts (Subedi 2006). However, benefit sharing mechanisms for non-carbon benefits are yet to be developed at the international level (Angelsen et al. 2012), though they will potentially come in the future. There are few examples of local or regional mechanisms that may be beneficially adopted. Payment for water resources to upstream users in Costa Rica (Pagiola 2008) is one. This could be adopted in Nepal because CFUGs supply water for drinking, hydropower generation and irrigation which could be included in the payment mechanism. However, currently the Forest Act of Nepal (GoN 1993) does not have clear provision for the PES. In practice, a payment transfer mechanism has been initiated by Kulekhani hydropower in Nepal but no proper institutional framework has been developed. Now, some revenue comes to local bodies through government systems (Joshi 2011) but this may not be channelled to the proper people as it expected in PES. Therefore an appropriate model needs to be adopted in the study areas where farmers who use irrigation water, Hydropower Company and the municipality who use water from the watershed are encouraged to pay for the sustainable supply of water for the long term operation of those systems. Recreation and biodiversity conservation are also important ecosystem services (Subedi & Singh 2008) that can be linked to a payment mechanism. However, there is no clear payment mechanism developed at the local level apart from some practices to take fees from tourists in national parks and distribute certain moneys to the surrounding communities for local development activities (Nepal 2000; Goodwin 2002).

As ecosystem services are an outcome more of landscape level forests management than individual CFs (Goldman *et al.* 2007), a proper institutional set up would be important covering all watershed level stakeholders. This landscape level institutional set up may be helpful for both conservation and benefit sharing practices; however, there is a possibility of undermining participatory decision making process and recentralising the devolved decision making authority from CFUGs. As discussed in literature, there is also a risk that forest management would be recentralised through the REDD+ mechanism (Phelps *et al.* 2010).

Behaviour of the CFUGs changed and was able to improve forest status after they took on forest management roles at the inception stage of CF (Acharya 2002; Kanel & Kandel 2004) and further changed to increase carbon stock in forests by sacrificing existing benefits for REDD+. This could be the outcome of the raised level of interest of CFUGs in keeping good forests for future generations and securing the supply of forest products for their own benefit through REDD+ carbon benefits. This study found that the potential for REDD+ benefits was one of the motivating factors for these changes; therefore, it is uncertain whether these management practices will be continued in future if the benefits are not experienced by communities.

As mentioned in other literatures regarding successful community forestry projects (Gautam *et al.* 2004; Pokharel 2011), favourable international and national policies and better market mechanism for carbon and non-carbon benefits are important to generate better REDD+ outcomes. If this enabling environment is not supportive for the REDD+ in CFs, it may provide perverse incentive to local communities and may not function well in the long-term.

8.10. Contemporary issues in REDD+ and REDD+ CFs

<u>Governance practice and multi-stakeholder co-ordination in REDD+ CFs</u>: The REDD+ incentive mechanism involves multiple stakeholders who have a stake or interest or right in the forest resources and include those that will be affected either negatively or positively through the mechanism (Gomes *et al.* 2010; UN-REDD 2012). It involves complex multi-level and multi-stakeholder processes (Cadman & Maraseni 2011; Corbera & Schroeder 2011) aimed at fulfilling multiple goals beyond emission reduction. There is a challenge to build common understanding among those stakeholders (Ihor & Keeton 2009). Even in CFs, there were several stakeholders involved before REDD+ including government agencies, forest users, private sector entities, non-government organisations, indigenous people and other

forest dependent communities. While CFUGs are getting involved in the REDD+ mechanism, each of them may have some specific interests in the mechanism therefore coordination and collective action would help deliver better outcomes for the project (Angelsen et al. 2012). However, in coordination process some stakeholders may need to compromise their interests for REDD+ (Thompson et al. 2011). Therefore participation and deliberation of all stakeholders in decision making and implementation of REDD+ is important to facilitate coordination and address such issues (Cadman & Maraseni 2012). Among these stakeholders, safeguarding the interests of local communities and indigenous peoples are considered a key focus (UNFCCC 2011); therefore, securing free and prior informed consent from local communities is suggested during the designing phase of the project (Lawlor & Huberman 2009). As a CFUG has many subgroups working together (Varughese & Ostrom 2001), the REDD+ governance mechanism is expected to give more emphasis to representation in the decision making process and the benefits received by vulnerable and resource dependent groups within a CFUG and to design a program which ensure their traditional rights and livelihood benefits. This may need to develop a new institutional set up, mechanism and process than existing CFs. Similarly, present land tenure system of CFs may need to further clarify because land tenure is important in REDD+ market mechanism (Karsenty et al. 2014).

<u>Appropriate scale of REDD+ project in CF:</u> Discussion about the appropriate right scale for REDD+ projects is going on at the global level. There are three proposals of geographical scale for REDD+ projects; these are the sub-national (including project scale), national and nested scale highlighted in the literatures (Angelsen *et al.* 2008; Clements 2010; Okereke & Dooley 2010; Minang & van Noordwijk 2013). A sub-national or project level mechanism covers a small landscape or project area in developing countries; national level takes all forests of a country as one project; and the nested scale combines both national and sub-national approaches. A country can make decisions about the scale of REDD+ applied to facilitate the mechanisms and most countries have initiated subnational level projects at the beginning (Angelsen *et al.* 2008).

While comparing possible governance practices, leakage possibility, equity and costs of these three scales, sub-national and nested options seem better for CFs (Table 8.4). As the sub national level REDD+ is small and comparatively easier to start at pilot stage, it could attract private sector involvements (Angelsen et al. 2008). However, there is a possibility of leakage where communities can shift their forest product extraction activities from the project area to outside forests. This would not be an issue in the case of a National approach (Herold & Skutsch 2011). If the three watersheds of the study decide to go for REDD+ projects, each watershed could be one sub-national project or the aggregated area of all three could be one project. Both leakage and permanence possibilities in sub-national REDD+ projects are always suspected (Pedroni et al. 2009). Therefore there is uncertainty about the capacity to maintain carbon stocks in the long term if no proper mechanism is developed to ensure that the forest product demands of communities are fulfilled by, by for example promoting alternative energy (biogas and solar) for fuel-wood reduction and encouraging plantations on unproductive private lands. Similarly, if proper benefit of the REDD+ is not channelled to CFUGs, people may reverse activities that have been modified to take advantage of the REDD+ benefits and carbon may be released back to the atmosphere. Since project level REDD+ could involve fewer stakeholders than national level projects, a focus on poor communities could provide equitable benefit to communities. However it is difficult at the national level at which elites of society may benefit from their access to information that flows from national structures in CFUGs (Kanel 2004). Similarly, a national approach could result in recentralisation of CFUGs' decisions making authority and the possible exclusion of forest dependent communities in project design and implementation and benefits sharing. This likely change in institutional and benefit sharing mechanism of CFs can create social instability. Therefore, there is a need for the careful design and implementation of REDD+ to deal with these aspects if a country wants to go for national approaches. In terms of transaction cost (administrative and MRV costs), sub-national and nested scale REDD+ mechanisms could require higher benefits per CO₂e than the national level due to the experts input and disaggregated data required (Angelsen et al. 2008). However, the MRV cost can be reduced by involving local communities rather than experts only (Fry 2011; Larrazábal et al. 2012). Therefore, an appropriate protocol about community monitoring at global and local levels to maximise community involvement in the MRV process would be beneficial (Larrazábal et al. 2012).

Considering positive and negative aspects of these approaches, a nested approach could be a better option for countries. The nested approach provides opportunities for both sub-national and national REDD+ projects (Angelsen *et al.* 2008). Now, Nepal has agreed on the nested approach of the REDD+ mechanism (GoN/MFSC/REDD-Cell 2011) which is a good decision that allows the design of REDD+ projects covering certain CFUGs at the sub-national level. This will likely not hamper existing decentralised and participatory forest management practices. However, as stated in the literature there are challenges in establishing harmony between national and project level REDD+ project activities (Angelsen *et al.* 2008). As mentioned in the literature (Bushley & Khatri 2011; Corbera & Schroeder 2011), proper multistakeholder coordination mechanisms and clear vertical and horizontal linkages between institutions would be important to addressing transparency issues, proper MRV, leakage control, the safeguarding of socio-economic and environmental benefits and marketing activities. This may ensure harmony between different REDD+ projects.

Project scale	Description	Governance	Leakage and permanence	Equity	Cost
National	Country level REDD+ project covering all forests	 government ownership less participatory re-centralization of CFs top down multi-stakeholder policy failure 	 address domestic leakage permanence depends in policy and practice of stakeholder 	 risk of elite capture difficult to set up a proper benefit sharing from central to local 	• reduce MRV and transactio n cost for per CO ₂ e
Sub- national (Project level)	REDD+ covering a certain areas or project within a country	 direct participation of local communities and IPs in decision making and benefits weak participation of government body 	 domestic leakage possibility permanence depends in policy and practice of communities 	 higher participation of poor better continuation of CF practices and benefit sharing 	• higher MRV and transactio n cost for per CO ₂ e
Nested	A flexible mechanism start with sub- national level REDD+ and gradually move to national level or co- existence of both	 harder in harmonisation between two approach 	• control leakage	 encourage participation and proper benefit sharing poor people can get benefit from REDD+ 	• higher MRV and transactio n cost for per CO2e (need disaggreg ated data)

Table 8.4 Scale of REDD+	project for	CFs and	possible issues
--------------------------	-------------	---------	-----------------

Source: Angelsen et al. (2008), Herold & Skutsch (2011), Pedroni et al. (2009)

Safeguarding non- carbon benefits in CFs: As forests provide goods and services to local poor people in developing countries (Vedeld et al. 2007), the REDD+ mechanism would not be expected to make their economic situation worse with the protection of forests. This is clearly mentioned in the UNFCCC agreement (UNFCCC 2011) about developing safeguard strategies in REDD+ projects. The contribution of forests to poverty reduction and environmental sustainability has been highlighted and agreed in many international forums, including article 20 of the CBD, UNFF and UNFCCC COP 16. According to the CBD, international support is needed to address the overriding socio-economic development and poverty reduction priorities of developing countries (CBD 1992). Similarly, enhancing the contribution of forests to achieve poverty reduction and environmental sustainability is an approach agreed in the UNFF (UNFF 2011). Therefore REDD+ mechanisms, as agreed in the UNFCCC COP 16 in Cancun (Ehara et al. 2013), need to respect these provisions while designing and implementing project activities. In this study. CFUGs are contributing to increases in carbon stock but are focusing on timber trees. The carbon oriented CFs may reduce the numbers of slow growing trees in forests, resulting also in the reduction in non-carbon benefits (socio-economic, biodiversity, water). In order to secure socio-economic benefits for local communities, it would be beneficial to implement Pareto optimisation approach. Pareto improvement is an important welfare economic concept in which the allocation of goods among individuals is designed to make at least one individual better off without making any other individual worse off (Chou & Talmain 1996). In REDD+ CFs, it has been expected that changing existing practices to increase biomass carbon stock would not make community's existing livelihood situations worse given the provision of alternatives and increased socio-economic security.

Suitable market mechanism for REDD+ in CFs: Two main payment mechanisms have been discussed for possible REDD+ financing; namely, fund based and market based (Okereke & Dooley 2010; Streck 2010). In the fund-based approach, a common fund is created and carbon credit is purchased by the fund which is mostly created by voluntary financial contribution and tax (Skutsch & McCall 2010). These funds have the flexibility to be used in supporting REDD+ activities, and/or delivering performance based payments, and can therefore also receive non carbon benefits (Brown et al. 2008). However, there is a possible issue around reduced payment capacity as the voluntary carbon price differs from a market based mechanism (World-Bank 2013). On the other hand, the market based mechanism is a mandatory mechanism of carbon trade (Busch et al. 2012) in which a cap and trade mechanism could possibly follow under market based REDD+ mechanism (Pedroni et al. 2009). Similar to the CDM, an emission reduction cap (i.e. certified emission reductions (CER) given to a country) has been achieved through buying credits from REDD+ projects of developing countries. Although, market based mechanisms for REDD+ can provide opportunities to sell large amount of carbon credits generated in developing countries, it needs to be highly competitive (Lubowski & Rose 2013). While global policy for market based REDD+ is evolving, FCPF (Westholm 2010), bilateral fund, Amazon fund (Moutinho et al. 2011) and green climate fund (Elias et al. 2014) have been initiated to fund REDD+ pilot activities in developing countries. However, understanding the operational aspects and outcomes of these funds in the development stages of the program is limited.

This study found community cost tends to be higher than the potential carbon revenue from REDD+ in the competitive market (World-Bank 2013), therefore CFUGs could benefit if they follow fund-based approaches to get better payment for their small holder forestry practices, generating non-carbon benefits and carbon benefits. As stated above, communities are highly dependent on forest resources and they need to change their forest product use behaviours for REDD+ carbon which would only continue in the long term if poor rural communities can get ex-ante incentives that could be possible in a fund-based approach. Similarly there are possibilities to generate over or under debiting the carbon credits in forestry projects (Murray *et al.* 2007) which needs a proper monitoring mechanism and a fair estimation of discounting rates.

8.11. Ideal CF for REDD+ benefits

Since REDD+ includes both reducing existing emission levels and increasing carbon sequestration capacity in forests (Angelsen *et al.* 2012), conserving biomass carbon in old aged trees and promoting growth of young aged trees in forests to attain a higher level of carbon stock are important. The study shows that it is possible to increase carbon stocks in CFs for the REDD+ incentive mechanism if only the carbon stock increment is considered, although all CFs are not equally benefited from REDD+ mechanism. Individual CFs differs in terms of their socio-economic and biophysical characteristics (Gilmour & Fisher 1991) and these characteristics

affect the biomass carbon stock differences. Comparatively, CFs with the following characteristics tends to have a better possibility for increasing carbon stock through a REDD+ project:

- CF with large per capita forests available: As noted, if CFUGs have larger per • capita forest available, they can increase carbon stocks to a greater extent than smaller ones (Table 6.5). As rural people are highly dependent on forest resources (Adhikari et al. 2004), they fulfil their needs from CFs, private forests, agriculture land, nearby national forests and purchase from markets (Thoms 2008). However, some CFUGs may have limited alternative supply source and can only fulfil their needs from their CFs. In this case, a large per capita forest would be better for REDD+ performance. It is obvious that, if communities are extracting the same quantity of forest products from the forests, they can better increase carbon stocks in larger per capita forests than in those of smaller size. The areas of CFs in Nepal range from less than 1 ha to above 4000 ha. Out of 17, 808 CFUGs of the countries, more than 50% are associated with CFs of less than 50 ha in area with less than 0.12 ha per capita allocation while less than 100 CFUGs are above 1000 ha area with above 4.0 ha per capita allocation (CFD/DoF 2013). Therefore the size of forests could play an important role in the outcomes of REDD+ projects in CFs. The productivity of forests also affects yield in forests (Pretzsch 2010); however, where productivity situations are similar, per capita forests affect total carbon stock in forests.
- CFUGs with large agricultural land areas: This study found that CFUGs with larger areas of agriculture land have higher carbon increment rates (Table 6.6). The majority of people in Nepal are farmers (CBS 2011b) and their rural farming practices are interdependent with forest resources (Adhikari et al. 2004). Adhikari et al. (2004) found that households with larger agriculture land holdings were keeping more livestock and using higher quantities of forest products from CFs than smaller land holdings. This result differs from the findings in this study. This is because local communities have initiated the keeping of productive livestock by reducing the numbers of unproductive livestock in the study CFUGs. This is also supported by a study conducted in the Hindu Kush region including Nepal in which people were replacing high numbers of non-productive livestock (Tulachan & Neupane 1999), which might have reduced the need for forest products. Additionally, people having larger areas of agricultural land can grow more agricultural products and have additional options to fulfil their forest products needs. Similar to other findings in the literature (Gilmour & Nurse 1991), people were planting trees on private uncultivated land, so if CFUGs have more non-cultivated lands, they can grow more forest products. For private plantations, REDD+ project and had provided seedlings technical guidance (ANSAB/ICIMOD/FECOFUN 2013).
- <u>CFUG with high proportions of households using alternative energy:</u> This study found higher carbon stock increment in CFs where a higher proportion of households use alternative energy. Most rural people living in developing countries, especially in Asia and Africa, use wood as the main source of household energy (May-Tobin 2011). This is higher in Nepal where more than 80% of people use fuel-wood for cooking and heating purposes (WHO

2006). In order to reduce emissions from fuel-wood extraction in CFs, the use of alternative energy sources such as biogas, petroleum energy and ICS would be beneficial (Bhattacharya & Abdul Salam 2002; Katuwal & Bohara 2009). However, while alternative energy use can reduce forestry sector emissions, burning of petroleum energy contributes to GHG emissions (Kumar *et al.* 2003). Therefore replacement or reduction of fuel-wood energy use by promoting ICS, solar, biogas or wind energy would be useful for REDD+ benefits rather than promoting LPG, kerosene, diesel, coal (Kumar *et al.* 2003; Zhou *et al.* 2009).

- Reduced extraction of timber and fuel-wood: Reduce extraction of timber and fuel-wood from forests can increase stocking rates, basal areas and height of trees in forests (Chettri et al. 2002; Kumar & Shahabuddin 2005). As mentioned by Chettri et al. (2002), fuel-wood extraction reduces wood biomass productivity in forests. The growth rate of trees is age and speciesspecific (Thomas et al. 1999) but if trees are dense, this may reduce overall carbon stock than that of sparse forests, probably due to competition for light, moisture, nutrients and minerals. The growth rates of trees are also affected by altitude and other environmental factors (edaphic, topographic, and climatic) (Khanna 2004). Therefore, if CFUGs are extracting fuel-wood from high density stands, extracting bent and unhealthy trees only may help the growth of remaining trees and increase carbon stocks. However communities are carrying out thinning and extraction activities in CFs and there is less chance of having highly dense forest stands. Therefore, the study found increasing biomass carbon in forests where extraction was less than those subject to higher extraction. The fundamental objectives of CFs were to provide a subsistence supply of forest products and to improve forests; existing forest policy allows CFUGs to harvest certain amounts of the increased quantity of forest products (CFD/DoF 2002). However, they may reduce extraction (less that the allowable cut) for REDD+, but reduction may not be possible because rural people are highly dependent on forest products and it is not fair to implement strategies to reduce extraction without providing alternatives for REDD+ carbon benefits. Moreover, if the REDD+ project does not allow local communities to harvest forest products in CFs without making alternative arrangements, communities may shift their forest product extraction to surrounding forests, increasing the possibility of leakage. Therefore, reducing extraction while providing alternative options is important to increase the carbon sequestration capacity of CFs.
- <u>Have more fast growing species</u>: Tree species can be categorised according to growth characteristics into fast, medium and slow growing species (Korning & Balslev 1994). A fast growing species can increase carbon stocks more rapidly than slow growing species; therefore, a CF with dominant fast growing species can result in higher carbon stocks (Smith *et al.* 1997). In this study, *Pine* and *Schima-Castanopsis* are fast growing trees at a younger age than *S. robusta* (Jackson 1994) and CFs with these trees dominant and at higher growth stage on the sigmoidal growth curve (Figure 8.1) will have higher increment rates. If a REDD+ project incentivises additional carbon only, these forests will be more beneficial. However, for non-carbon benefits particularly from a biodiversity conservation point of view, lower altitude forests with dominant *S. robusta* and mixed broadleaf forests in the study area

have higher tree species richness. The biodiverse forests provide many essential services to humans that are important for the future sustainability of the environment and society (Pimentel *et al.* 1997). Therefore it is important to develop a mechanism that can maintain multiple species in the REDD+ mechanism.

- Appropriate incentive mechanism for CFUGs: CFUGs can increase and maintain carbon stock in forests if they get REDD+ incentives to compensate for their sacrificed benefits. This study found higher trade-offs between foregone benefits of communities and carbon benefits. Therefore, while comparing with optimistic carbon price and the present increment of carbon stock, communities do not get payments to compensate their sacrificed amount for REDD+, going against welfare economic theory "Pareto improvements" (Hochman & Rodgers 1969; Chou & Talmain 1996). As carbon sequestration has global benefit, REDD+ should pay appropriately without worsening the existing economic situation of local forest user communities. It was also agreed in the 16th session of UNFCCC (2011) to safeguard socio-economic aspects of local communities and indigenous people while implementing REDD+. Similarly, in community based management systems, local people should have some economic motivation to conserve forests in the long run and their conservation activities should provide an income to them (Subedi 2006). REDD+ with proper compensation mechanisms, as well as increased access to alternative options to meet the forest products requirement of local people, is helpful from a welfare perspective to local poor people and also for the effective long term implementation of REDD+.
- Younger age of the forest stands: Tree growth follows a sigmoidal pattern by age (Birch 1999; Clark & Clark 1999). At the beginning of the juvenile stage, the growth rate of trees is slower and they may struggle for establishment. As shown in Figure (8-1), forests with trees of juvenile age had low growth rates forests with medium age have higher carbon stock growth and this was slower again in forests of older aged trees. Young forests may not get carbon benefits based on present carbon growth capacity; however, this is important for long term benefits. After the juvenile stage, established trees show a rapid carbon stock growth pattern. In this stage, CFs can get higher carbon gains with protection and management activities. In this study, most of the CFs fell into this category. However, most of the middle altitude area forests steeper rates of growth than others because they had higher populations and high degradation levels before the CF initiatives. In sigmoidal growth pattern, older aged forests can have less capacity to increase carbon stock. If CFs have higher numbers of old aged trees, they have maximum carbon stock compared to younger forests. This study found middle altitude forests have younger aged trees which have potential to higher growth rates and seem



Figure 8.2 Theoretical growth pattern of carbon stock in forests [Adopted from growth pattern of trees mentioned in literatures (Birch 1999; Clark & Clark 1999) which was found similar in this study]

Degradation stage of forest of CFs: Neither fully stocked nor highly degraded CFs are likely to be beneficial for REDD+ carbon benefits. Fully stocked old growth forests have limited space to increase stem volume (Smith *et al.* 1997) and carbon stock. Knowledge about the level of degradation of forests when CF and REDD+ CFs are initiated would be helpful in designing REDD+ projects. As most CFs in mid-hill regions were initiated to control the further degradation of forests (Gilmour *et al.* 1989), communities had initiated conservation activities and reversed the forest status. But if forests are highly degraded and have lower soil productivity (Islam & Weil 2000), they may not have the capacity to increase carbon stock. Based on the current condition of CFs, sparse forests tend to be more degraded. CFs that have younger trees, have the potential to grow higher carbon stocks by protecting them and reducing extraction. These CFUGs can get higher benefits from the REDD+ mechanism. There is high possibility that incentives may be given to people who have previously extracted old trees and only started conservation after the REDD+ project commenced.

CHAPTER NINE

SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

9.1. Introduction

The overarching goal of this study is to evaluate the sustainability of the REDD+ project in community forest (CF) systems by considering carbon stock and costs to communities. Within this goal, the study estimated carbon stock and annual carbon stock changes in CFs by dominant vegetation types, estimated technical potential carbon stocks of undisturbed forests and actual carbon stock in CFs by dominant vegetation types, identified and analysed key factors affecting carbon stock changes in CFs, and estimated trade-offs between carbon stocks and net sacrificed community benefits under REDD+ in CF. The study has carried out a comprehensive evaluation of a pilot project performance which was implemented in three watersheds (i.e. Kayerkhola Chitwan, Ludikhola Gorkha and Charnawati Dolakha) covering 105 CFs with five major dominant vegetation types (i.e. *Schorea robusta*, mixed broadleaf, *Schima-Castanopsis*, *Pine* and *Rhododendron-Quercus* forests) in Nepal.

Compared to other REDD+ studies, this study brings knowledge from pilot demonstration projects using repeated carbon pool measurement in CFs. Other studies review and discuss theoretical perspectives and cover mostly government management forests. Those studies have missed the CF system which is an important forestry system, engaging mostly forest dependent local people of developing countries and with potential to engage in the proposed REDD+ mechanism. CF is important from both socio-economic and environmental perspectives. While the policy framework for the REDD+ mechanism is in the preparation stage at international and national levels, learnings from pilot projects would be helpful for the design of appropriate policies to be implemented for REDD+ in CFs. This study provides a comprehensive knowledge base to REDD+ project developers and communities involved in designing REDD+ projects and to policy makers while formulating polices with regards to REDD+ in CFs.

Based on the previous chapters, this chapter provides a summary of the major findings of this study, puts forward research contributions, highlights research implications and makes suggestions for further research.

9.2. Summary of major findings

In order to address the above mentioned goal of the study, the study set four objectives. The first of these was to analyse biomass carbon stock and change in biomass in CFs by dominant vegetation type; the second was to identify gaps between technical potential biomass carbon stock in undisturbed natural forests and carbon stock in CFs; the third objective was to identify key factors affecting biomass carbon in CFs; and the fourth was to estimate the trade-off between community foregone benefits and carbon benefits. The detailed results were presented in chapters 4, 5, 6 and 7 and discussions in chapter 8. Here, the major findings of the study and the implications for the design of REDD+ projects and formulation of REDD+ policy for developing countries are summarised.

9.2.1. Distribution of different sized trees in forests and carbon stock dynamics in CFs by vegetation types

Among different sized trees, recruitment (i.e. sapling) stocking rates were found to be higher in all CFs with a gradual reduction toward larger sizes (i.e. trees). This distribution pattern generates a continuous supply of biomass because if larger trees are removed from the forests, younger will occupy the vacant space. However, the stocking rates of each size of trees differ by vegetation type. Compared to the reference year, 2010, both saplings and trees are increasing with REDD+ project activities in the years since in most of cases. This shows communities are able to change the structure of forests through their forest management activities. However, while regeneration in sparse Rhododendron-Quercus forests is fostered, stocking rates of sapling in sparse Schima-Castanopsis forests and Rhododendron-Quercus forests have declined after the implementation of REDD+ project activities. Therefore the protection of saplings in sparse forests during thinning and harvesting could be helpful in sparse forests of both vegetation types. In the 2010 reference year, carbon stock was higher in S. robusta forests followed by mixed broadleaf and lowest in Schima- Castanopsis and Pine forests whereas in case of carbon stock changes, *Pine* and *Schima-Castanopsis* forest increased to a greater extent than others. Therefore, both incentivisation of S. robusta dominant forests as well as Pine and Schima-Castanopsis forests are important for REDD+ aims. Similarly, carbon stock and change in carbon stock in CFs are found to differ by dominant vegetation types. Even within one vegetation type, dense canopy and dominant trees at higher growth stages (i.e. young age after juvenile stage) in forests can increase carbon stock to a greater extent than sparse forests and older aged stands.

9.2.2. Potential growth of carbon stock in CFs

Comparing the technical potential biomass carbon stock in *S. robusta* forests, CFs were found to perform less well in maintaining present carbon stocks. Lower aged forest stands showed less difference while there was a higher difference in older aged stands. This shows that communities were extracting or losing older trees at a greater rate than younger ones in CFs. Among dense and sparse canopy forests, dense forests had fewer gaps available than sparse forests with technical potential carbon stock. Therefore, there may be higher potential to increase carbon stocks in CFs up to the technical maximum level in sparse forests than in dense. Similarly, dense forests have comparatively higher stocking rates of trees and so higher carbon stocks in forests. Therefore the optimum stocking rates of trees in forest stands can be predicted and proper silviculture operations organised for both sparse and dense forest types. Community forest user groups (CFUGs) may be able to increase the biomass of carbon stock and reduce gaps to the theoretical maximum level of undisturbed forests by fully protecting forests and stopping biomass reduction factors.

Limited works were found to estimate theoretical maximum carbon stock in undisturbed forests for any of dominant mixed broadleaf forests, *Schima-Castanopsis*, *Pine*, *Rhododendron-Quercus* vegetation dominated forests. Therefore, this study compared two extreme percentile carbon stocks (90th and 10th percentile) in both dense and sparse categories. Both 10th and 90th percentile carbon stock values

were higher in dense canopy forests than sparse one. Large gaps between the carbon stocks in plots within each canopy category forests were found as well. In order to reduce such carbon stock differences within one stratum or between two strata in CFs by vegetation type, several important socio-economic and biophysical factors should considered and addressed. These factors are associated with community forest management practices and forest product use behaviour. These key factors are summarised in the following sections.

9.2.3. Key factors affecting carbon stock in CFs

Biophysical factors, mainly the elevation of forests (m asl), average age of dominant trees of forests stand and socio-economic factors including per capita forests, household level agriculture land holding, proportion of biogas using households, proportion of petroleum energy using households and the quantity of biomass (timber and firewood) extraction were responsible for carbon stock differences.

Though these factors affect carbon stocks in all CFs, the level of effect is contextual. For example, factors affecting lower altitude *S. robusta* forests are not the same in higher altitude *Rhododendron-Quercus* forests. CFUGs have successfully changed existing forest resource extraction behaviours and reduced damages in response to the REDD+ project. To achieve these changes, project interventions particularly in distributing improved cooking stoves (ICS) to reduce fuel wood consumption, plantation support, awareness activities and distributed incentives for carbon enhancement focusing on the poor and ethnic communities have a role.

9.2.4. Model CFs for REDD+ projects

CFs with the following characteristics could be ideal for REDD+ benefits:

- CF with large per capita forests
- CFUGs with large per household agriculture land area available
- CFUGs with a high proportion of households using alternative energy
- Reduced extraction of timber and fuel-wood
- CFs with higher growth stages by age
- Have more fast growing species such as Pine and Schima-Castanopsis
- Having appropriate incentives and payments from REDD+

9.2.5. Trade-offs between foregone community benefits and carbon benefits in REDD+ CFs

Local communities are using forest resources (timber, fuel wood, leaf litter, fodder and grass and grazing livestock) from CFs. In order to increase carbon stock in forests for the REDD+ mechanism, communities have changed existing practices but have incurred additional costs. Generally, these costs were of two types (i.e. costs due to additional efforts made for REDD+ activities), including foregone costs for changed forest management activities and foregone costs due to reduced forest product benefits. For the REDD+ project, communities had reduced the use of forest products through reduced extraction of forest products (i.e. timber, fuel wood, grass, and litter) and reduced livestock grazing (i.e. cow and goat) which has added costs. In species-wise comparisons in CFs, *Pine* forests had sacrificed the highest level of timber benefits followed by *S. robusta*-mixed broadleaf forests. However, total sacrificed benefits were highest in *S. robusta*-mixed broadleaf forests followed by *Pine* forests and lowest in *Rhododendron-Quercus* forests. There was no reduction, and even some increase, in NTFP income under the REDD+ project. As NTFPs share very little biomass carbon compared to other pools comprising the total biomass carbon of forests, the REDD+ project may encourage communities to increase their NTFPs income through conservation oriented extraction.

There are higher possible trade-offs between community foregone costs (the monetary value of sacrificed benefits and added contributions of communities) and carbon benefits (the monetary value of added carbon dioxide equivalent) in CFs for the REDD+ projects. On average, CFUGs get US\$3.76 benefits from biomass carbon by sacrificing US\$79.01 per hectare of forest which is much higher than expected. By forest type, CFUGs need to spend US\$38.99 in S. robusta- mixed broad leaf forests, US\$16.30 in Pine forests, US\$10.02 in Schima-Castanopsis forests and US\$3.62 in Rhododendron-Quercus forests to generate each dollar of benefit from the REDD+ project. This trade-off is less in *Rhododendron-Quercus* forests; however, these forests have lower carbon increment rates. Similarly, S. robusta mixed broadleaf forests have higher trade-offs, but these forests also have higher carbon stocks and also more valuable timber trees. Although all forests contributed increased carbon stocks under the REDD+ project activities, not all of these forests deliver benefits when trade-offs are considered. Therefore CFUGs may not gain benefits from the mechanism if it only considers carbon stocks as a commodity. CFs also generates social and environmental benefits. These non-carbon benefits should also be taken into account while making decisions about carbon prices, but these are currently not included in the market mechanism. The most suitable mechanism to follow for REDD+ in CFs is possibly a fund-based and voluntary market mechanism. Additionally, higher trade-offs were also associated with community wages for increased participation in forestry activities including meetings for REDD+; most of activities can be organized during the leisure time of the majority of participants when their opportunity costs are almost zero to reduce costs.

9.3. Contentions made before the study

This research was based on several contentions during data collection, analysis and interpretation. Most of these contentions were supported in this study.

• Forest biomass (and therefore carbon stocks and sequestration rates) in CF is affected by management practices. These include harvesting and other disturbance practices. Due to these disturbances, actual potential carbon stock in CFs is much lower than technical potential carbon stock in undisturbed natural forests.

Biomass carbon stock and changes in CFs were also affected by forest disturbances factors such as forest fire, biomass extraction and grazing. With the REDD+ project activities, communities reduced these disturbances thereby increasing the carbon stocks in forests. Compared with the technical potential carbon stock in undisturbed forests, biomass carbon in CFs was less indicating significant potential to increase biomass carbon in forests through the REDD+ project. There is higher potential to increase biomass carbon in sparse forests than dense. Forest management practices which most reduce the harvesting of forest products and control disturbances can deliver higher carbon benefits.

• Carbon stocks and sequestration rates are possibly affected by various biophysical and socio-economic factors which include altitude, age, forest canopy cover, species type, size of forests, caste heterogeneity, agriculture land holding size, disturbance levels, forest product extraction and use of alternative energy.

Of the various biophysical and socio-economic factors, altitude, age, forest canopy cover, species type, size of forests, agriculture land holding size, disturbance levels, forest product extraction and use of alternative energy affect biomass carbon stock and change in CFs. The study found that forests located at lower altitudes, forests with dominant old age trees, dense canopy forests, higher wood density species, larger per capita forests, larger per capita agriculture land holding forests, less disturbances (i.e. fire, harvesting of forest products, grazing) and/or higher proportions of household using alternative energy are likely to have higher carbon stocks. However, these factors are contextual; therefore, assessment of these factors will help in the design and implementation of REDD+ project activities in CFs. There were many caste groups working together in CFs but the study found that caste heterogeneity does not affect carbon stocks if CFUGs have shared goals and perform collective actions.

• *REDD*+ incentives may be insufficient, especially in the long term, to offset the economic losses from changing management practices of communities.

CFUG's costs for REDD+ were higher than potential carbon revenue. Communities need to spend more to generate additional carbon in their forests, although this differed across dominant vegetation types. Forests with valuable timber trees and high labour wages need to spend more to increase carbon stocks than others. Since rural communities who are managing forests are poor, they expect additional benefits or at least reimbursement of their costs for REDD+ activities in CFs. Therefore, if REDD+ only takes carbon benefits into account and goes for a market based mechanism, this is unlikely to compensate a communities may revert to previous behaviours for their livelihood support and emit CO_2 again in similar quantities.

9.4. Research contributions

This research contributes new knowledge in the following areas.

- <u>Research approach/framework:</u> This study assessed the REDD+ demonstration project in CFs using annual measurements of carbon stock changes, changed behaviour of local communities and possible outcomes in terms of carbon benefits and associated sacrificed benefits. This assessment process, by covering all aspects, can be useful in REDD+ project evaluation in CFs in developing countries.
- <u>Design and implementation of REDD+ project in CFs</u>: Carbon stock and change in carbon stock in CFs vary with vegetation type and canopy cover. Therefore, while designing the REDD+ project and benefit sharing mechanism, consideration should be given to forest vegetation and canopy type. This study brings new knowledge that even with community management, REDD+ performance differs and vegetation-specific project activities need to be designed and implemented.
- <u>Key factors to be considered in designing REDD+ project in CFs</u>: This study found that carbon stock and change in carbon stock in CFs differ with variation in socio-economic and biophysical factors. The impact of these factors differs with CF vegetation type. Therefore, species type, canopy cover, elevation, age, size of forests, agriculture land holding size, disturbance levels, forest product extraction and use of alternative energy are key factors that need to be considered in each vegetation types while designing REDD+ project activities.
- <u>Potential carbon increment in CFs with REDD+ project interventions</u>: Although all CFUGs contribute to increased carbon sequestration capacity in their forests, existing carbon stock in the CFs are currently less than the technical potential capacity of forests. Therefore, there is significant space available for the REDD+ project to increase carbon stock in CFs. Comparatively sparse canopy forests have the highest gap and greater potential for increase than dense forests. In age-wise comparisons, old aged forest stands in CFs have higher gaps than younger aged undisturbed forests. This means there are fewer old aged trees in CFs therefore CFUGs need to reduce the harvesting of old trees and create an environment which facilitates an increase in the proportion of old trees in forests to reduce that gap.
- <u>Trade-offs between sacrificed community benefits and carbon benefits in</u> <u>REDD+CFs</u>: This study estimated that CFUGs are able to increase biomass carbon in CFs by changing their existing forest management and resource use practices. These new practices have added costs for communities either through the loss of benefits or by demanding additional efforts for REDD+ activities in CFs. Analysing real time costs or sacrificed benefits to communities and potential carbon benefits indicates that the REDD+ project may not be beneficial in CFs. However, it could be beneficial if: a community's contribution (mostly participation and wages cost) could be

arranged during times of minimum opportunity cost (i.e. zero cost or free time of participants); other co-benefits could be bundled together with carbon benefits to get a premium price from REDD+ project in CFs; and possibilities could be explored for a PES mechanism at the local level for water and recreation services.

• <u>Benchmarking for further research</u>: The study of carbon stock increment in CFs involves ecological aspects of forests and behavioural aspects of local communities. There is a need for long term studies to confirm growth trends and the impact of community practices. This study covers four years of data which may not be enough to confirm growth patterns and changes in community behaviour for REDD+ carbon benefits. However, this study will be a benchmark for future long-term studies against which to compare changes in future.

9.5. Research implications

REDD+ is the likely incentive based mechanism developed to reduce forestry sector GHG emissions in developing countries. After the 13th session of UNFCCC (2007), REDD+ is on the agenda as a possible option in global climate change discussions. There is good indication for inclusion of REDD+ provisions in post Kyoto negotiations although no clear policy frameworks have been developed. Therefore several pilot initiatives have been implemented for REDD+ which are expected to generate knowledge and fill the policy vacuum. Some research implications are made based on this study's findings for better designing REDD+ for CFs in developing countries.

9.5.1. Implications for policy making level

Develop a benefits distribution mechanism accordiing to performance of CFUGs: As performance-based payment is the basis of REDD (ie, payment according to quantity of carbon added by the project), this study revealed that present CFs have high potential to add biomass carbon stock. Therefore, REDD+ mechanisms could help to incentivise local forest user communities for additional biomass carbon generated in their forests. Now, all CFs are placed in a single basket and treated in the same way with regard to forest management and subsistence use practices in a country. While considering ecosystem services (including REDD+ biomass carbon), different CFs can contribute differently with their different capacity to generate outcomes. Therefore, incentives are needed which accord to their performance through a fair benefit sharing mechanism. If a CF is generating higher carbon stock, they need to get more benefits than lower carbon forests. Policy should be developed by considering these aspects in order to make the REDD+ implementable in the long term.

<u>Consider degradation status of forests before CF and species domination while</u> <u>designing REDD+ mechanism:</u> Carbon stock is comparatively less in middle altitude forests in Nepal i.e. *Schima-Castanopsis* and *Pine* forests while the potential increase rate is higher in these forests. Similarly, sparse forests have higher potential increase rates of biomass carbon than dense forests. Therefore, REDD+ project should focus on *Schima-Castanopsis* and *Pine* forests and sparse canopy forests for optimal benefit. While REDD+ includes both addition and conservation of existing carbon stock in forests, lower altitude *S. Robusta-* mixed broadleaf and higher altitude *Rhododendron-Quercus* forests should also be considered during REDD+ policy formation.

National and international policies should emphasise the need to bring additional benefits at CFUG level by providing options for payments: CFUGs have changed their existing forest product use practices and forest management activities for REDD+. Due to these changed behaviours and efforts, they have lost benefits and/or increased costs. In comparison to the potential benefits from increased carbon stock, sacrificed foregone community benefits (i.e. costs) are higher. This shows REDD+ is not beneficial for CFUGs. However, if REDD+ benefits were to be expanded from carbon benefits by incorporating socio-economic and environmental benefits, the price of carbon could increase and provide additional income. International and national policy frameworks should be developed in favour of this provision.

9.5.2. Implications for project implementation level

<u>Knowledge to REDD+ project developers</u>: This result represents CFs in developing countries where communities are allowed to conserve and harvest forest products for their subsistence use. It indicates that there is potential to increase carbon stock in CFs in Nepal and community managed forests in other similar developing countries. Therefore, appropriate REDD+ activities and incentive mechanism may be able to increase carbon stock in majority CFs. These results can encourage REDD+ project developers to design and implement REDD+ activities in non-REDD+ CFs with similar forest types.

Possible activities for carbon stock increment in REDD+ CFs: The outcome of the REDD+ project in CFs depends on various factors. Among socio-economic factors, higher proportion of alternative energy using household in CFs and reduction in biomass extraction from forests are key interventions that could help to increase biomass carbon in CFs. These include alternative energy, mainly biogas, in lower and middle altitudes (mostly in warmer sites) with special focus on poor households and the promotion of improved cooking stoves (ICS) in all areas to reduce wood consumption during household cooking and heating. Poor people cannot afford the cost required to install and operate biogas plants therefore upfront financial support could be helpful to facilitate the adoption and use of that technology. ICS is a relatively cost effective intervention to reduce fuel wood consumption. REDD+ project activities should include technical and financial support to promote biogas and ICS wherever feasible in order to reduce fuel wood extraction from forests. In the case of other biomass, efforts to reduce illegal harvesting of trees and facilitate plantation activities (tree, fodder and grass) on unproductive agricultural lands will help to provide additional forest product supply options to communities. Therefore plantation activities and creating environments to grow trees in sparse canopy forests could enhance biomass carbon in REDD+ project.

<u>Considerations to minimise risk of perverse incentives to local communities in</u> <u>REDD+ CFs</u>: Communities have sacrificed different foregone benefits for REDD+ projects in each CF while increasing, by different amounts, the biomass carbon in forests. Therefore REDD+ projects need to develop a benefit distribution mechanism considering these aspects. If this consideration was not made, it would lead to a perverse incentive mechanism which may compromise the effectiveness of REDD+ in the long term. Accessing voluntary markets and seeking possible payments for non-carbon benefits would be helpful.

9.6. Limitations of the study

This study collected and analysed periodically collected carbon pool data in relation to the biophysical characteristics of CFs, community forest product use and management activities. While this study utilises scientific approaches to data collection and analysis, the following limitations should be taken into consideration:

- i. This study uses four years' data collected from the field after the REDD+ project activities in CFs in Nepal commenced. Finding on carbon stocks may differ with more data. However, the study gives an indication of carbon stocks in CFs under REDD+ project activities and also provides a basis for further studies.
- ii. This study collected real-time data on sacrificed benefits and additional contributions of communities for REDD+. The analysis of the trade-offs between communities' sacrificed benefits and REDD+ carbon benefits does not include indirect costs or benefits to local communities.
- iii. Climate change itself can have some effects on biomass growth and biodiversity; however, this has not been considered in this short term study, given lack of data on impacts.
- iv. Ages of the dominant tree species in plots are estimated based on observation by experienced local people and forestry technicians. This method is applied in other similar studies but it can give slightly higher or lower results than the exact age.
- v. Getting primary data was not possible therefore allometric equations used for the estimation of above ground biomass and so root: shoot ratio and wood density were based on information from the literature.
- vi. This study used a yield table developed for undisturbed natural *Shorea robusta* forests to predict technical potential carbon stock. However, undisturbed forest models were not available to predict the technical potential carbon stock for mixed broadleaf, *Schima-Castanopsis*, *Pine* and *Rhododendron-Quercus* dominated CFs. Therefore this study estimated only the gap between higher and lower carbon stocks in plots based on the percentile value of carbon stock in all vegetation types except *Shorea robusta* dominated forests.

9.7. Suggestions for further research

Long term research about carbon stock changes including soil carbon pool: Even though this study attempted to evaluate the REDD+ pilot project activities by analysing four years of inventory data and found CFUGs are contributing in generating additional carbon stock in forests, there is still scope to investigate the results in the longer term to confirm the change trends. Communities have changed their existing forest management activities and forest resource use practices and increased carbon stock in CFs for REDD+ carbon benefits. However, there are uncertainties as to whether these changed practices will persist in the long term or just for few years and whether added biomass carbon will be retained for the long term or release back to the atmosphere after a few years. Using the results of this study as a baseline, further studies generating time series data are recommended to identify trends for biomass carbon growth in CFs. That long term REDD+ projects with improving forest productivity and changes in forest use practices of communities.

Detailed economic analysis to assess REDD+ CFs to design economic models with various scenarios: This study included real-time added costs to communities for forest carbon enhancement. The direct cost due to changed forest management and the forest product use practices of communities are included in the accounting however, there could be other indirect costs involved. On the other hand, carbon benefits were estimated by considering the possibility of getting a premium price for a least developed country with respect to non-carbon benefits. Although analysis shows costs are higher than benefits, there is potential to emphasise some activities over others in order to reduce the cost to communities. A further research, particularly detailed economic analysis of various scenarios, are required to assess REDD+ options with various costs scenarios for CFs.

Long term study about trade-off between community cost and carbon benefits: The community costs estimated in this study were unexpectedly higher, possibly due to the early stage of the project. As communities' behaviour can change over time, it is likely that cost at later stages in the REDD implementation may also differ. Therefore, a long term study is needed to understand the trade-off between community cost and carbon benefits.

Share of potential costs and benefits of the REDD+ activities at different community <u>members within a CFUG</u>: This study does not include disaggregated analysis of potential cost and benefits proportions of REDD+ mechanism for individual groups (such as poor, women, marginalised groups) within a CFUG. This might be the same as indicated in other studies about CFs in Nepal which show that an inequitable proportion of the costs are borne by poor and marginalised members of the community (Neupane 2003; Richards *et al.* 2003; Adhikari *et al.* 2004; Dev *et al.* 2004; McDermott & Schreckenberg 2009; Sunam 2011). It is important to analyse both the costs associated with added or changed practices for the REDD+ and benefits received from payments. Therefore a detained study is needed to clarify whether poor and marginalised members are equally benefitted by the mechanism or not.

Technical potential carbon stock in undisturbed forests by vegetation type using time series data: Technical potential biomass carbon in undisturbed natural *S. robusta* forests was taken from the specific yield table for Nepal. But, similar information was not available for mixed broadleaf forests, *Schima-Castanopsis* forests, *Pine* forests or *Rhododendron-Quercus* forests. There is scope to contribute in these areas to enable the assessment of technical potential carbon stock gain in such forests with changed behaviour of communities in CFs. On the other hand, biomass carbon growth in CFs was estimated by using chronosequence data which could vary due to site productivity differences. Studies using time series data would give more accurate growth rates for undisturbed forests and CFs and could be an area for future research.

<u>Remote sensing and GIS based research in future to assess changes</u>: Similarly, carbon stock increases differ across the ranges of biophysical and socio-economic factors in CFs. Though the results of this study indicated key factors responsible for carbon stock, ecological study needs long term data and it is suggested that further studies to quantify the effects of particular factors on carbon stock changes be conducted. However, future studies can use this study's results as baseline data. Further research using spatial and temporal data including GIS and remote sensing data of all climatic, topographic and edaphic factors will be helpful in future.

<u>Impact of climate change on forest carbon stock changes</u>: Climate change is another important factor responsible for changing vegetation dynamics in forests. Although not included in this study, changing climate may affect the rates of carbon stock changes in forests. This study estimated changes over three years from the commencement of REDD+ activities in 2010 and assumed that any possible impacts of climate change on vegetation growth over this period were negligible. However, it can have significant impact in the long term and this factor needs to be added for long term carbon stock growth projections with REDD+ activities in forests. Further research is needed in this area.

<u>Effects of REDD+ projects in biodiversity</u>: CFUGs might be focusing on carbon enhancement for REDD+ incentives. Therefore, there is a possibility of reducing species richness in REDD+ CFs if communities do not get incentives to maintain species richness together with carbon benefits. Biodiversity and species turnover related studies need long term data. Therefore, further studies regarding the effect of REDD+ projects on biodiversity with reference to long time spans is suggested.

REFERENCES

Acharya, K 2003, 'Changing the strategy for community forestry in Nepal: The case for active management', *Journal of forest policy*, vol. 10, no. 1, pp. 43-50.

Acharya, K & Gentle, P 2006, *Improving the effectiveness of collective action: Sharing experiences from community forestry in Nepal*, Collective Action and Property Rights (CAPRi), Working Paper No. 54, International Food Policy Research Institute, Washington, D.C., USA

Acharya, KP 2002, 'Twenty-four years of community forestry in Nepal', *International Forestry Review*, vol. 4, no. 2, pp. 149-156.

Acharya, KP 2004, 'Does community forests management supports biodiversity conservation? Evidences from two community forests from the mid-hills of Nepal', *Journal of Forest and Livelihood*, vol. 4, no. 1, pp. 44-54.

Acharya, KP & Acharya, B 2004, *Early growth performance of natural Sal (Shorea robusta) forest in Central Nepal*, Department of Forest Research and Survey, Ministry of Forests and Soil Conservation, Kathmandu, Nepal.

Acharya, R 2005, 'Socio-economic impacts of community based forest enterprises in mid hills of Nepal-Case Study from Dolakha district', *Banko Janakari*, vol. 15, no. 2, pp. 43-47.

Adhikari, B, Di Falco, S & Lovett, JC 2004, 'Household characteristics and forest dependency: evidence from common property forest management in Nepal', *Ecological Economics*, vol. 48, no. 2, pp. 245-257.

Adhikari, B, Williams, F & Lovett, JC 2007, 'Local benefits from community forests in the middle hills of Nepal', *Forest Policy and Economics*, vol. 9, no. 5, pp. 464-478.

Adhikari, M & Nagata, S 2004, 'Rural household and forest: an evaluation of household's dependency on community forest in Nepal', *Journal of Forest Research*, vol. 9, no. 1, pp. 33-44.

Agarwal, B 2001, 'Participatory exclusions, community forestry, and gender: An analysis for South Asia and a conceptual framework', *World development*, vol. 29, no. 10, pp. 1623-1648.

Agrawal, A, Chhatre, A & Hardin, R 2008, 'Changing governance of the world's forests', *Science*, vol. 320, no.5882, pp. 1460–1462.

Agarwal, B 2009, 'Rule making in community forestry institutions: The difference women make', *Ecological Economics*, vol. 68, no. 8-9, pp. 2296-2308,

Agrawal, A & Ostrom, E 2001, 'Collective action, property rights, and decentralization in resource use in India and Nepal', *Politics & Society*, vol. 29, no. 4, pp. 485-514.

Agrawal, A & Goyal, S 2001, 'Group size and collective action third-party monitoring in common-pool resources', *Comparative Political Studies*, vol. 34, no. 1, pp. 63-93.

Agrawal, A & Angelsen, A 2009, 'Using community forest management to achieve REDD+ goals', in A Angelsen, M Brockhaus, M Kanninen, E Sills, WD Sunderlin & S Wertz-Kanounnikoff (eds), *Realising REDD*, CIFOR, Bogor, p. 201.

Agrawal, A, Nepstad, D & Chhatre, A 2011, 'Reducing emissions from deforestation and forest degradation', *Annual Review of Environment and Resources*, vol. 36, pp. 373-396.

Alamgir, M & Al-Amin, M 2008, 'Allometric models to estimate biomass organic carbon stock in forest vegetation', *Journal of Forestry Research*, vol. 19, no. 2, pp. 101-106.

Alcorn, JB 2014, Lessons learned from community forestry in Latin America and their relevance for REDD+, USAID-supported Forest Carbon, Markets and Communities (FCMC) Program. Washington, DC, USA.

Alexander, S, Nelson, CR, Aronson, J, Lamb, D, Cliquet, A, Erwin, KL, Finlayson, CM, de Groot, RS, Harris, JA, Higgs, ES, Hobbs, RJ, Robin Lewis, RR, Martinez, D & Murcia, C 2011, 'Opportunities and challenges for ecological restoration within REDD+', *Restoration Ecology*, vol. 19, no. 6, pp. 683-689.

Almeida, AC, Landsberg, JJ & Sands, PJ 2004, 'Parameterisation of 3-PG model for fast-growing Eucalyptus grandis plantations', *Forest Ecology and Management*, vol. 193, no. 1, pp. 179-195.

Alves, LF, Vieira, SA, Scaranello, MA, Camargo, PB, Santos, FAM, Joly, CA & Martinelli, LA 2010, 'Forest structure and live aboveground biomass variation along an elevational gradient of tropical Atlantic moist forest (Brazil)', *Forest Ecology and Management*, vol. 260, no. 5, pp. 679-691.

Angelsen, A 2008, 'How do we set the reference levels for REDD payments', in A Angelsen (ed), Moving ahead with REDD: issues, options and implications: 53-64.

Angelsen, A 2009, *Realising REDD+: National strategy and policy options*, Center for International Foretry Reserach, Bogor.

Angelsen, A & Kaimowitz, D 1999, 'Rethinking the causes of deforestation: lessons from economic models', *The world bank research observer*, vol. 14, no. 1, pp. 73-98.

Angelsen, A, Brockhaus, M, Sunderlin, WD & Verchot, LV 2012, *Analysing REDD+: Challenges and choices*, Center for International Foretry Reserach, Bogor.

Angelsen, A, Streck, C, Peskett, L, Brown, J & Luttrell, C 2008, 'What is the right scale for REDD', in A Angelsen (ed), *Moving ahead with REDD: issues, options and implications*, pp. 31-40.

Angelsen, A, Brockhaus, M, Kanninen, M, Sills, E, Sunderlin, W & Wertz-Kanounnikoff, S 2009, 'Realising REDD', *National strategy and policy options. Bogor, Indonesia: CIFOR.*

ANSAB/ICIMOD/FECOFUN 2013, Design and setting up of a governance and payment system for Nepal's Community Forest Management under Reduced Emission from Deforestation and Degradation (REDD), Project report, ANSAB/ICIMOD/FECOFUN, Kathmandu, Nepal.

Aplet, GH & Vitousek, PM 1994, 'An age-altitude matrix analysis of hawaiian rainforest succession', *Journal of Ecology*, vol. 82, no. 1, pp. 137-147.

Applegate, GB, Gilmour, DA & Mohns, B 1988a, 'Biomass productivity estimations for community forest management: A case study from the Middle Hills of Nepal. 1. Biomass and productivity of Chir Pine (*Pinus roxburghii* Sargent) plantations' *Biomass*, vol. 17, no. 2, pp.115-136.

Applegate, GB, Gilmour, DA & Mohns, B 1988b, 'The use of biomass estimation in the management of forest for fuelwood and fodder production', *Commonwealth Forestry Review*, vol. 67, no. 2, pp. 141-148.

Araújo, MB, Pearson, RG, Thuiller, W & Erhard, M 2005, 'Validation of speciesclimate impact models under climate change', *Global Change Biology*, vol. 11, no. 9, pp. 1504-1513.

Armenteras, D, Rudas, G, Rodriguez, N, Sua, S & Romero, M 2006, 'Patterns and causes of deforestation in the Colombian Amazon', *Ecological Indicators*, vol. 6, no. 2, pp. 353-368.

Arnold, J 1991, 'Community Forestry', *Ten years in review. Community Forestry notes*, no. 7.

Arthur, M, Paratley, R & Blankenship, B 1998, 'Single and repeated fires affect survival and regeneration of woody and herbaceous species in an oak-pine forest', *Journal of the Torrey Botanical Society*, pp. 225-236.

Aryal, K, Berg, Å & Ogle, B 2009, 'Uncultivated plants and livelihood support–A case study from the Chepang people of Nepal', *Ethnobotany Research & Applications*, vol. 7, pp. 409-422.

Apsey, M & Reed, L 1994, 'World timber resources outlook – current perceptions', A discussion paper, 2nd printing, Council of Forest Industries, Vancouver, BC, Canada.

Avitabile, V, Baccini, A, Friedl, MA & Schmullius, C 2012, 'Capabilities and limitations of Landsat and land cover data for aboveground woody biomass estimation of Uganda', *Remote Sensing of Environment*, vol. 117, pp. 366-380.

Ayres, MaL, MJ 2000, 'Assessing the consequences of global change for forest disturbance from herbivores and pathogens', *The science of the total environment*, vol. 262, pp. 263-286.

Bahuguna, VK 2000, 'Forests in the economy of the rural poor: An estimation of the dependency level', *AMBIO: A Journal of the Human Environment*, vol. 29, no. 3, pp. 126-129.

Baker, TR, Phillips, OL, Malhi, Y, Almeida, S, Arroyo, L, Di Fiore, A, Erwin, T, Killeen, TJ, Laurance, SG & Laurance, WF 2004, 'Variation in wood density determines spatial patterns inAmazonian forest biomass', *Global Change Biology*, vol. 10, no. 5, pp. 545-562.

Balch, JK, Nepstad, DC, Curran, LM, Brando, PM, Portela, O, Guilherme, P, Reuning-Scherer, JD & de Carvalho Jr, O 2011, 'Size, species, and fire behavior predict tree and liana mortality from experimental burns in the Brazilian Amazon', *Forest Ecology and Management*, vol. 261, no. 1, pp. 68-77.

Banerjee, A, Duflo, E & Qian, N 2012, *On the road: Access to transportation infrastructure and economic growth in China*, Working Paper No. 17897. Cambridge, MA: National Bureau of Economic Research.

Baral, JC & Subedi, BR 2000, 'Some community forestry issues in the Terai, Nepal: Where do we go from here?', *Forests, Trees and People Newsletter*, no. 42, pp. 20-25.

Barbier, EB & Rauscher, M 1994, 'Trade, tropical deforestation and policy interventions', *Environmental and Resource Economics*, vol. 4, no. 1, pp. 75-90.

Barbier, EB, Bockstael, N, Burgess, JC & Strand, I 1995, 'The linkages between the timber trade and tropical deforestation—Indonesia', *The World Economy*, vol. 18, no. 3, pp. 411-442.

Barlow, J, Peres, CA, Lagan, BO & Haugaasen, T 2003, 'Large tree mortality and the decline of forest biomass following Amazonian wildfires', *Ecology Letters*, vol. 6, no. 1, pp. 6-8.

Barraclough, SL & Ghimire, KB 1995, Forests and livelihoods: the social dynamics of deforestation in developing countries, Macmillan Press Ltd, London.

Basuki, TM, van Laake, PE, Skidmore, AK & Hussin, YA 2009, 'Allometric equations for estimating the above-ground biomass in tropical lowland Dipterocarp forests', *Forest Ecology and Management*, vol. 257, no. 8, pp. 1684-1694.

Beniston, M & Stephenson, DB 2004, 'Extreme climatic events and their evolution under changing climatic conditions', *Global and Planetary Change*, vol. 44, no. 1-4, pp. 1-9.

Bernard, HR 2000, *Social research methods: Qualitative and quantitative approaches*, Sage Publications, Inc.

Beyer, HL 2004, *Hawth's analysis tools for ArcGIS. Available from:* <*http://www.spatialecology.com/htools>.*

Bhatt, BP & Tomar, JMS 2002, 'Firewood properties of some Indian mountain tree and shrub species', *Biomass and Bioenergy*, vol. 23, no. 4, pp. 257-260.

Bhattacharya, S & Abdul Salam, P 2002, 'Low greenhouse gas biomass options for cooking in the developing countries', *Biomass and Bioenergy*, vol. 22, no. 4, pp. 305-317.

Bhattacharya, S, Abdul Salam, P & Sharma, M 2000, 'Emissions from biomass energy use in some selected Asian countries', *Energy*, vol. 25, no. 2, pp. 169-188.

Bhattarai, BP & Kindlmann, P 2012, 'Impact of livestock grazing on the vegetation and wild ungulates in the Barandabhar corridor forest, Nepal', in *Himalayan Biodiversity in the Changing World*, Springer, pp. 157-175.

Bhattarai, M & Hammig, M 2001, 'Institutions and the environmental Kuznets curve for deforestation: a crosscountry analysis for Latin America, Africa and Asia', *World development*, vol. 29, no. 6, pp. 995-1010.

Bhattarai, RC 2011, 'Economic impact of community forestry in Nepal: A case of mid-hill districts of Nepal', *Economic Journal of Development Issues*, vol. 13, pp. 75-96.

Biggs, S & Messerschmidt, D 2005, 'Social responsibility in the growing handmade paper industry of Nepal', *World development*, vol. 33, no. 11, pp. 1821-1843.

Birch, CP 1999, 'A new generalized logistic sigmoid growth equation compared with the Richards growth equation', *Annals of Botany*, vol. 83, no. 6, pp. 713-723.

Blackmore, M & Vitousek, PM 2000, 'Cattle grazing, forest loss, and fuel loading in a dry forest ecosystem at Pu'u Wa'aWa'a Ranch, Hawai'i1', *Biotropica*, vol. 32, no. 4a, pp. 625-632.

Blair, HW 1996, 'Democracy, equity and common property resource management in the Indian subcontinent', *Development and Change*, vol. 27, no. 3, pp. 475-499.

Blom, B, Sunderland, T & Murdiyarso, D 2010, 'Getting REDD to work locally: lessons learned from integrated conservation and development projects', *Environmental Science & Policy*, vol. 13, no. 2, pp. 164-172.

Bluffstone, R, Robinson, E & Guthiga, P 2013, 'REDD+and community-controlled forests in low-income countries: Any hope for a linkage?', *Ecological Economics*, vol. 87, pp. 43-52.

Bongers, F, Poorter, L, Van Rompaey, RSAR & Parren, MPE 1999, 'Distribution of twelve moist forest canopy tree species in Liberia and Côte d'Ivoire: response curves to a climatic gradient', *Journal of Vegetation Science*, vol. 10, no. 3, pp. 371-382.

Boojh, R & Ramakrishnan, P 1983, 'The growth pattern of two species of Schima', *Biotropica*, vol. 15, pp. 142-147.

Bornmann, L, Leydesdorff, L & Mutz, R 2013, 'The use of percentiles and percentile rank classes in the analysis of bibliometric data: opportunities and limits', *Journal of Informetrics*, vol. 7, no. 1, pp. 158-165.

Boyd, E & Schipper, EL 2002, 'The Marrakech Accord—At the crossroad to ratification: Seventh conference of the Parties to the United Nations Framework Convention on Climate Change', *The Journal of Environment & Development*, vol. 11, no. 2, pp. 184-190.

Bradford, JBBJB & Kastendick, DNKDN 2010, 'Age-related patterns of forest complexity and carbon storage in pine and aspen-birch ecosystems of northern Minnesota, USA', *Canadian Journal of Forest Research*, vol. 40, no. 3, pp. 401-9.

Bray, DB, Merino-Pérez, L & Barry, D 2005, The community forests of Mexico, JSTOR.

Breisinger, C & Diao X 2008, *Economic transformation in theory and practice: What are the messages for Africa?* ReSAKSS Working Paper 21. International Food policy Research Institute (IFPRI).

Brienen, RJ & Zuidema, PA 2005, 'Relating tree growth to rainfall in Bolivian rain forests: a test for six species using tree ring analysis', *Oecologia*, vol. 146, no. 1, pp. 1-12.

Brown, D 1999 'Principles & practice of forest Co- management: Evidence from West-Central Africa ', *EU Tropical Forestry Paper No.2, London: ODI*.

Brown, D, Seymour, F & Peskett, L 2008, 'How do we achieve REDD co-benefits and avoid doing harm', *Moving ahead with REDD: issues, options and implications*, pp. 107-118.

Brown, D, Malla, Y, Schreckenberg, K & Springate-Baginski, O 2002, 'From supervising 'subjects' to supporting 'citizens': recent developments in community forestry in Asia and Africa', *ODI natural resource perspectives*, vol. 75, Overseas Development Institute, London.

Brown, S 2002, 'Measuring carbon in forests: current status and future challenges', *Environmental pollution*, vol. 116, no. 3, pp. 363-372.

Brown, S & Lugo, AE 1992, 'Aboveground biomass estimates for tropical moist forests of the Brazilian Amazon', *Interciencia. Caracas*, vol. 17, no. 1, pp. 8-18.

Busch, J, Lubowski, RN, Godoy, F, Steininger, M, Yusuf, AA, Austin, K, Hewson, J, Juhn, D, Farid, M & Boltz, F 2012, 'Structuring economic incentives to reduce emissions from deforestation within Indonesia', *Proceedings of the National Academy of Sciences*, vol. 109, no. 4, pp. 1062-1067.

Bushley, BR & Khatri, D 2011, *REDD+: reversing, reinforcing or reconfiguring decentralized forest governance in Nepal*, Discussion Paper Series 11: 3, Forest Action Nepal

Bushnell, JB, Chong, H & Mansur, ET 2013, 'Profiting from regulation: Evidence from the european carbon market', *American Economic Journal: Economic Policy*, vol. 5, no. 4, pp. 78-106.

Cadman, T 2009, *Quality, legitimacy and global governance: A comparative analysis of four forest institutions.* PhD dissertation. School of Government, University of Tasmania, Tasmania, Australia.

Cadman, T & Maraseni, T 2011, 'The Governance of Climate Change: Evaluating the Governance Quality and Legitimacy of the United Nations' REDD-plus Programme', *International Journal of Climate Change: Impacts & Responses*, vol. 2, no. 3, pp. 103–123.

Cadman, T & Maraseni, T 2012, 'The governance of REDD+: an institutional analysis in the Asia Pacific region and beyond', *Journal of Environmental Planning and Management*, vol. 55, no. 5, pp. 617-635.

Cairns, MA, Brown, S, Helmer, EH & Baumgardner, GA 1997, 'Root biomass allocation in the world's upland forests', *Oecologia*, vol. 111, no. 1, pp. 1-11.

Campbell, J, Alberti, G, Martin, J & Law, B 2009, 'Carbon dynamics of a ponderosa pine plantation following a thinning treatment in the northern Sierra Nevada', *Forest Ecology and Management*, vol. 257, no. 2, pp. 453-463.

Canadell, JG & Raupach, MR 2008, 'Managing forests for climate change mitigation', *Science*, vol. 320, no. 5882, pp. 1456-1457.

Caplow, S, Jagger, P, Lawlor, K & Sills, E 2011, 'Evaluating land use and livelihood impacts of early forest carbon projects: Lessons for learning about REDD+', *Environmental Science & Policy*, vol. 14, no. 2, pp. 152-167.

Capoor, K & Ambrosi, P 2009, *State and trends of the carbon market 2009*, The World Bank, Washington, D.C.

CBD 1992, Agreed text for adoption in Convention on Biological Diversity 1992, www. biodiv.org/convention/articles.asp.

CBD 2001, Forest biodiversity introduction, CBD Secretariat, 2001.
CBS 2011a, *Nepal living standards survey 2010/11: Statistical report volume two*, Central Bureau of Statistics, National Planning Commission Secretariate, Government of Nepal.

CBS 2011b, *National population and housing census 2011 (National report)*, Central Bureau of Statistics, Government of Nepal.

Cerbu, GA, Swallow, BM & Thompson, DY 2011, 'Locating REDD: A global survey and analysis of REDD readiness and demonstration activities', *Environmental Science & Policy*, vol. 14, no. 2, pp. 168-180.

CFD/DoF 2002, *Forest inventory guidelines*, Ministry of forest and soil conservation, Government of Nepal.

CFD/DoF 2009, *Guidelines of community forestry*, Ministry of forest and soil conservation, Government of Nepal.

CFD/DoF 2013, *Community forestry database*, Ministry of forest and soil conservation, Government of Nepal.

Chadwick, BP 2006, 'Transaction costs and the clean development mechanism', *Natural Resources Forum*, vol.30, pp. 256-271.

Chapagain, TR, Sharma, RP & Bhandari, SK 2013, 'Modeling above-ground biomass for three tropical tree species at their juvenile stage', *Forest Science and Technology*, vol.10, no.2, pp. 51-60.

Chave, J, Olivier, J, Bongers, F, Châtelet, P, Forget, P-M, van der Meer, P, Norden, N, Riéra, B & Charles-Dominique, P 2008, 'Above-ground biomass and productivity in a rain forest of eastern South America', *Journal of Tropical Ecology*, vol. 24, no. 4, pp. 355-366.

Chave, J, Andalo, C, Brown, S, Cairns, M, Chambers, J, Eamus, D, Fölster, H, Fromard, F, Higuchi, N & Kira, T 2005, 'Tree allometry and improved estimation of carbon stocks and balance in tropical forests', *Oecologia*, vol. 145, no. 1, pp. 87-99.

Chen, J & Bradshaw, GA 1999, 'Forest structure in space: a case study of an old growth spruce-fir forest in Changbaishan Natural Reserve, PR China', *Forest Ecology and Management*, vol. 120, no. 1, pp. 219-233.

Chettri, N, Sharma, E, Deb, D & Sundriyal, R 2002, 'Impact of firewood extraction on tree structure, regeneration and woody biomass productivity in a trekking corridor of the Sikkim Himalaya', *Mountain Research and Development*, vol. 22, no. 2, pp. 150-158.

Chhabra, A, Palria, S & Dadhwal, VK 2002, 'Growing stock-based forest biomass estimate for India', *Biomass and Bioenergy*, vol. 22, no. 3, pp. 187-194.

Chhatre, A & Agrawal, A 2009, 'Trade-offs and synergies between carbon storage and livelihood benefits from forest commons', *Proceedings of the National Academy of Sciences*, vol. 106, no. 42, pp. 17667-17670.

Chhetri, BBK 2005, *Community forestry program in the hills of Nepal: determinants of user participation and household dependency*, MSc thesis, Norwegian University of Life Sciences, Norway.

Chiang, JM, McEwan, RW, Yaussy, DA & Brown, KJ 2008, 'The effects of prescribed fire and silvicultural thinning on the aboveground carbon stocks and net primary production of overstory trees in an oak-hickory ecosystem in southern Ohio', *Forest Ecology and Management*, vol. 255, no. 5-6, pp. 1584-1594.

Chomitz, KM 2000, Evaluating carbon offsets from forestry and energy projects: *How do they compare?*, vol. 2357, World Bank.

Chou, C-f & Talmain, G 1996, 'Redistribution and growth: Pareto improvements', *Journal of Economic Growth*, vol. 1, no. 4, pp. 505-523.

Clark, J 2001, 'The global wood market, prices and plantation investment: an examination drawing on the Australian experience.' *Environmental Conservation*, vol. 28, no.1, pp. 53-64.

Clark, DA & Clark, DB 1999, 'Assessing the growth of tropical rain forest trees: issues for forest modeling and management', *Ecological Applications*, vol. 9, no. 3, pp. 981-997.

Clements, T 2010, 'Reduced expectations: the political and institutional challenges of REDD+', *Oryx*, vol. 44, no. 3, pp. 309-310.

Cochrane, MA & Schulze, MD 1999, 'Fire as a recurrent event in tropical forests of the eastern Amazon: Effects on forest structure, biomass, and species composition', *Biotropica*, vol. 31, no. 1, pp. 2-16.

Comin, D & Hobijn, B 2004, 'Cross-country technology adoption: making the theories face the facts', *Journal of Monetary Economics*, vol. 51, no. 1, pp. 39-83.

Conley, A & Moote, MA 2003, 'Evaluating collaborative natural resourcemanagement', *Society &Natural Resources*, vol. 16, no. 5, pp. 371-386.

Connell, JH, Tracey, J & Webb, LJ 1984, 'Compensatory recruitment, growth, and mortality as factors maintaining rain forest tree diversity', *Ecological Monographs*, vol. 54, no. 2, pp. 141-164.

Cook, J, Nuccitelli, D, Green, SA, Richardson, M, Winkler, B, Painting, R, Way, R, Jacobs, P & Skuce, A 2013, 'Quantifying the consensus on anthropogenic global warming in the scientific literature', *Environmental Research Letters*, vol. 8, no. 2, p. 24024.

Cooke, P, Kohlin, G & Hyde, WF 2008, 'Fuelwood, forests and community management-evidence from household studies', *Environment and Development Economics*, vol. 13, no. 1, pp. 103-135.

Cooke, PA 1998, 'Intrahousehold labor allocation responses to environmental good scarcity: a case study from the hills of Nepal', *Economic Development and Cultural Change*, vol. 46, no. 4, pp. 807-830.

Corbera, E & Schroeder, H 2011, 'Governing and implementing REDD+', *Environmental Science & Policy*, vol. 14, no. 2, pp. 89-99.

Corbera, E, Estrada, M & Brown, K 2010, 'Reducing greenhouse gas emissions from deforestation and forest degradation in developing countries: revisiting the assumptions', *Climatic change*, vol. 100, no. 3, pp. 355-388.

Corbera, E, Estrada, M, May, P, Guillermo, N & Pablo, P 2011, 'Rights to land, forests and carbon in REDD+: insights from Mexico, Brazil and Costa Rica', *Forests*, vol. 2, no. 1, pp.301-342.

Costa, PM & Wilson, C 2000, 'An equivalence factor between CO₂ avoided emissions and sequestration–description and applications in forestry', *Mitigation and Adaptation Strategies for Global Change*, vol. 5, no. 1, pp. 51-60.

Curtis, RO 1995, 'Extended rotations and culmination age of coast Douglas-fir: old studies speak to current issues', *Res. Pap. PNW-RP-485. Portland, OR: U.S. Department of Agriculture, Forest Service, Pacific Northwest Research Station. 49 p.*

D'Amato, AW, Bradford, JB, Fraver, S & Palik, BJ 2011, 'Forest management for mitigation and adaptation to climate change: Insights from long-term silviculture experiments', *Forest Ecology and Management*, vol. 262, no. 5, pp. 803-816.

D'Antonio, CM 2000, 'Fire, plant invasions, and global changes', inH.A. Mooney & R.J. Hobbs (Eds.), *Invasive Species in a Changing World*, Island Press, pp. 65–93

Daily, G 1997, *Nature's services: societal dependence on natural ecosystems*, Island Press, Washington, DC.

Daniel, TW, Helms, JA & Baker, FS 1979, *Principles of silviculture*, McGraw-Hill Book Company, New York.

Danielsen, F, Skutsch, M, Burgess, ND, Jensen, PM, Andrianandrasana, H, Karky, B, Lewis, R, Lovett, JC, Massao, J & Ngaga, Y 2011, 'At the heart of REDD+: a role for local people in monitoring forests?', *Conservation Letters*, vol. 4, no. 2, pp. 158-167.

Das, R & Shivakoti, GP 2006, 'Livestock carrying capacity evaluation in an integrated farming system: A case study from the mid-hills of Nepal', *The International Journal of Sustainable Development and World Ecology*, vol. 13, no. 3, pp. 153-163.

Davidar, P, Arjunan, M & Puyravaud, J-P 2008, 'Why do local households harvest forest products? A case study from the southern Western Ghats, India', *Biological Conservation*, vol. 141, no. 7, pp. 1876-1884.

Davis, LS, Johnson, KN, Bettinger, PS & Howard, TE 2001, *Forest management: To sustain ecological economic and social value*, McGraw Hill, Boston.

de Jong, B, Anaya, C, Masera, O, Olguín, M, Paz, F, Etchevers, J, Martínez, RD, Guerrero, G & Balbontín, C 2010, 'Greenhouse gas emissions between 1993 and 2002 from land-use change and forestry in Mexico', *Forest Ecology and Management*, vol. 260, no. 10, pp. 1689-1701.

Deci, EL & Ryan, RM 2000, 'The" what" and" why" of goal pursuits: Human needs and the self-determination of behavior', *Psychological inquiry*, vol. 11, no. 4, pp. 227-268.

Deci, EL, Koestner, R & Ryan, RM 1999, 'A meta-analytic review of experiments examining the effects of extrinsic rewards on intrinsic motivation', *Psychological bulletin*, vol. 125, no. 6, p. 627.

Dev, OP, Yadav, NP, Springate-Baginski, O & Soussan, J 2003, 'Impacts of community forestry on livelihoods in the middle hills of Nepal', *Journal of Forest and Livelihood*, vol. 3, no.1, pp. 64-77

Devendra, C & Sevilla, C 2002, 'Availability and use of feed resources in cropanimal systems in Asia', *Agricultural Systems*, vol. 71, no. 1, pp. 59-73.

DFRS 1999, *National forest inventory report*, Departmenet of Forest Research and Survey, Nepal.

Dhakal, A 2013, Evolution, adoption and economic evaluation of an Agroforestrybased farming system with without carbon values: the case of Nepal, Universoty of Southern Queensland.

Dhakal, B, Bigsby, H & Cullen, R 2005, 'Impacts of community forestry development on livestock-based livelihood in Nepal', *Journal of Forest and Livelihood*, vol. 4, no. 2, pp. 43-49.

Dhakal, S & Raut, AK 2010, 'Potential and bottlenecks of the carbon market: The case of a developing country, Nepal', *Energy Policy*, vol. 38, no. 7, pp. 3781-3789.

DHM 2013, *Climatic data*, Department of Hydrology Meteorology, Government of Nepal., Kathmandu.

Dieter, M & Elsasser, P 2002, 'Carbon Stocks and Carbon Stock Changes in the Tree Biomass of Germany's Forests', *Forstwissenschaftliches Centralblatt*, vol. 121, no. 4, pp. 195-210.

Dixon, RK 1997, 'Silvicultural options to conserve and sequester carbon in forest systems: preliminary economic assessment', *Critical reviews in environmental science and technology*, vol. 27, no. S1, pp. 139-149.

DoF/MFSC 2005, *Hamro Ban*, Department of Forests, Ministry of Forests and Soil Conservation., Babarmahal, Kathmandu.

Dong, J, Kaufmann, RK, Myneni, RB, Tucker, CJ, Kauppi, PE, Liski, J, Buermann, W, Alexeyev, V & Hughes, MK 2003, 'Remote sensing estimates of boreal and temperate forest woody biomass: carbon pools, sources, and sinks', *Remote Sensing of Environment*, vol. 84, no. 3, pp. 393-410.

Dongol, C, Hughey, KF & Bigsby, HR 2002, 'Capital formation and sustainable community forestry in Nepal', *Mountain Research and Development*, vol. 22, no. 1, pp. 70-77.

Duchelle, AE,Cromberg, M, Gebara, MF, Guerra, R, Melo, T, Larson, A, Cronkleton, P, Börner, J, Sills, E, Wunder, S, Bauch, S, May, P, Selaya, G & Sunderlin, WD 2014, 'Linking Forest Tenure Reform, Environmental Compliance, and Incentives: Lessons from REDD+ Initiatives in the Brazilian Amazon', *World development*, vol 55, pp 53-67.

Dudley, N, Baldock, D, Nasi, R & Stolton, S 2005, 'Measuring biodiversity and sustainable management in forests and agricultural landscapes', *Philosophical Transactions of the Royal Society B: Biological Sciences*, vol. 360, no. 1454, pp. 457-470.

Dunlop, J 2009, *REDD, Tenure and local communities*, International Development Law Organization, Rome, Italy.

Eckert, S, Ratsimba, HR, Rakotondrasoa, LO, Rajoelison, LG & Ehrensperger, A 2011, 'Deforestation and forest degradation monitoring and assessment of biomass and carbon stock of lowland rainforest in the Analanjirofo region, Madagascar', *Forest Ecology and Management*, vol. 262, no. 11, pp. 1996-2007.

Edwards, D 1996, 'The trade in non-timber forest products from Nepal', *Mountain Research and Development*, pp. 383-394.

Edwards, VM & Steins, NA 1999, 'A framework for analysing contextual factors in common pool resource research', *Journal of environmental Policy and Planning*, vol. 1, no. 3, pp. 205-221.

Ehara, M, Hyakumura, K & Yokota, Y 2013, 'REDD+ initiatives for safeguarding biodiversity and ecosystem services: harmonizing sets of standards for national application', *Journal of Forest Research*, vol.19, no.5, pp. 427-436.

Elias, P, Leonard, S, Cando, L, Fedele, G, Gaveau, D, Locatelli, B, Martius, C, Murdiyarso, D, Sunderlin, W & Verchot, L 2014, *Synergies across a REDD+ landscape*, Bogor, Indonesia.

Elmendorf, WF & Luloff, A 2001, 'Using qualitative data collection methods when planning for community forests', *Journal of Arboriculture*, vol. 27, no. 3, pp. 139-151.

Eva, H, Carboni, S, Achard, F, Stach, N, Durieux, L, Faure, J-F & Mollicone, D 2010, 'Monitoring forest areas from continental to territorial levels using a sample of medium spatial resolution satellite imagery', *ISPRS Journal of Photogrammetry and Remote Sensing*, vol. 65, no. 2, pp. 191-197.

Everham III, EM, Myster, RW & VanDeGenachte, E 1996, 'Effects of light, moisture, temperature, and litter on the regeneration of five tree species in the tropical montane wet forest of Puerto Rico', *American Journal of Botany*, vol. 83, no.8, pp. 1063-1068.

Ewers, RM 2006, 'Interaction effects between economic development and forest cover determine deforestation rates', *Global Environmental Change*, vol. 16, no. 2, pp. 161-169.

Falconer, J 1990, 'Hungry season" food from the forests', *Unasylva*, vol. 41, no. 160, pp. 14-19.

FAO 1999, *Fra 2000: Forest resoruces of Nepal, Country report. Working paper 16.*, Food and Agricultural Organization of the United Nations, Rome.

FAO 2001, The global forest resources assessment 2000 (FRA 2000): Main report. Forestry paper no. 140, Rome: FAO.

FAO 2010b, 'Global forest resources assessment, 2010 Main report. FAO Forestry Paper 163. Rome, Italy'.

FAO 2011, State of the world's forests 2011, FAO, Rome, Italy.

FAO 2013, State of food insecurity in the world, FAO, Rome, Italy.

FAO 1997, *Estimating biomass and biomass change of tropical forests: a primer*, vol. 134, FAO, Rome, Italy.

FCPF/World-Bank 2013, Forest Carbon Partnership Facility 2013: Annual Report, World Bank, USA.

Fearnside, PM 1997, 'Monitoring needs to transform Amazonian forest maintenance into a global warming-mitigation option', *Mitigation and Adaptation Strategies for Global Change*, vol. 2, no. 2-3, pp. 285-302.

Fearnside, PM 2000, 'Global warming and tropical land-use change: greenhouse gas emissions from biomass burning, decomposition and soils in forest conversion, shifting cultivation and secondary vegetation', *Climatic change*, vol. 46, no. 1, pp. 115-58.

Fearnside, PM 2002, 'Why a 100-Year Time Horizon should be used for GlobalWarming Mitigation Calculations', *Mitigation and Adaptation Strategies for Global Change*, vol. 7, no. 1, pp. 19-30.

Fearnside, PM & Imbrozio Barbosa, R 1998, 'Soil carbon changes from conversion of forest to pasture in Brazilian Amazonia', *Forest Ecology and Management*, vol. 108, no. 1, pp. 147-166.

Finér, L, Mannerkoski, H, Piirainen, S & Starr, M 2003, 'Carbon and nitrogen pools in an old-growth, Norway spruce mixed forest in eastern Finland and changes associated with clear-cutting', *Forest Ecology and Management*, vol. 174, no. 1, pp. 51-63.

Flombaum, P & Sala, OE 2007, 'A non-destructive and rapid method to estimate biomass and aboveground net primary production in arid environments', *Journal of Arid Environments*, vol. 69, no. 2, pp. 352-358.

Foley, JA, DeFries, R, Asner, GP, Barford, C, Bonan, G, Carpenter, SR, Chapin, FS, Coe, MT, Daily, GC & Gibbs, HK 2005, 'Global consequences of land use', *Science*, vol. 309, no. 5734, pp. 570-574.

Foley, TG 2009, 'Extending rotation age for carbon sequestration: A cross-protocol comparison of North American forest offsets', *Forest Ecology and Management*, vol. 259, no. 2, pp. 201-209.

Foody, GM, Palubinskas, G, Lucas, RM, Curran, PJ & Honzak, M 1996, 'Identifying terrestrial carbon sinks: classification of successional stages in regenerating tropical forest from Landsat TM data', *Remote Sensing of Environment*, vol. 55, no. 3, pp. 205-216.

Fox, TR 2000, 'Sustained productivity in intensively managed forest plantations', *Forest Ecology and Management*, vol. 138, no. 1, pp. 187-202.

Fry, PB 2011, 'Community forest monitoring in REDD+: the [`]M' in MRV?', *Environmental Science & Policy*, vol. 14, no. 2, pp. 181-187.

Gagné, M & Deci, EL 2005, 'Self determination theory and work motivation', *Journal of Organizational Behavior*, vol. 26, no. 4, pp. 331-362.

Gardiner, BA & Quine, CP 2000, 'Management of forests to reduce the risk of abiotic damage — a review with particular reference to the effects of strong winds', *Forest Ecology and Management*, vol. 135, no. 1–3, pp. 261-277.

Gardner, TA, Burgess, ND, Aguilar-Amuchastegui, N, Barlow, J, Berenguer, E, Clements, T, Danielsen, F, Ferreira, J, Foden, W & Kapos, V 2012, 'A framework for integrating biodiversity concerns into national REDD+ programmes', *Biological Conservation*, vol. 154, pp. 61-71.

Garkoti, S & Singh, S 1995, 'Variation in net primary productivity and biomass of forests in the high mountains of Central Himalaya', *Journal of Vegetation Science*, vol. 6, no. 1, pp. 23-28.

Gautam, A, Shivakoti, G & Webb, E 2004, 'A review of forest policies, institutions, and changes in the resource condition in Nepal', *International Forestry Review*, vol. 6, no. 2, pp. 136-148.

Gautam, AP, Webb, EL, Shivakoti, GP & Zoebisch, MA 2003, 'Land use dynamics and landscape change pattern in a mountain watershed in Nepal', *Agriculture, ecosystems & environment*, vol. 99, no. 1, pp. 83-96.

Gautam, KH & Devoe, NN 2006, 'Ecological and anthropogenic niches of sal (Shorea robusta Gaertn. f.) forest and prospects for multiple-product forest management–a review', *Forestry*, vol. 79, no. 1, pp. 81-101.

Gégout, JC, Coudun, C, Bailly, G & Jabiol, B 2005, 'EcoPlant: a forest site database linking floristic data with soil and climate variables', *Journal of Vegetation Science*, vol. 16, no. 2, pp. 257-260.

Gehring, C, Denich, M & Vlek, PL 2005, 'Resilience of secondary forest regrowth after slash-and-burn agriculture in central Amazonia', *Journal of Tropical Ecology*, vol. 21, no. 5, pp. 519-527.

Geist, HJ & Lambin, EF 2002, 'Proximate causes and underlying driving forces of tropical deforestation', *BioScience*, vol. 52, no. 2, pp. 143-150.

Gentle, P 1997, 'A study of forest fires at Bara District of Nepal Terai. ', *Banko Jankari*, vol. 7, no. 1, pp. 36.

Gewali, MB & Bhandari, R 2005, 'Renewable energy technologies in Nepal', *World Review of Science, Technology and Sustainable Development*, vol. 2, no. 1, pp. 92-106.

Ghazoul, J, Butler, RA, Mateo-Vega, J & Koh, LP 2010, 'REDD: a reckoning of environment and development implications', *Trends in Ecology & Evolution*, vol. 25, no. 7, pp. 396-402.

Gibbs, HK, Brown, S, Niles, JO & Foley, JA 2007, 'Monitoring and estimating tropical forest carbon stocks: making REDD a reality', *Environmental Research Letters*, vol. 2, p. 045023.

Gibson, C, McKean, MA & Ostrom, E 2000, 'Explaining deforestation: the role of local institutions', in C Gibson, M McKean, & E Ostrom (Eds.), *People and forests: Communities, institutions, and governance*, MIT Press, Cambridge, MA

Gilmour, D 1990, 'Resource availability and indigenous forest management systems in Nepal', *Society & Natural Resources*, vol. 3, no. 2, pp. 145-58.

Gilmour, D 2003, "Retrospective and prospective view of community forestry in Nepal', *Journal of Forest and Livelihood*, vol. 2, no. 2, pp. 5-7.

Gilmour, D, Malla, Y & Nurse, M 2004, *Linkages between community forestry and poverty*, Regional Community Forestry Center for Asia and the Pacific Bangkok, Thailand.

Gilmour, DA & Fisher, RJ 1991, Villagers, forests, and foresters: The philosophy, process, and practice of community forestry in Nepal, Sahayogi Press Kathmandu.

Gilmour, DA & Nurse, M 1991, 'Farmer initiatives in increasing tree cover in central Nepal', *Mountain Research and Development*, vol. 11, no.4, pp. 329-337.

Gilmour, DA, King, G & Hobley, M 1989, 'Management of forests for local use in the hills of Nepal-I: Changing forest management paradigms', *Journal of world forest resource management*, vol. 4, no. 2, pp. 93-110.

Goldman, RL, Thompson, BH & Daily, GC 2007, 'Institutional incentives for managing the landscape: Inducing cooperation for the production of ecosystem services', *Ecological Economics*, vol. 64, no. 2, pp. 333-343.

Goldsmith, FB & Harrison, CM (eds) 1976, *Description and analysis of vegetation* Methods in plant ecology, Blackwell Science Publicer, Oxford.

Gomes, R, Bone, S, Cunha, M, Nahur, AC, Moreira, PF, Meneses-Filho, LC, Voivodic, M, Bonfante, T & Moutinho, P 2010, 'Exploring the bottom-up generation of REDD+ policy by forest-dependent peoples', *Nature*, vol. 2, no. 831, p. 836.

GoN 1993, Forest Act, Govenmnet of Nepal

GoN 2000, Forestry sector policy 2000, Govenmnet of Nepal

GoN 2002, *Nepal biodiversity strategy*, Ministry of Forests and Soil Conservation, Govenment of Nepal.

GoN 2009, *Subsidy policy for renewable (rural) energy*, Alternative Energy Promotion Centre 2009, Ministry of Environment, Government of Nepal.

GoN/MFSC/REDD-Cell 2011, *Study on REDD plus piloting in Nepal*. Ministry of Forests and Soil Conservation, Govenment of Nepal.

Goodwin, H 2002, 'Local community involvement in tourism around national parks: opportunities and constraints', *Current Issues in tourism*, vol. 5, no. 3-4, pp. 338-360.

Graichen, P 2005, 'Can forestry gain from emissions trading? Rules governing sinks projects under the UNFCCC and the EU Emissions Trading System', *Review of European Community & International Environmental Law*, vol. 14, no. 1, pp. 11-18.

Grainger, A 1993, *Controlling tropical deforestation*, Earthscan Publications Ltd, London.

Grainger, A & Obersteiner, M 2011, 'A framework for structuring the global forest monitoring landscape in the REDD+ era', *Environmental Science & Comp; Policy*, vol. 14, no. 2, pp. 127-139.

Grainger, A, Boucher, DH, Frumhoff, PC, Laurance, WF, Lovejoy, T, McNeely, J, Niekisch, M, Raven, P, Sodhi, NS & Venter, O 2009, 'Biodiversity and REDD at Copenhagen', *Current Biology*, vol. 19, no. 21, pp. 974-976.

Grau, HR, Gasparri, NI & Aide, TM 2005, 'Agriculture expansion and deforestation in seasonally dry forests of north-west Argentina', *Environmental Conservation*, vol. 32, no. 2, pp. 140-148.

Grieg-Gran, M, Porras, I & Wunder, S 2005, 'How can market mechanisms for forest environmental services help the poor? Preliminary lessons from Latin America', *World development*, vol. 33, no. 9, pp. 1511-1527.

Griffith, AL & Ram, BS 1943, 'Yield and stand tables for sal (Shorea robusta Gaertn. f.) high forest. ', *Indian Forestry Record (N.S.)*, Silviculture conference, vol. 4-A, Manager of Publications, Delhi, p. 287.

Guo, LB & Gifford, RM 2002, 'Soil carbon stocks and land use change: a meta analysis', *Global Change Biology*, vol. 8, no. 4, pp. 345-360.

Gurung, HB 2003, *Trident and Thunderbolt cultural dynamics in nepalese politics*, Social Science Baha, Kathmandu.

Haines, AM, Elledge, D, Wilsing, LK, Grabe, M, Barske, MD, Burke, N & Webb, SL 2012, 'Spatially explicit analysis of poaching activity as a conservation management tool', *Wildlife Society Bulletin*, vol.36, no.4, pp. 685-692.

Hajek, F, Ventresca, MJ, Scriven, J & Castro, A 2011, 'Regime-building for REDD+: Evidence from a cluster of local initiatives in south-eastern Peru', *Environmental Science & Policy*, vol 14, no. 2, pp. 201-215.

Hamilton, K, Sjardin, M, Marcello, T & Xu, G 2008, *Forging a frontier: State of the voluntary carbon markets 2008*, Ecosystem Market Place and New Carbon Finance, San Francisco and London

Hamilton, K, Sjardin, M, Shapiro, A & Marcello, T 2009, *Fortifying the foundation: State of the voluntary carbon markets 2009*, Ecosystem Marketplace and New Carbon Finance, USA

Hansen, J, Sato, M, Ruedy, R, Lo, K, Lea, DW & Medina-Elizade, M 2006, Global temperature change, *Proceedings of the National Academy of Sciences*, vol. 103, no. 39, p. 14288.

Hardin, G 1968, 'The tragedy of the commons', *Science*, vol.162, no. 3859, pp. 1243-1248.

Haripriya, GS 2000, 'Estimates of biomass in Indian forests', *Biomass and Bioenergy*, vol. 19, no. 4, pp. 245-258.

Harmon, ME, Moreno, A & Domingo, JB 2009, 'Effects of partial harvest on the carbon stores in Douglas-fir/western hemlock forests: a simulation study', *Ecosystems*, vol. 12, no. 5, pp. 777-791.

Hayashi, D & Michaelowa, A 2007, *Lessons from submission and approval process of large-scale energy efficiency CDM methodologies*, HWWI Research Paper no. 4-11, Hamburg, Germany.

Hayes, T & Persha, L 2010, 'Nesting local forestry initiatives: Revisiting community forest management in a REDD+ world', *Forest Policy and Economics*, vol. 12, no. 8, pp. 545-553.

Henry, M, Besnard, A, Asante, WA, Eshun, J, Adu-Bredu, S, Valentini, R, Bernoux, M & Saint-André, L 2010, 'Wood density, phytomass variations within and among trees, and allometric equations in a tropical rainforest of Africa', *Forest Ecology and Management*, vol. 260, no. 8, pp. 1375-1388.

Henry, M, Bombelli, A, Trotta, C, Alessandrini, A, Birigazzi, L, Sola, G, Vieilledent, G, Santenoise, P, Longuetaud, F & Valentini, R 2013, 'GlobAllomeTree: international platform for tree allometric equations to support volume, biomass and carbon assessment', *iForest-Biogeosciences and Forestry*, vol. 6, no. 1, p. 326.

Hepburn, C 2007, 'Carbon trading: a review of the Kyoto mechanisms', Annu. Rev. Environ. Resour., vol. 32, pp. 375-393.

Herold, M & Skutsch, M 2011, 'Monitoring, reporting and verification for national REDD+ programmes: two proposals', *Environmental Research Letters*, vol. 6, no. 1, p. 014002.

Hertel, D, Moser, G, Culmsee, H, Erasmi, S, Horna, V, Schuldt, B & Leuschner, C 2009, 'Below-and above-ground biomass and net primary production in a paleotropical natural forest (Sulawesi, Indonesia) as compared to neotropical forests', *Forest Ecology and Management*, vol. 258, no. 9, pp. 1904-1912.

Hetsch, S & Chang, J 2010, 'Key concepts for carbon accounting of REDD+ projects', *in Pathways for Implementing REDD, Experiences from carbon markets and communities. Perspectives Series* (pp. 29–39).Roskilde: UNEP Riso Centre.

Hett, C, Castella, JC, Heinimann, A, Messerli, P & Pfund, JL 2012, 'A landscape mosaics approach for characterizing swidden systems from a REDD+ perspective', *Applied Geography*, vol. 32, no. 2, pp. 608-618.

Hitimana, J, Legilisho Kiyiapi, J & Thairu Njunge, J 2004, 'Forest structure characteristics in disturbed and undisturbed sites of Mt. Elgon Moist Lower Montane Forest, western Kenya', *Forest Ecology and Management*, vol. 194, no. 1, pp. 269-291.

Hoang, MH, Do, TH, Pham, MT, van Noordwijk, M & Minang, PA 2013, 'Benefit distribution across scales to reduce emissions from deforestation and forest degradation (REDD+) in Vietnam', *Land Use Policy*, vol. 31, pp. 48-60.

Hobley, M 1996 'Participatory forestry: The process of change in India & Nepal ', *Rural Development Forestry Network Study Guide 3, London: ODI.*

Hochman, HM & Rodgers, JD 1969, 'Pareto optimal redistribution', *The American Economic Review*, vol 59, no.4, pp. 542-557.

Hoen, HF & Solberg, B 1994, 'Potential and economic efficiency of carbon sequestration in forest biomass through silvicultural management', *Forest Science*, vol. 40, no. 3, pp. 429-451.

Hoover, C & Stout, S 2007, 'The carbon consequences of thinning techniques: stand structure makes a difference', *Journal of Forestry*, vol. 105, no. 5, pp. 266-270.

Horn, CM, Gilmore, MP & Endress, BA 2012, 'Ecological and socio-economic factors influencing aguaje (Mauritia flexuosa) resource management in two indigenous communities in the Peruvian Amazon', *Forest Ecology and Management*, vol. 267, no. 0, pp. 93-103.

Houghton, R 2005, 'Aboveground forest biomass and the global carbon balance', *Global Change Biology*, vol. 11, no. 6, pp. 945-958.

Houghton, R, Skole, D, Nobre, CA, Hackler, J, Lawrence, K & Chomentowski, WH 2000, 'Annual fluxes of carbon from deforestation and regrowth in the Brazilian Amazon', *Nature*, vol. 403, no. 6767, pp. 301-304.

Houghton, RA, Lawrence, KT, Hackler, JL & Brown, S 2001, 'The spatial distribution of forest biomass in the Brazilian Amazon: a comparison of estimates', *Global Change Biology*, vol. 7, no. 7, pp. 731-746.

Huang, C-H & Kronrad, GD 2001, 'The cost of sequestering carbon on private forest lands', *Forest Policy and Economics*, vol. 2, no. 2, pp. 133-142.

Huettner, M 2012, 'Risks and opportunities of REDD+ implementation for environmental integrity and socio-economic compatibility', *Environmental Science & Policy*, vol. 15, no. 1, pp. 4-12.

Hugosson, M & Ingemarson, F 2004, 'Objectives and motivations of small-scale forest owners; theoretical modelling and qualitative assessment', *Silva Fennica*, vol. 38, no. 2, pp. 217-231.

Hurteau, MD, Koch, GW & Hungate, BA 2008, 'Carbon protection and fire risk reduction: toward a full accounting of forest carbon offsets', *Frontiers in Ecology and the Environment*, vol. 6, no. 9, pp. 493-498.

Husch, B, Beers, TW & Kershaw, JA 2003, *Forest mensuration*, John Wiley & Sons, Inc., Hoboken, New Jersey.

ICIMOD 2014, 'Alternative Energy Scheme in pilot sites of Nepal' Unpublished report, International Center for Integrated Mountain Development, Kathmandu

Ihor, PS & Keeton, WS (eds) 2009, *Ecological economics and sustainable forest management: Developing a trans-disciplinary approach for the Carpathian Mountains*, Ukrainian National Forestry University Press., Lviv.

ILO, 2010, *Labour and social trends in Nepal*, National Planning Commission, governmnet of Nepal and International Labour Office, Nepal

IPCC 2006, 2006 IPCC guidelines for national greenhouse gas inventories, prepared by the National Greenhouse Gas Inventories Programme, Eggleston H.S., Buendia L., Miwa K., Ngara T., Tanabe K.(eds.). IGES, Japan.

IPCC 2014, 'Climate change 2014: Mitigation of climate change. Working group III contribution to the IPCC 5th assessment report-

Changes to the underlying scientific/technical assessment. Chapter 11-Agriculture, forestry and other land use (AFOLU)' [Bustamante, M.,Smith, P., Ahammad, H., Clark, H., Dong, H., Elsiddig, E.A., Haberl, H., Happer, R., House, J., Jafari, M., Masera Ceruti, O., Mbow, C., Ravindranath,N., Rice, C., Robledo Abad, C., Romanovskaya, A., Sperling, F., Rubiello, F., Krug, T. and Nabburs, G.]. Intergovernmental Panel on Climate Change, Geneva. 179 pp. Available at: http://report.mitigation2014.org/drafts/final-draft-postplenary/ipcc_wg3_ar5_final-draft_postplenary_chapter11.pdf

Ishii, HT, Maleque, MA & Taniguchi, S 2008, 'Line thinning promotes stand growth and understory diversity in Japanese cedar (Cryptomeria japonica D. Don) plantations', *Journal of Forest Research*, vol. 13, no. 1, pp. 73-78.

Islam, K & Weil, R 2000, 'Land use effects on soil quality in a tropical forest ecosystem of Bangladesh', *Agriculture, ecosystems & environment*, vol. 79, no. 1, pp. 9-16.

Ives, JD & Messerli, B 1989, 'The Himalayan dilemma: reconciling development and conservation', Psychology Press.

Jackson, J 1994, *Manual of afforestation in Nepal (second edn)*, Forest Research and Survey Centre, Kathmandu, Nepal.

Jacoby, HG 2000, 'Access to markets and the benefits of rural roads', *The Economic Journal*, vol. 110, no. 465, pp. 713-737.

Jain, RK & Singh, B 1999, 'Fuelwood characteristics of selected indigenous tree species from central India', *Bioresource Technology*, vol. 68, no. 3, pp. 305-308.

Jalonen, J, Vanha-Majamaa, I & Tonteri, T 1998, 'Optimal sample and plot size for inventory of field and ground layer vegetation in a mature Myrtillus-type boreal spruce forest', *Annales Botanici Fennici*, Vol. 35, pp. 191-196 Finnish Zoological and botanical Publishing Board.

Jandl, R, Lindner, M, Vesterdal, L, Bauwens, B, Baritz, R, Hagedorn, F, Johnson, DW, Minkkinen, K & Byrne, KA 2007, 'How strongly can forest management influence soil carbon sequestration?', *Geoderma*, vol. 137, no. 3, pp. 253-268.

Jha, S & Bawa, KS 2006, 'Population growth, human development, and deforestation in biodiversity hotspots', *Conservation Biology*, vol. 20, no. 3, pp. 906-912.

Jia, B, He, H, Ma, F, Diao, M, Jiang, G, Zheng, Z, Cui, J & Fan, H 2014, 'Modeling aboveground biomass accumulation of cotton', *JAPS, Journal of Animal and Plant Sciences*, vol. 24, no. 1, pp. 280-289.

Jin, S & Sader, SA 2006, 'Effects of forest ownership and change on forest harvest rates, types and trends in northern Maine', *Forest Ecology and Management*, vol. 228, no. 1-3, pp. 177-186.

Johns, T, Merry, F, Stickler, C, Nepstad, D, Laporte, N & Goetz, S 2008, 'A threefund approach to incorporating government, public and private forest stewards into a REDD funding mechanism', *International Forestry Review*, pp. 458-464.

Johnson, B & Turner, LA 2003, 'Data collection strategies in mixed methods research', *Handbook of mixed methods in social and behavioral research*, pp. 297-319.

Johnson, I & Coburn, R 2010, 'Trees for carbon sequestration ', in *Primefact 981*. *Primefacts for profitable, adaptive and sustainable primary industries*, Department of Industry and Investment, NSW

Joshi, L 2011, 'A community-based PES scheme for forest preservation and sediment control in Kulekhani, Nepal', *Payments for Ecosystem Services and Food Security*, pp. 198-203.

Kanel, KR 2004, 'Twenty five years of community forestry: contribution to millennium development goals', in R.K. Kanel, P. Mathema, B.R. Kandel, D.R. Niraula, A.R. Sharma, M. Gautam (Eds.), *Twenty-five years of community forestry: proceedings of the fourth National Workshop on Community Forestry* 4–6 August, 2004. Kathmandu, Ministry of Forests and Soil Conservation, Govennmet of Nepal, pp. 4–18

Kanel, KR 2006, 'Current status of community forestry in Nepal', submitted to Regional Community Forestry Training Center for Asia and the Pacific Bangkok, Thailand 26p.

Kanel, KR & Kandel, BR 2004, 'Community forestry in Nepal: achievement and challenges', *Journal of Forest and Livelihood*, vol. 4, no. 4, pp. 55-63.

Kanowski, J, Catterall, CP & Wardell-Johnson, GW 2005, 'Consequences of broadscale timber plantations for biodiversity in cleared rainforest landscapes of tropical and subtropical Australia', *Forest Ecology and Management*, vol. 208, no. 1, pp. 359-372.

Kanowski, PJ, McDermott, CL & Cashore, BW 2011, 'Implementing REDD+: lessons from analysis of forest governance', *Environmental Science & Policy*, vol. 14, no. 2, pp. 111-117,

Karky, BS & Skutsch, M 2010, 'The cost of carbon abatement through community forest management in Nepal Himalaya', *Ecological Economics*, vol. 69, no. 3, pp. 666-672.

Karousakis, K 2009, *Promoting biodiversity co-benefits in REDD*, OECD Publishing.

Karsenty, A, Vogel, A, Castell, F 2014, ' "Carbon rights", REDD+ and payments for environmental services', *Environmental Science & Policy*, vol. 35, pp.20-29.

Kasischke, ES, Christensen Jr, N & Stocks, BJ 1995, 'Fire, global warming, and the carbon balance of boreal forests', *Ecological Applications*, vol. 5, no. 2, pp. 437-451.

Kataki, R & Konwer, D 2002, 'Fuelwood characteristics of indigenous tree species of north-east India', *Biomass and Bioenergy*, vol. 22, no. 6, pp. 433-437.

Katuwal, H & Bohara, AK 2009, 'Biogas: a promising renewable technology and its impact on rural households in Nepal', *Renewable and Sustainable Energy Reviews*, vol. 13, no. 9, pp. 2668-2674.

Kaul, M, Dadhwal, VK & Mohren, GMJ 2009, 'Land use change and net C flux in Indian forests', *Forest Ecology and Management*, vol. 258, no. 2, pp. 100-108.

Kauppi, PE, Mielikäinen, K & Kuusela, K 1992, 'Biomass and carbon budget of European forests, 1971 to 1990', *Science*, vol. 256, no. 5053, pp. 70-74,

KC, S, Khanal, SK, Shrestha, P & Lamsal, B 2011, 'Current status of renewable energy in Nepal: Opportunities and challenges', *Renewable and Sustainable Energy Reviews*, vol. 15, no. 8, pp. 4107-4117.

Keith, H, Mackey, BG & Lindenmayer, DB 2009, 'Re-evaluation of forest biomass carbon stocks and lessons from the world's most carbon-dense forests', *Proceedings of the National Academy of Sciences*, vol. 106, no. 28, pp. 11635-11640.

Kelkar, G & Nathan, D 2005, *Gender relations and the energy transition in rural Asi*', New Delhi, India: UNIFEM.

Kellert, SR, Mehta, JN, Ebbin, SA & Lichtenfeld, LL 2000, 'Community natural resource management: Promise, rhetoric, and reality', *Society & Natural Resources*, vol. 13, no. 8, pp. 705-715.

Kenzo, T, Ichie, T, Hattori, D, Kendawang, JJ, Sakurai, K & Ninomiya, I 2010, 'Changes in above- and belowground biomass in early successional tropical secondary forests after shifting cultivation in Sarawak, Malaysia', *Forest Ecology and Management*, vol. 260, no. 5, pp. 875-882.

Khanna, L 2004, 'Theory and practice of Indian silvicultural systems', Khanna Bandhu, Dehradun.

Kitzinger, J 1994, 'The methodology of focus groups: the importance of interaction between research participants', *Sociology of health & illness*, vol. 16, no. 1, pp. 103-121.

Kollmuss, A, Zink, H & Polycarp, C 2008, *Making sense of the voluntary carbon market: A comparison of carbon offset standards*, WWF Germany.

Koop, G & Tole, L 2001, 'Deforestation, distribution and development', *Global Environmental Change*, vol. 11, no. 3, pp. 193-202.

Korhonen, KT, Sharma, ER & Rajbhandari, RD 1992, *Diameter growth and height models for forest trees in Kapilbastu District*, HMGN Ministry of Forest and Environment, Forest Survey and Statistics Division, FSIPS/FRIS. Technical Report.

Korning, J & Balslev, H 1994, 'Growth rates and mortality patterns of tropical lowland tree species and the relation to forest structure in Amazonian Ecuador', *Journal of Tropical Ecology*, vol. 10, no.2, pp. 151-166.

Köthke, M, Schröppel, B & Elsasser, P 2014, 'National REDD+ reference levels deduced from the global deforestation curve', *Forest Policy and Economics*, vol. 43, pp. 18-28.

Kuehne, C, Nosko, P, Horwath, T & Bauhus, J 2014, 'A comparative study of physiological and morphological seedling traits associated with shade tolerance in introduced red oak (Quercus rubra) and native hardwood tree species in southwestern Germany', *Tree physiology*, vol. 34, no. 2, pp. 184-193.

Kumar, A, Bhattacharya, SC & Pham, HL 2003, 'Greenhouse gas mitigation potential of biomass energy technologies in Vietnam using the long range energy alternative planning system model', *Energy*, vol. 28, no. 7, pp. 627-654.

Kumar, R & Shahabuddin, G 2005, 'Effects of biomass extraction on vegetation structure, diversity and composition of forests in Sariska Tiger Reserve, India', *Environmental Conservation*, vol. 32, no. 03, pp. 248-259.

Kyoto-Protocol 1997, *The Kyoto protocol to the united nations framework convention on climate change. Kyoto*, UNEP/WMO.

Laamanen, R, Joshi, MR & Sharma, SP 1995, *Biomass and volume models for Sal (Shorea robusta) in the central Terai of Nepal*, Forest Resource Information System Project (FRISP), HMGN/FINNIDA, Finnish Forest and Park Service. Kathmandu, Nepal. FRIS Project Paper No. 7.

Lal, R 2006, 'Enhancing crop yields in the developing countries through restoration of the soil organic carbon pool in agricultural lands', *Land Degradation & Development*, vol. 17, no. 2, pp. 197-209.

Lambin, EF 1999, 'Monitoring forest degradation in tropical regions by remote sensing: some methodological issues', *Global ecology and biogeography*, vol. 8, no. 3-4, pp. 191-198.

Lambin, EF, Turner, BL, Geist, HJ, Agbola, SB, Angelsen, A, Bruce, JW, Coomes, OT, Dirzo, R, Fischer, G & Folke, C 2001, 'The causes of land-use and land-cover change: moving beyond the myths', *Global Environmental Change*, vol. 11, no. 4, pp. 261-269.

Landsberg, J & Waring, R 1997, 'A generalised model of forest productivity using simplified concepts of radiation-use efficiency, carbon balance and partitioning', *Forest Ecology and Management*, vol. 95, no. 3, pp. 209-228.

Lanier, A, Romsos, C & Goldfinger, C 2007, 'Seafloor habitat mapping on the Oregon continental margin: A spatially nested GIS approach to mapping scale, mapping methods, and accuracy quantification', *Marine Geodesy*, vol. 30, no. 1-2, pp. 51-76.

Larrazábal, A, McCall, MK, Mwampamba, TH & Skutsch, M 2012, 'The role of community carbon monitoring for REDD+: a review of experiences', *Current Opinion in Environmental Sustainability*, vol. 4, no. 6, pp. 707-716.

Larson, AM 2011, 'Forest tenure reform in the age of climate change: Lessons for REDD+', *Global Environmental Change*, vol. 21, no. 2, pp. 540-549.

Laurance, WF 2008, 'Can carbon trading save vanishing forests', *BioScience*, vol. 58, no. 4, pp. 286-7,

Laurance, WF, Alonso, A, Lee, M & Campbell, P 2006, 'Challenges for forest conservation in Gabon, Central Africa', *Futures*, vol. 38, no. 4, pp. 454-470.

Laurance, WF, Fearnside, PM, Laurance, SG, Delamonica, P, Lovejoy, TE, Rankinde Merona, JM, Chambers, JQ & Gascon, C 1999, 'Relationship between soils and Amazon forest biomass: a landscape-scale study', *Forest Ecology and Management*, vol. 118, no. 1, pp. 127-138.

Lawlor, K & Huberman, D 2009, 'Reduced emissions from deforestation and forest degradation (REDD) and human rights', in J Campese, TCH Sunderland, T Greiber & G Oviedo (eds), *Rights-based Approaches: Exploring Issues and Opportunities for Conservation*, CIFOR, Bogor, pp. 269-286.

Lederer, M 2011, 'From CDM to REDD+ — What do we know for setting up effective and legitimate carbon governance?', *Ecological Economics*, vol. 70, no. 11, pp. 1900-1907.

Lee, C-C 2005, 'Energy consumption and GDP in developing countries: a cointegrated panel analysis', *Energy Economics*, vol. 27, no. 3, pp. 415-427.

Leggett, M & Lovell, H 2012, 'Community perceptions of REDD+: a case study from Papua New Guinea', *Climate Policy*, vol. 12, no. 1, pp. 115-134.

Leuschner, C 2000, 'Are high elevations in tropical mountains arid environments for plants?', *Ecology*, vol. 81, no. 5, pp. 1425-1436.

Lin, L, Pattanayak, SK, Sills, EO & Sunderlin, WD 2012, *Site selection for forest carbon projects*, CIFOR, Bogor, Indonesia.

Lindhjem, H, Aronsen, I, Bråten, KG & Gleinsvik, A 2011, *Experiences with benefit sharing: Issues and options for REDD-plus*, Pöyry Management Consulting (Norway) AS.

Liski, J, Pussinen, A, Pingoud, K, Mäkipää, R & Karjalainen, T 2001, 'Which rotation length is favourable to carbon sequestration?', *Canadian Journal of Forest Research*, vol. 31, no. 11, pp. 2004-2013.

Litton, CM, Ryan, MG, Tinker, DB & Knight, DH 2003, 'Belowground and aboveground biomass in young postfire lodgepole pine forests of contrasting tree density', *Canadian Journal of Forest Research*, vol. 33, no. 2, pp. 351-363.

Liu, AM & Walker, A 1998, 'Evaluation of project outcomes', *Construction Management & Economics*, vol. 16, no. 2, pp. 209-19.

Lodhiyal, N & Lodhiyal, L 2003, 'Biomass and net primary productivity of Bhabar Shisham forests in Central Himalaya, India', *Forest Ecology and Management*, vol. 176, no. 1-3, pp. 217-235.

Lu, S-Y, Cheng, JD & Brooks, KN 2001, 'Managing forests for watershed protection in Taiwan', *Forest Ecology and Management*, vol. 143, no. 1, pp. 77-85.

Luckert, MK 1999, 'Are community forests the key to sustainable forest management? Some economic considerations', *The Forestry Chronicle*, vol. 75, no. 5, pp. 789-792.

Luyssaert, S, Schulze, E-D, Börner, A, Knohl, A, Hessenmöller, D, Law, BE, Ciais, P & Grace, J 2008, 'Old-growth forests as global carbon sinks', *Nature*, vol. 455, no. 7210, pp. 213-215.

Lynn, B, Dahal, DR & Govindasamy, P 2008, *Caste, ethnic and regional identity in Nepal: Further analysis of the 2006 Nepal demographic and health survey*, Calverton, Maryland, USA: Macro International Inc.

Lyster, R 2011, 'REDD+, transparency, participation and resource rights: the role of law', *Environmental Science & Policy*, vol. 14, no. 2, pp. 118-126,

MacDicken, K 1997, A guide to monitoring carbon storage in forestry and agro forestry projects, Forest Carbon Monitoring Program Winrock International Institute for Agricultural Development, VA., USA, International Institute for Agricultural Development, US Agency for International Development. http://www. fcarbonsinks. gov. cn/thjl/Winrock% 20International, vol. 20.

Magnussen, S & Boudewyn, P 1998, 'Derivations of stand heights from airborne laser scanner data with canopy-based quantile estimators', *Canadian Journal of Forest Research*, vol. 28, no. 7, pp. 1016-1031.

Mahanty, S, Gronow, J, Nurse, M & Malla, Y 2009, 'Reducing poverty through community based forest management in Asia', *Journal of Forest and Livelihood*, vol. 5, no. 1, pp. 78-89.

Mailly, D, Kimmins, JP & Busing, RT 2000, 'Disturbance and succession in a coniferous forest of northwestern North America: simulations with dryades, a spatial gap model', *Ecological Modelling*, vol. 127, no. 2–3, pp. 183-205.

Malhi, Y, Baldocchi, D & Jarvis, P 1999, 'The carbon balance of tropical, temperate and boreal forests', *Plant, Cell & Environment*, vol. 22, no. 6, pp. 715-740.

Malla, G 2009, 'Climate change and its impact on Nepalese agriculture', *Journal of Agriculture and Environment*, vol. 9, pp. 62-71.

Malla, YB, Neupene, HR & Branney, PJ 2003, 'Why aren't poor people benefiting more from community forestry', *Journal of Forest and Livelihood*, vol. 3, 1, pp.78-92.

Maltsoglou, I & Taniguchi, K 2004, *Poverty, livestock and household typologies in Nepal*, Agriculture and Developmenet Economics Division, The Food and Agriculture Organization of the United Nations.

Manandhar, NP 2002, Plants and people of Nepal, Timber Press, Portland, OR

Manhas, R, Negi, J, Kumar, R & Chauhan, P 2006, 'Temporal assessment of growing stock, biomass and carbon stock of Indian forests', *Climatic change*, vol. 74, no. 1-3, pp. 191-221.

Mani, S & Parthasarathy, N 2009, 'Tree population and above-ground biomass changes in two disturbed tropical dry evergreen forests of peninsular India', *Tropical Ecology*, vol. 50, no. 2, p. 249.

Marahatta, S, Dangol, BS & Gurung, GB 2009, *Temporal and spatial variability of climate change over Nepal (1976 - 2005)*. Practical Action Nepal.

Maraseni, TN 2007, *Re-evaluating land use choices to incorporate carbon values: A case study in the south burnett region of Queensland, Australia, Ph.D thesis, University of Southern Queensland, Australia.*

Maraseni, TN & Xinquan, G 2011, 'An analysis of Chinese perceptions on unilateral Clean Development Mechanism (uCDM) projects', *Environmental Science & Policy*, vol. 14, no. 3, pp. 339-346.

Maraseni, TN & Cockfield, G 2011, 'Crops, cows or timber? Including carbon values in land use choices', *Agriculture, ecosystems & environment*, vol. 140, no. 1–2, pp. 280-288.

Maraseni, TN, Cockfield, G & Apan, A 2005, 'Community based forest management systems in developing countries and eligibility for Clean Development Mechanism', *Journal of Forest and Livelihood*, vol. 4, no. 2, pp. 31-42.

Maraseni, TN, Cockfield, G & Maroulis, J 2010, 'An assessment of greenhouse gas emissions: implications for the Australian cotton industry', *Journal of Agricultural Science*, vol. 148, no. 5, pp. 501-510.

Maraseni, TN, Neupane, PR, Lopez-Casero, F & Cadman, T 2014, 'An assessment of the impacts of the REDD+ pilot project on community forests user groups (CFUGs) and their community forests in Nepal', *Journal of Environmental Management*, vol. 136, pp. 37-46.

Marshall, M 1996, 'The key informant technique', *Family Practice*, vol. 13, no. 1, pp. 92-97.

Maskey, V, Gebremedhin, TG & Dalton, TJ 2006, 'Social and cultural determinants of collective management of community forest in Nepal', *Journal of Forest Economics*, vol. 11, no. 4, pp. 261-274.

Mather, A & Needle, C 2000, 'The relationships of population and forest trends', *The Geographical Journal*, vol. 166, no. 1, pp. 2-13.

May-Tobin, C 2011, 'Wood for fuel', in D Boucher, P Elias, K Lininger, C May-Tobin, S Roquemore & E Saxon (eds), *The Root of the Problem: What's Driving Tropical Deforestation Today*, Union of Concerned Scientists, Cambridge,MA 02138-3780, p. 11.

McDermott, CL, 2014, 'REDDuced: From sustainability to legality to units of carbon—The search for common interests in international forest governance.' *Environmental Science & Policy*, vol 35, pp. 12-19.

McDermott, M & Schreckenberg, K 2009, 'Equity in community forestry: insights from North and South', *International Forestry Review*, vol. 11, no. 2, pp. 157-170,

McDonald, GT & Lane, MB 2004, 'Converging global indicators for sustainable forest management', *Forest Policy and Economics*, vol. 6, no. 1, pp. 63-70.

McRoberts, RE, Wendt, DG, Nelson, MD & Hansen, MH 2002, 'Using a land cover classification based on satellite imagery to improve the precision of forest inventory area estimates', *Remote Sensing of Environment*, vol. 81, no. 1, pp. 36-44.

Mehta, J 1996, 'Forest fires and land degradation in Uttarakhand', *Land utilization in the Central Himalaya: problems and management options*, no. 8, p. 125.

Meinshausen, M 2006, 'What does a 2^oC target mean for greenhouse gas concentrations? A brief analysis based on multi-gas emission pathways and several climate sensitivity uncertainty estimates', In: HJ Schellnhuber, W Cramer, N Nakicenovic, T Wigley, G Yohe (Eds.), *Avoiding Dangerous Climate Change*. Cambridge University Press, Cambridge, pp. 265–279.

Metz, JJ 1990, 'Forest-product use in upland Nepal', *Geographical Review*, vol 80, no.3, pp. 279-287.

Metz, JJ 1994, 'Forest product use at an upper elevation village in Nepal', *Environmental Management*, vol. 18, no. 3, pp. 371-390.

MFSC 1988, Master plan for the forestry sector Nepal: Forest resources information status and development plan, Ministry of Forest and Soil Conservation, Govenment of Nepal.

MFSC 2011, *Study on REDD plus piloting in Nepal*, REDD forestry and Climate Change Cell, Ministry of Forests and Soil Conservation, Govenment of Nepal.

MFSC 2013, *Persistence and change : Review of 30 years of community forestry in Nepal*, Ministry of Forests and Soil Conservation, Nepal, Govenment of Nepal.

Michalek, J, French, N, Kasischke, E, Johnson, R & Colwell, J 2000, 'Using Landsat TM data to estimate carbon release from burned biomass in an Alaskan spruce forest complex', *International Journal of Remote Sensing*, vol. 21, no. 2, pp. 323-338,

Miles, L 2010, *Implications of the REDD negotiations for forest restoration*, UNEP, World Conservation Monitoring Centre.

Millar, CI, Stephenson, NL & Stephens, SL 2007, 'Climate change and forests of the future: managing in the face of uncertainty', *Ecological Applications*, vol. 17, no. 8, pp. 2145-2151.

MillenniumEcosystemAssessment 2003, *Ecosystems and human well-being: A famework for assessment*, Millennium Ecosystem Assessment. Island Press, Washington, D.C., USA.

Miller, CC & Cardinal, LB 1994, 'Strategic planning and firm performance: a synthesis of more than two decades of research', *Academy of Management Journal*, vol. 37, no. 6, pp. 1649-1665.

Minang, PA & van Noordwijk, M 2013, 'Design challenges for achieving reduced emissions from deforestation and forest degradation through conservation: leveraging multiple paradigms at the tropical forest margins', *Land Use Policy*, vol. Applegate, GB 31, pp. 61-70.

Mohns, B, Applegate, GB & Gilmour, DA 1989, 'Biomass and productivity estimations for community forest management: A case study from the Middle Hills of Nepal. 2. Dry matter production in mixed young stands of Chir Pine (*Pinus*)

roxburghii Sargent) and secondary succession broadleaf species', *Biomass'*, vol. 17, no. 3, pp.165-184.

Mokany, K, Raison, R & Prokushkin, AS 2006, 'Critical analysis of root: shoot ratios in terrestrial biomes', *Global Change Biology*, vol. 12, no. 1, pp. 84-96.

Mon, MS, Mizoue, N, Htun, NZ, Kajisa, T & Yoshida, S 2012, 'Factors affecting deforestation and forest degradation in selectively logged production forest: A case study in Myanmar', *Forest Ecology and Management*, vol. 267, pp. 190-198.

Monserud, RA 2003, 'Evaluating forest models in a sustainable forest management context', *Forest Biometry, Modelling and Information Sciences*, vol. 1, pp. 35-47.

Montagnini, F & Ashton, MS 2010, *The silvicultural basis for agroforestry systems*, CRC Press.

Moser, LE & Hoveland, CS 1996, A cool-season grass overview, Crop Science Society of America, Madison, WI

Moutinho, P, Martins, OS, Christovam, M, Lima, A, Nepstad, D & Crisostomo, AC 2011, 'The emerging REDD+ regime of Brazil', *Carbon Management*, vol. 2, no. 5, pp. 587-602.

MPFS 1988, Master plan for forestry sector Nepal (Appendix table 2.2 forest types, representative species, uses and wood density), Ministry of forest and soil conservation, Government of Nepal.

Murray, BC, Sohngen, B & Ross, MT 2007, 'Economic consequences of consideration of permanence, leakage and additionality for soil carbon sequestration projects', *Climatic change*, vol. 80, no. 1-2, pp. 127-143.

Mwendera, E, Saleem, M & Woldu, Z 1997, 'Vegetation response to cattle grazing in the Ethiopian highlands', *Agriculture, ecosystems & environment*, vol. 64, no. 1, pp. 43-51.

Næsset, E, Gobakken, T, Solberg, S, Gregoire, TG, Nelson, R, Ståhl, G & Weydahl, D 2011, 'Model-assisted regional forest biomass estimation using LiDAR and InSAR as auxiliary data: A case study from a boreal forest area', *Remote Sensing of Environment*, vol. 115, no. 12, pp. 3599-3614.

Nagendra, H 2002, 'Tenure and forest conditions: community forestry in the Nepal Terai', *Environmental Conservation*, vol. 29, no. 4, pp. 530-539.

Nagendra, H 2007, 'Drivers of reforestation in human-dominated forests', *Proceedings of the National Academy of Sciences*, vol. 104, no. 39, p. 15218.

Nagendra, H, Karmacharya, M & Karna, B 2005, 'Evaluating forest management in Nepal: views across space and time', *Ecology and Society*, vol. 10, no. 1, p. 24.

Nelson, E, Mendoza, G, Regetz, J, Polasky, S, Tallis, H, Cameron, D, Chan, KM, Daily, GC, Goldstein, J & Kareiva, PM 2009, 'Modeling multiple ecosystem services, biodiversity conservation, commodity production, and tradeoffs at landscape scales', *Frontiers in Ecology and the Environment*, vol. 7, no. 1, pp. 4-11.

Nepal, SK 2000, 'Tourism in protected areas: the Nepalese Himalaya', Annals of Tourism Research, vol. 27, no. 3, pp. 661-681.

Nepstad, DC, Verissimo, A, Alencar, A, Nobre, C, Lima, E, Lefebvre, P, Schlesinger, P, Potter, C, Moutinho, P, Mendoza, E, M., C & V., B 1999, 'Large-scale impoverishment of Amazonian forests by logging and fire', *Nature*, vol. 398, pp. 505-508.

Neupane, H 2003, 'Contested impact of community forestry on equity: Some evidence from Nepal', *Journal of Forest and Livelihood*, vol. 2, no. 2, pp. 55-61.

Ngugi, MR, Johnson, RW & McDonald, WJF 2011, 'Restoration of ecosystems for biodiversity and carbon sequestration: Simulating growth dynamics of brigalow vegetation communities in Australia', *Ecological Modelling*, vol. 222, no. 3, pp. 785-794.

Nordh, N-E & Verwijst, T 2004, 'Above-ground biomass assessments and first cutting cycle production in willow (Salix sp.) coppice—a comparison between destructive and non-destructive methods', *Biomass and Bioenergy*, vol. 27, no. 1, pp. 1-8.

Nowak, DJ & Crane, DE 2002, 'Carbon storage and sequestration by urban trees in the USA', *Environmental pollution*, vol. 116, no. 3, pp. 381-389.

Nowak, DJ, Stevens, JC, Sisinni, SM & Luley, CJ 2002, 'Effects of urban tree management and species selection on atmospheric carbon dioxide', *Journal of Arboriculture*, vol. 28, no. 3, pp. 113-122.

NPC/GON 2002, *Tenth Plan* (2002-2007), National Planning Commission, Government of Nepal.

Nunery, JS & Keeton, WS 2010, 'Forest carbon storage in the northeastern United States: Net effects of harvesting frequency, post-harvest retention, and wood products', *Forest Ecology and Management*, vol. 259, no. 8, pp. 1363-1375.

Nurse, M & Malla, Y 2006, 'Advances in community forestry in Asia', *About the Organisations*, p. 25.

Nyland, RD 1996, Silviculture: concepts and applications, McGraw-Hill, New York.

O'Brien, EK, Denham, AJ & Ayre, DJ 2014, 'Patterns of genotypic diversity suggest a long history of clonality and population isolation in the Australian arid zone shrub Acacia carneorum', *Plant Ecology*, vol. 215, no. 1, pp. 55-71.

Ojha, HR, Persha, L & Chhatre, A 2009, 'Community forestry in Nepal: A policy innovation for local livelihoods', *Proven Successes in Agricultural Development*, p. 123.

Okereke, C & Dooley, K 2010, 'Principles of justice in proposals and policy approaches to avoided deforestation: Towards a post-Kyoto climate agreement', *Global Environmental Change*, vol. 20, no. 1, pp. 82-95.

Olander, LP, Galik, CS & Kissinger, GA 2012, 'Operationalizing REDD+: scope of reduced emissions from deforestation and forest degradation', *Current Opinion in Environmental Sustainability*, vol. 4, no. 6, pp. 661-669.

Oliver, CD & Larson, BC 1990, Forest stand dynamics, McGraw-Hill, Inc.

Ordóñez, JAB, de Jong, BHJ, García-Oliva, F, Aviña, FL, Pérez, JV, Guerrero, G, Martínez, R & Masera, O 2008, 'Carbon content in vegetation, litter, and soil under 10 different land-use and land-cover classes in the Central Highlands of Michoacan, Mexico', *Forest Ecology and Management*, vol. 255, no. 7, pp. 2074-2084.

Ostrom, E 1990, *Governing the commons: The evolution of institutions for collective action*, Cambridge Univ Pr.

Ostrom, E 1999, 'Coping with tragedies of the commons', *Annual review of political science*, vol. 2, no. 1, pp. 493-535.

Ostrom, E & Gardner, R 1993, 'Coping with asymmetries in the commons: Selfgoverning irrigation systems can work', *Journal of Economic Perspectives*, vol. Volume 7, , no. Number 4-Fall 1993, pp. 93-112.

Ostrom, E, Gardner, R & Walker, J 1994, *Rules, games, and common-pool resources*, Univ of Michigan Pr.

Pabst, RJ, Goslin, MN, Garman, SL & Spies, TA 2008, 'Calibrating and testing a gap model for simulating forest management in the Oregon Coast Range', *Forest Ecology and Management*, vol. 256, no. 5, pp. 958-972.

Pagdee, A, Kim, Y-s & Daugherty, P 2006, 'What makes community forest management successful: A meta-study from community forests throughout the world', *Society and Natural Resources*, vol. 19, no. 1, pp. 33-52.

Pagiola, S 2008, 'Payments for environmental services in Costa Rica', *Ecological Economics*, vol. 65, no. 4, pp. 712-724.

Pahari, K & Murai, S 1999, 'Modelling for prediction of global deforestation based on the growth of human population', *ISPRS Journal of Photogrammetry and Remote Sensing*, vol. 54, no. 5-6, pp. 317-324.

Palmer, C 2011, 'Property rights and liability for deforestation under REDD+: Implications for 'permanence'in policy design', *Ecological Economics*, vol. 70, no. 4, pp. 571-576.

Pandey, S, Subedi, B & Dhungana, H 2011, 'Economic potential of forest resources of Nepal', *Banko Janakari*, vol. 20, no. 2, pp. 48-52.

Pandey, SS 2007, *Tree species diversity in existing community based forest management systems in central mid-hills of Nepal*, MSc thesis, Uppsala, Sweden: Swedish Biodiversity Centre. Swedish University of Agricultural Sciences.

Pandey, SS, Cockfield, G & Maraseni, TN 2013, 'Major drivers of deforestation and forest degradation in developing countries and REDD+ ', *International Journal of Forest Usufructs Management*, vol. 14 no. 1 pp. 99-107.

Pandit, R & Bevilacqua, E 2011, 'Forest users and environmental impacts of community forestry in the hills of Nepal', *Forest Policy and Economics*, vol. 13, no. 5, pp. 345-352.

Parikka, M 2004, 'Global biomass fuel resources', *Biomass and Bioenergy*, vol. 27, no. 6, pp. 613-620.

Paudel, KC 2005, '20 Forests for poverty alleviation: the changing role of R&D institutions in Nepal', *Forest for proverty reduction*, p. 137,

Paul, K & Polglase, P 2004, 'Prediction of decomposition of litter under eucalypts and pines using the FullCAM model', *Forest Ecology and Management*, vol. 191, no. 1, pp. 73-92.

Paul, K, Polglase, P, Nyakuengama, J & Khanna, P 2002, 'Change in soil carbon following afforestation', *Forest Ecology and Management*, vol. 168, no. 1, pp. 241-257.

Pearce, Dw, Atkinson, G & Mourato, S 2006, *Cost-benefit analysis and the environment: recent developments*, Organisation for Economic Cooperation and Development, Paris.

Pedroni, L, Dutschke, M, Streck, C & Porrúa, ME 2009, 'Creating incentives for avoiding further deforestation: the nested approach', *Climate Policy*, vol. 9, no. 2, pp. 207-220.

Peichl, M & Arain, MA 2007, 'Allometry and partitioning of above-and belowground tree biomass in an age-sequence of white pine forests', *Forest Ecology and Management*, vol. 253, no. 1-3, pp. 68-80.

Pelletier, J, Ramankutty, N & Potvin, C 2011, 'Diagnosing the uncertainty and detectability of emission reductions for REDD+ under current capabilities: an example for Panama', *Environmental Research Letters*, vol. 6, no. 2, p. 024005.

Peng, C, Liu, J, Dang, Q, Apps, MJ & Jiang, H 2002, 'TRIPLEX: a generic hybrid model for predicting forest growth and carbon and nitrogen dynamics', *Ecological Modelling*, vol. 153, no. 1-2, pp. 109-130.

Perrin, C, Michel, C & Andréassian, V 2001, 'Does a large number of parameters enhance model performance? Comparative assessment of common catchment model structures on 429 catchments', *Journal of Hydrology*, vol. 242, no. 3, pp. 275-301.

Peskett,L, Schreckenberg, K & Brown, J 2011, 'Institutional approaches for carbon financing in the forest sector: learning lessons for REDD+ from forest carbon projects in Uganda', *Environmental Science & Policy*, vol. 14, no.2, pp. 216-229.

Peters-Stanley, M & Yin, D 2013, *Maneuvering the mosaic state of the voluntary carbon markets 2013*, Ecosystem Marketplace, Bloomberg New Energy Finance, Washington, DC.

Phelps, J, Webb, EL & Agrawal, A 2010, 'Does REDD+ threaten to recentralize forest governance?', *Science*, vol. 328, no. 5976, pp. 312-313.

Phelps, J, Guerrero, M, Dalabajan, D, Young, B & Webb, E 2010, 'What makes a REDD country?', *Global Environmental Change*, vol. 20, no. 2, pp. 322-332,

Pimentel, D, Wilson, C, McCullum, C, Huang, R, Dwen, P, Flack, J, Tran, Q, Saltman, T & Cliff, B 1997, 'Economic and environmental benefits of biodiversity', *BioScience*, vol. 47, no. 11, pp. 747-757.

Pirard, R 2008, 'The fight against deforestation (REDD+): economic implications of market-based funding', *Paris: IDDRI*.

Pokharel, R 2011, 'Factors influencing the management regime of Nepal's community forestry', *Forest Policy and Economics*, vol 17, pp.13-17.

Pokharel, RK 2008, *Nepal's community forestry funds: Do they benefit the poor?*, South Asian Network for Development and Environmental Economics, Kathmandu, Nepal.

Pokharel, S & Chandrashekar, M 1994, 'Biomass resources as energy in Nepal', *Natural Resources Forum*, vol. 18, no. 3, pp. 225-230.

Poorter, L 2001, 'Light-dependent changes in biomass allocation and their importance for growth of rain forest tree species', *Ecology*, vol. 15, pp. 113-123.

Poteete, AR & Ostrom, E 2004, 'Heterogeneity, group size and collective action: The role of institutions in forest management', *Development and Change*, vol. 35, no. 3, pp. 435-461.

Powers, RF 2001, 'Assessing potential sustainable wood yield', *The forests handbook*, vol. 2, pp. 105-128.

Pregitzer, KS & Euskirchen, ES 2004, 'Carbon cycling and storage in world forests: biome patterns related to forest age', *Global Change Biology*, vol. 10, no. 12, pp. 2052-2077.

Pretzsch, H 2010, Forest dynamics, growth, and yield, Springer, Berlin, Heidelberg.

Pukkala, T, Lähde, E & Laiho, O 2009, 'Growth and yield models for uneven-sized forest stands in Finland', *Forest Ecology and Management*, vol. 258, no. 3, pp. 207-216,

Randalls, S 2010, 'History of the 2°C climate target', *Wiley Interdisciplinary Reviews: Climate Change*, vol. 1, no. 4, pp. 598-605.

Rautiainen, O 1995, Growth and yield models for uniform sal (Shorea robusta Gaertn. f.) forests in the Bhabar – Terai in Nepal, Forest Management and Utilization Development Project, HMGN/The Government of Finland. Kathmandu, Nepal. FMUDP Technical Report No. 17.

Rautiainen, O 1999, 'Spatial yield model for Shorea robusta in Nepal', *Forest Ecology and Management*, vol. 119, no. 1-3, pp. 151-162.

Rautiainen, O, Pukkala, T & Miina, J 2000, 'Optimising the management of evenaged Shorea robusta stands in southern Nepal using individual tree growth models', *Forest Ecology and Management*, vol. 126, no. 3, pp. 417-429.

Reyes, G, Brown, S, Chapman, J & Lugo, AE 1992, *Wood densities of tropical tree species*, US Department of Agriculture, Forest Service, Southern Forest Experiment Station.

Resosudarmo, IAP, Duchelle, AE, Ekaputri, AD & Sunderlin, W 2012, '*Local hopes and worries about REDD+ projects*' in A Angelsen, M Brockaus, WD Sunderlin & LV Verchot (Eds.), Analysing REDD+: Challensges and Choice, CIFOR, Indonesia.

Resosudarmo, IAP, Atmadja, S, Ekaputri, AD, Intarini, DY, Indriatmoko, Y, & Astri, P 2014, 'Does Tenure Security Lead to REDD+ Project Effectiveness? Reflections from Five Emerging Sites in Indonesia', *World development*, vol.55, pp. 68-83.

Richards, M, Maharjan, M & Kanel, K 2003, 'Economics, poverty and transparency: measuring equity in forest user groups', *Journal of Forest and Livelihood*, vol. 3, no. 1, pp. 91-106.

Risch, AC, Heiri, C & Bugmann, H 2005, 'Simulating structural forest patterns with a forest gap model: a model evaluation', *Ecological Modelling*, vol. 181, no. 2–3, pp. 161-172.

Rist, J, Milner-Gulland, E, Cowlishaw, G & Rowcliffe, M 2010, 'Hunter reporting of catch per unit effort as a monitoring tool in a bushmeat harvesting system', *Conservation Biology*, vol. 24, no. 2, pp. 489-499.

Robert, J 1991, 'Motivation and education: The self-determination perspective', *Educational Psychologist*, vol. 26, no. 3-4, 325-346.

Robinson, WC & Ross, JA 2007, *The global family planning revolution: three decades of population policies and programs*, World Bank Publications.

Rogers, R & Schumm, S 1991, 'The effect of sparse vegetative cover on erosion and sediment yield', *Journal of Hydrology*, vol. 123, no. 1, pp. 19-24.

Romijn, E, Herold, M, Kooistra, L, Murdiyarso, D & Verchot, L 2012, 'Assessing capacities of non-Annex I countries for national forest monitoring in the context of REDD+', *Environmental Science & Policy*, vol. 19–20, pp. 33-48.

Rossi, S, Tremblay, M-J, Morin, H & Savard, G 2009, 'Growth and productivity of black spruce in even- and uneven-aged stands at the limit of the closed boreal forest', *Forest Ecology and Management*, vol. 258, no. 9, pp. 2153-2161.

Roy, P 2003, 'Forest fire and degradation assessment using satellite remote sensing and geographic information system', *Paper presented at the Satellite Remote Sensing and GIS Applications in Agricultural Meteorology*, Dehra Dun, India p. 361.

Rudel, TK 1989, 'Population, development, and tropical deforestation: A crossnational study.' *Rural sociology*, vol. 54, no. 3, pp. 327-338.

Rudel, TK, Coomes, OT, Moran, E, Achard, F, Angelsen, A, Xu, J & Lambin, E 2005, 'Forest transitions: towards a global understanding of land use change', *Global Environmental Change*, vol. 15, no. 1, pp. 23-31.

Sachs, JD & McArthur, JW 2005, 'The millennium project: a plan for meeting the millennium development goals', *The Lancet*, vol. 365, no. 9456, pp. 347-53.

Sands, P 2000, *3PGPJS–a user-friendly interface to 3-PG, the Landsberg and Waring model of forest productivity*, CSIRO, Australia.

Sapkota, I, Tigabu, M & Christer Odén, P 2009, 'Species diversity and regeneration of old-growth seasonally dry Shorea robusta forests following gap formation', *Journal of Forestry Research*, vol. 20, no. 1, pp. 7-14.

Sapkota, IP & Odén, PC 2008, 'Household characteristics and dependency on community forests in Terai of Nepal', *International journal of social forestry*, vol. 1, no. 2, pp. 123-44,

Sapkota, P & Meilby, H 2009, 'Modelling the growth of Shorea robusta using growth ring measurements', *Banko Janakari*, vol. 19, no. 2, pp. 25-32.

Sasaki, N, Abe, I, Khun, V, Chan, S,Ninomiya, H & Chheng, K 2013, 'Reducing carbon emissions through improved forest management in Cambodia', *Low Carbon Economy*, vol. 4, pp.55-67.

Schimel, DS, House, J, Hibbard, K, Bousquet, P, Ciais, P, Peylin, P, Braswell, BH, Apps, MJ, Baker, D & Bondeau, A 2001, 'Recent patterns and mechanisms of carbon exchange by terrestrial ecosystems', *Nature*, vol. 414, no. 6860, pp. 169-172.

Schlager, E & Ostrom, E 1992, 'Property-rights regimes and natural resources: a conceptual analysis', *Land economics*, pp. 249-262.

Schlamadinger, B, Bosquet, B, Streck, C, Noble, I, Dutschke, M & Bird, N 2005, 'Can the EU emission trading scheme support CDM forestry?', *Climate Policy*, vol. 5, no. 2, pp. 199-208.

Schlesinger, WH & Lichter, J 2001, 'Limited carbon storage in soil and litter of experimental forest plots under increased atmospheric CO₂', *Nature*, vol. 411, no. 6836, pp. 466-469.

Schmitt, CB, Burgess, ND, Coad, L, Belokurov, A, Besançon, C, Boisrobert, L, Campbell, A, Fish, L, Gliddon, D & Humphries, K 2009, 'Global analysis of the protection status of the world's forests', *Biological Conservation*, vol. 142, no. 10, pp. 2122-2130.

Scrieciu, SS 2007, 'Can economic causes of tropical deforestation be identified at a global level?', *Ecological Economics*, vol. 62, no. 3–4, pp. 603-612.

Seddon, D, Gurung, G & Adhikari, J 1998, 'Foreign labour migration and the remittance economy of Nepal', *Himalaya, the Journal of the Association for Nepal and Himalayan Studies*, vol. 18, no. 2, pp.3-10.

Segura, O & Kanninen, M 2005, 'Allometric models for estimating volume and total aboveground biomass of seven dominant tree species in a tropical humid forest in Costa Rica', *Biotropica*, vol. 37, pp. 2-8.

Seidl, I & Tisdell, CA 1999, 'Carrying capacity reconsidered: from Malthus' population theory to cultural carrying capacity', *Ecological Economics*, vol. 31, no. 3, pp. 395-408.

Sharma, CM, Baduni, NP, Gairola, S, Ghildiyal, SK & Suyal, S 2010, 'Tree diversity and carbon stocks of some major forest types of Garhwal Himalaya, India', *Forest Ecology and Management*, vol. 260, no. 12, pp. 2170-2179.

Sharma, ER & Pukkala, T 1990, *Volume equations and biomass prediction of forest trees of Nepal*, Forest Survey and Statistics Division, Ministry of Forests and Soil Conservation. Babar Mahal, Kathmandu, Nepal. Publication 47.

Shi, W, Wang, S & Yang, Q 2010, 'Climate change and global warming', *Reviews in Environmental Science and Biotechnology*, vol. 9, no. 2, pp. 99-102.

Shrestha, AB & Aryal, R 2011, 'Climate change in Nepal and its impact on Himalayan glaciers', *Regional Environmental Change*, vol. 11, no. 1, pp. 65-77.

Shrestha, BB 2005, 'Fuelwood harvest, management and regeneration of two community forests in Central Nepal', *Himalayan Journal of Sciences*, vol. 3, no. 5, pp. 75-80.

Shrestha, BB, Uprety, Y & Jha, PK 2006, 'Wood properties in relation to foliar phenology of some planted tree species at Kirtipur, central Nepal', *Tropical Ecology*, vol. 47, pp. 201-209.

Shrestha, NR & Conway, D 1985, 'Issues in population pressure, land resettlement, and development: The case of Nepal', *Studies in Comparative International Development (SCID)*, vol. 20, no. 1, pp. 55-82.

Shrestha, PM & Dhillion, SS 2006, 'Diversity and traditional knowledge concerning wild food species in a locally managed forest in Nepal', *Agroforestry Systems*, vol. 66, no. 1, pp. 55-63.

Sills, E, Madeira, EM, Sunderlin, WD & Wertz-Kanounnikoff, S 2009, '*The evolving landscape of REDD+ projects*', Edited by: Angelsen A, Brockhaus M, Kanninen M, Sills E, Sunderlin WD, Wertz-Kanounnikoff S. Realising REDD+: National strategy and policy options. CIFOR, Bogor, Indonesia, pp. 265-280.

Silver, WL, Kueppers, LM, Lugo, AE, Ostertag, R & Matzek, V 2004, 'Carbon sequestration and plant community dynamics following reforestation of tropical pasture', *Ecological Applications*, vol. 14, no. 4, pp. 1115-1127.

Singh, B & Chapagain, D 2006, *Trends in forest ownership, forest resources tenure and institutional arrangements: are they contributing to better forest management and poverty reduction?*, Understanding forest tenure in South and Southeast Asia, FAO, Rome.

Singh, K, Kumar, S, Rai, L & Krishna, A 2003, 'Rhododendrons conservation in the Sikkim Himalaya', *Current Schience Bangalore*, vol. 85, no. 5, pp. 602-606.

Singh, KJ & Sooch, SS 2004, 'Comparative study of economics of different models of family size biogas plants for state of Punjab, India', *Energy Conversion and Management*, vol. 45, no. 9–10, pp. 1329-1341.

Singh, PP 2008, 'Exploring biodiversity and climate change benefits of communitybased forest management', *Global Environmental Change*, vol. 18, no. 3, pp. 468-478.

Singh, S & Singh, S 1997, 'Environmental degradation in Uttarakhand Himalaya', in P Nag, VK Kumra & J Sing (Eds.), *Geography and Environment*, Concept Publishing Company, New Delhi.

Singh, SP, Adhikari, BS & Zobel, DB 1994, 'Biomass, productivity, leaf longevity, and forest structure in the central Himalaya', *Ecological Monographs*, vol. 64, no. 4, pp. 401-421.

Siry, JP, Cubbage, FW & Ahmed, MR 2005, 'Sustainable forest management: global trends and opportunities', *Forest Policy and Economics*, vol. 7, no. 4, pp. 551-561.

Skutsch, MM 2005, 'Reducing carbon transaction costs in community-based forest management', *Climate Policy*, vol. 5, no. 4, pp. 433-443.

Skutsch, MM & McCall, MK 2010, 'Reassessing REDD: governance, markets and the hype cycle', *Climatic change*, vol. 100, no. 3, pp. 395-402.

Sloan, S & Pelletier, J 2012, 'How accurately may we project tropical forest-cover change? A validation of a forward-looking baseline for REDD', *Global Environmental Change*, vol. 22, no. 2, pp. 440-453.

Smith, DM, Larson, BC, Kelty, MJ & Ashton, PMS 1997, *The practice of silviculture: applied forest ecology*, John Wiley and Sons, Inc., New York, USA.

Smith Olsen, C & Overgaard Larsen, H 2003, 'Alpine medicinal plant trade and Himalayan mountain livelihood strategies', *The Geographical Journal*, vol. 169, no. 3, pp. 243-254.

Solberg, S, Astrup, R, Gobakken, T, Næsset, E & Weydahl, DJ 2010, 'Estimating spruce and pine biomass with interferometric X-band SAR', *Remote Sensing of Environment*, vol. 114, no. 10, pp. 2353-2360.

Specht, A & West, PW 2003, 'Estimation of biomass and sequestered carbon on farm forest plantations in northern New South Wales, Australia', *Biomass and Bioenergy*, vol. 25, no. 4, pp. 363-379.

Springate-Baginski, O, Springate-Baginski, OW & Wollenberg, E 2010, *REDD*, *forest governance and rural livelihoods: the emerging agenda*, CIFOR, Indonesia

Springate-Baginski, O, Dev, OP, Yadav, NP & Soussan, J 2003, 'Community forest management in the middle hills of Nepal: the changing context', *Journal of Forest and Livelihood*, vol. 3, no. 1, pp. 5-20.

Steininger, M 2000, 'Satellite estimation of tropical secondary forest above-ground biomass: data from Brazil and Bolivia', *International Journal of Remote Sensing*, vol. 21, no. 6-7, pp. 1139-1157.

Stern, N 2007, *The economics of climate change: the Stern review*, Cambridge University Press.

Stewart, DW, Shamdasani, PN & Rook, DW 2007, *Focus groups: Theory and practice*, vol. 20, Sage publications, Newbury Park,CA.

Streck, C 2010, 'Reducing emissions from deforestation and forest degradation: national implementation of REDD schemes', *Climatic change*, vol. 100, no. 3, pp. 389-394.

Streck, C 2012, 'Financing REDD+: matching needs and ends', *Current Opinion in Environmental Sustainability*, vol. 4, no. 6, pp. 628-637.

Subedi, B & Singh, S 2008, 'Ecosystem services of Forests in Nepal and Uttarakhand Himalayas: A few observations based on a Pilot Study', in M Bhatnagar (ed.),

Payment for environmental services: Some concepts and experiences, The Icfai University Press, India.

Subedi, BP 2006, *Linking plant-based enterprises and local communities to biodiversity conservation*, Adroit Publishers, New Delhi, India.

Subedi, BP, Pandey, SS, Pandey, A, Rana, EB, Bhattarai, S, Banskota, TR, Charmakar, S & Tamrakar, R 2010, *Forest carbon stock measurement: Guidelines for measuring carbon stocks in community-managed forests*, ANSAB/ICIMOD/FECOFUN.

Sunam, R 2011, Promoting equity in community forestry: recognition of the marginalized people matters, Forest Action Nepal.

Sunderlin, WD, Hatcher, J & Liddle, M 2008, From exclusion to ownership? Challenges and opportunities in advancing forest tenure reform, Rights and Resources Initiative, Washington, DC.

Sunderlin, WD, Angelsen, A, Belcher, B, Burgers, P, Nasi, R, Santoso, L & Wunder, S 2005, 'Livelihoods, forests, and conservation in developing countries: an overview', *World development*, vol. 33, no. 9, pp. 1383-1402.

Szolgayová, J, Golub, A & Fuss, S 2014, 'Innovation and risk-averse firms: Options on carbon allowances as a hedging tool', *Energy Policy*, vol. 70, no. 0, pp. 227-235.

Taiyab, N 2006, *Exploring the market for voluntary carbon offsets*, International Institute for Environment and Development, London.

Tamrakar, PR 2000, Biomass and volume tables with species description for community forest managment, DFRS/MoFSC, NARMSAP-TISC.

Tan, Z-H, Zhang, Y-P, Schaefer, D, Yu, G-R, Liang, N & Song, Q-H 2011, 'An oldgrowth subtropical Asian evergreen forest as a large carbon sink', *Atmospheric Environment*, vol. 45, no. 8, pp. 1548-1554.

Tan, Z, Zhang, Y, Yu, G, Sha, L, Tang, J, Deng, X & Song, Q 2010, 'Carbon balance of a primary tropical seasonal rain forest', *Journal of Geophysical Research*, vol. 115, no. D4, 10.1029/2009JD012913.

Tanvir, MA, Siddiqui, MT & Shah, AH 2002, 'Growth and Price Trend of *Eucalyptus camaldulensis* in Central Punjab', *International Journal of Agriculture & Biology*, vol. 4, no. 3, pp. 344-346.

Tashakkori, A & Teddlie, C 2003, *Handbook of mixed methods in social & behavioral research*, Sage Publications, Thousand Oaks, California.

Tett, SF, Jones, GS, Stott, PA, Hill, DC, Mitchell, JF, Allen, MR, Ingram, WJ, Johns, TC, Johnson, CE & Jones, A 2002, 'Estimation of natural and anthropogenic contributions to twentieth century temperature change', *Journal of Geophysical Research*, vol. 107, no. D16, 10.1029/2000JD000028.

Thapa, GB & Paudel, GS 2000, 'Evaluation of the livestock carrying capacity of land resources in the Hills of Nepal based on total digestive nutrient analysis', *Agriculture, ecosystems & environment*, vol. 78, no. 3, pp. 223-235.

World Bank 2011a, *Migration and remittances factbook*, The World Bank, Washington DC.

World Bank 2011b, Large-scale Migration and Remittances in Nepal: Issues, Challenges and Opportunities. Kathmandu: Poverty Reduction and Economic Management Sector Unit, South Asia Region, Report No. 55390-NP, The World Bank, Washington DC.

The World Bank 2011c, *Fossil Fuel energy Consuption (% of total)*, The World Bank, Washington DC. Details are available from the website <<u>http://data.worldbank.org/indicator/EG.USE.COMM.FO.ZS</u>>

Thomas, CD, Cameron, A, Green, RE, Bakkenes, M, Beaumont, LJ, Collingham, YC, Erasmus, BF, De Siqueira, MF, Grainger, A & Hannah, L 2004, 'Extinction risk from climate change', *Nature*, vol. 427, no. 6970, pp. 145-148.

Thomas, S, Dargusch, P, Harrison, S & Herbohn, J 2010, 'Why are there so few afforestation and reforestation Clean Development Mechanism projects?', *Land Use Policy*, vol. 27, no. 3, pp. 880-7.

Thomas, SC, Halpern, CB, Falk, DA, Liguori, DA & Austin, KA 1999, 'Plant diversity in managed forests: understory responses to thinning and fertilization', *Ecological Applications*, vol. 9, no. 3, pp. 864-879.

Thompson, IS 1990, 'Biomass tables and inventory', *Banko Janakari*, vol. 2, no. 4, pp. 356-362.

Thompson, IS, Tamraker, PR & Mathema, P 1990, 'Community forestry: some indication of yield in katus/chilaune forest', *Banko Janakari* vol. 2, no. 4, pp. 341-344.

Thompson, MC, Baruah, M & Carr, ER 2011, 'Seeing REDD+ as a project of environmental governance', *Environmental Science & Policy*, vol. 14, no. 2, pp. 100-110.

Thoms, CA 2008, 'Community control of resources and the challenge of improving local livelihoods: A critical examination of community forestry in Nepal', *Geoforum*, vol. 39, no. 3, pp. 1452-1465.

Thorne, PJ & Tanner, JC 2002, 'Livestock and nutrient cycling in crop–animal systems in Asia', *Agricultural Systems*, vol. 71, no. 1–2, pp. 111-126.

Thornley, J & Cannell, M 2000, 'Managing forests for wood yield and carbon storage: a theoretical study', *Tree physiology*, vol. 20, no. 7, p. 477.

Timsina, N 2005, *Supporting livelihoods through employment: The Chaubas-Bhumlu community sawmill, Nepal*, ITTO, Forest Trends, RECOFTC, Rights and Resource.

Titlyanova, A, Romanova, I, Kosykh, N & Mironycheva-Tokareva, N 1999, 'Pattern and process in above ground and below ground components of grassland ecosystems', *Journal of Vegetation Science*, vol. 10, no. 3, pp. 307-320.

Tol, RS 2008, 'The social cost of carbon: trends, outliers and catastrophes', *Economics: The Open-Access, Open-Assessment E-Journal*, vol. 2, no. 2008-25, pp. 1-22.

Troup, RS 1921, 'The silviculture of Indian trees, Vol. 1', *The Silviculture of Indian Trees, Vol. 1*, pp. 785-1172.

Tulachan, PM & Neupane, A 1999, *Livestock in mixed farming systems of the Hindu Kush-Himalayas, Trends and Sustainability.* ICIMOD, Kathmandu, Nepal, p. 116.

Turner, DP, Koerper, GJ, Harmon, ME & Lee, JJ 1995, 'A carbon budget for forests of the conterminous United States', *Ecological Applications*, vol. 5, no. 2, pp. 421-436.

Tyagp, K, Sharma, S & Tyagi, P 2009, 'Development of biomass and productivity in an age series of Dalbergia sissoo plantations in sodic lands of Uttar Pradesh', *Annal of Forestry*, vol. 17, no. 2, pp. 219-233,

UNFCCC 2007, 'Report on the Conference of the parties on its thirteenth sessions, held in Bali from 3 to 15 December 2007.

UNFCCC 2010, 'Copenhagen accord', http://unfccc.int/resource/docs/2009/cop15/eng/11a01.pdf (accessed 20, September 2011).

UNFCCC 2011, 'Report of the conference of the parties on its sixteenth session, held in Cancun from 29 November to 10 December 2010 ', http://unfccc.int/files/meetings/cop_16/application/pdf/cop16_lca.pdf (accessed 26 June 2014).

UNFCCC 2014, UNFCCC report of the conference of the parties on its nineteenth session, held in Warsaw from 11 to 23 November 2013 Bonn: Germany (2014)

UNFCCC 2014, Statistics of the CDM projects (end of June 2014), UNFCCC.

UNFF 2011, United nations forum on forests: fact sheet.

United-Nations 2007, The millennium development goals report 2007. New York: United Nations, Department of Economic and Social Affairs.

UN-REDD 2012, Guidelines on stakeholder engagement in REDD+ readiness with a focus on the participation of indigenous peoples and other forest-dependent communities. UN-REDD Programme, Secretariate, Geneva

UN-REDD 2014, About the UN-REDD programmes, UN-REDD+ programme.

Usuga, JCL, Toro, JAR, Alzate, MVR & de Jesús Lema Tapias, Á 2010, 'Estimation of biomass and carbon stocks in plants, soil and forest floor in different tropical forests', *Forest Ecology and Management*, vol. 260, no. 10, pp. 1906-1913.

Vaganov, E, Hughes, M, Kirdyanov, A, Schweingruber, F & Silkin, P 1999, 'Influence of snowfall and melt timing on tree growth in subarctic Eurasia', *Nature*, vol. 400, no. 6740, pp. 149-151.

Van der Werf, G, Morton, DC, DeFries, RS, Olivier, JGJ, Kasibhatla, PS, Jackson, RB, Collatz, G & Randerson, J 2009, 'CO₂ emissions from forest loss', *Nature Geoscience*, vol. 2, no. 11, pp. 737-378.

Van Kanten, R, Schroth, G, Beer, J & Jiménez, F 2005, 'Fine-root dynamics of coffee in association with two shade trees in Costa Rica', *Agroforestry Systems*, vol. 63, no. 3, pp. 247-261.

Van Leeuwen, M, Hilker, T, Coops, NC, Frazer, G, Wulder, MA, Newnham, GJ & Culvenor, DS 2011, 'Assessment of standing wood and fiber quality using ground and airborne laser scanning: A review', *Forest Ecology and Management*, vol. 261, no. 9, pp. 1467-1478.

Vanclay, JK & Skovsgaard, JP 1997, 'Evaluating forest growth models', *Ecological Modelling*, vol. 98, no. 1, pp. 1-12.

Varughese, G & Ostrom, E 2001, 'The contested role of heterogeneity in collective action: Some evidence from community forestry in Nepal', *World development*, vol. 29, no. 5, pp. 747-766.

Vatn, A & Angelsen, A 2009, 'Options for a national REDD+ architecture', Angelsen, A. with Brockhaus, M., Kanninen, M., Sills, E., Sunderlin, WD and Wertz-Kanounnikoff, S.(eds.): Realising REDD+ national strategy and policy options. CIFOR, Bogor (Indonesia), pp. 57-74.

Vedeld, P, Angelsen, A, Bojö, J, Sjaastad, E & Kobugabe Berg, G 2007, 'Forest environmental incomes and the rural poor', *Forest Policy and Economics*, vol. 9, no. 7, pp. 869-879.

Verwijst, T & Telenius, B 1999, 'Biomass estimation procedures in short rotation forestry', *Forest Ecology and Management*, vol. 121, no. 1, pp. 137-146,

Vetaas, OR & Grytnes, JA 2002, 'Distribution of vascular plant species richness and endemic richness along the Himalayan elevation gradient in Nepal', *Global ecology and biogeography*, vol. 11, no. 4, pp. 291-301.

Victor, D, Zhou, D, Ahmed, EHM, Dadhich, PK, Olivier, J, Rogner, H-H, Sheikho, K & Yamaguchi, M 2014, 'Introductory chapter', in O Edenhofer, R Pichs-Madruga, Y Sokona, E Farahani, S Kadner, K Seyboth, A Adler, I Baum, S Brunner, P Eickemeier, B Kriemann, J Savolainen, S Schlömer, C von-Stechow, T Zwickel & JC Minx (eds), *Climate change 2014: Mitigation of climate change. Contribution of working group III to the fifth assessment report of the Intergovernmental Panel on Climate Change* Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA. Figure 1.X.

Vinton, MA, Hartnett, DC, Finck, EJ & Briggs, JM 1993, 'Interactive effects of fire, bison (Bison bison) grazing and plant community composition in tallgrass prairie', *American Midland Naturalist*, pp. 10-18.

Visseren-Hamakers, IJ, McDermott, C, Vijge, MJ & Cashore, B 2012, 'Trade-offs, co-benefits and safeguards: current debates on the breadth of REDD+', *Current Opinion in Environmental Sustainability*, vol. 4, no. 6, pp. 646-653.

Vogt, KA, Vogt, DJ & Bloomfield, J 1998, 'Analysis of some direct and indirect methods for estimating root biomass and production of forests at an ecosystem level', in *Root Demographics and Their Efficiencies in Sustainable Agriculture, Grasslands and Forest Ecosystems*, Springer, pp. 687-720.

Waddell, KL 2002, 'Sampling coarse woody debris for multiple attributes in extensive resource inventories', *Ecological Indicators*, vol. 1, no. 3, pp. 139-153.

Walther, GR, Post, E, Convey, P, Menzel, A, Parmesan, C, Beebee, TJC, Fromentin, JM, Hoegh-Guldberg, O & Bairlein, F 2002, 'Ecological responses to recent climate change', *Nature*, vol. 416, no. 6879, pp. 389-395.

Wang, X, Fang, J & Zhu, B 2008, 'Forest biomass and root-shoot allocation in northeast China', *Forest Ecology and Management*, vol. 255, no. 12, pp. 4007-4020.

WECS 2010, 'Energy sector synopsis report 2010', Water and Energy Commission Secretariat, Kathmandu, Nepal.

Weiner, J & Thomas, SC 2001, "The nature of tree growth and the" age-related decline in forest productivity", *Oikos*, pp. 374-376.

Weiskittel, AR, Hann, DW, Kershaw Jr, JA & Vanclay, JK 2011, *Forest growth and yield modeling*, John Wiley & Sons.

Westholm, L 2010, Getting ready for REDD, Gothenburg.

White, A & Martin, A 2002, *Who owns the world's forests? Forest tenure and public forests in transition*, Washington, DC, USA.

WHO 2006, *Fuel for life : household energy and health*, World Health Organization, 20 Avenue Appia, 1211 Geneva 27, Switzerland.
Woerdman, E 2000, 'Implementing the Kyoto protocol: why JI and CDM show more promise than international emissions trading', *Energy Policy*, vol. 28, no. 1, pp. 29-38.

Woo, J, Kay, R & Nicholls, M 1991, 'Environmental temperature and stroke in a subtropical climate', *Neuroepidemiology*, vol. 10, no. 5-6, pp. 260-265.

World-Bank 2013, *Mapping carbon pricing initiatives: Developments and prospects 2013*, ECOFYS/World Bank, Washington DC, May 2013.

Wright, SJ & Cornejo, FH 1990, 'Seasonal drought and leaf fall in a tropical forest', *Ecology*, pp. 1165-1175.

Wunder, S 2001, 'Poverty alleviation and tropical forests—What scope for synergies?', *World development*, vol. 29, no. 11, pp. 1817-1833.

Yadav, NP, Dev, OP, Springate-Baginski, O & Soussan, J 2003, 'Forest management and utilization under community forestry', *Journal of Forest and Livelihood*, vol. 3, no. 1, pp. 37-50.

Yang, Y, Watanabe, M, Li, F, Zhang, J, Zhang, W & Zhai, J 2006, 'Factors affecting forest growth and possible effects of climate change in the Taihang Mountains, northern China', *Forestry*, vol. 79, no. 1, pp. 135-147.

Zanchi, G, Frieden, D, Pucker, J, Bird, DN, Buchholz, T & Windhorst, K 2013, 'Climate benefits from alternative energy uses of biomass plantations in Uganda', *Biomass and Bioenergy*, vol. 59, no. 0, pp. 128-136.

Zhang, J, Chu, Z, Ge, Y, Zhou, X, Jiang, H, Chang, J, Peng, C, Zheng, J, Jiang, B, Zhu, J & Yu, S 2008, 'TRIPLEX model testing and application for predicting forest growth and biomass production in the subtropical forest zone of China's Zhejiang Province', *Ecological Modelling*, vol. 219, no. 3-4, pp. 264-275.

Zhao, M & Zhou, G-S 2006, 'Carbon storage of forest vegetation in China and its relationship with climatic factors', *Climatic change*, vol. 74, no. 1-3, pp. 175-189.

Zhou, Z, Wu, W, Wang, X, Chen, Q & Wang, O 2009, 'Analysis of changes in the structure of rural household energy consumption in northern China: A case study', *Renewable and Sustainable Energy Reviews*, vol. 13, no. 1, pp. 187-193,

APPENDICES

Appendix A: Checklist for key informant survey

This checklist used for key informant consultation to identify factors affecting carbon stock and change in carbon stock

- 1. Encroachment and expansion of agriculture land
- 2. Shifting cultivation
- 3. Frequency of biomass reducing factors (forest fire, timber harvesting, firewood collection, fodder collection, grass collection, grazing)
- 4. Weather condition
- 5. Climate change
- 6. Resettlement program
- 7. Infrastructure development
- 8. Elevation of forests from mean sea level
- 9. Proximity of forests from motor-able road and from settlement
- 10. Average age of dominant trees in forest stands
- 11. Per capita forest areas
- 12. Caste heterogeneity
- 13. Household level landholding
- 14. Average change in livestock
- 15. Biogas use
- 16. Petroleum energy use
- 17. Biomass extraction
- 18. Grazing
- 19. Quantity of grass and fodder collection
- 20. Fodder and grass collection
- 21. Litter collection
- 22. Others

Appendix B: Checklist used to collect forest related information from documents available with CFUGs

Descriptions

1.	Hand over date (date when CF was legally initiated)
2.	Population (Total in CFUG)
3.	Forest condition before CF and REDD+
4.	Household numbers (total and caste-wise)
5.	Average age of forests (average age of dominant trees in CF)
6.	Meeting and general assembly (Frequency in year and participation numbers)
7.	Income sources
8.	Sources of forest products supply
9.	Harvesting practices
10.	Average agricultural land holding size per household in CFUG
11.	Alternative energy (type of energy and adopting households number) use in CFUG
12.	Average live-stock numbers per HH (buffalo, cow, goat)

Benefits	(Quantity	6/7	7/8	8/9	9/10	10/11	11/12	Price/unit
extraction	n)							(2012)
a.	Timber							
	(species-							
	wise							
	quantity)							
b.	Grasses							
с.	Fodder							
d.	Litter							
e.	Fuel wood							
	(species-							
	wise							
	quantity)							
f.	NTFPs (total							
	income)							
g.	Grazing							
	(cow,							
	buffalo,							
	goat)							
Costs (la	bour unit and							
other cos	t in monetary							
value)	-							
A.	General							
	assembly							
B.	Executive							
	committee							
	meeting							
C.	Plantation							
D.	Forest fire							
	control							
E.	Silviculture							
	operation							
F.	Guarding							
G.	Harvesting							

Appendix C: Checklist used to collect costs and benefits data from documents available with CFUGs

Appendix D: Checklist of the questions asked in group discussion

Key questions

1.	How was lorest status before CF?
2.	Why do you want to protect forests? (reasons)
3.	Do your group generate incomes from forest? What do you do with
	forest incomes?
4.	What benefits are you getting from forests?
5.	What are the major sources of income of community members?
6.	What was labour rate in 2012? (per man-day)
7.	How do you make forest management related decisions?
8.	Which are the timber tree species and preferred timber tree species
	available in your forests?
9.	How communities are fulfilling their forest product needs? (supply
	sources of forest resources)
10.	What do you know about the REDD+?
11.	What changes have been made after the REDD+ project activities?
	Particularly forest product use practices (timber, fire wood, fodder,
	grass, fodder and NTFPs), decision making processes (meeting
	frequency and participation), forest improvement activities (guarding,
	plantation and silviculture operation) and forest degradation activities
	(fire, grazing, illegal harvesting)
12.	How much grass is required for each livestock (i.e. for one cow, one
	buffalo, one goat) during grazing day and non-grazing day?
13.	What have you done to increase carbon stocks in CFs?
14.	What is the perspective benefits of the REDD+?
15.	What is important to make the REDD+ incentive mechanism more
	beneficial at community level for long term benefit?
16.	What is the observed age of the forests? (based on dominant trees)
17.	In your opinion, which forests (i.e. nearby motor able road or far
	from the road; nearby settlement or far from the settlement) is better
	(in term of stock and bigger trees in unit area).

- How was forest status before CE? 1
- 2.
- 3. with
- 4.
- 5)
- 6.
- 7
- 8. cies
- 9. pply
- 1(

- 15 ore 16 ees)
- 17 ar better

Appendix E: A format used to collect forest resource inventory data in each sample plot

(Source: Subedi et al. 2010)

1. Dackground miormation		
Name of CFUG:	Date:	Referen ces of plot
Name of forest (based on species):	Forest type: Natural/Plantation	
Forest area: Natural forestha. Plantation forestha.	Strata type:	
Name of forest block:		
Plot no: Name of data recorder:	GPS coordinates:	

1. Background information

2. Plot information

Altitude (m):		
Forest type (Major	Forest disturbar	ce incidence
spp):	Forest fire symbol: yes/no	Fodder collection: yes/no
	Grazing: yes/no	Firewood collection: yes/no
	Timber collection: yes/no	Forest encroachment: yes/no
Dense/Open	Other information:	

		Diameter	Angle f Height	rom eye (⁰)	Distance	Slope	Tree
S.No.	Ѕрр	(cm)	Top angle	Bottom angle	between standing point to tree (m)	angle(⁰)	outward appearance
1							
2							
3							
4							
5							
6							
7							
8							
9							
10							
11							
12							
13							
14							

3. Trees (> 5 cm DBH) measurement in 8.92 m circular plot

4. Saplings (1-5 cm DBH) measurement in 5.64 m cicular plot

S.No.	Species	Diameter in cm at breast height	Height (m)
1			
2			
3			
4			
5			
6			
7			
8			
9			
10			
11			
12			
13			
14			

S.No	Species	Total no.]	S.No	Species	Total no.
1				8		
2				9		
3				10		
4				11		
5				12		
6				13		
7				14		

5. Regenerations counted in 1m cicular plots

6. Information to be collected within 0.56 m radius subplot

Herbs and grasses to b woody plants < 1 cm d	e weighed within 0.56 m radis sub plot (all non iameter)
Sample packet ID	Total weight of herbs and grasses in plot (Kg):
(100 sample to lab at	
first year):	
Leaf litters to be weigh	ed within 0.56 m radius sub plot
Sample packet ID	Total weight of leaf litters in plot (Kg):
(100 sample to lab at	
first year):	

			S	apling Bi (Mg /ł	omass 1a)							Tree Bio (Mg/	omass ha)				I	Litter B (Mg	iomass /ha)	s]	Oth Bion (Mg)	er hass /ha)]	Total Bioma Mg / h	ss a)
Canopy	2	010	20	11	20	12	20	13	2	010	20	11	20	12	20	13											
cover	* G S B	**B G SB	A G S B	B G S B	A G S B	B G S B	A G S B	B G S B	# A G T B	##B G TB	A G T B	B G T B	A G T B	B G T B	A G T B	B G T B	2010	2011	2012	2013	2010	2011	2012	2013	2010	2011	2013
Dense Shorea forests	4.6 (4.5)	1.2 (1.2)	5.7 (4.6)	1.5 (1.2)	5.2 (4.7)	1.4 (1.3)	4.4 (3.2)	1.2 (0.9)	209.7(140.6)	56.7 (38.1)	224.6(139.2)	60.7(37.7)	231.1(135.5)	62.5 (36.7)	237.6(133.4)	64.2 (36.2)	4.0 (3.9)	4.9 (2.7)	4.7 (2.5)	4.6 (3.4)	0.2 (0.3)	0.2 (0.3)	0.3(0.3)	0.4 (0.5)	276.4 (177.8)	297.6 (176.2) 305.1 (172.0)	312.3 (169.3)
Sparse Shorea forests	4.8 (4.6)	1.3 (1.2)	6.9 (5.3)	1.9 (1.4)	5.1 (3.6)	1.4 (1.0)	5.5 (4.9)	1.5 (1.3)	142.6(100.1)	38.5 (27.0)	146.6 (103.5)	39.6 (28.0)	150.1(98.2)	40.5 (26.5)	153.9(93.4)	41.6 (25.2)	2.2 (2.6)	5.3 (2.6)	5.0 (2.7)	5.1 (3.8)	0.3 (0.3)	0.3 (0.3)	0.4~(0.3)	0.5(0.5)	189.7 (127.3)	200.6 (130.7)	208.1 (119.6)
Total Shorea	4.7 (4.5)	1.3 (1.20	6.0 (4.8)	1.6 (1.3)	5.2 (4.5)	1.4 (1.2)	4.6 (3.6)	1.3 (1.0)	195.3(135.6)	52.8 (36.7)	207.9 (136.0)	56.2 (36.8)	213.7 (132.5)	57.8 (35.9)	219.7 (130.3)	59.4 (35.3)	3.6 (3.7)	5.0 (2.7)	4.8 (2.5)	4.7 (3.5)	0.2 (0.3)	0.2 (0.3)	0.3 (0.3)	0.4(0.5)	257.8 (171.7)	2/6.9 (1/1.9) 283 1 (168 1)	290.0 (165.4)
Dense Mixed broad leaf forests	5.8 (4.0)	1.6 (1.1)	6.3 (3.5)	1.7 (0.9)	4.7 (3.4)	1.3 (0.9)	5.9 (3.1)	1.6(0.8)	189.1 (140.3)	51.6 (38.6)	203.1 (141.5)	55.4 (39.1)	209.4 (138.7)	57.1 (38.2)	218.5 (141.5)	59.6 (38.9)	2.6 (3.6)	4.3 (2.4)	3.7 (2.4)	5.1 (4.0)	0.4 (0.4)	0.3 (0.2)	0.4(0.3)	0.7~(0.5)	251.1 (179.7)	(180.8) (180.8)	291.3 (179.5)

Appendix F: Comparison of biomass (both above ground and below ground) using four year measurement data by vegetation types and canopy strata of CFs in the study areas

Sparse Pine forests	Dense Pine forests	Total Schima- Castanops is forests	Sparse Schima- Castanops is forests	Dense Schima- Castanops is forests	Total Mixed broad leaf forests	Sparse Mixed broad leaf forests
2.0 (2.2)	3.1 (5.2)	4.8 (3.9)	4.7 (4.5)	4.8 (3.8)	5.9 (4.1)	6.4 (4.6)
0.6 (0.6)	0.9 (1.5)	1.3 (1.1)	1.3 (1.2)	1.3 (1.1)	1.6 (1.1)	1.7 (1.2)
3.8 (4.1)	2.9 (3.6)	6.4 (5.0)	4.6 (3.5)	6.7 (5.2)	5.9 (3.5)	4.2 (3.6)
1.0 (1.1)	0.8 (1.0)	1.8 (1.4)	1.3 (0.9)	1.8 (1.4)	1.6 (1.0)	1.1 (1.0)
3.7 (3.2)	4.2 (5.3)	6.2 (5.3)	4.5 (3.7)	6.5 (5.4)	4.5 (3.2)	3.2 (1.7)
1.0 (0.9)	1.2 (1.5)	1.7 (1.4)	1.2 (1.0)	1.8 (1.5)	1.2 (0.9)	0.9 (0.5)
5.1 (4.9)	3.9 (5.7)	6.1 (5.6)	5.0 (6.0)	6.3 (5.5)	5.8 (3.1)	5.3 (3.0)
1.4 (1.4)	1.1 (1.6)	1.7 (1.5)	1.4 (1.6)	1.8 (1.5)	1.6 (0.8)	1.4 (0.8)
101.3 (79.2)	165.1 (94.9)	139.3 (99.9)	76.0 (60.3)	150.9 (101.6)	176.3(136.1)	108.7 (88.1)
28.5 (23.9)	46.9 (28.0)	38.5 (27.9)	20.7 (16.4)	41.7 (28.4)	48.1 (37.7)	29.4 923.8)
97.3 (75.2)	183.1 (93.0)	152.1 (102.8)	79.4 (64.4)	165.5 (103.1)	188.0 (138.1)	108.4 (86.2)
27.6 (22.8)	51.9 (27.4)	42.0 (28.6)	21.6 (17.4)	45.7 (28.8)	51.3 (38.1)	29.3 (23.3)
97.9 (76.4)	196.5 (91.1)	157.5 (104.3)	74.5 (56.5)	172.8 (104.0)	193.5 (135.4)	109.7 (77.0)
27.8 (23.1)	55.8 (26.9)	43.4 (28.9)	20.3 (15.3)	47.7 (28.8)	52.8 (37.3)	29.6 (20.8)
111.9 (77.6)	203.0 (90.0)	168.0 (104.5)	83.4 (58.5)	183.6(103.8)	203.2 (137.9)	121.9 (80.5)
31.0 (23.3)	57.5 (26.6)	46.3 (29.1)	22.7 (15.8)	50.7 (28.9)	55.4 (37.9)	32.9 (21.7)
1.7 (2.2)	2.9 (3.8)	3.0 (3.9)	1.2 (2.1)	3.3 (4.1)	2.4 (3.4)	1.1 (2.1)
2.5 (2.9)	4.4 (3.5)	4.4 (2.8)	3.1 (2.8)	4.6 (2.8)	4.2 (2.4)	3.6 (2.6)
4.6 (2.4)	4.6 (3.2)	4.2 (2.6)	3.3 (2.0)	4.4 (2.6)	3.7 (2.3)	3.6 (1.2)
2.9 (3.3)	4.3 (4.4)	3.7 (3.9)	2.2 (2.6)	4.0 (4.0)	5.2 (3.8)	6.0 (2.6)
0.3~(0.3)	0.4 (0.4)	0.4 (0.4)	0.4 (0.4)	0.4 (0.4)	0.4 (0.4)	0.5(0.4)
0.6(0.4)	0.5 (0.4)	0.4 (0.3)	0.5(0.3)	0.4(0.3)	0.3 (0.3)	0.3 (0.3)
0.9 (0.5)	0.5 (0.4)	0.4 (0.4)	0.5 (0.4)	0.4 (0.4)	0.4 (0.3)	0.3 (0.3)
0.3 (0.3) 132.8 (102.8)	0.5 (0.4) 219.2 (120.7)	(C.U C.U (0.82.0) (128.0)	(C.U) C.U (74.7)	0.0) 0.0) 202.5 (130.0)	(c.0) /.0 234.7 (174.3)	0.7 (0.7) 147.8 (113.3)
137.0 (98.6)	243.6 (120.5)	207.0 (131.6)	110.5 (83.3)	224.7 (131.4)	251.3 (176.7)	146.8 (109.3)
144.4 (105.2)	262.8 (116.9)	215.5 (134.2)	117.7 (97.7)	233.5 (132.6)	256.1 (173.2)	147.3 (97.3)
150.0 (98.9)	270.1 (115.5)	226.5 (133.9)	115.1 (74.9)	246.9 (132.5)	271.8 (175.1)	168.3 (103.9)

Total	Total sparse	Total dense	Total Rhododen dron- Quercus forests	Sparse Rhododen dron- Quercus forests	Dense Rhododen dron- Quercus forests	Total <i>Pine</i> forests
4.9 (4.5)	4.6 (4.3)	5.0 (4.6)	6.1 (5.3)	5.0 (4.2)	6.3 (5.5)	2.8 (4.5)
1.4 (1.3)	1.3 (1.2)	1.4 (1.3)	1.8 (1.6)	1.5 (1.2)	(7.1) 6.1	0.8 (1.3)
6.0 (4.8)	5.6 (4.6)	6.1 (4.8)	7.1 (5.4)	5.4 (3.7)	7.5 (5.6)	3.2 (3.8)
1.7 (1.3)	1.5 (1.3)	1.7 (1.4)	2.1 (1.6)	1.6 (1.1)	2.2 (1.7)	0.9 (1.0)
5.5 (4.8)	4.9 (3.8)	5.6 (5.1)	6.9 (5.9)	7.0 (5.5)	6.8 (6.0)	4.1 (4.7)
1.5 (1.4)	1.3 (1.1)	1.6 (1.4)	2.0 91.8)	2.1 (1.6)	2.0 (1.8)	1.2 (1.4)
5.5 (4.8)	5.7 (5.2)	5.5 (4.7)	7.3 (6.1)	7.8 (6.4)	7.2 (6.1)	4.3 (5.4)
1.5 (1.3)	1.6 (1.4)	1.5 (1.3)	2.2 (1.8)	2.3 (1.8)	2.2 (1.8)	1.2 (1.5)
170.1(123.5)	111.9 (90.5)	184.1 (126.4)	159.2 (114.7)	74.4 (79.6)	177.6 (113.4)	146.8 (94.4)
47.0 (34.2)	30.6 (24.7)	51.0 (35.0)	47.3 (34.2)	21.3 (22.6)	52.9 (33.8)	41.7 (27.9)
181.6 (125.4)	113.4 (93.0)	198.0 (126.7)	165.4 (119.6)	71.5 (81.9)	185.8 (117.1)	158.6 (95.8)
50.2 (34.7)	31.0 (25.4)	54.8(35.1)	49.2 (35.6)	20.4 (23.1)	55.4 (34.9)	45.0 (28.1)
186.5 (123.5)	114.0 (89.7)	203.9 (124.2)	163.8 (115.4)	70.6 (81.2)	183.9 (112.2)	168.3 (97.3)
51.5 (34.1)	31.1 (24.5)	56.4 (34.3)	48.6 (34.3)	20.2 (22.8)	54.8 (33.4)	47.8 (28.6)
194.0 (122.3)	121.7 (87.4)	211.4 (123.1)	168.4 (112.9)	79.3 (83.1)	187.6 (109.7)	177.0 (95.3)
53.6 (33.8)	33.2 (23.9)	58.5 (34.0)	50.0 (33.6)	22.6 (23.3)	55.9 (32.6)	49.9 (28.2)
3.3 (3.8)	1.6 (2.3)	3.7 (4.0)	4.0 (4.5)	0.2 (0.9)	4.8 (4.5)	2.5 (3.4)
4.7 (3.0)	4.2 (3.0)	4.8 (2.9)	5.6 (3.7)	4.3 (3.8)	5.9 (3.7)	3.8 (3.4)
4.4 (2.6)	4.1 (2.4)	4.5 (2.7)	4.5 (2.9)	2.6 (2.1)	4.9 (2.9)	4.6 (3.0)
4.5 (3.9)	4.1 (3.5)	4.6 (3.9)	5.1 (4.4)	3.0 (3.0)	5.5 (4.6)	3.9 (4.1)
0.3 (0.3)	0.3~(0.3)	0.3 (0.4)	0.4 (0.4)	0.3 (0.2)	0.5(0.4)	0.3~(0.3)
0.3 (0.3)	0.4 (0.3)	0.3 (0.3)	0.4 (0.3)	0.3 (0.4)	0.4 (0.3)	0.5 (0.4)
0.4 (0.3)	0.4 (0.4)	0.3 (0.3)	0.3 (0.3)	0.2 (0.4)	0.3 (0.3)	0.6 (0.5)
0.5 (0.5)	0.5 (0.5)	0.5 (0.5)	0.4 (0.6)	0.2 (0.4)	0.5 (0.7)	0.3 (0.4) 104 5 (101 2)
244.6 (159.7)	(1.0.1) (14) (119.1) (119.1) (119.1)	265.7 (161.1)	229.7 (155.2)	102.5 (107.5)	257.1 (151.0)	213.1 (123.5)
250.5 (157.3)	159.6 (117.3)	272.4 (158.0)	226.1 (149.7)	102.7 (105.2)	252.9 (144.9)	229.0 (124.7)
259.5 (155.6)	166.3 (112.3)	282.0 (156.3)	233.3 (146.0)	115.3 (108.2)	258.9 (141.0)	235.8 (122.8)

*AGSB- Above Ground Sapling Biomass, **BGSB-Below Ground Sapling Biomass, #AGTB- Above Ground Tree Biomass, ## Below Ground Tree Biomass, N= number of plots

CF name	А	В	С	D	E	F	G	Н	Ι	J	K	L	Μ	Ν	0	Р	Q	R	S	Т	U
AamswaraBhawa																					
nipakha_G	9.1	4.2	4.9	0.0	0.0	0.0	0.0	891.7	950.6	964.5	999.0	59.0	13.8	34.5	6.5	1.5	3.8	11.7	97.6	3.9	1
Amalekharka_D	6.6	0.0	0.0	1.5	5.1	0.0	0.0	452.3	482.8	526.3	546.2	30.5	43.5	19.9	4.6	6.6	3.0	14.2	68.6	4.7	2
Badahare_G	25.8	10.7	15.1	0.0	0.0	0.0	0.0	2454.6	2616.5	2652.7	2748.9	161.9	36.2	96.2	6.3	1.4	3.7	11.4	95.2	3.8	1
Bagpani_G	68.2	67.6	0.6	0.0	0.0	0.0	0.0	8492.4	9067.6	9256.0	9549.5	575.2	188.5	293.5	8.4	2.8	4.3	15.5	124.6	5.2	1
Barshedadapari_ D	35.4	0.0	0.0	11.6	23.8	0.0	0.0	2511.2	2723.9	2961.1	3085.3	212.7	237.2	124.1	6.0	6.7	3.5	16.2	70.9	5.4	3
Batauli_C	155.8	91.3	64.5	0.0	0.0	0.0	0.0	16193.5	17271.9	17553.7	18161.5	1078.4	281.9	607.8	6.9	1.8	3.9	12.6	104.0	4.2	1
BhageristanGhant ari_G	5.2	3.2	2.1	0.0	0.0	0.0	0.0	548.7	585.3	595.0	615.5	36.6	9.7	20.5	7.0	1.8	3.9	12.7	104.8	4.2	1
Bhakare_D	104.4	0.0	0.0	76.3	28.2	0.0	0.0	8437.8	9654.2	10401.4	10974.0	1216.5	747.2	572.6	11.6	7.2	5.5	24.3	80.8	8.1	2
BhalukholaSoti_ G	107.6	53.5	1.1	53.0	0.0	0.0	0.0	11403.1	12679.3	13224.1	13819.4	1276.2	544.8	595.3	11.9	5.1	5.5	22.5	106.0	7.5	2
Bhasmepakha_D	10.9	0.0	0.0	6.2	4.8	0.0	0.0	838.6	940.3	1016.5	1067.5	101.8	76.2	51.0	9.3	7.0	4.7	20.9	76.7	7.0	2
BhirmuniDevitha n_D	6.0	0.0	0.0	0.0	6.0	0.0	0.0	376.3	384.8	422.7	434.0	8.4	37.9	11.3	1.4	6.3	1.9	9.6	62.9	3.2	4
Bhiteripakha_D	542.6	0.0	0.0	0.0	0.0	377.7	165.0	53620.1	55977.4	56192.6	57951.5	2357.4	215.1	1759.0	4.3	0.4	3.2	8.0	98.8	2.7	2
Bhumethan_D	46.7	0.0	0.0	15.9	30.0	0.8	0.0	3369.8	3663.1	3971.7	4139.1	293.3	308.6	167.5	6.3	6.6	3.6	16.5	72.2	5.5	4
Bichaur_D	47.7	0.0	0.0	7.3	40.4	0.0	0.0	3180.9	3350.4	3660.8	3787.1	169.5	310.4	126.3	3.6	6.5	2.6	12.7	66.7	4.2	2
Birenchok_G	83.6	69.9	13.7	0.0	0.0	0.0	0.0	9753.3	10410.1	10610.7	10957.6	656.8	200.5	347.0	7.9	2.4	4.2	14.4	116.7	4.8	1
Boldesetidevi_D	172.1	0.0	0.0	0.0	0.0	113.7	58.4	16673.4	17363.1	17432.0	17993.9	689.6	68.9	561.9	4.0	0.4	3.3	7.7	96.9	2.6	3
Budhabhimsen_D	42.0	0.0	0.0	9.1	32.9	0.0	0.0	2864.4	3051.5	3327.6	3452.0	187.1	276.1	124.4	4.5	6.6	3.0	14.0	68.2	4.7	2
Charnawati_D	819.4	0.0	0.0	0.0	73.6	733.7	12.1	90056.9	95444.6	96183.6	98597.9	5387.7	739.0	2414.4	6.6	0.9	2.9	10.4	109.9	3.5	2
Charnawati_DFas ku	55.1	29.1	0.0	14.0	12.0	0.0	0.0	5620.6	6100.4	6361.9	6605.7	479.8	261.6	243.8	8.7	4.7	4.4	17.9	102.0	6.0	2
Chelibeti_C	64.8	59.6	6.2	0.0	0.0	0.0	0.0	7911.9	8446.2	8614.9	8892.6	534.2	168.7	277.7	8.2	2.6	4.3	15.1	122.1	5.0	1
Chisapani_G	50.0	45.2	4.9	0.0	0.0	0.0	0.0	6008.0	6413.6	6541.5	6752.5	405.6	127.9	211.0	8.1	2.6	4.2	14.9	120.1	5.0	1
Chitadhunda_D	51.5	0.0	0.0	0.0	51.5	0.0	0.0	3241.8	3314.5	3640.8	3738.4	72.7	326.3	97.7	1.4	6.3	1.9	9.6	62.9	3.2	4
Chitramkaminchu	313.9	19.2	14.5	213. 7	66.5	0.0	0.0	26338.7	29961.7	32036.4	33748.5	3623.0	2074.7	1712.1	11.5	6.6	5.5	23.6	83.9	7.9	2

Appendix G: Average carbon stock (MgC/ha) in individual CF in the study areas

li_C																					
Chuchchedhunga																					\mathbf{r}
D	8.9	0.0	0.0	0.0	8.9	0.0	0.0	560.0	572.5	628.9	645.8	12.6	56.4	16.9	1.4	6.3	1.9	9.6	62.9	3.2	
Chyanedada_D	64.9	0.0	0.0	0.0	0.0	33.1	31.8	5749.0	5915.8	5942.8	6161.1	166.8	27.0	218.3	2.6	0.4	3.4	6.4	88.6	2.1	2
ChyaseBhagawati																					2
_D	30.3	0.0	0.0	23.8	6.5	0.0	0.0	2490.9	2867.6	3086.4	3260.9	376.7	218.8	174.5	12.4	7.2	5.8	25.4	82.2	8.5	2
Deujar_C	278.9	52.2	85.3	131. 7	9.5	0.0	0.0	24949.5	27840.3	29064.5	30487.0	2890.8	1224.2	1422.5	10.4	4.4	5.1	19.9	89.5	6.6	2
Devidhunda_C	189.1	152.0	36.1	0.0	0.0	0.0	0.0	21684.8	23143.4	23582.4	24358.1	1458.7	439.0	775.7	7.7	2.3	4.1	14.1	114.7	4.7	1
Devithan_D	43.9	0.0	0.0	14.4	29.6	0.0	0.0	3116.6	3379.9	3674.4	3828.2	263.3	294.5	153.8	6.0	6.7	3.5	16.2	70.9	5.4	2
Dhade_D	29.2	0.0	0.0	0.2	29.0	0.0	0.0	1840.5	1884.1	2069.1	2125.3	43.6	185.0	56.2	1.5	6.3	1.9	9.8	63.1	3.3	2
Dhadesinghadevi																					
_D	343.7	0.0	0.0	0.0	0.0	229.5	114.2	33431.2	34831.9	34969.2	36089.7	1400.8	137.3	1120.5	4.1	0.4	3.3	7.7	97.3	2.6	4
Dharapani_C	147.1	142.2	5.0	0.0	0.0	0.0	0.0	18150.0	19378.2	19776.4	20406.5	1228.2	398.3	630.1	8.3	2.7	4.3	15.3	123.4	5.1	1
Dimal_D	38.2	0.0	0.0	33.2	3.5	1.5	0.0	3293.5	3821.2	4091.6	4328.7	527.7	270.4	237.1	13.8	7.1	6.2	27.1	86.2	9.0	3
Eklepakha_D	197.3	0.0	0.0	0.0	18.3	157.8	21.3	20663.9	21783.7	21967.1	22560.4	1119.8	183.4	593.3	5.7	0.9	3.0	9.6	104.7	3.2	2
GahateBaghkhor_																					-
D	5.5	0.0	0.0	0.2	5.4	0.0	0.0	353.3	363.8	399.1	410.5	10.5	35.3	11.4	1.9	6.4	2.1	10.3	63.7	3.4	2
GairiJungal_D	131.1	0.0	0.0	26.0	0.0	100.0	5.1	14124.6	15237.6	15470.1	15969.8	1113.0	232.5	499.7	8.5	1.8	3.8	14.1	107.7	4.7	2
Gangatepakha_G	173.6	156.8	13.2	0.0	3.7	0.0	0.0	20816.7	22212.7	22678.4	23405.7	1396.0	465.7	727.3	8.0	2.7	4.2	14.9	119.9	5.0	1
GhaledandaRana																					•
khola_G	181.7	131.5	25.0	15.1	10.1	0.0	0.0	20249.5	21728.7	22282.0	23054.1	1479.2	553.3	772.1	8.1	3.0	4.3	15.4	111.5	5.1	2
Goldada_G	46.0	45.6	0.4	0.0	0.0	0.0	0.0	5733.3	6121.6	6248.9	6447.0	388.3	127.3	198.1	8.4	2.8	4.3	15.5	124.7	5.2	1
Golmeswor D	215.2	0.0	0.0	101. 0	114. 2	0.0	0.0	16011.6	17730.5	19206.9	20110.8	1718.8	1476.5	903.8	8.0	6.9	4.2	19.1	74.4	6.4	3
Gothpani D	23.5	0.0	0.0	17.4	6.1	0.0	0.0	1904.0	2181.1	2349.4	2479.4	277.1	168.3	130.0	11.8	7.2	5.5	24.5	81.0	8.2	3
Indreni C	172.2	155.6	11.0	0.0	5.6	0.0	0.0	20622.4	22000.4	22474.0	23192.4	1377.9	473.6	718.4	8.0	2.8	4.2	14.9	119.8	5.0	1
Jamuna C	34.5	10.9	23.7	0.0	0.0	0.0	0.0	3111.3	3315.2	3355.6	3481.0	203.9	40.4	125.3	5.9	1.2	3.6	10.7	90.1	3.6	1
Janapragati C	118.8	97.2	21.7	0.0	0.0	0.0	0.0	13757.8	14683.6	14963.6	15454.8	925.8	280.0	491.2	7.8	2.4	4.1	14.3	115.8	4.8	1
Jharna_C	34.5	23.5	11.1	0.0	0.0	0.0	0.0	3755.1	4006.3	4076.4	4214.4	251.2	70.1	138.0	7.3	2.0	4.0	13.3	108.7	4.4	1
Jugedarkha_D	125.6	0.0	0.0	0.0	0.0	101.5	24.1	13181.0	13860.8	13909.1	14306.9	679.8	48.3	397.8	5.4	0.4	3.2	9.0	104.9	3.0	3
Jyamire D	70.0	0.0	1.4	58.4	10.3	0.0	0.0	5850.0	6771.8	7272.8	7694.2	921.8	501.0	421.5	13.2	7.2	6.0	26.3	83.5	8.8	3

Kalchhe_D	21.5	3.4	0.0	13.0	5.1	0.0	0.0	1880.1	2116.1	2254.7	2367.3	236.0	138.7	112.6	11.0	6.5	5.2	22.7	87.5	7.6	3
Kalika_C	213.8	206.2	7.6	0.0	0.0	0.0	0.0	26342.3	28124.7	28702.3	29617.0	1782.5	577.6	914.7	8.3	2.7	4.3	15.3	123.2	5.1	1
Kamalamai_D	71.8	0.0	4.3	15.3	52.2	0.0	0.0	4942.2	5272.5	5718.9	5936.4	330.4	446.4	217.5	4.6	6.2	3.0	13.8	68.8	4.6	2
Kankali_C	91.6	78.5	13.1	0.0	0.0	0.0	0.0	10786.6	11513.6	11737.9	12120.0	727.0	224.3	382.2	7.9	2.4	4.2	14.6	117.8	4.9	1
Kharkandepakha_ G	47.8	45.8	2.0	0.0	0.0	0.0	0.0	5878.5	6276.2	6404.7	6609.0	397.7	128.5	204.4	8.3	2.7	4.3	15.3	122.9	5.1	3
Kharkopakha_G	51.2	44.4	6.8	0.0	0.0	0.0	0.0	6050.2	6458.1	6584.5	6798.5	407.9	126.5	213.9	8.0	2.5	4.2	14.6	118.3	4.9	3
Kopila_D	96.1	0.0	0.0	88.2	7.8	0.0	0.0	8204.7	9577.4	10285.1	10900.6	1372.6	707.7	615.6	14.3	7.4	6.4	28.1	85.4	9.4	3
Kuprisalleri_D	42.0	0.0	0.0	1.6	40.4	0.0	0.0	2684.5	2766.3	3034.4	3122.0	81.9	268.0	87.6	1.9	6.4	2.1	10.4	63.9	3.5	3
Kuwadi_G	92.3	83.8	8.5	0.0	0.0	0.0	0.0	11105.0	11854.9	12091.9	12481.5	749.9	237.0	389.7	8.1	2.6	4.2	14.9	120.4	5.0	1
Kyamundanda_G	58.7	56.6	2.1	0.0	0.0	0.0	0.0	7235.5	7725.1	7883.7	8135.0	489.6	158.6	251.3	8.3	2.7	4.3	15.3	123.2	5.1	1
LaliGurans_D	35.5	0.0	0.0	10.3	25.2	0.0	0.0	2488.7	2683.7	2920.3	3038.4	194.9	236.7	118.1	5.5	6.7	3.3	15.5	70.1	5.2	2
Lamidanda_G	61.6	59.0	2.6	0.0	0.0	0.0	0.0	7569.5	8081.5	8247.0	8510.2	512.1	165.4	263.2	8.3	2.7	4.3	15.3	122.9	5.1	1
LaxmiMahila_G	8.7	8.1	0.6	0.0	0.0	0.0	0.0	1058.4	1129.9	1152.7	1189.7	71.5	22.8	37.0	8.2	2.6	4.2	15.1	121.3	5.0	1
Lodhini_D	50.7	0.0	0.0	46.6	4.0	0.0	0.0	4329.9	5055.2	5428.6	5753.7	725.4	373.4	325.2	14.3	7.4	6.4	28.1	85.5	9.4	2
Ludidamgade_G	270.7	221.4	44.1	0.0	5.2	0.0	0.0	31284.6	33376.5	34045.2	35156.8	2091.9	668.7	1111.6	7.7	2.5	4.1	14.3	115.6	4.8	1
Ludikhola_G	17.4	5.8	11.7	0.0	0.0	0.0	0.0	1584.9	1688.9	1709.9	1773.5	104.0	21.0	63.6	6.0	1.2	3.6	10.8	91.0	3.6	2
Mahabhir_D	50.3	0.0	0.0	33.1	2.3	14.9	0.0	4753.4	5375.3	5642.0	5917.0	621.9	266.7	275.0	12.4	5.3	5.5	23.2	94.6	7.7	2
MahakalSahele_ D	39.4	0.0	0.0	26.7	12.7	0.0	0.0	3131.3	3561.0	3840.4	4046.2	429.7	279.4	205.8	10.9	7.1	5.2	23.2	79.5	7.7	3
Mahalaxmi_G	64.0	38.1	25.9	0.0	0.0	0.0	0.0	6678.4	7123.4	7240.4	7490.6	444.9	117.1	250.1	7.0	1.8	3.9	12.7	104.4	4.2	1
Maithan_D	28.3	0.0	0.0	3.5	24.9	0.0	0.0	1870.7	1959.9	2143.5	2214.5	89.2	183.6	71.0	3.1	6.5	2.5	12.1	66.0	4.0	2
MajhikholaSimre ndanda_G	6.0	0.3	5.7	0.0	0.0	0.0	0.0	461.2	490.8	494.2	514.4	29.6	3.3	20.2	4.9	0.6	3.4	8.9	76.9	3.0	1
Majhkharka_lisep aniD	174.2	0.0	0.0	0.0	0.0	145.7	28.4	18550.4	19540.5	19606.9	20155.2	990.0	66.4	548.3	5.7	0.4	3.1	9.2	106.5	3.1	2
Mathani_D	28.3	0.0	0.0	22.5	5.8	0.0	0.0	2331.3	2687.0	2891.5	3055.7	355.6	204.5	164.3	12.6	7.2	5.8	25.6	82.4	8.5	2
Napkenmara_D	152.5	0.0	0.0	0.0	12.1	82.6	57.8	13803.5	14286.4	14420.5	14908.4	482.9	134.1	487.9	3.2	0.9	3.2	7.2	90.5	2.4	3
Nibuwatar_C	329.2	315.2	14.0	0.0	0.0	0.0	0.0	40455.6	43192.3	44076.5	45483.1	2736.8	884.2	1406.6	8.3	2.7	4.3	15.3	122.9	5.1	1
PalekoBan_D	1.5	0.0	0.0	0.0	0.0	1.0	0.5	146.2	152.7	153.3	158.1	6.4	0.6	4.8	4.3	0.4	3.2	8.0	98.5	2.7	4

PalungMahila_D	10.3	0.0	0.0	0.3	10.0	0.0	0.0	654.8	673.8	739.2	760.3	19.0	65.5	21.1	1.8	6.4	2.0	10.3	63.7	3.4	4
Patalchhape_G	8.2	7.4	0.7	0.0	0.0	0.0	0.0	983.8	1050.2	1071.3	1105.8	66.4	21.0	34.5	8.1	2.6	4.2	14.9	120.5	5.0	1
Pauwa_D	58.6	0.0	0.0	33.6	8.5	8.6	8.2	4956.1	5529.0	5840.1	6141.1	573.0	311.1	301.0	9.8	5.3	5.1	20.2	84.5	6.7	2
Pokhari_D	23.6	0.0	0.0	10.8	5.6	7.3	0.0	2128.9	2355.1	2473.4	2579.2	226.3	118.3	105.8	9.6	5.0	4.5	19.1	90.2	6.4	3
Pragati_C	115.5	70.8	44.7	0.0	0.0	0.0	0.0	12162.7	12973.7	13190.0	13643.7	811.0	216.3	453.7	7.0	1.9	3.9	12.8	105.3	4.3	1
Pungche_G	18.1	15.4	2.7	0.0	0.0	0.0	0.0	2130.2	2273.7	2317.9	2393.5	143.5	44.2	75.5	7.9	2.4	4.2	14.5	117.5	4.8	1
Ramite_D	13.6	0.0	0.0	13.2	0.4	0.0	0.0	1178.1	1381.9	1482.9	1573.3	203.8	100.9	90.5	15.0	7.4	6.7	29.1	86.6	9.7	2
RamLaxman_G	13.2	12.8	0.5	0.0	0.0	0.0	0.0	1633.1	1743.6	1779.4	1836.1	110.5	35.8	56.7	8.3	2.7	4.3	15.3	123.3	5.1	1
Salleri_D	92.3	0.0	0.0	26.7	65.6	0.0	0.0	6459.8	6964.2	7578.6	7884.6	504.4	614.4	306.0	5.5	6.7	3.3	15.4	70.0	5.1	2
Samfrang_C	63.9	26.8	37.1	0.0	0.0	0.0	0.0	6101.6	6504.2	6594.8	6833.5	402.6	90.6	238.7	6.3	1.4	3.7	11.5	95.5	3.8	1
SandanBisaune_ G	50.6	48.7	2.0	0.0	0.0	0.0	0.0	6230.9	6652.5	6788.9	7005.4	421.6	136.4	216.5	8.3	2.7	4.3	15.3	123.1	5.1	2
Sankhadevi_D	305.3	0.0	0.0	0.0	0.0	247.4	57.9	32074.3	33733.2	33850.5	34816.8	1658.9	117.3	966.3	5.4	0.4	3.2	9.0	105.1	3.0	2
Sanobotle_D	35.1	0.0	0.0	0.0	0.0	18.3	16.8	3129.2	3223.1	3237.6	3355.4	93.9	14.6	117.7	2.7	0.4	3.4	6.5	89.3	2.2	2
Satkanya_C	58.3	56.0	2.3	0.0	0.0	0.0	0.0	7169.8	7654.9	7811.7	8060.9	485.1	156.9	249.2	8.3	2.7	4.3	15.3	123.0	5.1	1
Setidevi_D	421.7	0.0	0.0	0.0	0.0	192.6	229.1	36155.2	37026.5	37204.5	38638.4	871.4	177.9	1433.9	2.1	0.4	3.4	5.9	85.7	2.0	2
Shikhar_G	50.8	42.5	8.4	0.0	0.0	0.0	0.0	5932.1	6331.5	6453.4	6664.5	399.5	121.9	211.1	7.9	2.4	4.2	14.4	116.7	4.8	1
ShikharBarbhanjy ang_G	55.5	55.4	0.0	0.0	0.0	0.0	0.0	6928.4	7397.8	7552.2	7791.3	469.4	154.3	239.1	8.5	2.8	4.3	15.5	124.8	5.2	1
Shikhardanda_G	30.3	16.3	14.1	0.0	0.0	0.0	0.0	3079.6	3284.2	3335.5	3452.5	204.6	51.4	117.0	6.7	1.7	3.9	12.3	101.5	4.1	1
Simpani_D	64.4	0.0	0.0	8.1	56.4	0.0	0.0	4249.8	4453.5	4870.5	5032.2	203.7	417.0	161.6	3.2	6.5	2.5	12.1	66.0	4.0	2
Simsungure_D	33.4	0.0	0.0	3.9	29.5	0.0	0.0	2192.9	2294.2	2509.7	2591.9	101.3	215.5	82.2	3.0	6.5	2.5	12.0	65.8	4.0	2
Siraute_G	60.3	56.2	4.1	0.0	0.0	0.0	0.0	7335.1	7830.9	7989.2	8245.5	495.7	158.3	256.3	8.2	2.6	4.2	15.1	121.6	5.0	2
Sitakunda_D	141.3	0.0	45.1	15.7	80.5	0.0	0.0	9779.6	10348.9	10995.1	11404.3	569.3	646.2	409.2	4.0	4.6	2.9	11.5	69.2	3.8	2
Sitalupakha_G	5.7	0.0	0.0	4.1	1.6	0.0	0.0	458.7	524.5	565.1	596.2	65.8	40.7	31.0	11.6	7.1	5.5	24.2	80.7	8.1	2
Srijana_D	264.2	0.0	0.0	0.0	0.0	209.9	54.3	27528.6	28924.1	29026.1	29865.2	1395.6	102.0	839.1	5.3	0.4	3.2	8.8	104.2	2.9	2
Sundarimai_D	13.0	0.0	0.0	4.5	8.5	0.0	0.0	927.2	1008.7	1096.0	1142.8	81.5	87.3	46.8	6.3	6.7	3.6	16.6	71.5	5.5	2
Taksartari_G	89.3	83.1	6.2	0.0	0.0	0.0	0.0	10851.6	11585.0	11819.1	12198.3	733.4	234.1	379.2	8.2	2.6	4.2	15.1	121.5	5.0	1
Thanhsadeurali_ D	124.4	0.0	0.0	0.0	0.0	59.3	65.3	10811.2	11091.3	11143.6	11565.6	280.1	52.3	422.0	2.3	0.4	3.4	6.1	86.9	2.0	4

Tharlange_D	204.0	0.0	0.0	35.9	20.1	148.1	0.0	21487.8	23141.3	23589.6	24321.8	1653.5	448.4	732.2	8.1	2.2	3.6	13.9	105.3	4.6	2
Thoknebhanjyang																					
_G	76.2	73.5	2.7	0.0	0.0	0.0	0.0	9393.3	10029.0	10235.0	10561.1	635.6	206.0	326.1	8.3	2.7	4.3	15.3	123.3	5.1	1
Thumkadada_D	40.8	0.0	0.0	0.0	0.0	20.6	20.2	3601.9	3704.6	3721.6	3859.0	102.7	17.0	137.4	2.5	0.4	3.4	6.3	88.3	2.1	3
Thutemane_D	23.6	0.0	0.0	0.0	2.1	8.6	12.9	1910.2	1946.1	1968.3	2046.3	35.9	22.2	78.0	1.5	0.9	3.3	5.8	81.0	1.9	4
TimureTinsale_D	67.1	0.0	0.0	23.5	43.6	0.0	0.0	4798.1	5222.1	5673.6	5916.2	424.0	451.5	242.6	6.3	6.7	3.6	16.7	71.5	5.6	4

Note:

A- Total area of community forests (ha)

B- Total area of forests in<1000m (\geq 70% canopy) (ha) C-Total area of forests in <1000m (<70% canopy) (ha) D-Total area of forests in 1000-2000m (\geq 70% canopy) (ha) E-Total area of forests in 1000-2000m (<70% canopy) (ha) F-Total area of forests in >2000m (\geq 70% canopy) (ha) G-Total area of forests in <2000m (<70% canopy) (ha) H-Total Carbon stock (Mg) in CF in 2010 I-Total Carbon stock (Mg) in CF in year 2011 J-Total Carbon stock (Mg) in CF in year 2012 K- Total Carbon stock (Mg) in CF in year 2013 L-Carbon stock change (Mg) in year 2010-2011 M- Carbon stock change (Mg) in year 2012-2012 N-Carbon stock change (Mg t) in year 2012-2013

O- Carbon stock change (MgC/ha) in CF in year 2010-2011

P- Carbon stock change (MgC/ha) in CF in year 2011-2012 Q-Carbon stock change (MgC/ha) in CF in year 2012-2013

R- Carbon stock change (MgC/ha) in CF in year 2012-2013 R- Carbon stock change (MgC/ha) in CF in year 2010-2013

S- Carbon stock (MgC/ha) in 2010

T-Average annual stock change (MgC/ha) in CF

Appendix H: Summary of local practices, changes in CFs and local perceptions about REDD+ in the study areas

1. Status of forest before CFs

Most of the CFs in the mid-hills (*Schima-Castanopsis* forests, *Pine* forests) were degraded when legal community management activities started. Barren soil was visible from a distance due to the sparse canopy of the forests. Most of the large trees with cylindrical boles were gone. Management commenced with the collective decision of all members in CFs.

S. robusta and mixed broadleaf forests were comparatively better in terms of density; however, forests were treated as government property and local people were not taking care of them. They were collecting forest products illegally using haphazard practices. The government gave management authority of *S. robusta* forests to communities years later than was the case for higher and middle altitude forests. Moreover, settlements in areas surrounding *S. robusta* and mixed broadleaf forests of lower altitude areas have many newly migrated households (about 25–30 years). Therefore, they do not have established forest management practices compared to higher altitude forests.

2. Forest management practices of communities

According to communities, the key aims of forest management were to supply forest products for their own use, maintain forests for future generations and regulate rainfall. After community forestry practices commenced, they observed improved forests status and availability of forest products (firewood, fodder and grass). Now communities have good practices enabling the distribution of forest products in all CFs.

3. Benefits obtained from forests

In all CFs, there are provisions to facilitate income generating activities for poor members. These include a sustainable supply of forest products for their own use and raw materials for cottage industries. Handmade paper, essential oil distillation plants and beehive briquettes are the main interests in middle and high altitude CFs (*Schima-Castanopsis, Pine* and *Rhododendron-Quercus* forests). In lower altitude areas (*S. robusta* and mixed broad leaf), income from firewood and the sale of timber are main income streams. Possible income from ecosystem services including carbon sequestration is of increasing interest. Communities are willing to change their existing forest management and use practices to gain increasing carbon and non-carbon products if they get appropriate monetary benefits.

4. Decision making

Consensus building was the main approach for making decisions in CFs. In most CFs, the Chairman and Secretary call an executive committee meeting and provide an agenda, giving a chance for all participants to give their opinion, before arriving at a decision.

In a general assembly of the CFUG, the presence of all members is mandatory. The executive committee put out an agenda and take opinions from participants of the assembly to arrive at a decision. However, there is less chance to collect the opinions of communities due to time constraints particularly in terai (*S. robusta* and mixed broad leaf) where CFUG households are larger in numbers and more heterogeneous. However, mostly middle altitude CFUGs are small (by household numbers) and homogeneous; therefore, people have more chance to speak and include their voice in the decisions.

5. Preferred timber species

S. robusta is the preferred timber tree available in lower altitudes whereas *Schima-wallichi, Michelia, Pine* spp. and *Alnus nepalensis* are the main timber trees in middle and higher altitude areas. However, rich people of the hilly areas also use *S. robusta* timber brought from the terai region because it is comparatively durable; this is too expensive for poor people.

6. <u>Source of forest products</u>

CFs, private lands and leasehold forests (a patch of government forests handed to a group of poor households for the production of forest products, use and to generate income—e.g. fodder and grass production in leasehold forests for livestock is in practice in Kayerkhola watershed, Chitwan) are the main sources of forest products. CFs and leasehold forests were the main source in lower altitude terai (*S. robusta* and mixed broad leaf forests) whereas CFs and private land in hill areas (*Schima-Castanopsis, Pine* and *Rhododendron-Quercus* forests). In hilly regions, private land is less productive for agriculture and trees are grown in such lands. The collection of forest products from Government forests is limited because these forests are not in close to settlements. Timber products are generally obtained from the CFs and very little from private land. Timber extraction is only done for needy people after a decision of the executive committee of a CFUG.

7. <u>Knowledge of REDD+ at communities</u>

The majority of people were able to explain that the conservation of forests generates carbon and that will generate monetary benefits from carbon trading. They said the REDD+ is coming up and that they are implementing a pilot project which may not bring carbon money. However, they were happy to take initiatives for carbon enhancement in their forests as a trial. The level of understanding about REDD+ increased in all CFs after project activities.

8. Key changes made for REDD+

Communities have changed their existing forest product use behaviours for the REDD+. The main behaviours were:

- Most of CFUGs have reduced timber and green tree harvest in forests
- They have made more systematic and careful collection of grasses and fodder to reduce possible damages

- They have not reduced NTFP income but have tried to promote this in a sustainable way
- Decisions were made in executive committee meetings and assemblies. The numbers of meetings increased for REDD+
- Additional efforts were made for silviculture operations, guarding against illegal harvesting and fire control activities.
- Plantations on non-arable private lands and sparse areas in CFs has increased
- Reduced grazing in CFs
- Reduced numbers of livestock in some CFs

9. Perspective benefits of REDD+

Improvement of forests, sustainable supply of forest products and monetary benefit for CFUGs for use in development activities such as road, school, community building and alternative energy (improved cooking stock, biogass, solar) promotion etc. REDD+ carbon money can also be used for income generation activities particularly for poor.

10. <u>Suggestions for making the REDD+ project effective in CFs</u>

- Encourage all members by proving adequate knowledge of the REDD+
- Compensate/ benefit all users with priority to poor users
- Provide alternatives for household energy needs
- Provide alternative income sources that help to reduce livestock number therefore reduce fodder and grass requirements
- Institutional setup to ensure the flow of carbon benefits and trade

11. <u>Opinion about biomass quantity in proximity of forests from settlement and</u> road head

In lowland terai, forests located close to a roadhead have high levels of illegal harvesting activity and may have less biomass whereas in the hills, this is not the case. In a majority of forests located close to a settlement, there is more care and people do not engage in illegal activities in that area; therefore biomass growth can be higher. However, lower altitude *S. robusta* forests and broadleaf forests had heterogeneous communities and comparatively new settlements which may have led to some extraction going on.

Bhari	A load that an adult male/female carries on his/her back. An average Bhari of fuelwood is 30 kg, Bhari of fodder is 25 kg, average Bhari of litter is 20 kg
Caste	A form of social stratification characterized by hereditary transmission which may involve different occupations and beliefs
Co-benefits/non-carbon benefits	Benefits arising from REDD schemes (other than reducing GHG emissions), such as alleviating poverty, protecting environment, enhancing biodiversity, improving forest governance and protecting human rights
Community forestry	Forest management model in which a national forest handed over to an user group for its development, conservation and utilization for the collective interest following forest policy provisions
Current annual increment (CAI)	Growth observed in a stand in a specific one year period
Forest ecosystem services	Multiple benefits that a forest can provide to human being. It includes provisioning services such as food, raw material, medicine, genetic resources, biodiversity etc; regulating services such carbon sequestration, water/air purification; and cultural services such as recreational and historical value. In this study ecosystem services refers other ecosystem services than carbon sequestration
Leakage	Displaced emissions from REDD+ project area
Livestock unit (LU)	A way of comparing the nutritional requirements of grazing animals. Equivalent of 1Buffalo=1, 1 Cow=0.7, and 1 Goat=0.1
Maximum sustained yield (MSY)	This is theoretically largest yield that can be taken from a species stock over a long period
Mean annual increment (MAI)	Average growth per year a stand of trees has exhibited to a specified age

Appendix I: Definition of some terms used in this thesis

Non-permanence	The risks that carbon removals are reversed after the credits have been created
REDD+	A likely incentive mechanism for reducing emissions from deforestation and forest degradation, conservation, sustainable management of forests and enhancement of forest carbon stock in developing countries
Sustained yield	The amount of a resource obtained from such a schedule without depleting the resources